

Performance Analysis of Optimization Methods for Machine Learning

HADI ABBASZADEHPEIVASTI



Performance Analysis of Optimization Methods for Machine Learning

PROEFSCHRIFT

ter verkrijging van de graad van doctor aan Tilburg University op gezag van de rector magnificus, prof. dr. W.B.H.J. van de Donk, in het openbaar te verdedigen ten overstaan van een door het college voor promoties aangewezen commissie in de Aula van de Universiteit op

vrijdag 4 oktober 2024 om 13.30 uur

door

Hadi Abbaszadehpeivasti,

geboren te Tabriz, Iran

prof. dr. E. de Klerk (Tilburg University)
prof. dr. M. Laurent (Tilburg University)
prof. dr. I. Birbil (University of Amsterdam)
prof. dr. F. Glineur (Université Catholique de Louvain)
dr. M. Hochstenbach (TU Eindhoven)
prof. dr. J.C. Vera Lizcano (Tilburg University)
prof. dr. ir. R. Sotirov (Tilburg University)

This project was supported by the Dutch Scientific Council (NWO) Grant Optimization for and with Machine Learning (OPTIMAL), OCENW.GROOT.2019.015.

©2024 Hadi Abbaszadehpeivasti, The Netherlands. All rights reserved. No parts of this thesis may be reproduced, stored in a retrieval system or transmitted in any form or by any means without permission of the author. Alle rechten voorbehouden. Niets uit deze uitgave mag worden vermenigvuldigd, in enige vorm of op enige wijze, zonder voorafgaande schriftelijke toestemming van de auteur. To the loving souls who have enriched my life with their presence

Acknowledgments

This thesis is a clear reflection of strong support from many people who have been vital to my academic journey. My deepest gratitude goes to my supervisor, Etienne de Klerk, and co-supervisor, Monique Laurent, for their exceptional guidance, and the wisdom they shared. Moslem Zamani, postdoctoral researcher, deserves special recognition for his invaluable assistance and insights. I would also like to express my appreciation to the members of my defense committee, Ilker Birbil, François Glineur, Michiel Hochstenbach, Juan Vera Lizcano, and Renata Sotirov, for their rigorous evaluation, which has undoubtedly improved the quality of this dissertation.

I would like to express my heartfelt gratitude to my mother, Batoul, for her unwavering support and inspiration throughout my academic journey. Her encouragement and belief in my abilities have been a constant source of motivation. I am deeply thankful for her boundless love and guidance, which have played a pivotal role in my accomplishments. Her presence and support have been invaluable, and I owe a significant part of my success to her. Additionally, I would like to thank my sister, Hamideh, for her support during my academic journey.

I owe immeasurable thanks to my family and friends, who provided unending encouragement and understanding throughout this challenging pursuit. To my fellow graduate students and colleagues, your mutual support and shared experiences have been a constant source of inspiration. I must acknowledge the support provided by Tilburg University, which furnished the necessary resources and facilities for this research.

To all those who have contributed in ways both seen and unseen, I am profoundly grateful for your roles in this important chapter of my life.

Contents

Li	List of Figures xi		
Li	st of '	Tables 2	viii
No	otatio	n	xv
1	Intr	oduction	1
	1.1	Nonlinear programming	2
	1.2	Classes of functions	2
	1.3	Iterative first-order methods	3
	1.4	Convergence rate	4
	1.5	Performance estimation problems (PEPs)	4
	1.6	Semidefinite programming	5
	1.7	Machine learning and optimization	6
	1.8	Thesis overview	6
	1.9	Contributions to the Literature	10
2	Prel	iminaries and interpolation theorems	13
	2.1	Preliminaries	13
		2.1.1 Convex functions	15
		2.1.2 Subgradient	16
		2.1.3 Conjugate function	18
		2.1.4 L-smooth functions	20
		2.1.5 Strong convexity	22
		2.1.6 Non-convex non-smooth functions	24
	2.2	Semidefinite programming	24
	2.3	Interpolation theorems	26
	2.4	Conclusion	32

3	Perf	ormance estimation problems (PEPs)	33
	3.1	Worst case formulation of convergence rate of iterative first-order	
		methods	34
	3.2	Gram matrix reformulation	36
	3.3	Dual multipliers	40
	3.4	A simple example	41
	3.5	Some recent works on/with PEP	43
4	The	exact worst-case convergence rate of the gradient method with	
	fixe	d step lengths for L-smooth functions	47
	4.1	Introduction	48
	4.2	Performance estimation	49
	4.3	Worst-case convergence rate	52
	4.4	Extension of L-smooth functions	57
	4.5	Concluding remarks	60
5 Conditions for linear convergence of the gradient method		ditions for linear convergence of the gradient method for non-	
	con	vex optimization	61
	5.1	Introduction	62
	5.2	Linear convergence under the PŁ inequality	64
	5.3	The PŁ inequality: relation to some classes of functions	74
	5.4	Concluding remarks	81
6 Convergence rate analysis of the randomized and cycli		vergence rate analysis of the randomized and cyclic coordinate	
	dese	cent methods for convex optimization through semidefinite pro-	
	grai	nming	83
	6.1	Introduction	84
	6.2	Convergence rate of randomized coordinate descent	85
		6.2.1 The case of <i>L</i> -smooth functions	85
		6.2.2 The case of convex quadratic functions	89
		6.2.3 The case of μ -strongly convex <i>L</i> -smooth functions	91
	6.3	Cyclic coordinate descent	91
		6.3.1 Relation to the Gauss–Seidel method	95
		6.3.2 Weighted-Jacobi method	97
	6.4	Conclusion	101

7	Convergence rate analysis of the gradient descent-ascent method for					
	conv	convex-concave saddle-point problems 103				
	7.1	Introduction	04			
	7.2	The gradient descent-ascent method10	06			
		7.2.1 Numerical illustration	16			
	7.3	Linear convergence without strong convexity	17			
	7.4	Concluding remarks	22			
8	On t	he convergence rate of the difference-of-convex algorithm (DCA)12	23			
	8.1	Introduction	24			
	8.2	Basic Definitions and Preliminaries	27			
		8.2.1 The DC problem	29			
		8.2.2 One iteration of DCA on an example	31			
	8.3	Performance analysis of the DCA for smooth f_1 or f_2	33			
		8.3.1 An example to prove tightness	39			
		8.3.2 Convergence rates for the iterates	40			
		8.3.3 The gradient descent method14	41			
		8.3.4 Proximal gradient method	42			
	8.4	Performance estimation using a convergence criterion for critical				
points in the nonsmooth case		points in the nonsmooth case	44			
		Performance analysis of DCA with regularization	47			
		Linear convergence of the DCA under the Polyak-				
		Łojasiewicz inequality1	50			
	8.7	Conclusion	52			
9 The exact worst-case convergence rate of the alternating direction						
	met	hod of multipliers 15	55			
	9.1	Introduction	56			
	9.2	Performance estimation	59			
	9.3	Worst-case convergence rate	53			
	9.4	Linear convergence of ADMM1	75			
	9.5	<i>R</i> -linear convergence of ADMM	34			
	9.6	Concluding remarks	38			
10	Con	clusion and outlook 19	91			

10 Conclusion and outlook

A Appendices		197	
	A.1	Nonnegativity of multipliers in Theorem 7.2	197
A.2 Identity used in the proof of Theorem 7.2		Identity used in the proof of Theorem 7.2	200
A.3 Proof for positive semidefniteness of the matrix presented in Tl			
		orem 9.6	201
Index		205	
Bil	Bibliography 2		

List of Figures

2.1	Three points x^1, x^2, x^3 along with their corresponding function values and gradients	27
2.2	A μ -strongly convex and <i>L</i> -smooth function, with $\mu = 0.21$ and $L = 3.74$ modules, that passes through the points with the corresponding function values and gradients shown in Figure 2.1	28
4.1	Plot of the function f in (4.14) for different parameters and $t_k = \frac{1}{L}$. (Dotted lines denote the endpoints of intervals.)	56
5.1	Convergence rate computed by performance estimation (red line) and the bound given by (5.19) (blue line) for $\frac{\mu_g}{L} \in (0, 1)$.	76
6.1	Convergence rate for Algorithm 6.2 computed by performance estimation problem (6.7) (dashed lines) and the bound given by (6.3) (thick lines) for $L = 2$, $l_{max} = 1$, $t = 1$, $\Delta = 1$ and different $n \dots$	88
6.2	Convergence rate for Algorithm 6.2 computed by performance estimation problem (6.7) (dashed lines) and the bound given by (6.3) (thick lines) for $L = 4$, $l_{\text{max}} = 2$, $t = 0.5$, $\Delta = 1$ and different n	88
6.3	Convergence rate for Algorithm 6.2 computed by performance estimation problem for quadratic functions (red line) and the bound given by (6.7) (blue line) for $n = 10, L = 2, l_{max} = 1, t = 1, \Delta = 1$.	90
6.4	Convergence rate for Algorithm 6.2 computed by performance estimation problem for quadratic functions (red line) and the bound given by (6.7) (blue line) for $n = 2, L = 2, l_{max} = 1, t = 1, \Delta = 1$	90
6.5	Convergence rate Algorithm 6.3 computed by performance estima- tion problem (6.15) (blue), the bound given by (6.13) (red) and the bound for quadratic functions (green) for $n = 2, L = 2, \ell_1 =$	
	$1, \ell_2 = 1, t = 0.5, \Delta = 1. \dots $	94

6.6	Convergence rate for Algorithm 6.3 computed by performance esti- mation problem of quadratic functions for $\ell_i = 1$ $i \in \{1, \dots, n\}, L = \sum_{i=1}^{n} \ell_i, t = 1, \Delta = 1$ and different n
7.1	Mean values of $ x^k - x^* ^2 + y^k - y^* ^2$ for 100 randomly generated instances for each iteration <i>k</i> using the two different step lengths <i>t</i> 117
8.1	Illustration of the function (8.10)
8.2	Illustration of the functions f_1 and f_2 of problem (8.11) 131
8.3	Illustration of the subproblem (8.12)
8.4	Illustration of the first iteration of the Algorithm 8.1 on the function
	(8.10)

List of Tables

9.1	Scenarios leading to linear convergence rates	183
10.1	The summary of the main results on the convergence rates pre-	
	sented in the thesis	195

Notation

Set notation

the empty set
the union of the sets A and B
the intersection of the sets A and B
A is a subset of B
Set difference, i.e., $\{a \in A : a \notin B\}$
the closure of <i>A</i>
the interior of <i>A</i>
the relative interior of A
domain of function f
epigraph of function f
the convex hull of $A \subseteq \mathbb{R}^n$

Special sets

Nthe natural numbers \mathbb{N}^n the set of <i>n</i> -tuples of the natural numbers \mathbb{Z} the integer numbers \mathbb{R} the integer numbers \mathbb{R}_+ the non-negative reals \mathbb{R}^n the <i>n</i> -dimensional vectors of reals \mathbb{R}^n the <i>n</i> -dimensional vectors of non-negative real
\mathbb{N}^n the set of <i>n</i> -tuples of the natural numbers \mathbb{Z} the integer numbers \mathbb{R} the real numbers \mathbb{R}_+ the non-negative reals \mathbb{R}^n the <i>n</i> -dimensional vectors of reals \mathbb{R}^n the <i>n</i> -dimensional vectors of non-negative real
\mathbb{Z} the integer numbers \mathbb{R} the real numbers \mathbb{R}_+ the non-negative reals \mathbb{R}^n the <i>n</i> -dimensional vectors of reals \mathbb{R}^n the <i>n</i> -dimensional vectors of non-negative real
\mathbb{R} the real numbers \mathbb{R}_+ the non-negative reals \mathbb{R}^n the <i>n</i> -dimensional vectors of reals \mathbb{R}^n the <i>n</i> -dimensional vectors of non-negative real
\mathbb{R}_+ the non-negative reals \mathbb{R}^n the <i>n</i> -dimensional vectors of reals \mathbb{R}^n the <i>n</i> -dimensional vectors of non-negative real
\mathbb{R}^n the <i>n</i> -dimensional vectors of reals \mathbb{R}^n the <i>n</i> -dimensional vectors of non-negative real
\mathbb{R}^n the <i>n</i> -dimensional vectors of non-negative real
¹ / ₊ the <i>n</i> -unitensional vectors of non-negative real
\mathbb{S}^n the set of $n \times n$ real symmetric matrices
\mathbb{S}^n_+ the set of $n \times n$ positive semidefinite matrices

Functions

 f^* conjugate of function f

Inner Products, dualities and Norms

$\langle \cdot, \cdot \rangle_X$	bilinear function for dual system (X, X^*)
$\langle \cdot, \cdot \rangle_x$	local inner product at <i>x</i>
$\ \cdot\ _x$	norm induced by $\langle \cdot, \cdot \rangle_x$

Linear Algebra

Ι	identity matrix (of fitting size)
I _n	$n \times n$ identity matrix
J	all-ones matrix (of fitting size)
J_n	$n \times n$ all-ones matrix
е	all-ones vector
e _i	<i>i</i> -th standard unit vector
Tr	the trace operator
$A \circ B$	Hadamard product of matrices A and B
span(A)	real span of elements of A
$v^{ op}$	transpose of vector v (or matrix)
$A \succ 0$	the matrix A is positive definite
$A \succeq 0$	the matrix A is positive semidefinite
$\lambda_{\min}(A)$	minimum eigenvalue of matrix A
$\lambda_{\max}(A)$	maximum eigenvalue of matrix A
A^{-1}	inverse of matrix A
A_{ij}	i-th and j -th element of matrix A

What we hope ever to do with ease, we must learn first to do with diligence. Samuel Johnson

Introduction

Preamble

Optimization plays a crucial role in various fields, ranging from engineering and economics to machine learning and data analysis. Most of the optimization problems are hard to be solved analytically (or in some cases it is not efficient to do so), therefore they are solved using iterative methods to approximate the solution. As a result, the development of efficient algorithms for solving optimization problems is crucial. To understand if an algorithm is efficient for a specific class of problems it is important to understand the behaviour of the algorithm. Henceforth, analysis of convergence of algorithms is a fundamental research area in optimization theory. In this context, different mathematical frameworks and tools have emerged to study and estimate the performance of optimization algorithms.

In this section we will overview general optimization problems as well as some concepts that we will use in this thesis. We will end the chapter by providing the societal and scientific relevance of the topic. Then, we will present the outcomes of this research that appeared in peer-reviewed journals or are under review.

1.1 Nonlinear programming

The general form of a nonlinear programming problem is as follows

$$\inf_{x \in \mathbb{R}^n} f(x)$$
s.t. $g_i(x) \le 0$, for $i = 1, 2, \dots, m$
 $h_j(x) = 0$, for $j = 1, 2, \dots, p$

where $x = (x_1, x_2, ..., x_n)$ represents the decision variables, f(x) is the objective function to be minimized, $g_i(x)$ are the inequality constraints, and $h_j(x)$ are the equality constraints [LY⁺84]. Any point that satisfies the constraints is called a feasible point, and the set of feasible points is called the feasible set. Alternatively, the feasible set can be represented by X, defined as

$$\mathbb{X} = \left\{ x \in \mathbb{R}^n \mid g_i(x) \le 0, \ h_j(x) = 0, \ i \in \{1, \dots, m\}, \ j \in \{1, \dots, p\} \right\}.$$

Therefore the optimization problem can be written as

$$\inf_{x\in\mathbb{X}}f(x).$$

Analyzing such a general optimization problem is difficult, therefore we review different classes of functions facilitating studying them by considering their properties.

1.2 Classes of functions

Suppose that we want to solve the following optimization problem

$$\inf_{x\in\mathbb{R}^n}f(x).$$

It is almost impossible to talk about the behaviour of an algorithm for a general optimization problem as above [Nes18, Chapter 2], even for convergence of the algorithm to a local minimum. Therefore, it is important to introduce some assumptions on the function in order to facilitate the study of the function class. In other words, we assume that the function under the study belongs to some function classes \mathcal{F} and try to make these assumptions as little restrictive as possible.

In the literature convex or *L*-smooth functions are among the most important function classes. A function *f* is called *L*-smooth in \mathbb{R}^n if for some L > 0 it satisfies the following condition,

$$\|\nabla f(x) - \nabla f(y)\| \le L \|x - y\|$$
 for all $x, y \in \mathbb{R}^n$.

There are also other classes such as μ -strongly convex functions, where a function f is called μ -strongly convex if, for some $\mu \ge 0$, the function $f(x) - \frac{\mu}{2} ||x||^2$ is convex. Note that a 0-strongly convex function is just a convex function.

We will also consider the class of μ -strongly convex, *L*-smooth functions, denoted by $\mathcal{F}_{\mu,L}(\mathbb{R}^n)$. A function f belongs to this class if it is both *L*-smooth and μ -strongly convex. Note that the class $\mathcal{F}_{\mu,L}(\mathbb{R}^n)$ is a convex set in a suitable function space.

As we will discuss these function classes and their properties in Chapter 2 in detail, for further discussion we refer the interested reader to look at Chapter 2 and [Nes18, Bec17].

1.3 Iterative first-order methods

Iterative methods are among the most widely used algorithms to solve an optimization problem. The class of iterative algorithms ranges from zero-order to higher order algorithms. These algorithms start from an initial point x^0 and iteratively generate subsequent points (iterates). *First-order methods* are among the iterative methods. First-order methods use only the information of the gradient and function value at a given point. These methods can be categorized by three strategies that they use to generate the next point, namely line search, trust region, and fixed step lengths [NW06]. In this thesis we mainly focus on methods that use fixed step lengths.

Line-search algorithms, in general, move in a search direction d^k from the previous point x^k to generate the new point x^{k+1} . In other words, they solve the following problem

$$x^{k+1} = \operatorname{argmin}_t f(x^k + td^k).$$

One choice for the search direction is the steepest descent direction $-\nabla f(x^k)$ [NW06, Section 2.2].

Sometimes, it is too expensive to solve the above problem. One way to deal with this problem is to choose the step length in advance, which sometimes is called fixed step length. The gradient descent method is a widely used fixed step first-order iterative method. It updates the decision variables by taking steps proportional to the negative gradient of the objective function, i.e.

$$x^{k+1} = x^k - t_k \nabla f(x^k).$$

To have an overview on first-order methods we refer the reader to some reference books, e.g., [NW06, Nes18, Bec17].

Nocedal and Wright [NW06] name three properties that an algorithm should possess to be called a good algorithm, namely robustness, efficiency and accuracy. Studying worst-case convergence allows us to understand the weaknesses and strengths of an algorithm.

1.4 Convergence rate

Convergence rate in the worst case is a crucial aspect of analyzing optimization algorithms' performance. In this section we present some fundamental concepts and definitions about convergence rates.

Suppose that a sequence $\{x^k\}$ has a limit point x^* . We are interested in how fast $x^k - x^*$ goes to zero. We consider two types of *convergence*, *Q*-convergence and *R*-convergence [BGLS06].

Convergence of the quotient series $q^k := \frac{\|x^{k+1}-x^*\|}{\|x^k-x^*\|}$ as $k \to \infty$ is called *Q*-convergence of the sequence $\{x^k\}$. We say the sequence $\{x^k\}$ converges to x^* *Q*-linearly if $\limsup_{k\to\infty} q^k < 1$. If $\limsup_{k\to\infty} q^k = 0$ we call $\{x^k\}$ *Q*-super-linearly convergent [BGLS06, Section 1.5]. We usually omit the *Q* when we talk about convergence rates, if it does not cause any confusion.

On the other hand, if $||x^k - x^*|| \le v^k$ and v^k is *Q*-convergent, then the sequence $\{x^k\}$ is called *R*-convergent [NW06, Section A.2]. We will refer back to *R*-convergence in Section 9.5. It is obvious that *Q*-convergence implies *R*-convergence.

Moreover, if $\frac{\|x^k - x^*\|}{f(x^0) - f(x^*)} \leq \frac{c}{k^{\alpha}}$ for some constants *c* and $\alpha > 0$, the convergence rate of $\{x^k\}$ is called *sub-linear*, and it is denoted by $\mathcal{O}\left(\frac{1}{k^{\alpha}}\right)$ [Bec17].

For a comprehensive exploration of convergence rates, we recommend the interested reader to delve into the references [NW06] and [BGLS06]. These books provide valuable insights and in-depth analyses on the topic of convergence rates in the context of optimization.

1.5 Performance estimation problems (PEPs)

The Performance Estimation Problem (PEP) is a mathematical framework that aims to estimate the worst-case convergence rates and performance of optimization algorithms, originally for calculating convergence rates of first-order algorithms. This strong tool was first introduced by Drori and Teboulle [DT14]. This method was improved by the work of Taylor et. al [THG17c] by introducing necessary and sufficient interpolation conditions for μ -strongly convex *L*-smooth functions. Since then, many scholars used this method to find the convergence rate of different algorithms; see Section 3.5 for the recent works on PEP.

Briefly speaking, PEP tries to model the convergence rate of the algorithm by modeling the convergence rate problem as a semidefinite programming problem. The idea is to find the worst objective function from a given class for a given algorithm. In other words, PEP aims to identify the worst-case input for a given iterative optimization algorithm. This is in itself an optimization problem in some function space, and the surprising thing is that it may sometimes be reformulated as a semidefinite program. We describe PEP in detail in Chapter 3.

1.6 Semidefinite programming

As mentioned above, PEP transforms convergence rate analysis of an iterative algorithm into a semidefinite programming problem. In this section, we briefly introduce semidefinite programming (SDP); more details are presented in Section 2.2.

Semidefinite programming is a generalization of linear programming to handle optimization problems involving positive semidefinite matrix variables. More precisely, SDP deals with optimizing a linear objective function over the set of positive semidefinite matrices subject to some linear constraints [VB96].

Mathematically, SDP can be formulated as

$$p^* = \sup_{X} \langle C, X \rangle$$

s.t. $\langle A_i, X \rangle = b_i, \quad i = 1, \dots, m,$
 $X \succeq 0,$

where matrix *X* is the variable, C, A_1, \ldots, A_m are given symmetric matrices, and $b \in \mathbb{R}^m$ is a given vector [LV12].

SDP has been extensively studied in the field of optimization. The works by De Klerk [DK06], and Vandenberghe and Boyd [VB96] provide a comprehensive overview of SDP and its applications.

1.7 Machine learning and optimization

Machine learning typically involves a training phase, where the model learns from data, followed by a testing phase to evaluate its performance on unseen data, ensuring it can generalize well. Neural networks, which are central to deep learning, consist of interconnected layers of neurons that learn complex patterns [GBC16]. Machine learning algorithms can be broadly divided into four main categories as supervised, semi-supervised, unsupervised, and reinforcement learning [SCZZ19]. Typically, these algorithms are used for solving optimization problems during the training phase. This process aims to determine the optimal parameters of a model by minimizing a loss function.

A loss function, often denoted as $\mathcal{L}(\theta)$, depends on the model parameters θ . The optimization problem is then to find the values of the model parameters that minimize this loss function. Mathematically, it can be expressed as

$$\min_{\theta} \mathcal{L}(\theta).$$

For a comprehensive review on loss functions we refer the interested reader to [WMZT20]. In the context of training deep neural networks, the objective is to minimize a cost function, which can be formulated as

$$\min_{\theta} J(\theta) := \mathbb{E}_{(x,y) \sim \hat{p}_{\text{data}}} [\mathcal{L}(f(x;\theta), y)].$$

Here, $\mathcal{L}(\cdot)$ represents the loss function, $f(x;\theta)$ denotes the predicted value for the input x, θ stands for the model parameters, y represents the actual output, and \hat{p}_{data} corresponds to the empirical distribution [GBC16].

First-order methods in deep learning, such as Gradient Descent, SGD (Stochastic Gradient Descent), Momentum, and Adam, are crucial for optimizing neural networks. They work by using the gradient of the loss function to iteratively adjust the network's parameters, minimizing loss [BB24, GBC16].

The importance of the use of first-order methods in machine learning forms an additional motivation for gaining a better understanding of their convergence analysis.

1.8 Thesis overview

We now give a brief overview of the contents of this thesis, chapter by chapter. Moreover, a summary of the main results on the convergence rate in the thesis will be given later in Table 10.1.

Chapter 2

This chapter is dedicated to some preliminaries that will be used in the other chapters. In particular, we present some fundamental theorems on functions, semidefinite programming and most importantly, we present interpolation theorems that are foundations for the performance estimation method. These interpolation theorems are in the spirit of Whitney-type extension theorems that characterize when a function may be extended to a larger domain while keeping certain properties (like continuity, convexity, or L-smoothness).

Chapter 3

In this chapter, we introduce the performance estimation method. We show how to get a convergence rate of a first-order algorithm using this method. We define the Gram matrix semidefinite programming formulation and continue with dual presentation of the performance estimation problem. We present a simple example to show how the performance estimation method works. We conclude the chapter by reviewing some recent works on the performance estimation method.

Chapter 4

In this chapter, we study the convergence rate of the gradient (or steepest descent) method with fixed step lengths for finding a stationary point of an *L*-smooth function. We establish a new worst-case convergence rate, and show that the bound may be exact in some cases, in particular when all step lengths lie in the interval (0, 1/L]. In addition, we derive an optimal step length with respect to the new bound. In addition, we present an extension of *L*-smooth functions from a open convex set.

Chapter 5

In this chapter, we derive a new linear convergence rate for the gradient method with fixed step lengths for non-convex smooth optimization problems satisfying the Polyak-Łojasiewicz (PŁ) inequality. We establish that the PŁ inequality is a necessary and sufficient condition for linear convergence to the optimal value for this class of problems. We list some related classes of functions for which the gradient method may enjoy a linear convergence rate. Moreover, we investigate their relationship with the PŁ inequality.

Chapter 6

In this chapter, we study randomized and cyclic coordinate descent for convex unconstrained optimization problems. We improve the known convergence rates in some cases by using the numerical semidefinite programming performance estimation method. As a spin-off we provide a method to analyse the worst-case performance of the Gauss–Seidel iterative method for linear systems where the coefficient matrix is positive semidefinite with a positive diagonal. Moreover, we study the weighted Jacobi method for solving quadratic programming problems and revisit some well-known results in the literature.

Chapter 7

In this chapter, we study the gradient descent-ascent method for convex-concave saddle-point problems. We derive a new non-asymptotic global convergence rate in terms of distance to the solution set by using the semidefinite programming performance estimation method. The given convergence rate incorporates most parameters of the problem and it is exact for a large class of strongly convex-strongly concave saddle-point problems for one iteration. We also investigate the algorithm without strong convexity and we provide some necessary and sufficient conditions under which the gradient descent-ascent enjoys linear convergence.

Chapter 8

In this chapter, we study the non-asymptotic convergence rate of the DCA (differenceof-convex algorithm), also known as the convex–concave procedure, with two different termination criteria that are suitable for smooth and non-smooth decompositions, respectively. The DCA is a popular algorithm for difference-of-convex (DC) problems and known to converge to a stationary point of the objective under some assumptions. We derive a $O(1/\sqrt{N})$ worst-case convergence rate of the objective gradient norm after *N* iterations for certain classes of DC problems, without assuming strong convexity in the DC decomposition, and give an example which shows the convergence rate is exact. We also provide a new O(1/N)convergence rate for the DCA with another termination criterion to deal with the non-smooth case. Furthermore, we study the convergence rate for the proximal gradient method and the gradient descent method. Additionally, we study the impact of using regularization on DCA. Moreover, we derive a new linear convergence rate result for the DCA under the assumption of the Polyak–Łojasiewicz inequality. The novel aspect of our analysis is that it employs semidefinite programming performance estimation.

Chapter 9

In this chapter, we derive new non-ergodic convergence rates for the alternating direction method of multipliers (ADMM) by using performance estimation. We give some examples which show the exactness of the given bounds. We also study the linear and R-linear convergence of ADMM. We establish that ADMM enjoys a global linear convergence rate if and only if the dual objective satisfies the Polyak-Łojasiewicz (PŁ) inequality in the presence of strong convexity. In addition, we give an explicit formula for the linear convergence rate factor. Moreover, we study the *R*-linear convergence of ADMM under two new scenarios related to function classes and the rank of the matrix.

Chapter 10

We end the thesis by providing some concluding remarks and possible research questions for future work. Moreover, we provide a table summarizing the main results of the thesis.

Societal and scientific relevance of the thesis topics

In this section, we examine the influence of our endeavors on both society and the scientific community. As it is mentioned, optimization techniques are essential tools in a spectrum of fields, including mathematics, engineering, economics, computer science, and more.

From the society's perspective, the rising demand for optimization in various fields from healthcare, environmental sustainability, to financial market and machine learning, increases the demand for more efficient algorithms to solve these problems in an efficient way. With the increasing complexity of the systems and the proliferation of big data, there is an ever growing demand for more efficient and faster algorithms. The first-order algorithms demonstrate a good performance both in practice and in theory. They usually are easy to implement and have low

costs. On the other hand, different problems have different structures that may fit better with one algorithm than the other. Studying first-order methods provides practitioners with a good perspective on selecting the most suitable algorithm for their problem with a theoretical guarantee. Moreover, first-order methods are the back-bone for optimizing the models used in machine learning and it goes without saying how important machine learning is in the modern world.

1.9 Contributions to the Literature

This thesis is based on the following articles:

- [AdKZ22] Hadi Abbaszadehpeivasti, Etienne de Klerk, and Moslem Zamani. The exact worst-case convergence rate of the gradient method with fixed step lengths for L-smooth functions. *Optimization Letters*, 16(6):1649–1661, 2022.
- [AdKZ23a] Hadi Abbaszadehpeivasti, Etienne de Klerk, and Moslem Zamani. Conditions for linear convergence of the gradient method for nonconvex optimization. Optimization Letters, 17(5):1105–1125, 2023.
- [AdKZ23b] Hadi Abbaszadehpeivasti, Etienne de Klerk, and Moslem Zamani. Convergence rate analysis of randomized and cyclic coordinate descent for convex optimization through semidefinite programming. *Applied Set-Valued Analysis and Optimization*, 5(2):141–153, 2023.
- [ZAdK24] Moslem Zamani, Hadi Abbaszadehpeivasti, and Etienne de Klerk. Convergence rate analysis of the gradient descent-ascent method for convex-concave saddle-point problems. Optimization Methods and Software, 2024.
- [AdKZ23c] Hadi Abbaszadehpeivasti, Etienne de Klerk, and Moslem Zamani. On the rate of convergence of the difference-of-convex algorithm (DCA). *Journal of Optimization Theory and Applications*, pages 1–22, 2023.
- [ZAdK23] Moslem Zamani, Hadi Abbaszadehpeivasti, and Etienne de Klerk. The exact worst-case convergence rate of the alternating direction method of multipliers. *Mathematical Programming*, 2023.

These articles are used in the chapters of this thesis as follows:

Chapter 4	Based on [AdKZ22]
Chapter 5	Based on [AdKZ23a]

Chapter 6	Based on [AdKZ23b]
Chapter 7	Based on [ZAdK24]
Chapter 8	Based on [AdKZ23c]
Chapter 9	Based on [ZAdK23]

Now is no time to think of what you do not have. Think of what you can do with what there is.

The Old Man and the Sea, Ernest Hemingway

2

Preliminaries and interpolation theorems

Preamble

In this chapter, we delve into the study of interpolation theorems, a crucial ingredient extensively utilized throughout this thesis to analyze the complexity of first-order methods. In particular, these theorems sometimes enable us to reformulate the membership problem for some function classes as finite interpolation conditions.

In recent years, interpolation theorems have gained significant attention from researchers who aim to extend or leverage them for their analyses, particularly in the context of performance estimation. Before delving into these theorems, we will cover fundamental concepts in convex and smooth analysis. By establishing a solid foundation in these areas, we can better grasp the subsequent theorems and their implications.

2.1 Preliminaries

In this section, we introduce fundamental mathematical concepts and definitions that will serve as the foundation for our thesis. We provide a concise overview of

these key notions to establish the groundwork for our study. For a more in-depth exploration of these mathematical concepts and the proofs, interested readers are encouraged to refer to supplementary reference books (e.g., [Roc97, Bec17, Nes18, HUL13]).

Let us begin with the definition of convex sets.

Definition 2.1. A set $\mathbb{S} \subseteq \mathbb{R}^n$ is said to be *convex* if, for all $x, y \in \mathbb{S}$ and $\lambda \in [0, 1]$, we have:

$$\lambda x + (1 - \lambda)y \in \mathbb{S}.$$

In simpler terms, this definition implies that any line segment connecting two points within the set also lies entirely within the set.

The notation \mathbb{R}^n represents the Euclidean space of *n* dimensions. Now, let us define the inner product and *induced norm* with respect to a given matrix $A \in \mathbb{R}^{n \times m}$. Given vectors $x \in \mathbb{R}^n$ and $y \in \mathbb{R}^m$, their inner product with respect to matrix *A*, denoted as $\langle x, y \rangle_A$, is defined as:

$$\langle x, y \rangle_A = \langle x, Ay \rangle,$$

where $\langle \cdot, \cdot \rangle$ denotes a reference inner product. Note that in the Euclidean space the reference inner product is defined by $\langle x, y \rangle = x^T y$ for any given vectors $x, y \in \mathbb{R}^n$. The *seminorm* with respect to matrix *A* and the *inner product* $\langle \cdot, \cdot \rangle$, denoted as $||x||_A$, is defined as:

$$\|x\|_A = \|Ax\| = \sqrt{\langle Ax, Ax \rangle}.$$

It is worth noting that if *A* has independent columns this is an induced norm; see [HJ12, Section 5.2] for more discussion on seminorms.

Now, let us briefly discuss some properties of a function. Consider an extended real-valued function $f : \mathbb{R}^n \to \mathbb{R} \cup \{+\infty\}$. The effective domain of the function is defined as:

dom
$$f = \{x \in \mathbb{R}^n : f(x) < +\infty\}$$

To define the notion of relative interior we need to first define the affine hull of a set \mathbb{S} .

Definition 2.2. The *affine hull* of a set $\mathbb{S} \subseteq \mathbb{R}^n$ is given by

aff
$$\mathbb{S} = \{\theta_1 x_1 + \theta_2 x_2 + \dots + \theta_n x_n \mid x_1, \dots, x_n \in \mathbb{S}, \ \theta_1 + \dots + \theta_n = 1, \ \theta_1, \dots, \theta_n \in \mathbb{R}\}.$$

Now we define the relative interior of the set S.

Definition 2.3. The *relative interior* of a set S is defined by

ri
$$\mathbb{S} = \{x \in \text{aff } \mathbb{S} \mid B(x, \epsilon) \cap \text{aff } \mathbb{S} \subseteq \mathbb{S} \text{ for some } \epsilon > 0\},\$$

where $B(x, \epsilon) = \{y \mid ||y - x|| \le \epsilon\}$ is the ball around x with radius ϵ .

In case that the set S is convex, alternatively, $x \in ri(S)$ if and only if, for every $y \in S \setminus \{x\}$, there exists a $z \in S$ such that x is on the line segment connecting y and z.

We end this subsection with the definition of proper functions.

Definition 2.4. A function f is called *proper* if there exists $x \in \mathbb{R}^n$ such that $f(x) < +\infty$.

Definition 2.5. For a set $X \subseteq \mathbb{R}^n$, we denote the *distance function* to the set X by $d_X(x) := \inf_{y \in X} ||y - x||$ for any $x \in \mathbb{R}^n$. The set-valued mapping $\Pi_X(x)$ stands for the *projection* of x on X, that is, $\Pi_X(x) = \{y : ||y - x|| = d_X(x)\}$.

Note that, if *X* is a non-empty closed set, then $\Pi_X(x)$ is non-empty and well-defined.

2.1.1 Convex functions

Convex functions play a pivotal role in optimization and mathematical modeling. Intuitively, a univariate function is considered convex if, for any two points in its domain, the value of the function at any point on the line segment connecting these two points lies below or on the line connecting the function values at those points.

There are some alternative ways of defining a convex function mathematically which are considered interchangeable. One definition which is very intuitive is as follows.

Definition 2.6. Let $\mathbb{D} \subseteq \mathbb{R}^n$ be a convex set. A function $f : \mathbb{D} \to \mathbb{R}$ is *convex* if, for all x, y in its domain and for any λ with $0 \le \lambda \le 1$, the following inequality holds:

$$f(\lambda x + (1 - \lambda)y) \le \lambda f(x) + (1 - \lambda)f(y),$$

where \mathbb{D} is the domain of the function f.

A function is called *strictly convex* if the above inequality strictly holds for $x \neq y$ and $0 < \lambda < 1$. To introduce the other definition we need to define the epigraph of a function.

Definition 2.7. For a function $f : \mathbb{D} \to \mathbb{R}$, the *epigraph* of f is defined by

 $epi f = \{(x, t) \in \mathbb{D} \times \mathbb{R} \mid f(x) \le t\}.$

We now define convex and closed functions using the epigraph of the function.

Definition 2.8. A function *f* is *convex* if its epigraph is a convex set.

Definition 2.9. A function *f* is *closed* if its epigraph is a closed set.

We can characterize lower semi-continuous functions through the definition of closed functions. In other words, a function $f : \mathbb{R}^n \to (-\infty, \infty]$ is *lower semi-continuous* function if and only if it is a closed function; see [Roc97, Theorem 7.1].

2.1.2 Subgradient

Another essential concept utilized in this thesis is the notion of subgradient. To formally define the subgradient, we first need to introduce the concept of dual spaces. Let us consider a vector space \mathbb{E} , and the set of all linear functionals on \mathbb{E} , termed the *dual space*, and denoted as \mathbb{E}^* . The norm in the dual space is called the *dual norm*, and it is defined as follows:

$$||y||_* = \sup_{x \in \mathbb{E}} \{y(x) : ||x|| \le 1\}, \quad y \in \mathbb{E}^*.$$

In this thesis, we focus on utilizing the Euclidean norm. An essential characteristic of the Euclidean norm is its self-duality, expressed as $\|\cdot\| = \|\cdot\|_*$. Additionally, our study centers on the vector space $\mathbb{E} = \mathbb{R}^n$, which possesses a dual space denoted as $\mathbb{E}^* = \mathbb{R}^n$. In this case, any *y* in \mathbb{E}^* may be identified with $x \mapsto \langle x, y \rangle$ for some *y* in \mathbb{R}^n and $\langle \cdot, \cdot \rangle$ the Euclidean inner product on \mathbb{R}^n . In other words, \mathbb{E}^* is isomorphic to \mathbb{R}^n .

This understanding provides us with the necessary tools to present a formal definition of the subgradient.

Definition 2.10. Let $f : \mathbb{R}^n \to (-\infty, +\infty]$ be a proper function. If $x \in \text{dom}(f)$, a vector $g \in \mathbb{R}^n$ is called *subgradient* of f at x if

$$f(y) \ge f(x) + \langle g, y - x \rangle \ \forall y \in \mathbb{R}^n.$$

In a more intuitive sense, this inequality implies that the subgradient at a given point corresponds to an affine (linear) function, which underestimates the original function.

Now we can provide the formal definition of subdifferential.

Definition 2.11. The *subdifferential* of a function f at point x in its domain is the set of all its subgradients, denoted by $\partial f(x)$. In other words,

$$\partial f(x) = \{g \in \mathbb{R}^n : f(y) \ge f(x) + \langle g, y - x \rangle \ \forall y \in \mathbb{R}^n \}.$$

This definition allows us to handle cases where the function may have more than one subgradient associated with the given point. However, when a proper convex function is differentiable and $x \in int(dom(f))$, the set reduces to a singleton, equal to $\{\nabla f(x)\}$; e.g. see [Bec17, Theorem 3.33]. The subdifferential possesses several important properties. For a proper function $f : \mathbb{R}^n \to (-\infty, +\infty]$, its subdifferential at a specific point $x \in \mathbb{R}^n$ is a closed and convex set. Additionally, a proper function is said to be subdifferentiable at a point x if $\partial f(x) \neq \emptyset$, indicating that there exists at least one subgradient at that point. Another significant theorem related to the subdifferential of a function is known as Fermat's optimality condition, which can be stated as follows.

Theorem 2.12. [E.g. Bec17, Theorem 3.63] Let $f : \mathbb{R}^n \to (-\infty, +\infty]$ be a proper convex function. Then, a point x^* is a global minimizer of the function f, i.e., $x^* \in \arg\min_x \{f(x) : x \in \mathbb{R}^n\}$, if and only if $0 \in \partial f(x^*)$.

In other words, Fermat's optimality condition asserts that a point x^* is a global minimizer of the function f if and only if the subdifferential of f at that point contains the zero vector. This condition serves as a fundamental guideline to identify optimal solutions in convex optimization problems.

An important rule which is used in this thesis is the inclusion known as the weak sum rule of subdifferential calculus:

$$\sum_{i=1}^{m} \partial f_i(x) \subseteq \partial \left(\sum_{i=1}^{m} f_i \right)(x),$$

where $f_i : \mathbb{R}^n \to \mathbb{R}$ for all $i \in \{1, ..., m\}$ are proper convex functions. Moreover, equality holds if the intersection of the relative interiors of the domains of the functions is nonempty, i.e., $\bigcap_{i=1}^{m} \operatorname{ri}(\operatorname{dom}(f_i)) \neq \emptyset$.

Now, let us consider the following constrained optimization problem:

$$\inf\{f(x): x \in C\},\tag{2.1}$$

where $f : \mathbb{R}^n \to (-\infty, +\infty]$ is an extended real-valued proper convex function, and the set *C* is a convex set. The necessary and sufficient condition for optimality in this problem can be expressed through the following theorem:

Theorem 2.13. [E.g. Bec17, Theorem 3.68] Let $f : \mathbb{R}^n \to (-\infty, +\infty]$ be an extended real-valued proper convex function, and let C be a convex set where $ri(dom(f)) \cap ri(C) \neq \emptyset$. Then, $x^* \in C$ is an optimal solution of problem (2.1) if and only if there exists $g \in \partial f(x^*)$ such that $\langle g, x - x^* \rangle \geq 0$ for any $x \in C$.

In other words, x^* is an optimal solution of the constrained optimization problem if and only if there exists a subgradient g of f at x^* such that the inner product of g with any direction $(x - x^*)$ in the feasible set C is nonnegative. This condition is a fundamental result that helps identify the optimal solutions in convex constrained optimization problems.

For a comprehensive exploration of further properties related to the subdifferential and subgradient and the formal proofs of the theorems given in this section, we recommend interested readers to consult relevant reference books such as [Bec17].

2.1.3 Conjugate function

Another fundamental concept heavily employed in this thesis is the notion of the conjugate function.

Definition 2.14. Let $f : \mathbb{R}^n \to (-\infty, +\infty]$ be an extended real-valued function. The function $f^* : \mathbb{R}^n \to (-\infty, +\infty]$, defined by

$$f^*(y) = \sup_{x \in \mathbb{R}^n} \{ \langle y, x \rangle - f(x) \}, \ y \in \mathbb{R}^n,$$

is called the *conjugate* of f.

Hence, the conjugate function of f transforms a given point in the dual space \mathbb{R}^n into a real value. The conjugate function plays a pivotal role in various mathematical and optimization analyses. The most important property of the conjugate function is its closeness and convexity, which can be formally stated as follows.

Theorem 2.15. [*E.g.* Bec17, Theorem 4.3] Let $f : \mathbb{R}^n \to (-\infty, +\infty]$ be an extended real-valued function. Then the conjugate function f^* is closed and convex.

Moreover, if f is a proper convex function, its conjugate is also proper. By definition of the conjugate, the following theorem can be derived, which is also known as *Fenchel's inequality*.
Theorem 2.16. [E.g. Bec17, Theorem 4.6] Let $f : \mathbb{R}^n \to (-\infty, +\infty]$ be an extended real-valued proper function. For any $x \in \mathbb{R}^n$ and $y \in \mathbb{R}^n$, we have

$$f(x) + f^*(y) \ge \langle y, x \rangle.$$

One might wonder what happens if we take the conjugate of the conjugate function, which is known as the *biconjugate*, denoted by f^{**} and defined by

$$f^{**}(x) = \sup_{y \in \mathbb{R}^n} \{ \langle x, y \rangle - f^*(y) \}, \quad x \in \mathbb{R}^n.$$

The biconjugate also possesses some interesting properties. For any extended real-valued function f and any x, we have $f(x) \ge f^{**}(x)$. Moreover, if f is a proper, closed, and convex function, then equality holds [Bec17, Theorem 4.8].

We can now establish a connection between the subdifferential and the conjugate function through the conjugate subgradient theorem.

Theorem 2.17. [E.g. Roc97, Theorem 23.5] Let $f : \mathbb{R}^n \to (-\infty, +\infty]$ be a proper and convex function. The following two claims are equivalent for any $x \in \mathbb{R}^n$ and $y \in \mathbb{R}^n$:

(1) $\langle x, y \rangle = f(x) + f^*(y)$.

(2)
$$y \in \partial f(x)$$
.

If, in addition, f is closed, then (1) and (2) are equivalent to:

(3)
$$x \in \partial f^*(y)$$
.

In simpler terms, the conjugate subgradient theorem establishes the equivalence between three statements for a proper and closed convex function f, its conjugate function f^* , and their respective subdifferentials. This theorem provides essential insights into the relationship between the primal and dual spaces in optimization.

Now, let us discuss some calculation rules applicable to conjugate functions. We begin with the summation of separable functions:

Theorem 2.18. [E.g. Bec17, Theorem 4.12] Let $g(x_1, x_2, ..., x_p) = \sum_{i=1}^p f_i(x_i)$ be a seperable function, where $f_i : \mathbb{R}^{n_i} \to (-\infty, +\infty]$ for $i \in 1, 2, ..., p$ are proper functions. Then, we have:

$$g^*(y_1, y_2, \dots, y_p) = \sum_{i=1}^p f_i^*(y_i) \text{ for any } y_i \in \mathbb{R}^{n_i}, i \in \{1, 2, \dots, p\}$$

This theorem allows us to express the conjugate of separable functions as the sum of the conjugates of each function. Next, we present another useful property of conjugate functions.

Theorem 2.19. [*E.g.* Bec17, Theorem 4.14] Let $f : \mathbb{R}^n \to (-\infty, +\infty]$ be an extended real-valued function, and $\alpha \in \mathbb{R}_{++}$.

• If
$$g(x) = \alpha f(x)$$
, then $g^*(y) = \alpha f^*\left(\frac{y}{\alpha}\right)$, for $y \in \mathbb{R}^n$.

• If
$$h(x) = \alpha f\left(\frac{x}{\alpha}\right)$$
, then $h^*(y) = \alpha f^*(y)$, for $y \in \mathbb{R}^n$.

These properties allow us to manipulate the conjugate function when the original function is scaled.

We conclude this section with Fenchel's duality theorem.

Theorem 2.20. [E.g. Roc97, Theorem 31.1] Let f and g be extended real-valued proper convex functions. If the intersection of the relative interiors of the domains of f and g is nonempty, i.e., $ri(dom(f))\cap ri(dom(g)) \neq \emptyset$, then the following relationship holds:

$$\inf_{x\in\mathbb{R}^n} \{f(x) + g(x)\} = \sup_{y\in\mathbb{R}^n} \{-f^*(y) - g^*(-y)\},\$$

and the supremum in the right-hand problem is attained whenever it is finite.

There are several other properties for conjugate functions, which will be discussed in the relevant chapters.

2.1.4 L-smooth functions

In this section, we will review the definition and properties of L-smooth functions, which constitute one of the essential classes of functions studied in this thesis. Let us begin by formally defining L-smooth functions.

Definition 2.21. Let $f : \mathbb{R}^n \to (-\infty, +\infty]$ and open set $\mathbb{D} \subseteq \mathbb{R}^n$ where f is differentiable over \mathbb{D} . f is called *L*-smooth over \mathbb{D} for some L > 0 if it satisfies the following condition:

$$\|\nabla f(x) - \nabla f(y)\| \le L \|x - y\| \text{ for all } x, y \in \mathbb{D}.$$

The constant *L* in the definition is referred to as the smoothness modulus or parameter of the function. The definition of smoothness allows us to observe that if a function *f* is L_1 -smooth, it is also L_2 -smooth for every $L_2 \ge L_1$. Consequently,

when we refer to a function as L-smooth, we specifically mean the smallest possible value of L that satisfies the smoothness condition.

The *descent lemma* is one of the most important properties of *L*-smooth functions.

Lemma 2.22. [Nes03, Lemma 1.2.3] Suppose that $f : \mathbb{R}^n \to (-\infty, \infty]$ is an *L*-smooth function over an open convex set \mathbb{D} for some L > 0. Then, for every $x, y \in \mathbb{D}$, we have:

$$f(y) \le f(x) + \langle \nabla f(x), y - x \rangle + \frac{L}{2} ||x - y||^2$$

In particular, if $y = x - 1/L\nabla f(x)$, then

$$\frac{1}{2L} \|\nabla f(x)\|^2 \le f(x) - f\left(x - \frac{1}{L} \nabla f(x)\right).$$

Additionally, we list some key properties of *L*-smooth functions.

Theorem 2.23. [Nes18, Theorem 2.1.5] Let f be a differentiable convex function over \mathbb{R}^n , and let L > 0. Then all of the following statements are equivalent:

• f is L-smooth.

•
$$f(y) \le f(x) + \langle \nabla f(x), y - x \rangle + \frac{L}{2} ||x - y||^2$$
 for all $x, y \in \mathbb{R}^n$.

•
$$f(y) \ge f(x) + \langle \nabla f(x), y - x \rangle + \frac{1}{2L} \| \nabla f(x) - \nabla f(y) \|^2$$
 for all $x, y \in \mathbb{R}^n$.

•
$$\langle \nabla f(x) - \nabla f(y), x - y \rangle \ge \frac{1}{L} \| \nabla f(x) - \nabla f(y) \|^2$$
 for all $x, y \in \mathbb{R}^n$.

• $f(\lambda x + (1-\lambda)y) \ge \lambda f(x) + (1-\lambda)f(y) - \frac{L}{2}\lambda(1-\lambda)||x-y||^2$ for all $x, y \in \mathbb{R}^n$.

Furthermore, if the Hessian of a function f exists, then the module of smoothness for the function can be defined by spectral radius of the Hessian. In other words, the spectral radius of the Hessian lies within the interval [-L, L]. It's important to note that the spectrum of a Hessian depends on the choice of the underlying inner product. Consequently, the value of L is dependent on the specific inner product used.

The following proposition states a well-known characterization of *L*-smooth functions that follows, e.g., from [Nes03, Lemma 1.2.3], [Nes03, Theorem 2.1.5] and [THG17a, Lemma 3.9].

Proposition 2.24. Let L > 0. The function $f : \mathbb{R}^n \to \mathbb{R}$ is L-smooth and the gradient exists, if and only if it has an L-Lipschitz gradient.

We denote the class of *L*-smooth functions by $\mathcal{F}_{-L,L}(\mathbb{D})$ where $\mathbb{D} \subseteq \mathbb{R}^n$.

2.1.5 Strong convexity

We begin this section by providing the formal definition of strongly convex functions.

Definition 2.25. [Nes18, Definition 2.1.3] Let $\mu \ge 0$. Suppose that $f : \mathbb{R}^n \to \mathbb{R}$ is a continuously differentiable convex function over its domain, and for every $x, y \in \text{dom}(f)$ the inequality

$$f(y) \ge f(x) + \langle \nabla f(x), y - x \rangle + \frac{\mu}{2} ||x - y||^2$$

holds. In this case, f is called a μ -strongly convex function.

This definition implies that a function f is μ -strongly convex if and only if the function $f(\cdot) - \frac{\mu}{2} \| \cdot \|^2$ is convex. Similar to the *L*-smooth case, if f is strongly convex with modulus μ_1 , it is also μ_2 -strongly convex for every $0 < \mu_2 < \mu_1$. Hence, it is essential to find the largest strongly convex modulus. Furthermore, it is evident that a convex function is a 0-strongly convex function.

It is straightforward to observe that if $f : \mathbb{R}^n \to \mathbb{R}$ is a μ -strongly convex function for some $\mu > 0$ and $g : \mathbb{R}^n \to \mathbb{R}$ is a convex function, then the sum f + g is also a μ -strongly convex function.

The following theorem presents some essential characteristics of μ -strongly convex functions.

Theorem 2.26. [E.g. Bec17, Theorem 5.24] Let $f : \mathbb{R}^n \to \mathbb{R}$ be a proper, closed, and convex function, and let $\mu > 0$. The following statements are equivalent:

- *f* is a μ-strongly convex function.
- For any $x \in dom(\partial f)$, $y \in dom(f)$, and $g \in \partial f(x)$, the inequality

$$f(y) \ge f(x) + \langle g, y - x \rangle + \frac{\mu}{2} ||y - x||^2$$

holds.

• For any $x, y \in dom(\partial f)$ and $g_x \in \partial f(x)$, $g_y \in \partial f(y)$, the inequality

$$\langle g_x - g_y, x - y \rangle \ge \mu \|x - y\|^2$$

holds.

One of the most significant characteristics of strongly convex functions is their possession of a unique minimizer. The following theorem formalizes this property.

Theorem 2.27. [E.g. Bec17, Theorem 5.25] If $f : \mathbb{R}^n \to \mathbb{R}$ is a proper, closed, and μ -strongly convex function for some $\mu > 0$, then f has a unique minimizer. Moreover, the following inequality holds for any $x \in \text{dom}(f)$ and x^* , which is the unique minimizer of f:

$$f(x) - f(x^*) \ge \frac{\mu}{2} ||x - x^*||^2$$

Suppose that the function f is twice continuously differentiable and the Hessian of the function is positive definite. Similar to *L*-smooth functions, the modulus of strong convexity of f is equal to the minimum eigenvalue of the Hessian. Therefore, one has

$$\mu I \preceq \nabla^2 f(x) \preceq LI.$$

The connection between strong convexity and *L*-smoothness of a function can be established using conjugate functions. The following theorem presents this connection formally.

Theorem 2.28. [E.g. Bec17, Theorem 5.26] Suppose that $\mu > 0$. Then:

- If $f : \mathbb{R}^n \to \mathbb{R}$ is a $\frac{1}{\mu}$ -smooth convex function, then its conjugate function f^* is a μ -strongly convex function.
- If f : ℝⁿ → ℝ is a proper closed μ-strongly convex function, then its conjugate function f* is a ¹/_μ-smooth function.

The set of functions that are both μ -strongly convex and *L*-smooth is denoted by $\mathcal{F}_{\mu,L}(\mathbb{R}^n)$. Notably, it is worth mentioning that for μ -strongly convex and *L*smooth functions, $\mu \leq L$ always holds. Furthermore, the class of μ -strongly convex functions which are not smooth is denoted by $\mathcal{F}_{\mu,\infty}(\mathbb{R}^n)$.

The definitions of *L*-smoothness and μ -strong convexity can be influenced by the chosen norm, raising the question of whether these definitions can be generalized. To address this concern, Lu et al. introduced the concepts of *relative smoothness* and *relative strong convexity* with respect to a reference function [LFN18]. Inspired by their work, we define *c*-strongly convex functions as follows:

Definition 2.29. Let $f : \mathbb{R}^n \to (-\infty, \infty]$ be a closed proper function, and let $A \in \mathbb{R}^{n \times m}$. We say f is c-strongly convex relative to $\|\cdot\|_A$ if the function $f - \frac{c}{2} \|\cdot\|_A^2$ is convex.

It is worth noting that in Definition 2.29 the value of *c* depends on the seminorm $\|\cdot\|_A$ that we use. Moreover, setting A = I in the definition allows us to deduce the definition of μ -strong convexity. It is noteworthy that any function that is μ -strongly convex is also $\frac{\mu}{\lambda_{\max}(A^TA)}$ -strongly convex with respect to the seminorm $\|\cdot\|_A$. However, the converse is not necessarily true unless *A* has full column rank. For further details on the strong convexity in relation to a given function, we refer the reader to [LFN18, BBT17].

The set of *c*-strongly convex functions relative to the seminorm $\|\cdot\|_A$ on \mathbb{R}^n is denoted as $\mathcal{F}_c^A(\mathbb{R}^n)$.

2.1.6 Non-convex non-smooth functions

If a function f is non-convex and non-smooth, we will also need a more general notion of subgradients than in the convex case.

Definition 2.30. Let $f : \mathbb{R}^n \to \mathbb{R}$ be lower semi-continuous.

• The vector $g \in \mathbb{R}^n$ is called a *regular subgradient* of f at \bar{x} , written $g \in \hat{\partial}_L f(\bar{x})$, if for all x in some neighborhood of \bar{x}

$$f(x) \ge f(\bar{x}) + \langle g, x - \bar{x} \rangle + o(||x - \bar{x}||).$$

• The vector $g \in \mathbb{R}^n$ is called a *general subgradient* of f at \bar{x} , written $g \in \partial_L f(\bar{x})$, if there exist sequences $\{x^i\}$ and $\{g^i\}$ with $g^i \in \hat{\partial}_L f(x^i)$ such that

$$x^i \to \bar{x}, f(x^i) \to f(\bar{x}), g^i \to g \text{ as } i \to \infty.$$

It is worth mentioning that $\hat{\partial}_L f(\bar{x})$ is a closed convex set. In addition, $\partial_L f(\bar{x})$ is also closed but not necessarily convex. Note that when f is closed proper convex, then $\partial f(x) = \hat{\partial}_L f(x) = \partial_L f(x)$ for $x \in \text{dom}(f)$. We refer the interested reader to [RW09, Chapter 8] for more discussions on regular and general subdifferentials.

2.2 Semidefinite programming

In this section we discuss some concepts on *semidefinite programming* (SDP); for more details see [LV12, VB96]. The standard form of semidefinite programming

as used in [LV12] (more common is 'inf' instead of 'sup') is given by

$$p^{*} = \sup_{X} \langle C, X \rangle$$

s.t. $\langle A_{i}, X \rangle = b_{i}, \quad i = 1, \dots, m,$
 $X \succeq 0,$ (2.2)

where matrix *X* is the variable, C, A_1, \ldots, A_m are given symmetric matrices, and $b \in \mathbb{R}^m$ is a given vector. The inner product $\langle \cdot, \cdot \rangle$ denotes the *Frobenius inner product*, that is $\langle A, B \rangle = \text{tr}(AB)$. If there is a positive definite matrix *X* that satisfies all the constraints and $X \succ 0$, then we call the problem strictly feasible; sometimes this is called Slater's condition [BV04, Chapter 5].

Similar to the linear programming setting we define the dual form of the primal problem 2.2, which is given by

$$d^{\star} = \inf_{y} \quad b^{T} y$$

s.t.
$$\sum_{i=1}^{m} y_{i} A_{i} - C \succeq 0, \quad i = 1, \dots, m,$$
$$y \in \mathbb{R}^{m},$$
(2.3)

where *y* is the variable and the constraint $\sum_{i=1}^{m} y_i A_i - C \succeq 0$ is also known as a linear matrix inequality (LMI). Now we provide some results on SDP which we will use in the following chapters.

Lemma 2.31. [E.g. LV12, Lemma 2.1.1] Assume that X is a feasible solution to Problem 2.2 and y is a feasible solution to Problem (2.3). Then we have

- $p^* \leq d^*$, which is known as weak duality.
- If $\langle C, X \rangle = b^T y$ then $p^* = d^* = \langle C, X \rangle = b^T y$.

The difference between the supremum in the primal problem and the infimum in the dual problem is called the *duality gap*, defined as $d^* - p^*$. If there is no duality gap, then we say that *strong duality* holds. The next theorem states this in formal way.

Theorem 2.32. [E.g. DK06, Theorem 2.2] Assume that X is a feasible solution to Problem 2.2 and y is a feasible solution to Problem (2.3). Then we have

- If the problem (2.3) is bounded from below (i.e., d^{*} > -∞) and is strictly feasible, then the primal problem (2.2) attains its supremum and there is no duality gap.
- If the problem (2.2) is bounded from above (i.e., p^{*} < ∞) and is strictly feasible, then the primal problem (2.3) attains its infimum and there is no duality gap.

Another important concept is complementary slackness.

Theorem 2.33. Assume that X is a feasible solution of the primal problem (2.2) and y is a feasible solution of dual problem (2.3). If (X, y) satisfy

$$\left(\sum_{i=1}^m y_i A_i - C\right) X = 0,$$

which is known as the complementary slackness condition, then X is an optimal solution of the primal problem and y is an optimal solution of the dual problem.

Consider vectors $u_1, u_2, \ldots, u_k \in \mathbb{R}^n$ with $k \ge 1$. Define the matrix $X = \text{Gram}(u_1, u_2, \ldots, u_k)$ such that $X_{ij} = u_i^T u_j$ for $i, j \in \{1, \ldots, k\}$. This matrix is referred to as the Gram matrix of u_1, u_2, \ldots, u_k . It is noteworthy that a matrix $X \in \mathbb{S}^k$ is positive semidefinite if and only if it can be expressed as the Gram matrix of some vectors $u_1, u_2, \ldots, u_k \in \mathbb{R}^n$ and some $n \ge k$.

See [DK06] for a comprehensive overview of SDP and [MHA20] for new developments on SDP and its applications. Additionally, it's important to note that the MOSEK toolbox in MATLAB [ApS19] is used to solve the semidefinite programming (SDP) problems that appear in this thesis.

2.3 Interpolation theorems

In this section, our main focus is on interpolation theorems, which play a crucial role in formulating the performance estimation problem, as discussed in Chapter 3. The primary aim is to express a specific class of functions using a finite number of triplets, each containing a point, the function value, and the (sub-)gradient of the function at that point. Let us consider \mathcal{I} be a finite index set, this enables us to determine, based on a finite number of such triplets $\{(x^i; f^i; g^i)\}_{i \in \mathcal{I}}$, whether there is a function with $f(x^i) = f^i$ and $g^i \in \partial f(x^i)$ that belongs to a specific class of functions. The formal definition of interpolability is given by the following definition.

Definition 2.34. A set of triplets $\{(x^i; f^i; g^i)\}_{i \in \mathcal{I}} \subseteq \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n$ is called $\mathcal{F}_{\mu,L}(\mathbb{R}^n)$ interpolable if there is a function $f \in \mathcal{F}_{\mu,L}(\mathbb{R}^n)$ such that for $i \in \mathcal{I}$, $f(x^i) = f^i$ and $g^i \in \partial f(x^i)$.

To illustrate interpolation, let us consider the following example.

Example 2.35. Consider $\{(x^1, x^2, x^3) = (-1, 1.5, 4)\} \subseteq \mathbb{R}^3$, the corresponding function values $\{(f^1, f^2, f^3) = (3.07, 0.52, 1.91)\}$ and the corresponding gradients $\{(g^1, g^2, g^3) = (-1.6, -0.65, 2.8)\} \subseteq \mathbb{R}^3$ as illustrated in Figure 2.1. Our goal is to determine if there exists a μ -strongly convex and L-smooth interpolating function with $\mu = 0.21$ and L = 3.74. If such a function exists, we say that the triplet $\{(x^i; f^i; g^i)\}_{i \in \mathcal{I}}$, where $\mathcal{I} = \{1, 2, 3\}$, is $\mathcal{F}_{\mu, L}(\mathbb{R})$ -interpolable.



Figure 2.1: Three points x^1, x^2, x^3 along with their corresponding function values and gradients

In this example, we find that there does indeed exist a function that satisfies the conditions. It is presented in Figure 2.2 and it is given by

$$f(x) = \begin{cases} 0.7167x^2 + 2.3887 & x \in (-\infty, -1.5) \\ 0.02x^4 - 0.0533x^3 + .16x^2 - 1.04x + 1.8 & x \in [-1.5, 4.5] \\ 0.4944x^2 - 6.3112 & x \in (4.5, \infty). \end{cases}$$



Figure 2.2: A μ -strongly convex and *L*-smooth function, with $\mu = 0.21$ and L = 3.74 modules, that passes through the points with the corresponding function values and gradients shown in Figure 2.1.

Now, we present the interpolation theorems utilized in this thesis. For a comprehensive understanding and detailed proofs, we direct the interested reader to [Tay17], unless specified otherwise.

The following theorem provides *convex interpolation*, which serves as the basis for the other theorems.

Theorem 2.36. [THG17c, Theorem 1] The following statements are equivalent:

- 1. $\{(x^i; f^i; g^i)\}_{i \in \mathcal{T}}$ is $\mathcal{F}_{0,\infty}(\mathbb{R}^n)$ -interpolable.
- *2.* The following inequality holds for all $i, j \in \mathcal{I}$:

$$f^i \ge f^j + \left\langle g^j, x^i - x^j \right\rangle.$$

The proof of the theorem comes from the definition of subgradient. We move on to a more general theorem, which provides necessary and sufficient conditions for the class of μ -strongly convex *L*-smooth functions. To prove these theorems one may use the properties of conjugate functions.

Theorem 2.37. [THG17c, Theorem 4] Let $0 \le \mu < L \le \infty$. The following statements are equivalent:

- 1. $\{(x^i; f^i; g^i)\}_{i \in I}$ is $\mathcal{F}_{\mu,L}(\mathbb{R}^n)$ -interpolable.
- 2. The following inequality holds for all $i, j \in I$:

$$\frac{1}{2\left(1-\frac{\mu}{L}\right)} \left(\frac{1}{L} \left\|g^{i}-g^{j}\right\|^{2} + \mu \left\|x^{i}-x^{j}\right\|^{2} - \frac{2\mu}{L} \left\langle g^{j}-g^{i}, x^{j}-x^{i} \right\rangle \right) \leq f^{i} - f^{j} - \left\langle g^{j}, x^{i}-x^{j} \right\rangle.$$
(2.4)

The proof of this theorem relies on the properties of conjugate functions. Using this theorem, the following corollaries can be deduced.

Corollary 2.38. The following statements are equivalent:

- 1. $\{(x^i; f^i; g^i)\}_{i \in I}$ is $\mathcal{F}_{0,L}(\mathbb{R}^n)$ -interpolable.
- 2. The following inequality holds for all $i, j \in I$:

$$f^{i} \ge f^{j} + \langle g^{j}, x^{i} - x^{j} \rangle + \frac{1}{2L} \|g^{i} - g^{j}\|^{2}.$$

Corollary 2.39. The following statements are equivalent:

- 1. $\{(x^i; f^i; g^i)\}_{i \in I}$ is $\mathcal{F}_{\mu,\infty}(\mathbb{R}^n)$ -interpolable.
- 2. The following inequality holds for all $i, j \in I$:

$$f^{i} \ge f^{j} + \langle g^{j}, x^{i} - x^{j} \rangle + \frac{\mu}{2} ||x^{i} - x^{j}||^{2}.$$

The next theorem gives necessary and sufficient conditions for $\mathcal{F}^{A}_{c,\infty}$ -interpolablity. Analogous to that of [THG17c, Theorem 4] (Theorem 2.37) we have the following interpolation theorem.

Theorem 2.40. Let $c \in [0, \infty)$. The set $\{(x^i; f^i; g^i)\}_{i \in \mathcal{I}} \subseteq \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n$ is $\mathcal{F}^A_{c,\infty}$ -interpolable if and only if for any $i, j \in \mathcal{I}$, we have

$$\frac{c}{2} \left\| x^{i} - x^{j} \right\|_{A}^{2} \le f^{i} - f^{j} - \left\langle g^{j}, x^{i} - x^{j} \right\rangle.$$
(2.5)

Proof. The triple $\{(x^i; f^i; g^i)\}_{i \in \mathcal{I}}$ is $\mathcal{F}^A_{c,\infty}$ -interpolable if and only if the triple $\{(x^i; f^i - \frac{c}{2} ||x^i||_A^2; g^i - cA^T A x^i)\}_{i \in \mathcal{I}}$ is $\mathcal{F}_{0,\infty}$ -interpolable. By Theorem 2.36, $\{(x^i; f^i - \frac{c}{2} ||x^i||_A^2; g^i - cA^T A x^i)\}_{i \in \mathcal{I}}$ is $\mathcal{F}_{0,\infty}$ -interpolable if and only if

$$f^{i} - \frac{c}{2} \|x^{i}\|_{A}^{2} \ge f^{j} - \frac{c}{2} \|x^{j}\|_{A}^{2} - \langle g^{j} - cA^{T}Ax^{j}, x^{i} - x^{j} \rangle$$

which implies the desired inequality.

н		

The other interpolation theorem of interest pertains to the class of non-convex smooth functions.

Theorem 2.41. [THG17a, Theorem 3.10][DS20, Theorem 7 in Appendix] Let $\{(x^i; f^i; g^i)\}_{i \in \mathcal{I}} \subseteq \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n$ with a given finite index set \mathcal{I} and L > 0. There exists an L-smooth function f with

$$f(x^{i}) = f^{i}, \nabla f(x^{i}) = g^{i} \quad \forall i \in \mathcal{I},$$
(2.6)

if and only if

$$\frac{1}{2L} \left\| g^{i} - g^{j} \right\|^{2} - \frac{L}{4} \left\| x^{i} - x^{j} - \frac{1}{L} (g^{i} - g^{j}) \right\|^{2} \le f^{i} - f^{j} - \left\langle g^{j}, x^{i} - x^{j} \right\rangle \quad \forall i, j \in \mathcal{I}.$$
(2.7)

In addition, if $\{(x^i; f^i; g^i)\}_{i \in \mathcal{I}}$ satisfies (2.7), then there exists a *L*-smooth function f for which (2.6) holds and $\min_{x \in \mathbb{R}^n} f(x) = \min_{i \in \mathcal{I}} f_i - \frac{1}{2L} ||g^i||^2$. Moreover, letting $i^* \in \arg\min_{i \in \mathcal{I}} f_i - \frac{1}{2L} ||g^i||^2$, a global minimizer of this function is given by $x^* = x^{i^*} - \frac{1}{L}g^{i^*}$.

Another proof of the first part of the above-mentioned theorem may be found in [Wel73, Theorem 2. Page 148].

Now we provide a theorem that extends the result to a more general case. Analogous to strongly convex functions, we define the concept of 'minimum curvature' with modulus μ for the function f if the function $f(\cdot) - \frac{\mu}{2} || \cdot ||^2$ is convex. Notably, in this case, the minimum curvature parameter is allowed to take negative values, unlike in the convex case where μ can only take non-negative values. The class of functions that exhibit minimum curvature μ and are L-smooth is termed 'hypo-convex functions' and is denoted by $\mathcal{H}_{\mu,L}(\mathbb{R}^n)$. It is worth mentioning that when $\mu = -L$, this class of functions reduces to non-convex L-smooth functions. The following theorem provides the necessary and sufficient conditions for the interpolability of this class of functions.

Theorem 2.42. [*RGP22, Theorem 3.1*] Let $\{(x^i; f^i; g^i)\}_{i \in \mathcal{I}} \subseteq \mathbb{R}^n \times \mathbb{R} \times \mathbb{R}^n$ with a given index set \mathcal{I} and let $L \in (0, \infty]$ and $\mu \in (-\infty, L)$. There exists a function $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$ with

$$f(x^i) = f^i, \nabla f(x^i) = g^i \quad i \in \mathcal{I},$$

if and only if for every $i, j \in \mathcal{I}$

$$\frac{1}{2(1-\frac{\mu}{L})} \left(\frac{1}{L} \left\|g^{i}-g^{j}\right\|^{2} + \mu \left\|x^{i}-x^{j}\right\|^{2} - \frac{2\mu}{L} \left\langle g^{j}-g^{i},x^{j}-x^{i} \right\rangle \right) \leq f^{i} - f^{j} - \left\langle g^{j},x^{i}-x^{j} \right\rangle.$$
(2.8)

Homogeneous quadratic functions have a specific structure that allows us to simplify the interpolation theorem for them, making it worth mentioning. Consider the quadratic function $f(x) := \frac{1}{2}x^TQx$ where $\mu I \leq Q \leq LI$ for some $0 \leq \mu \leq L$. The gradient at a given point x can be calculated as $\nabla f(x) = Qx$. Moreover, the function can be expressed as $f(x) = \frac{1}{2}x^T\nabla f(x)$. Utilizing this representation, we can substitute the function value in the interpolation Theorem 2.37; see [BHG22] for more details. Consequently, the interpolation theorem takes the form:

$$\frac{1}{2\left(1-\frac{\mu}{L}\right)}\left(\frac{1}{L}\left\|g^{i}-g^{j}\right\|^{2}+\mu\left\|x^{i}-x^{j}\right\|^{2}-\frac{2\mu}{L}\left\langle g^{j}-g^{i},x^{j}-x^{i}\right\rangle\right)\leq \frac{1}{2}\left\langle g^{i},x^{i}\right\rangle -\frac{1}{2}\left\langle g^{j},x^{j}\right\rangle -\left\langle g^{j},x^{i}-x^{j}\right\rangle.$$

We conclude this section by providing the following theorem for μ -strongly convex, *L*-smooth functions that are twice continuously differentiable in an open convex set $\mathbb{D} \subseteq \mathbb{R}^n$, denoted by $\mathcal{F}_{\mu,L}(\mathbb{D})$.

Theorem 2.43. [DKGT20] Let $f : \mathbb{D} \to \mathbb{R}$ be twice continuously differentiable, defined on an open convex set \mathbb{D} . The following statements are equivalent:

- 1. $f \in \mathcal{F}_{u,L}(\mathbb{D})$.
- 2. for all $x \in \mathbb{D}$, $\mu I \preceq \nabla^2 f(x) \preceq LI$.
- 3. On the set \mathbb{D} , the functions $f(\cdot) \frac{\mu}{2} \|\cdot\|^2$ and $\frac{L}{2} \|\cdot\|^2 f(\cdot)$ are convex.
- 4. For all $x, y \in \mathbb{D}$

$$\frac{1}{1-\frac{\mu}{L}} \left(\frac{1}{L} \|\nabla f(x) - \nabla f(y)\|^2 + \mu \|x - y\|^2 - \frac{2\mu}{L} \langle \nabla f(y) - \nabla f(x), y - x \rangle \right) \le \langle \nabla f(x) - \nabla f(y), x - y \rangle.$$

By the example given by [Dro20], it is known that in the case that $\mathbb{D} \subsetneq \mathbb{R}^n$ the inequality (2.4) does not necessarily hold. In other words, in case that $\mathbb{D} = \mathbb{R}^n$, the last inequality can be replaced by stronger inequality (2.4). Indeed, finding interpolation theorems over general open convex sets is an open problem; see the discussion after Lemma 3.1 in [DKGT20].

2.4 Conclusion

In this chapter, we have presented some fundamental theorems that will serve as building blocks for the subsequent chapters. Initially, we reviewed key theorems and definitions in convex analysis and smooth functions. Then, we briefly introduced semidefinite programming. Subsequently, we focused on interpolation theorems, that are essential for performance estimation problem—the fundamental tool for conducting our worst-case performance analysis in the upcoming chapters. We ended the chapter by providing some general definition of subgradients which we will use in Chapter 6. A wise man proportions his belief to the evidence.

David Hume

3

Performance estimation problems (PEPs)

Preamble

It is important to know that, in general, optimization problems are considered unsolvable; see [Nes18, Introduction] and [NY83, Chapter 1]. Consequently, optimization algorithms are developed to find at least stationary points for certain classes of functions. Some optimization algorithms are iterative methods, generating feasible points from an initial guess until, hopefully, an optimal solution is reached. These methods differ based on the information used, like the objective function, constraint functions, and derivatives. Some use past data, while others rely on just the information gathered from the current iteration [NW06]. It is important to know how rapidly or with what level of accuracy an algorithm approaches the optimal solution after a certain number of iterations, if it does converge. Therefore, it is important to understand the behaviour of these algorithm for different class of functions.

In this chapter, we present the performance estimation problem (PEP), which serves as the foundation for our subsequent convergence rate analysis. Let us consider the following optimization problem,

$$\inf_{\mathbf{x}\in\mathbb{R}^n}f(\mathbf{x}).\tag{3.1}$$

Here, *f* denotes a lower semi-continuous function, potentially convex and smooth, which is lower-bounded by some $f^* > -\infty$.

In this thesis, we focus on the worst case convergence rate of first-order methods from black-box perspective. This means that we assume the availability of the (sub-)gradient and function value at a given point. To analyze the worst-case convergence of a function, we utilize the performance estimation problem (PEP) — described below — which uses interpolation theorems discussed in Chapter 2.

Generally, expressing the worst-case convergence rate of an algorithm as an optimization problem results in an intractable task. However, PEP provides us with valuable tool to reformulate the problem as a semidefinite problem, offering a tractable problem that is independent of the dimension of the problem (3.1).

In the subsequent sections, we begin by expressing the worst-case convergence rate of an algorithm as a mathematical programming problem. We then transform this problem into a semidefinite programming formulation. Finally, we demonstrate how to derive a convergence rate by solving its Lagrangian dual. For simplicity, in the rest of the chapter we consider that the function f is L-smooth with finite L; for the cases that the function is not necessarily L-smooth we refer the reader to Chapters 8 and 9.

3.1 Worst case formulation of convergence rate of iterative first-order methods

Performance estimation problems were introduced in the seminal paper by Drori and Teboulle [DT14]. We focus on studying the convergence rate of first-order methods from an oracle viewpoint. Studying complexity from the oracle viewpoint was initially introduced by Nemirovski and Yudin in their book [NY83]. This viewpoint involves considering the function value and the (sub-)gradient of the function at a given point, denoted as f^i , g^i , where g^i represents the (sub-)gradient of the function at point x^i .

Let us consider an iterative first-order algorithm \mathcal{M} that starts from the initial point x^0 . In each iteration, first-order algorithms use the function value and the gradient of the points generated in the previous iterations, as well as the initial point x^0 . This combination is known as the *first-order oracle*, denoted by \mathcal{O}_f . In other words, the first-order oracle of function f at point x is defined as $\mathcal{O}_f(x) = \{f(x), \nabla f(x)\}$. Using this notation, we can express the points generated by *black*-

box first-order methods after N iterations as follows.

$$\begin{cases} x^{1} = \mathcal{M}(x^{0}, \mathcal{O}_{f}(x^{0})) \\ x^{2} = \mathcal{M}(x^{0}, \mathcal{O}_{f}(x^{0}), \mathcal{O}_{f}(x^{1})) \\ \vdots \\ x^{N} = \mathcal{M}(x^{0}, \mathcal{O}_{f}(x^{0}), \mathcal{O}_{f}(x^{1}), \dots, \mathcal{O}_{f}(x^{N-1})) \end{cases}$$

To analyze an algorithm we need to determine a performance measure or criterion. Natural and common performance measures which are used in the literature, depending on the problem class, are distance of the point generated by the algorithm in last iterate from the optimal solution, $||x^N - x^*||$, the distance of the function value from optimal value, $f(x^N) - f^*$, and the norm of the gradient, $||\nabla f(x^N)||$, as well as other measures.

Let us consider a performance measure denoted by $\mathcal{P}(\mathcal{O}_f, x^0, x^1, \dots, x^N, x^*)$. In order to analyze an algorithm, we must establish an initial condition for the starting point x^0 to measure the algorithm's performance. A commonly used initial condition is to restrict the distance of the initial point from the optimal solution, given by $||x^0 - x^*|| \leq \Delta$, where Δ represents a predetermined constant. It is readily seen that this is problem-specific and it can be different for different algorithms and different function classes; for more initial conditions see the subsequent chapters.

For the remainder of this chapter, we will focus exclusively on the case where the function class is restricted to smooth convex or smooth strongly convex functions, for the sake of simplicity and illustration. Since the original problem (3.1) is a minimization problem, when evaluating the worst-case convergence rate, we aim to maximize the performance measure to evaluate the worst case scenario. Thus, we can formulate the worst-case convergence of the given algorithm \mathcal{M} after N iterations within a specific class of functions \mathcal{F} as follows.

$$w(\mathcal{F}, \Delta, \mathcal{M}, N, \mathcal{P}) = \sup_{f, x^1, x^2, \dots, x^N, x^*} \mathcal{P}(\mathcal{O}_f, x^0, x^1, \dots, x^N, x^*)$$

s.t. $f \in \mathcal{F}$
 x^* is an optimal solution of f (3.2)
 x^1, x^2, \dots, x^N are generated by the algorithm \mathcal{M}
given x^0 and \mathcal{O}_f
 $||x^0 - x^*|| \leq \Delta$.

This problem is intractable in general because we aim to optimize over the, typically infinite-dimensional, function class. To solve this issue, we use the interpolation theorems which are discussed in Chapter 2. Using these results we can rewrite the problem 3.2 as a finite-dimensional problem.

$$w_{f}(\Delta, \mathcal{M}, N, \mathcal{P}) = \sup_{\{x^{i}; f^{i}; g^{i}\}_{i \in \mathcal{I}} \in (\mathbb{R}^{n} \times \mathbb{R} \times \mathbb{R}^{n})^{N+2}} \mathcal{P}\left(\{x^{i}; f^{i}; g^{i}\}_{i \in \mathcal{I}}\right)$$

s.t. $\{x^{i}; f^{i}; g^{i}\}_{i \in \mathcal{I}}$ are \mathcal{F} -interpolable
i.e. $f^{i} = f(x^{i})$ and $g^{i} = \nabla f(x^{i})$ for $i \in \mathcal{I}$
 x^{*} is an optimal solution of f (3.3)
 $x^{1}, x^{2} \dots, x^{N}$ are generated by the algorithm \mathcal{M}
 $||x^{0} - x^{*}|| \leq \Delta,$

where $\mathcal{I} = \{0, 1, ..., N, \star\}$. Using interpolation theorems that give necessary and sufficient conditions for different class of functions, it can be easily seen that problems (3.2) and (3.3) are equivalent as the only difference is the first constraint. For a comprehensive understanding of this topic, we recommend interested readers to refer to Taylor's thesis on performance estimation problems [Tay17, Chapter 4] and Drori's thesis [Dro14, Chapter 2].

Problem (3.3) is finite-dimensional but still it is not a convex problem due to the non-convex quadratic constraints imposed by interpolation conditions. In the next section, we will provide a Gram matrix reformulation of the problem to make it a semidefinite programming problem.

3.2 Gram matrix reformulation

In this section, our objective is to formulate worst-case convergence rates of iterative first-order methods with a fixed step length as a mathematical programming problem. We start by introducing fixed step first-order methods. In general, *fixed step iterative methods* are characterized by the fact that the current point generated by the algorithm depends only on the starting point and the gradients of the points generated by the algorithm, using a pre-determined step length. Mathematically, the iterative process is expressed as follows.

$$x^{k} = x^{0} - \sum_{i=1}^{k} t_{ki} g^{i-1},$$
(3.4)

where:

- x^0 represents the starting point,
- g^i denotes the gradient of the point x^i , and
- *t_{ki}* are fixed step lengths for the iterations.

One of the most well-known first-order methods is the *gradient descent method with fixed step-size* described in Algorithm 3.1.

Algorithm 3.1 Gradient method with fixed step lengths

Set *N* and $\{t_k\}_{k=1}^N$ (step lengths) and pick $x^0 \in \mathbb{R}^n$. For k = 1, 2, ..., N perform the following step: 1. $x^k = x^{k-1} - t_k \nabla f(x^{k-1})$

For more details about this algorithm see Chapter 4 and Chapter 5. Note that Algorithm 3.1 fits in the general framework (3.4) by setting $t_{kk} = t_k$ and $t_{ki} = 0$ if $i \neq k$.

The function classes $\mathcal{F}_{\mu,L}$ along with the iterative first-order methods, are invariant concerning translations of the domain and shifts in function values, i.e.

$$f \in \mathcal{F}_{u,L}(\mathbb{R}^n) \Longrightarrow x \mapsto f(x+c) + d \in \mathcal{F}_{u,L}(\mathbb{R}^n),$$

where $c \in \mathbb{R}^n$ and $d \in \mathbb{R}$ are fixed.

Remark 3.1. Without loss of generality, we may therefore assume that $f^* = 0$ and x^* is the zero vector. Additionally, when studying unconstrained optimization problems, so in problem 3.3 we can set $x^* = 0$, $g^* = 0$, $f^* = 0$ without loss of generality because of the optimality conditions for the unconstrained problem 3.1.

To proceed with performance estimation for this class of algorithms, we introduce a *Gram matrix*. Consider the following $n \times (N + 2)$ matrix

$$P = [g^0 \ g^1 \ \dots \ g^N \ x^0], \tag{3.5}$$

where g^i represents the gradient of the function at point x^i , and x^0 denotes the starting point. Now, let us define *Gram matrix* $G = \text{Gram}(g^0, g^1, \dots, g^N, x^0) = P^T P \in \mathbb{S}^{N+2}$, where \mathbb{S}^{N+2} denotes the set of symmetric $(N+2) \times (N+2)$ matrices. The elements of *G* can be expressed as follows

$$G = \begin{pmatrix} \|g^{0}\|^{2} & \dots & \langle g^{0}, g^{N} \rangle & \langle g^{0}, x^{0} \rangle \\ \vdots & \ddots & \vdots & \vdots \\ \langle g^{0}, g^{N} \rangle & \dots & \|g^{N}\|^{2} & \langle g^{N}, x^{0} \rangle \\ \langle g^{0}, x^{0} \rangle & \dots & \langle g^{N}, x^{0} \rangle & \|x^{0}\|^{2} \end{pmatrix},$$
(3.6)

it is easily seen that this matrix is independent of the dimension of the x^0 and g^i .

We observe that by using (3.4), it becomes possible to express x^k as a linear combination of x^0 and g^i . This allows for the reformulation of the constraints within the context of problem (3.3) using both the Gram matrix *G* and the function values of f^i .

With this foundation in place, we can now proceed to introduce a semidefinite programming problem to address the performance estimation problem. In this regard, we consider the general case for the performance measure. We consider the performance measure as combination of function value as well as any quadratic function of the variable x^N and the gradient of the function at the last iterate. It can be easily seen that this can be generalized to other cases for example when we consider ergodic convergence rate, but for the simplicity in this part we just consider the most common criterion. In other words, we consider

$$\mathcal{P}\left(\left\{x^{i};f^{i};g^{i}\right\}_{i\in\mathcal{I}}\right) = b(f(x^{N}) - f^{\star}) + c_{1}\|g^{N}\|^{2} + c_{2}\left\|x^{0} - \sum_{i=1}^{N} t_{i}g^{i-1} - x^{\star}\right\|^{2},$$

where b, c_1 and c_2 are some real numbers. By considering this criteria the semidefinite programming problem can be written as

$$w_{f}(\Delta, \mathcal{M}, N, \mathcal{P}) = \sup_{\{G \in \mathbb{S}^{N+2}, f \in \mathbb{R}^{N+1}\}} bf^{N} + \operatorname{tr}(GC)$$
s.t. $f^{j} - f^{i} + \operatorname{tr}(GA_{ij}) \leq 0 \quad \forall i, j \in \mathcal{I}$

$$\operatorname{tr}(GA_{\Delta}) \leq \Delta^{2} \qquad (3.7)$$
 $G \geq 0$

$$\operatorname{rank}(G) \leq n.$$

In this conceptual framework, the matrices denoted as A_{ij} arise from interpolation theorems, while the matrix A_{Δ} is related to the initial conditions. To show how to construct the matrices A_{ij} we use the same notation as used in [Tay17, Section 4.2.3]. Let us define $t_i \in \mathbb{R}^{N+2}$ for $i \in \{0, \ldots, N, \star\}$ by

$$t_i^T = (-t_{i,1}, -t_{i,2}, \dots, -t_{i,i}, 0, \dots, 0, 1), t_{\star}^T = (0, \dots, 0)$$

By this definition and the definition of *P* as in (3.5) one can easily see that $x_i = Pt_i$. Also, define $u_i = e_{i+1} \in \mathbb{R}^{N+2}$ for i = 0, 1, ..., N+1 and $u_* = 0^{N+2}$. For the purpose of easy reference, consider the interpolation constraint given by the interpolation Theorem 2.4,

$$\frac{1}{2\left(1-\frac{\mu}{L}\right)} \left(\frac{1}{L} \left\|g^{i}-g^{j}\right\|^{2} + \mu \left\|x^{i}-x^{j}\right\|^{2} - \frac{2\mu}{L} \left\langle g^{j}-g^{i}, x^{j}-x^{i} \right\rangle \right) \leq f^{i}-f^{j}-\left\langle g^{j}, x^{i}-x^{j} \right\rangle.$$

By the defined Gram matrix (3.6), we can write this as follows.

$$f_{i} \geq f_{j} + \frac{L}{L-\mu} \left(u_{j}^{T}Gt_{i} - u_{j}^{T}Gt_{j} \right) + \frac{1}{2(L-\mu)} \left(u_{i} - u_{j} \right)^{T} G\left(u_{i} - u_{j} \right) + \frac{\mu}{L-\mu} \left(u_{i}^{T}Gt_{j} - u_{i}^{T}Gt_{i} \right) + \frac{L\mu}{2(L-\mu)} \left(t_{i} - t_{j} \right)^{T} G\left(t_{i} - t_{j} \right).$$

Using this inequality, A_{ij} is defined as

$$2A_{ij} = \frac{L}{L-\mu} \left(u_j \left(t_i - t_j \right)^T + \left(t_i - t_j \right) u_j^T \right) + \frac{1}{L-\mu} \left(u_i - u_j \right) \left(u_i - u_j \right)^T + \frac{\mu}{L-\mu} \left(u_i \left(t_j - t_i \right)^T + \left(t_j - t_i \right) u_i^T \right) + \frac{L\mu}{L-\mu} \left(t_i - t_j \right) \left(t_i - t_j \right)^T.$$

The matrix A_{Δ} can be defined similarly as

$$A_{\Delta} = u_{N+1} u_{N+1}^T.$$

Also, the matrix *C* is constructed in the same way using values of c_1 , c_2 , and the recursion formula for x^N .

As it is discussed the initial conditions can be different based on the problem under study. In this context, we primarily focus on introducing the commonly employed initial condition for the sake of simplicity. Subsequent chapters will delve into specific problems, each accompanied by distinct initial conditions related to the particular problems.

If *n* is equal to or greater than N + 2, it is possible to reformulate the problem as a standard semidefinite programming problem [THG17c, Theorem 5]. This reformulation eliminates the rank constraint, effectively transforming the problem into a convex programming problem. Consequently, it can be solved to an optimal solution, if such a solution exists.

$$w_{f}^{\mathrm{sdp}}(\Delta, \mathcal{M}, N, \mathcal{P}) = \sup_{\{G \in \mathbb{S}^{N+2}, f \in \mathbb{R}^{N+1}\}} bf^{N+1} + \mathrm{tr}(GC)$$

s.t. $f^{j} - f^{i} + \mathrm{tr}(GA_{ij}) \leq 0, \quad \forall i, j \in \mathcal{I}$
 $\mathrm{tr}(GA_{\Delta}) \leq \Delta^{2}$ (3.8)
 $G \geq 0,$

where $I = \{0, 1, ..., N + 1, \star\}.$

In this thesis we just consider the case that $n \ge N+2$. As it is shown in Section 2.2, under this condition, the Problems (3.7) and (3.8) attain the same optimal values. Otherwise, when this condition is not met, Problem (3.8) is a relaxation of Problem (3.7) and we get an upper bound for the convergence rate by solving Problem (3.8). The assumption $n \ge N + 2$ is satisfied for many real-world problems. Moreover, from the complexity perspective, one is usually interested in the asymptotic performance as n grows to infinity. For a more comprehensive understanding of how the matrices are constructed to establish problem (3.8), we provide a simple example in Section 3.4; also see subsequent chapters where each chapter is dedicated to a specific problem.

3.3 Dual multipliers

Solving the dual of (3.8) provides us with useful information about the convergence rate of a given algorithm by providing dual multipliers, which help to deduce an analytical proof for the convergence rate. According to the weak duality theorem (see Lemma 2.31), the dual formulation (problem (3.9) below) can yield an upper bound for problem (3.8). As our goal is to find an upper bound on the convergence rate of an algorithm, a solution to the dual problem offers precisely that; see Section 2.2. Within this section, we proceed to present the Lagrangian dual formulation of the problem (3.8) as

$$\inf_{\lambda_{ij},\tau} \tau \Delta^{2}
s.t. \tau A_{\Delta} - C + \sum_{i,j \in \mathcal{I}} \lambda_{ij} A_{ij} \succeq 0
b - \sum_{i,j \in \mathcal{I}} \lambda_{ij} (u_{j} - u_{i}) = 0
\lambda_{ij} \ge 0, \quad \forall i, j \in \mathcal{I}
\tau \ge 0,$$
(3.9)

where $\mathcal{I} = \{0, 1, ..., N + 1, \star\}$, u_i is defined as in Section 3.2, the dual variable τ refers to the constraint related to the initial condition in the primal problem, and dual variable λ_{ij} refers to the constraints related to the interpolation constraints which corresponds to the pairs of the points *i* and *j*. The parameter t_{ij} is step length defined in relation (3.4). It is shown in the next theorem that under the

assumption $t_{i,i} \neq 0$ for i = 1..., N, there is no duality gap between problem (3.8) and (3.9), since the Slater condition holds in this case (see Section 2.2 and [BV04] for more details); for more details see the proof of Theorem 4.7 in [Tay17, Section 4.2.4]. Recall that the assumption $t_{i,i} \neq 0$ means that each generated point depends on the previously generated point.

Theorem 3.2. [THG17c, Theorem 6] If $t_{i,i} \neq 0$ for all $i \in \{1, ..., N\}$ and $0 \leq \mu < L < \infty$, the optimal values of problem (3.8) and (3.9) are the same and finite.

To prove this theorem, Taylor et al. provide a quadratic function that satisfies the constraints of Problem (3.8) with $G \succ 0$. This means that the Slater condition holds and the duality gap between both problems is equal to zero [THG17c]. The question that naturally arises at this point is whether this method exclusively provides us with a numerical solution for a specific set of parameter values. In essence, how can one derive an analytical solution for the algorithm's convergence rate in a general case that can be mathematically verified?

To determine a convergence rate through this method, one must solve problem (3.9) for various parameters that are Δ , N, $t_{i,j}$, L, μ . By using these diverse combinations, one can estimate the optimal values and dual multipliers for the primal problem. Subsequently, applying weak duality, it becomes possible to construct an analytical proof for the convergence rate. To illustrate this procedure we provide a simple example.

3.4 A simple example

In this section, we will demonstrate the proofs of the performance estimation method by applying it to a straightforward example introduced in [THG17c]. Our objective is to determine the worst-case performance of a single iteration of the gradient descent method (see Algorithm 3.1) with a step length of $t = \frac{3}{2L}$ when addressing the unconstrained optimization problem (3.1). We make the assumption that the function f is a smooth convex function with Lipschitz constant L. As performance measure, we will use the metric $f(x^1) - f^*$, along with the initial condition $||x^1 - x^*|| = \Delta$.

The optimal solution to the corresponding dual SDP problem (3.9) associated with this problem is $\frac{L\Delta^2}{8}$, and it is achieved with the dual variables $\lambda_{01} = \lambda_{\star 0} = \lambda_{\star 1} = \frac{1}{2}$ and $\tau = \frac{L}{8}$. Additionally, the optimal positive semidefinite dual slack

matrix is

$$S = \frac{1}{2} \begin{pmatrix} \frac{1}{L} & \frac{1}{L} & \frac{-1}{2} \\ \frac{1}{L} & \frac{1}{L} & \frac{-1}{2} \\ \frac{-1}{2} & \frac{-1}{2} & \frac{L}{4} \end{pmatrix} \succeq 0,$$

which is a rank one matrix. After retrieving the dual multipliers corresponding to the constraints, one can prove the inequality below (3.10) by constructing the following inequality, which can be obtained by multiplication of the corresponding dual variables to the constraints

$$\begin{split} f(x^{1}) - f(x^{\star}) &- \frac{L}{8} \|x^{0} - x^{\star}\|^{2} - \\ \frac{1}{2} \left(f(x^{1}) - f(x^{0}) + \langle \nabla f(x^{1}), x^{0} - x^{1} \rangle + \frac{1}{2L} \| \nabla f(x^{0}) - \nabla f(x^{1}) \|^{2} \right) - \\ \frac{1}{2} \left(f(x^{0}) - f(x^{\star}) + \langle \nabla f(x^{0}), x^{\star} - x^{0} \rangle + \frac{1}{2L} \| \nabla f(x^{0}) - \nabla f(x^{\star}) \|^{2} \right) - \\ \frac{1}{2} \left(f(x^{1}) - f(x^{\star}) + \langle \nabla f(x^{1}), x^{\star} - x^{1} \rangle + \frac{1}{2L} \| \nabla f(x^{1}) - \nabla f(x^{\star}) \|^{2} \right) = \\ - \frac{L}{2} \left\| \frac{1}{2} (x^{0} - x^{\star}) - \frac{\nabla f(x^{0})}{L} - \frac{\nabla f(x^{1})}{L} \right\|^{2} \le 0. \end{split}$$

To verify the validity of the equality, one can set $\nabla f(x^*) = 0$ and through elementary calculus check it. Given that the terms related to the interpolation constraints are negative, we can deduce the following result

$$f(x^{1}) - f(x^{\star}) \le \frac{L}{8} \|x^{0} - x^{\star}\|^{2}.$$
(3.10)

It is important to notice that this example is just for illustration purpose. If the criteria and initial condition are not the same, one cannot deduce a general convergence rate only after studying one iteration of the algorithm. To do so, it is necessary to provide the proof for an arbitrary number of iterations N. Here, for the purpose of illustration we just provided the proof for only one iteration which shows the behaviour of the function value with respect to the distance of the initial point from the optimal solution.

To determine the tightness of this bound, it is necessary to identify a function that achieves this bound following a single iteration of the algorithm. To find such a function, we may need to solve the primal problem (3.8). If the optimal value of the primal problem equals the optimal value of the corresponding dual problem, then by strong duality theorem, we can identify the worst-case function. In the case of this example, the following primal solution is optimal:

$$f^{0} = \frac{L\Delta^{2}}{2}, f^{1} = \frac{L\Delta^{2}}{8}, \text{ and } G = L\Delta^{2} \begin{pmatrix} L & \frac{-L}{2} & 1\\ \frac{-L}{2} & \frac{L}{4} & \frac{-1}{2}\\ 2 & \frac{-1}{2} & L \end{pmatrix} \succeq 0,$$

where *G* is a rank one matrix. Now one can set the values $f(x^0) = \frac{L\Delta^2}{2}$, $f(x^1) = \frac{L\Delta^2}{8}$, $\nabla f(x^0) = L\Delta$, $\nabla f(x^1) = \frac{-L\Delta}{2}$, and $x^0 = \Delta$. This corresponds to the function $f(x) = \frac{L}{2}x^2$. In other words, it can be checked that one iteration of the algorithm over the function $f(x) = \frac{L}{2}x^2$ attains the bound (3.10) if the step length is equal to $\frac{3}{2L}$, i.e. $x^1 = x^0 - \frac{3}{2L}\nabla f(x^0)$ with the starting point $x^0 = \Delta$.

One might wonder about the process of generating a function that achieves the bound and belongs to the function class. Generally, this process is based on an educated guess. Initially, it involves identifying suitable function values, gradients, and points that meet the optimal solution of Problem (3.8). Subsequently, using the properties of subgradients and conjugate functions, it is possible to conjecture a function that belongs to the specified function class and obtains the worst-case convergence rate.

As it is discussed in the previous section, PEP is considered as a computerassisted method for finding worst-case convergence rates. To this end, we formulate the corresponding dual problem. Subsequently, we try to solve the problem using different settings of the parameters. This helps us to guess parametric optimal solutions to the dual problem as well as the optimal dual multipliers. Following this, one needs to verify the proof analytically, similar to the one in the example. Therefore, the computer is just used for finding the dual variables but the proofs are independent of the computer.

3.5 Some recent works on/with PEP

To conclude this chapter, in this section we provide a brief overview on some recent research done by other scholars using PEP without getting into all the details. As mentioned, the performance estimation method is a strong tool in worst-case convergence rate analysis of first-order methods, used by a growing number of scholars to evaluate the convergence rate of different algorithms in different settings, initially introduced in Drori and Teboulle's seminal paper [DT14].

The gradient descent method, originally introduced by Cauchy in 1847 [Cau47], has been a topic of interest for researchers aiming to understand its computational

complexity for different classes of functions. Rotaru et al. studied this method [RGP22]. They extended the result provided in [AdKZ22] to the generalization of $\mathcal{F}_{\mu,L}$, where μ is allowed to take negative values, and proved that the given bound is tight for this class of functions with some step lengths [RGP22].

In the setting of strongly convex functions, the gradient descent method with fixed step-length is studied in [Tay17] and when the line search is implemented in each iteration [dKGT17] proved a tight convergence rate by providing a convex quadratic function that attains the bound. Moreover, the worst case convergence rate for *L*-smooth convex functions with step-length less than $\frac{1}{L}$ is given by [DT14]. Recently, using PEP, Grimmer has shown that for *L*-smooth convex functions instead of considering fixed step-length, taking long steps periodically in the iterations of the gradient descent method improves the convergence rate [Gri23]. In the same spirit, Grimmer et al. [GSW23] introduced long steps for the gradient method and provided a new accelerated convergence rate of the algorithm for smooth convex functions. Also, more recently, Altschuler and Parrilo provided new results on the long step gradient descent method [AP23a, AP23b].

One variation on the gradient method is the coordinate descent method; see Chapter 6 for more details of the algorithm. For the class of smooth convex functions, [KHG23] proposed a method to study the convergence of this type of algorithm.

Using PEP Zamani and Glineur managed to find the tight convergence rate for the subgradient methods considering the last iterate of the algorithm [ZG23]. Fixing the number of iterations *N*, they provide the optimal constant step size based on the convergence rate. The optimal subgradient method introduced in the paper matches the best known lower bound.

Bousselmi et al. extended the PEP framework to analyze linear operators [BHG23]. Using this framework they were able to analyze composed objective function and also the Chambolle-Pock method.

Beyond first-order methods, de Klerk et al. using PEP managed to find worstcase convergence rate for one iteration of Newton method which is a second order method [DKGT20]. This line of work shows that in some cases PEP may be used to find convergence rate of second order methods as well.

Ryu et al. extended performance estimation to find the tight contraction factors for operator splitting methods [RTBG20]. They proposed some interpolation theorems and called their methodology *operator splitting performance estimation*.

Kim and Fessler [KF18a] presented a relaxed PEP to recover the results of fast iterative shrinkage/thresholding algorithm (FISTA), originally proposed in

[BT09], and introduced a generalized version of the algorithm. The same authors in another paper used PEP to generalize the optimized gradient method [KF18b]. In other papers, Kim and Fessler considered the step length of the first-order methods as a variable. In this case, PEP transforms to a bilinear optimization problem to find the best step length that minimizes the worst-case convergence of the algorithm with regard to the criterion. With this, they could manage to find new algorithms for convex smooth optimization problem [KF16, KF21]. To design new first-order algorithms, Gupta et al. [DGVPR23] developed a branch-and-bound performance estimation programming (BnB-PEP) method to efficiently deal with nonconvexity that arises in optimizing a first-order method. Moreover, using this tool, Jang et al. presented a new method called OptISTA which is developed based on FISTA method [JGR23].

Finding Lyapunov functions is of importance in studying first-order methods. Recently, Moucer et al. [MTB23] developed a systematic way using PEP to find and verify Lyapunov functions. PEP is also extended by Upadhyaya et al. to find quadratic Lyapunov inequalities to derive tight convergence rates for some classes of first-order methods [UBTG23].

Under negative comonotonicity assumptions Gorbunov et al. studied some algorithms for variational inequality and min-max optimization using PEP [GTHG23]. Moreover, Gorbunov et al. using this method could manage to derive the known convergence rate for the Past Extragradient (PEG) method with fewer assumptions [GTG22], in fact they removed the assumptions on the Lipschitz Jacobian which was used to prove the convergence rate. Moreover, for maximally monotone operators, Kim developed an accelerated proximal point using PEP [Kim21]. On the other hand, Gu and Yang could manage to find tight ergodic convergence rate of a proximal point algorithm for monotone variational inequalities [GY22].

Furthermore, performance estimation method and integral quadratic constraints (IQC) formulation is combined to find new algorithms by Lessard et al. [LRP16].

To make the use of this computer-assisted method easy, in two papers the authors developed a package (toolbox) to be used in Python (MATLAB) [GMG⁺22, THG17b], called PEPit (PESTO). Using these packages the user can easily numerically compute the complexity of some of the well-known first-order methods for different settings of the parameters.

There are some other works that are done using PEP; e.g. [PR23, DT20, DTdB21, TD23, THG18, TVSL18]. This list is not exhaustive, though.

Men pass away, but their deeds abide.

Augustin-Louis Cauchy

4

The exact worst-case convergence rate of the gradient method with fixed step lengths for L-smooth functions

Preamble

In this chapter, we study the convergence rate of the gradient (also known as steepest descent) method with fixed step lengths, which is introduced earlier in Algorithm 3.1, for finding a stationary point of an *L*-smooth function. We establish a new convergence rate, and show that the bound may be exact in some cases, in particular when all step lengths lie in the interval (0, 1/L]. In addition, we derive an optimal step length with respect to the new bound. This chapter is based on the paper [AdKZ22], except for Section 4.4 that deals with extensions of *L*-smooth functions.

4.1 Introduction

We consider the non-convex unconstrained optimization problem

$$\inf_{x\in\mathbb{R}^n}f(x),\tag{4.1}$$

where $f : \mathbb{R}^n \to \mathbb{R}$ is bounded from below, and let a real number f^* denote a lower bound for problem (4.1). In addition, we assume throughout the chapter that f has an *L*-Lipschitz gradient, that is, f satisfies the conditions in Definition 2.21 for some (known) Lipschitz constant L > 0.

Problem (4.1) arises naturally in many applications including machine learning, signal and image processing, to name but a few [BCN18, JK⁺17]. One of the historic solution methods for problem (4.1) is the gradient method, proposed by Cauchy in 1847 [Cau47].

The gradient method with fixed step lengths may be described as follows.

Algorithm 4.1 Gradient method with fixed step lengths	
Set N and $\{t_k\}_{k=1}^N$ (step lengths) and pick $x^0 \in \mathbb{R}^n$.	
For $k = 1, 2,, N$ perform the following step:	
1. $x^k = x^{k-1} - t_k \nabla f(x^{k-1})$	

Nesterov [Nes03, page 28] gives the following convergence rate (to a stationary point) for Algorithm 4.1 when $t_k \in (0, \frac{2}{L}), k \in \{1, ..., N\}$:

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \le \left(\frac{f(x^0) - f^*}{\left(\sum_{k=1}^N t_k (1 - \frac{1}{2} L t_k) \right) + \frac{1}{2L}} \right)^{1/2}$$

In the special case $t_k = \frac{1}{L}$, $k \in \{1, ..., N\}$, the last bound becomes

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \le \left(\frac{2L(f(x^0) - f^*)}{N+1} \right)^{1/2}$$

By employing the performance estimation method, Taylor [Tay17, page 190], without giving a proof, states the following convergence rate

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \le \left(\frac{4L(f(x^0) - f^*)}{3N} \right)^{1/2}, \tag{4.2}$$

for $t_k = \frac{1}{L}$, $k \in \{1, ..., N\}$. Drori and Shamir [DS20, Corollary 1 in Appendix] considers the case that all step lengths are smaller than $\frac{1}{L}$, and proves the following convergence rate

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \le \left(\frac{4(f(x^0) - f^*)}{\sum_{k=1}^N t_k (4 - Lt_k)} \right)^{1/2}.$$
(4.3)

It can be observed that when the step lengths are the same for each iteration and tend to $\frac{1}{L}$, the bound (4.3) reduces to Taylor's convergence rate.

In this chapter, we investigate the convergence rate of Algorithm 4.1 further. By using the performance estimation method, we provide a converge rate, which is tighter than all aforementioned bounds. For example, as a part of our main result in Theorem 4.3, we improve on (4.3) by showing, for any choice of $t_k \in (0, \sqrt{3}/L)$ ($k \in \{1, ..., N\}$), that

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \le \left(\frac{4(f(x^0) - f^*)}{\sum_{k=1}^N \min(-L^2 t_k^3 + 4t_k, -Lt_k^2 + 4t_k) + \frac{2}{L}} \right)^{1/2}.$$
(4.4)

As a consequence, we also prove and improve on (4.2) by showing, in the special case where all $t_k = 1/L$ ($k \in \{1, ..., N\}$), that

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \le \left(\frac{4L(f(x^0) - f^*)}{3N + 2} \right)^{1/2}.$$

In addition, we construct an *L*-smooth function that attains the given bound in Theorem 4.3 for certain step lengths. We also propose an optimal step length that minimizes the right-hand-side of the bound (4.4), namely $t_k = \frac{\sqrt{4/3}}{L}$ for all $k \in \{1, ..., N\}$.

Outline

The chapter is organized as follows. We describe the performance estimation technique for this specific problem in Section 4.2. In Section 4.3, we study the convergence rate by using performance estimation. Section 4.4 is dedicated to study of extension of L-smooth functions. Finally, we conclude the chapter with a conjecture.

4.2 Performance estimation

As it is mentioned in Chapter 3, computation of the worst-case convergence rate for a given iterative method and a given class of functions is an infinite-dimensional optimization problem; for more details of the performance estimation see Chapter 3.

Similar to problem (P) in [DT14], the worst-case convergence rate of Algorithm 4.1 may be formulated as the following abstract optimization problem,

$$\max\left(\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \right)$$

s.t. $f(x^0) - f^* \le \Delta$
 x^N, x^{N-1}, \dots, x^1 are generated by Algorithm 4.1 w.r.t. f, x^0 (4.5)
 $f(x) \ge f^* \quad \forall x \in \mathbb{R}^n$
 $f \in \mathcal{F}_{-L,L}(\mathbb{R}^n)$
 $x^0 \in \mathbb{R}^n$,

where $\Delta \ge 0$ denote the difference between the given lower bound, f^* , and the value of f at the starting point. In problem (4.5), f and x^0 are decision variables. This is an infinite-dimensional optimization problem with infinite number of constraints, and consequently intractable in general. In what follows, we provide a semidefinite programming relaxation for the problem.

The following well-known result is a fundamental property of gradient descent for *L*-smooth functions, if the step length 1/L is used.

Proposition 4.1. [*Nes03, page 26*] If $f : \mathbb{R}^n \to \mathbb{R}$ is *L*-smooth, and $x \in \mathbb{R}^n$, then

$$f\left(x - \frac{1}{L}\nabla f(x)\right) \le f(x) - \frac{1}{2L} \|\nabla f(x)\|^2.$$

Using this and Theorem 2.41, we will formulate problem (4.5) as a finite dimensional optimization problem.

$$\max \left(\min_{0 \le k \le N} \left\| g^k \right\| \right)$$

s.t. $\frac{1}{2L} \left\| g^i - g^j \right\|^2 - \frac{L}{4} \left\| x^i - x^j - \frac{1}{L} (g^i - g^j) \right\|^2 \le f^i - f^j - \langle g^j, x^i - x^j \rangle \ i, j \in \{0, \dots, N\}$
 $x^k = x^{k-1} - t_k g^{k-1} \ k \in \{1, \dots, N\}$
 $f^k \ge f^* \ k \in \{0, \dots, N\}$
 $f^0 - f^* \le \Delta.$ (4.6)

In the above formulation, x^k , g^k , f^k , $k \in \{0, ..., N\}$, are decision variables. Note that in the above formulation, the constraints $f(x) \ge f^*$ for each $x \in \mathbb{R}^n$ are replaced by the weaker condition $f^k \ge f^*$, $k \in \{0, ..., N\}$. Therefore, the optimal value of (4.5) and (4.6) may not be equal in general. However, if an optimal solution of problem (4.6) satisfies $f^* = \min_{0 \le k \le N} f^k - \frac{1}{2L} ||g^k||^2$, then the formulation will be exact; see the second part of Theorem 2.41. By Proposition 4.1, we have $f(x) - \frac{1}{2L} ||\nabla f(x)||^2 \ge f^*$ for $x \in \mathbb{R}^n$. Hence, we replace the constraint $f^k \ge f^*$ by $f^k - \frac{1}{2L} ||g^k||^2 \ge f^*$ and consider the following problem:

$$\max\left(\min_{0 \le k \le N} \left\|g^{k}\right\|\right)$$

s.t. $\frac{1}{2L} \left\|g^{i} - g^{j}\right\|^{2} - \frac{L}{4} \left\|x^{i} - x^{j} - \frac{1}{L}(g^{i} - g^{j})\right\|^{2} \le f^{i} - f^{j} - \langle g^{j}, x^{i} - x^{j} \rangle \quad i, j \in \{0, \dots, N-1, N\}$
 $x^{k} = x^{k-1} - t_{k}g^{k-1} \quad k \in \{1, \dots, N\}$
 $f^{k} - \frac{1}{2L} \|g^{k}\|^{2} - f^{*} \ge 0 \quad k \in \{0, \dots, N\}$
 $f^{0} - f^{*} \le \Delta.$
(4.7)

From the constraint $x^k = x^{k-1} - t_k g^{k-1}$, we get $x^i = x^0 + \sum_{k=0}^{i-1} t_{k+1} g^k$, $i \in \{1, ..., N\}$. By using this relation to eliminate the x^i ($i \in \{1, ..., N\}$), problem (4.7) may be written as follows:

 $\max \ell$

$$s.t. f^{i} - f^{j} - \frac{1}{2L} \left\| g^{i} - g^{j} \right\|^{2} + \frac{L}{4} \left\| -\sum_{k=j}^{i-1} t_{k+1} g^{k} + \frac{1}{L} (g^{i} - g^{j}) \right\|^{2} + \left\langle g^{j}, \sum_{k=j}^{i-1} t_{k+1} g^{k} \right\rangle \ge 0 \quad i > j$$

$$f^{i} - f^{j} - \frac{1}{2L} \left\| g^{i} - g^{j} \right\|^{2} + \frac{L}{4} \left\| \sum_{k=i}^{j-1} t_{k+1} g^{k} - \frac{1}{L} (g^{i} - g^{j}) \right\|^{2} - \left\langle g^{j}, \sum_{k=i}^{j-1} t_{k+1} g^{k} \right\rangle \ge 0 \quad i < j$$

$$f^{k} - \frac{1}{2L} \left\| g^{k} \right\|^{2} - f^{*} \ge 0 \quad k \in \{0, \dots, N\}$$

$$f^{*} - f^{0} + \Delta \ge 0$$

$$\left\| g^{k} \right\|^{2} - \ell \ge 0 \quad k \in \{0, \dots, N\},$$

$$(4.8)$$

where ℓ is an auxiliary variable to convert problem (4.7) into a quadratic program. Problem (4.8) is a non-convex quadratic program with quadratic constraints. In the following proposition, we show that the optimal values of problems (4.5) and (4.7) (or equivalently problem (4.8)) are the same for step lengths in the interval (0, $\frac{2}{L}$).

Proposition 4.2. If $t_k \in (0, \frac{2}{L})$, $k \in \{1, ..., N\}$, then problems (4.5) and (4.7) (or equivalently problem (4.8)) share the same optimal value.

Proof. Clearly, problem (4.7) is a relaxation of problem (4.5). Therefore, we only need to show that, for any feasible solution of (4.7), say $\{(\bar{x}^i; \bar{g}^i; \bar{f}^i)\}_0^N$, there exists an *L*-smooth function *f* with

$$f(\bar{x}^i) = \bar{f}^i, \ \nabla f(\bar{x}^i) = \bar{g}^i, \quad 0 \le i \le N,$$

and $\min_{x \in \mathbb{R}^n} f(x) \ge f^*$. The existence such of a function follows from Theorem 2.41, as all assumptions of Theorem 2.41 are satisfied.

To obtain a tractable form of problem (4.8), we relax it to a semidefinite program, similar to that of Chapter 3. To this end, we define the $(N + 1) \times (N + 1)$ positive semidefinite matrix *G* as,

$$G = \begin{pmatrix} \begin{pmatrix} g^0 \end{pmatrix}^T \\ \vdots \\ \begin{pmatrix} g^N \end{pmatrix}^T \end{pmatrix} \begin{pmatrix} g^0 & \dots & g^N \end{pmatrix} = \begin{pmatrix} \| g^0 \|^2 & \dots & \langle g^0, g^N \rangle \\ \vdots & \ddots & \vdots \\ \langle g^0, g^N \rangle & \dots & \| g^N \|^2 \end{pmatrix}.$$

We may now formulate the following semidefinite program,

$$\max \ell \text{s.t. } f^{i} - f^{j} + \operatorname{tr}(A^{ij}G) \ge 0 \quad i \ne j \in \{0, \dots, N\} f^{k} - \frac{1}{2L}G_{kk} - f^{*} \ge 0 \quad k \in \{0, \dots, N\} f^{*} - f^{1} + \Delta \ge 0 G_{kk} - \ell \ge 0 \quad k \in \{0, \dots, N\} G \ge 0,$$
 (4.9)

where the $(N+1)\times(N+1)$ matrices A^{ij} , $i \neq j \in \{0, ..., N\}$, are formed according to the constraints (4.8), and $G, \ell, f^i, i \in \{0, ..., N\}$, are decision variables. Problem (4.9) is a relaxation of (4.8), but if $n \ge N + 1$ the relaxation is exact, that is the optimal values of (4.8) and (4.9) are the same. Indeed, if $n \ge N + 1$, and *G* is a feasible matrix in (4.9), then *G* is the Gram matrix of N + 1 vectors in \mathbb{R}^n , and these vectors may be identified with g^0, \ldots, g^N ; see Chapter 3 for more details.

4.3 Worst-case convergence rate

In this section, we investigate the convergence rate of gradient method with fixed step lengths. The next theorem gives the worst-case convergence rate of Algorithm 4.1 to a stationary point of an *L*-smooth function. The technique of the proof, as

is usual for SDP performance estimation, is to use weak duality. In particular, we will in fact construct a feasible solution to the dual SDP problem of (4.9), and thus derive an upper bound for problem (4.8).

In practice, this dual feasible solution is constructed in a computer-assisted manner, by solving the primal and dual SDP problems for different fixed values of the parameters, and subsequently guessing the values of the dual multipliers. In the proof of Theorem 4.3, we simply verify that these 'guesses' are correct.

Theorem 4.3. Let $t_k \in (0, \frac{\sqrt{3}}{L})$ for $k \in \{1, ..., N\}$. Consider N iterations of Algorithm 4.1 with step lengths t_k ($k \in \{1, ..., N\}$), applied to some L-smooth function f with minimum value f^* , with the starting point x^0 satisfying $f(x^0) - f^* \leq \Delta$, for some given $\Delta > 0$.

Then, if x^1, \ldots, x^N denote the iterates of Algorithm 4.1, one has

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \le \left(\frac{4\Delta}{\sum_{k=1}^N \min(-L^2 t_k^3 + 4t_k, -Lt_k^2 + 4t_k) + \frac{2}{L}} \right)^{1/2}.$$
 (4.10)

In particular, if $t_k = \frac{\sqrt{4/3}}{L}$ for $k \in \{1, \dots, N\}$, we get

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \le \left(\frac{6\sqrt{3}L(f(x^0) - f^*)}{8N + 3\sqrt{3}} \right)^{1/2}.$$
(4.11)

Similarly, if $t_k = \frac{1}{L}$ for $k \in \{1, \dots, N\}$, one has

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| \le \left(\frac{4L(f(x^0) - f^*)}{3N + 2} \right)^{1/2}.$$
(4.12)

Proof. Let *U* denote the square of the right-side of inequality (4.10) and let $B = \frac{U}{\Delta}$. To establish this bound, we show that *U* is an upper bound for problem (4.8). Consider the feasible point $(\{g^k; f^k\}_0^N; \ell)$ for problem (4.8). Suppose that

$$a_k = \frac{B}{2} \max\{2, t_k L + 1\} \quad k \in \{1, \dots, N\}$$

In addition, we define σ_1 and σ_k , respectively, as follows:

$$\sigma_{1} = \frac{B}{4} \min \left\{ -Lt_{1}^{2} + 3t_{1}, -L^{2}t_{1}^{3} + 3t_{1} \right\},\$$

$$\sigma_{k} = \frac{B}{4} \min \left\{ -Lt_{k}^{2} + 3t_{k} + t_{k-1}, -L^{2}t_{k}^{3} + 3t_{k} + t_{k-1} \right\} \quad k \in \{2, \dots, N\},\$$

and $\sigma_{N+1} = 1 - \sum_{k=1}^{N} \sigma_k = \frac{B}{4L}(2 + Lt_N)$. As $t_k \in (0, \frac{\sqrt{3}}{L})$ for $k \in \{1, \dots, N\}$, the σ_k 's will be non-negative. It is seen that

$$\sigma_k + (2\alpha_k - B)\frac{Lt_k^2}{4} - \frac{Bt_k}{2} = \frac{B}{4}(t_k + t_{k-1}) \quad k \in \{2, \dots, N\}.$$

By using the last equality, one may verify directly through elementary algebra that

$$\begin{split} \ell - U + \sum_{k=1}^{N+1} \sigma_k \left(\left\| g^{k-1} \right\|^2 - \ell \right) + B \left(f^* - f^0 + \Delta \right) + B \left(f^N - \frac{1}{2L} \| g^N \|^2 - f^* \right) \\ + \sum_{k=1}^N \alpha_k \left(f^{k-1} - f^k - \frac{1}{2L} \| g^{k-1} - g^k \|^2 + \frac{L}{4} \| t_k g^{k-1} - \frac{1}{L} \left(g^{k-1} - g^k \right) \|^2 \\ - \left\langle g^k, t_k g^{k-1} \right\rangle \right) + \sum_{k=1}^N (\alpha_k - B) \left(f^k - f^{k-1} - \frac{1}{2L} \| g^k - g^{k-1} \|^2 \\ + \frac{L}{4} \| - t_k g^{k-1} - \frac{1}{L} \left(g^k - g^{k-1} \right) \|^2 - \left\langle g^{k-1}, -t_k g^{k-1} \right\rangle \right) = \frac{-(2\alpha_1 - B)}{4L} \| g^0 - g^1 \|^2 \\ + \frac{Bt_1}{4} \| g^0 \|^2 - \frac{Bt_1}{2} \left\langle g^0, g^1 \right\rangle + \frac{Bt_N}{4} \| g^N \|^2 \\ + \sum_{k=2}^N \left(\frac{-(2\alpha_k - B)}{4L} \| g^{k-1} - g^k \|^2 + \frac{B(t_k + t_{k-1})}{4} \| g^{k-1} \|^2 - \frac{Bt_k}{2} \left\langle g^{k-1}, g^k \right\rangle \right) = \\ - \sum_{k=1}^N Q_k, \end{split}$$

where

$$Q_{k} = \begin{cases} \frac{B}{4} \left(\frac{1}{L} - t_{k}\right) \left\| g^{k-1} - g^{k} \right\|^{2} & t_{k} < \frac{1}{L} \\ 0 & t_{k} \ge \frac{1}{L} \end{cases}$$

Since $\sum_{k=1}^{N} Q_k$ is a non-negative quadratic function and the given dual multipliers are non-negative, we have $\ell \leq U$ for any feasible solution of (4.8).

The special step length $t_k = \frac{\sqrt{4/3}}{L}$ for $k \in \{1, ..., N\}$ used to obtain (4.11) will be motivated later in Theorem 4.5. Note that (4.12) gives a formal proof (with a small improvement) of the bound claimed by Taylor [Tay17, page 190]; see (4.2).

An important question concerning the bound (4.10) is its difference with the optimal value of (4.5). It is known that the lower bound for Algorithm 4.1 is of the order $\Omega\left(\frac{1}{\sqrt{N}}\right)$ [CGT10, CDHS20]. In what follows, we establish that the bound (4.10) is exact in some cases.

Proposition 4.4. The value

$$\left(\frac{4\Delta}{\sum_{k=1}^{N}\min\left(-L^{2}t_{k}^{3}+4t_{k},-Lt_{k}^{2}+4t_{k}\right)+\frac{2}{L}}\right)^{1/2}$$

is the optimal value of (4.5) when all step lengths satisfy $t_k \in \{0, \frac{1}{L}\}, k \in \{1, \dots, N\}$.
Proof. It suffices for a given N to demonstrate an L-smooth function f and a point x^1 such that

$$\min_{0 \le k \le N} \left\| \nabla f(x^k) \right\| = \left(\frac{4\Delta}{\sum_{k=1}^N \min\left(-L^2 t_k^3 + 4t_k, -L t_k^2 + 4t_k \right) + \frac{2}{L}} \right)^{1/2}.$$
 (4.13)

Suppose now that $t_k \in (0, \frac{1}{L}], k \in \{1, ..., N\}$, and *U* denotes the right-hand-side of equality (4.13). We set $t_{N+1} = \frac{1}{L}$. Let

$$l_{i} = U\left(\sum_{k=i}^{N} t_{k+1}\right), \quad f^{i} = \Delta - \frac{U^{2}}{4}\left(\sum_{k=1}^{i} -Lt_{k}^{2} + 4t_{k}\right) \qquad i \in \{0, \dots, N\},$$

and $l_{N+2} = 0$. By elementary calculus, one can check that the function $f : \mathbb{R} \to \mathbb{R}$ given by

$$f(x) = \begin{cases} \frac{L}{2}(x-l_0)^2 + U(x-l_0) + f^0 & x \in \left[\frac{1}{2}(l_0+l_1), \infty\right) \\ \frac{-L}{2}(x-l_i)^2 + U(x-l_i) + f^i & x \in \left[l_i, \frac{1}{2}(l_{i-1}+l_i)\right] \\ \frac{L}{2}(x-l_i)^2 + U(x-l_i) + f^i & x \in \left[\frac{1}{2}(l_i+l_{i+1}), l_i\right] \\ \frac{L}{2}x^2 & x \in \left(-\infty, \frac{1}{2}l_N\right] \end{cases}$$
(4.14)

for $i \in \{1, ..., N\}$, is *L*-smooth with the optimal value $f^* = 0$ and the optimal solution $x^* = 0$. In addition, we have equality (4.13) for $x^1 = l_1$. Indeed,

$$\begin{aligned} x^{i} &= l_{i} & i \in \{0, \dots, N\} \\ \nabla f(x^{i}) &= U & i \in \{0, \dots, N\} \\ f(x^{i}) &= f^{i} & i \in \{0, \dots, N\}. \end{aligned}$$

Figure 4.1 represents the plot of function *f* as constructed in the proof of Proposition 4.4 for different parameters and the fixed step length $t_k = \frac{1}{L}$ for all *k*.

Note that, though we have only shown the exactness of the bound (4.10) for step lengths in the interval $(0, \frac{1}{L}]$, we also conjecture that the bound (4.10) is in fact exact for all step lengths in the interval $(0, \frac{\sqrt{3}}{L})$.

By minimizing the right-hand-side of (4.10), the next theorem gives the 'optimal' step lengths with respect to the bound.

Theorem 4.5. Let f be an L-smooth function. Then the optimal step size for gradient method with respect to bound (4.10) is given by

$$t_k = \frac{\sqrt{\frac{4}{3}}}{L} \quad \forall k \in \{1, \dots, N\},$$



Figure 4.1: Plot of the function f in (4.14) for different parameters and $t_k = \frac{1}{L}$. (Dotted lines denote the endpoints of intervals.)

provided that $t_k \in (0, \frac{\sqrt{3}}{L})$ for all $k \in \{1, \dots, N\}$.

Proof. We minimize the right-hand-side of (4.10), that is

$$\min_{t_k \in \{0, \frac{\sqrt{3}}{L}\}} \left(\frac{4\Delta}{\sum_{k=1}^N \min(-L^2 t_k^3 + 4t_k, -L t_k^2 + 4t_k) + \frac{2}{L}} \right)^{1/2},$$

which is equivalent to maximizing

$$\max_{t \in \left(0, \frac{\sqrt{3}}{L}\right)^{N}} H(t) := \sum_{k=1}^{N} \min\left(-L^{2}t_{k}^{3} + 4t_{k}, -Lt_{k}^{2} + 4t_{k}\right).$$

Since *H* is a strictly concave function on $\left(0, \frac{\sqrt{3}}{L}\right)^N$ and at \bar{t} given by

$$\bar{t}_k = \frac{\sqrt{\frac{4}{3}}}{L} \quad \forall k \in \{1, \dots, N\},$$

we have $\nabla H(\bar{t}) = 0$, which shows that \bar{t} is the unique maximum solution of H over $\left(0, \frac{\sqrt{3}}{L}\right)^N$.

The step length $\frac{1}{L}$ commonly is regarded as the optimal step length in the literature; see [Nes03, Chapter 1]. Due to the example introduced in (4.14), we see that the worst-case convergence rate for the step length $\frac{1}{L}$ cannot be better than $\left(\frac{4L(f(x^0)-f^*)}{3N+2}\right)^{1/2}$. By our analysis, it follows that, for the step length $\frac{\sqrt{\frac{4}{3}}}{L}$, we get the convergence rate (4.11), which is better than $\left(\frac{4L(f(x^0)-f^*)}{3N+2}\right)^{1/2}$, since the constant in the bound improves from ca. $\frac{4}{3} \approx 1.333$ to $\frac{6\sqrt{3}}{8} \approx 1.299$.

In the next section we provide an upperbound on the module of smoothness of a function that is the extension of an *L*-smooth function over a convex set $\mathbb{D} \subseteq \mathbb{R}^n$.

4.4 Extension of L-smooth functions

In this section, we try to extend any *L*-smooth function on $\mathbb{D} \subseteq \mathbb{R}^n$ to an \overline{L} -smooth function on \mathbb{R}^n . Results of this type are usually called *extension theorems*, with the most famous extension theorem being due to Whitney [Whi34], known as *Whitney extension theorem*. This allows us to extend our results to any $\mathbb{D} \subseteq \mathbb{R}^n$ rather than \mathbb{R}^n .

Let $\mathbb{D} \subseteq \mathbb{R}^n$ be a given set and let $\alpha : \mathbb{D} \to \mathbb{R}$ and $\nu : \mathbb{D} \to \mathbb{R}^n$. In this section, we list some key results concerning the existence of an *L*-smooth function $f : \mathbb{R}^n \to \mathbb{R}$ with

$$f(x) = \alpha(x), \ \nabla f(x) = \nu(x), \ x \in \mathbb{D}.$$

Before we get to the results, we need to introduce the constant $\Gamma(\mathbb{D}, (\alpha, \nu))$, which is defined as

$$\Gamma(\mathbb{D}, (\alpha, \nu)) = \sup_{s_1, s_2 \in \mathbb{D}, s_1 \neq s_2} \left(\sqrt{A_{s_1 s_2}^2 + B_{s_1 s_2}^2} + |A_{s_1 s_2}| \right)$$
(4.15)

where

$$A_{s_1s_2} = \frac{2(\alpha(s_1) - \alpha(s_2)) + \langle \nu(s_1) + \nu(s_2), s_2 - s_1 \rangle}{\|s_1 - s_2\|^2}, \ B_{s_1s_2} = \frac{\|\nu(s_1) - \nu(s_2)\|}{\|s_1 - s_2\|}.$$

The following is an extension theorem for L-smooth functions.

Theorem 4.6. [*Gru09, Theorem 2.6*] If $\overline{L} := \Gamma(\mathbb{D}, (\alpha, \nu)) < \infty$, then there exists an \overline{L} -smooth function $f : \mathbb{R}^n \to \mathbb{R}$ with

$$f(x) = \alpha(x), \ \nabla f(x) = \nu(x) \ x \in \mathbb{D}.$$

One may wonder if there exists an *L*-smooth function $g : \mathbb{R}^n \to \mathbb{R}$ with $L < \Gamma(\mathbb{D}, (\alpha, \nu))$ such that

$$g(x) = \alpha(x), \ \nabla g(x) = \nu(x), \ x \in \mathbb{D},$$

if α is an *L*-smooth function on \mathbb{D} . The answer is negative when \mathbb{D} is compact; see [Gru09, Theorem 3.2].

Under the assumptions of Theorem 4.6, Daniilidis et al. [DHLGL18] introduce an *L*-smooth function $g : \mathbb{R}^n \to \mathbb{R}$ with explicit formula such that

$$g(x) = \alpha(x), \ \nabla g(x) = \nu(x), \ x \in \mathbb{D}.$$

Moreover, $\bar{L} \leq \left(\frac{5+\sqrt{29}}{2}\right)L$; see [DHLGL18, Theorem 3.1].

In the following proposition, we give a bound for $\Gamma(\mathbb{D}, (\alpha, \nu))$ when \mathbb{D} is an open convex set and we study the extension of an *L*-smooth function on \mathbb{D} .

Corollary 4.7. Let \mathbb{D} be an open convex set. If $f : \mathbb{D} \to \mathbb{R}$ is an L-smooth function on \mathbb{D} then

$$\Gamma(\mathbb{D}, (f, \nabla f)) \le (\sqrt{2} + 1)L$$

Proof. Let $s_1, s_2 \in \mathbb{D}$. First we show that $B_{s_1s_2} \leq L$. To this end, note that

$$B_{s_{1}s_{2}} = \frac{\|\nabla f(s_{1}) - \nabla f(s_{2})\|}{\|s_{1} - s_{2}\|} \le \frac{L \|s_{1} - s_{2}\|}{\|s_{1} - s_{2}\|} \le L,$$

where the last inequality is due to the *L*-smooth property of f. On the other hand, by the fundamental theorem of calculus,

$$f(s_1) - f(s_2) = \int_0^1 \langle \nabla f(\lambda s_1 + (1 - \lambda)s_2), s_1 - s_2 \rangle d\lambda.$$

It follows that

$$\begin{split} |A_{s_{1}s_{2}}| &= \frac{\left|2\left(\int_{0}^{1}\left\langle \nabla f\left(\lambda s_{1}+(1-\lambda)s_{2}\right),s_{1}-s_{2}\right\rangle d\lambda\right)+\left\langle \nabla f\left(s_{1}\right)+\nabla f\left(s_{2}\right),s_{2}-s_{1}\right\rangle\right|}{\|s_{1}-s_{2}\|^{2}} \\ &= \frac{\left|\int_{0}^{1}\left\langle \nabla f\left(\lambda s_{1}+(1-\lambda)s_{2}\right)-\nabla f\left(s_{1}\right),s_{1}-s_{2}\right\rangle d\lambda+\right|}{\int_{0}^{1}\left\langle \nabla f\left(\lambda s_{1}+(1-\lambda)s_{2}\right)-\nabla f\left(s_{2}\right),s_{1}-s_{2}\right\rangle d\lambda}\right|} \\ &\leq \frac{\int_{0}^{1}L(1-\lambda)\left\|s_{1}-s_{2}\right\|^{2} d\lambda+\int_{0}^{1}L\lambda\left\|s_{1}-s_{2}\right\|^{2} d\lambda}{\left\|s_{1}-s_{2}\right\|^{2}} = L. \end{split}$$

Hence, in view of (4.15), $\Gamma(\mathbb{D}, (f, \nabla f)) \leq (\sqrt{2}+1)L$, and the proof is complete.

Using Corollary 4.7, we can extend our convergence rate results for *L*-smooth functions on open convex sets. In particular, one can derive the following corollary.

Corollary 4.8. If the function f is L-smooth on the open convex set \mathbb{D} and the level set $\{x : f(x) < f(x_0)\}$ is contained in the set \mathbb{D} , then the gradient method with step lengths $\frac{1}{L}$ has convergence rate of

$$\min_{0 \le k \le N} \left\| \nabla f\left(x^k \right) \right\| \le \left(\frac{4(\sqrt{2}+1)L(f(x^0)-f^*)}{3N+2} \right)^{1/2}.$$

Proof. By the descent lemma 2.22, the points generated by the gradient descent method for step length 1/L lies on the level set $\{x : f(x) < f(x_0)\}$. By Corollary 4.7, f can be extended to a $(\sqrt{2}+1)L$ -smooth function from \mathbb{R}^n to \mathbb{R} . Using Theorem 4.3 the desired results now follow.

We end this chapter by some concluding remarks and a conjecture.

4.5 Concluding remarks

In this chapter, we studied the convergence rate of the gradient method for *L*-smooth functions and we provided a new convergence rate when the step lengths belong to the interval $(0, \frac{\sqrt{3}}{L})$. Moreover, we have shown that this convergence rate is tight for step lengths in the interval $(0, \frac{1}{L}]$.

It is not knowledge, but the act of learning, not possession but the act of getting there, which grants the greatest enjoyment.

Carl Friedrich Gauss

5

Conditions for linear convergence of the gradient method for non-convex optimization

Preamble

In the previous chapter, we studied convergence rate of the gradient method for *L*-smooth functions and derived sub-linear convergence rate for this class of functions. In this chapter, we derive a new linear convergence rate for the gradient method with fixed step lengths for non-convex smooth optimization problems satisfying the Polyak-Łojasiewicz (PŁ) inequality. We establish that the PŁ inequality is a necessary and sufficient condition for linear convergence to the optimal value for this class of problems. We list some related classes of functions for which the gradient method may enjoy linear convergence rate. Moreover, we investigate their relationship with the PŁ inequality. This chapter is based on the paper [AdKZ23a].

5.1 Introduction

We consider the gradient method for the unconstrained optimization problem

$$f^* := \inf_{x \in \mathbb{R}^n} f(x), \tag{5.1}$$

where $f : \mathbb{R}^n \to \mathbb{R}$ is differentiable, and f^* is finite. The gradient method with fixed step lengths may be described as follows, which is the same as Algorithm 4.1.

Algorithm 5.1 Gradient method with fixed step lengths	
Set N and $\{t_k\}_{k=1}^N$ (step lengths) and pick $x^0 \in \mathbb{R}^n$.	
For $k = 1, 2,, N$ perform the following step:	
1. $x^{k} = x^{k-1} - t_{k} \nabla f(x^{k-1})$	

In addition, we assume that f has a maximum curvature $L \in (0, \infty)$ and a minimum curvature $\mu \in (-\infty, L)$. Recall that f has a maximum curvature L if $\frac{L}{2} \|.\|^2 - f$ is convex. Similarly, f has a minimum curvature μ if $f - \frac{\mu}{2} \|.\|^2$ is convex. We denote smooth functions with curvature belonging to the interval $[\mu, L]$ by $\mathcal{H}_{\mu,L}(\mathbb{R}^n)$. The class $\mathcal{H}_{\mu,L}(\mathbb{R}^n)$ includes all smooth functions with Lipschitz gradient (note that $\mu \ge 0$ corresponds to convexity). Indeed, f is L-smooth on \mathbb{R}^n if and only if f has a maximum and minimum curvature $\overline{L} > 0$ and $\overline{\mu}$, respectively, with $\max(\overline{L}, |\overline{\mu}|) \le L$; recall the discussion before Theorem 2.42. This class of functions is broad and appears naturally in many models in machine learning, see [DD19] and the references therein.

For $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$, we have the following inequalities for $x, y \in \mathbb{R}^n$

$$f(y) \le f(x) + \langle \nabla f(x), y - x \rangle + \frac{L}{2} ||y - x||^2,$$
 (5.2)

$$f(y) \ge f(x) + \langle \nabla f(x), y - x \rangle + \frac{\mu}{2} ||y - x||^2;$$
 (5.3)

see Lemma 2.5 in [RGP22], where (5.2) and (5.3) are similar to Lemma 2.22 and Definition 2.25, respectively.

It is known that the number of iterations of first-order methods, which is needed to be performed for obtaining an ϵ -stationary¹ point, is of the order $\Omega(\epsilon^{-2})$ for *L*-smooth functions [CDHS20]. Hence, it is of interest to investigate the classes of functions for which the gradient method enjoys linear convergence rate. This subject has been investigated by some scholars and some classes of functions

¹A point *x* is called ϵ -stationary if $\|\nabla f(x)\| \le \epsilon$.

have been introduced where linear convergence is possible; see [HSS20, KNS16, HSL21, DDG⁺22] and the references therein. This includes the class of functions satisfying the Polyak-Łojasiewicz (PŁ) inequality [KNS16, Pol63].

Definition 5.1. A function f is said to satisfy the *PL inequality* on $\mathbb{X} \subseteq \mathbb{R}^n$, where $x^* \in \mathbb{X}$ is a minimizer of f over \mathbb{R}^n , if there exists $\mu_p > 0$ such that

$$f(x) - f^* \le \frac{1}{2\mu_p} \|\nabla f(x)\|^2, \quad \forall x \in \mathbb{X}.$$
(5.4)

Note that the PŁ inequality is also known as *gradient dominated*; see [Nes18, Definition 4.1.3]. Strongly convex functions satisfy the PŁ inequality, but some classes of non-convex functions also fulfill this inequality. For instance, the following proposition provides an example, which is a slightly more general case than the example presented by [Nes18, Example 4.1.3].

Proposition 5.2. Consider a differentiable function $G : \mathbb{R}^n \to \mathbb{R}^m$ with $m \leq n$ and let $J_G(x)$ be the Jacobian matrix of G at x. If

$$\inf_{x\in\mathbb{X}}\lambda_{\min}(J_G(x)J_G(x)^T)=\alpha>0,$$

for some $\mathbb{X} \subseteq \mathbb{R}^n$, then the function $f(x) = ||G(x)||^2$ fulfils the PL inequality (5.4) with constant $\mu_p = 2\alpha$.

Proof. Let $f(x) = \sum_{j=1}^{m} G_j^2(x)$ and $(J_G^T(x))_{ij} = \frac{\partial}{\partial x_i} G_j(x)$ where $i \in \{1, ..., n\}$ and $j \in \{1, ..., m\}$. Moreover,

$$\nabla f(x) = 2J_G^T(x)G(x) \in \mathbb{R}^n.$$

Assume that $J_G(x)J_G^T(x) \succeq \alpha I_{m \times m}$ for all $x \in \mathbb{X}$ and some $\alpha > 0$. Then,

$$\frac{1}{4} \|\nabla f(x)\|^2 = G^T(x) J_G(x) J_G^T(x) G(x) \ge \alpha \|J_G(x)\|^2 = \alpha f(x).$$

Thus,

$$f(x) - f^* \le \frac{1}{4\alpha} \|\nabla f(x)\|^2 - f^* \le \frac{1}{4\alpha} \|\nabla f(x)\|^2,$$

where the last inequality is due to $f^* \ge 0$.

In other words, nonlinear least squares problems sometimes correspond to instances of (5.1) where the objective satisfies the PŁ inequality.

The following classical theorem provides a linear convergence rate for Algorithm 5.1 under the PL inequality.

Theorem 5.3. [Pol63, Theorem 4] Let f be L-smooth and let f satisfy PL inequality on $\mathbb{X} = \{x : f(x) \le f(x^0)\}$. If $t_1 \in (0, \frac{2}{L})$ and x^1 is generated by Algorithm 5.1, then

$$f(x^{1}) - f^{\star} \leq \left(1 - t_{1}\mu_{p}(2 - t_{1}L)\right) \left(f(x^{0}) - f^{\star}\right).$$
(5.5)

In particular, if $t_1 = \frac{1}{L}$, we have

$$f(x^{1}) - f^{\star} \le (1 - \frac{\mu_{p}}{L}) \left(f(x^{0}) - f^{\star} \right).$$
(5.6)

In this chapter we will sharpen this bound; see Theorem 5.4. Under the assumptions of Theorem 5.3, Karimi et al. [KNS16] established linear convergence rates for some other methods including the randomized coordinate descent. We refer the interested reader to the recent survey [DDG⁺22] for more details on the convergence of algorithms under the PŁ inequality.

In this chapter, we study the convergence rate of Algorithm 5.1 by using performance estimation; see Chapter 3.

The rest of the chapter is organized as follows. In Section 5.2, we consider problem (5.1) when f satisfies the PŁ inequality. We derive a new linear convergence rate for Algorithm 5.1 by using performance estimation. Furthermore, we provide an optimal step length with respect to the given bound. We also show that the PŁ inequality is necessary and sufficient for linear convergence, in a well-defined sense. Section 5.3 lists some other situations where Algorithm 5.1 is linearly convergent. Moreover, we study the relationships between these situations. Finally, we conclude the chapter with some remarks and questions for future research.

5.2 Linear convergence under the PŁ inequality

In this section we study linear convergence of the gradient descent for $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$ under the PŁ inequality. It is readily seen that the PŁ inequality implies that every stationary point is a global minimum on *X*. By virtue of the descent lemma [Nes18, Page 29] and Lemma 2.22, we have

$$f(x) - f^* \ge \frac{1}{2L} \|\nabla f(x)\|^2, \ \forall x \in \mathbb{R}^n.$$
(5.7)

Hence, μ_p can only take values in (0, L] by (5.7). On the other hand, if f in $\mathcal{H}_{\mu,L}(\mathbb{R}^n)$ we may assume without loss of generality $\mu \leq \mu_p$ where μ is the

minimum curvature parameter. The inequality is trivial if $\mu \leq 0$, and we therefore assume that $\mu > 0$. By taking the minimum with respect to *y* from both side of inequality (5.3), we get

$$f(x) - f^* \le \frac{1}{2\mu} \|\nabla f(x)\|^2.$$

Hence, one may assume without loss of generality $\mu_p = \max{\{\mu, \mu_p\}}$ in inequality (5.4).

In what follows, we employ performance estimation to get a new bound under the assumptions of Theorem 5.3. In this setting, the worst-case convergence rate of Algorithm 5.1 may be cast as the following optimization problem,

$$\max \frac{f(x^{1}) - f^{\star}}{f(x^{0}) - f^{\star}}$$

$$x^{1} \text{ is generated by Algorithm 5.1 w.r.t. } f, x^{0} \qquad (5.8)$$

$$f(x) \ge f^{\star} \quad \forall x \in \mathbb{R}^{n}$$

$$f(x) - f^{\star} \le \frac{1}{2\mu_{p}} \|\nabla f(x)\|^{2}, \quad \forall x \in X$$

$$f \in \mathcal{H}_{\mu,L}(\mathbb{R}^{n})$$

$$x^{0} \in \mathbb{R}^{n}.$$

In problem (5.8), f and x^0 are decision variables and $X = \{x : f(x) \le f(x^0)\}$. We may replace the infinite dimensional condition $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$ by a finite set of constraints, by using interpolation. Theorem 2.42 gives some necessary and sufficient conditions for the interpolation of given data by some $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$.

It is worth noting that Theorem 2.42 addresses non-smooth functions as well. In fact, $L = \infty$ covers non-smooth functions. Note that we only investigate the smooth case in this chapter, that is $L \in (0, \infty)$ and $\mu \in (-\infty, 0]$.

By Theorem 2.42, problem (5.8) may be relaxed as follows,

$$\max \frac{f^{1} - f^{\star}}{f^{0} - f^{\star}}$$
s.t.
$$\frac{1}{2(1 - \frac{\mu}{L})} \left(\frac{1}{L} \left\| g^{i} - g^{j} \right\|^{2} + \mu \left\| x^{i} - x^{j} \right\|^{2} - \frac{2\mu}{L} \left\langle g^{j} - g^{i}, x^{j} - x^{i} \right\rangle \right) \le$$

$$f^{i} - f^{j} - \left\langle g^{j}, x^{i} - x^{j} \right\rangle \quad i, j \in \{0, 1\}$$

$$x^{1} = x^{0} - t_{1}g^{0}$$

$$f^{k} \ge f^{\star} \quad k \in \{0, 1\}$$

$$f^{k} - f^{\star} \le \frac{1}{2\mu_{p}} \|g^{k}\|^{2}, \quad k \in \{0, 1\}.$$

$$(5.9)$$

As we replace the constraint $f(x)-f^* \leq \frac{1}{2\mu_p} \|\nabla f(x)\|^2$ for each $x \in X$ by $f^0-f^* \leq \frac{1}{2\mu_p} \|g^0\|^2$ and $f^1-f^* \leq \frac{1}{2\mu_p} \|g^1\|^2$, problem (5.9) is a relaxation of problem (5.8). By using the constraint $x^1 = x^0 - t_1 g^0$, problem (5.9) may be reformulated as,

$$\begin{aligned} \max \frac{f^{1} - f^{\star}}{f^{0} - f^{\star}} \\ \text{s.t.} & \frac{1}{2(L - \mu)} \left(\|g^{1}\|^{2} + (1 + \mu L t_{1}^{2} - 2\mu t_{1})\|g^{0}\|^{2} + 2(\mu t_{1} - 1)\langle g^{0}, g^{1} \rangle \right) - \\ & f^{1} + f^{0} - \langle g^{0}, t_{1}g^{0} \rangle \leq 0 \\ & \frac{1}{2(L - \mu)} \left(\|g^{1}\|^{2} + (1 + \mu L t_{1}^{2} - 2\mu t_{1})\|g^{0}\|^{2} + 2(\mu t_{1} - 1)\langle g^{0}, g^{1} \rangle \right) - \\ & f^{0} + f^{1} + \langle g^{1}, t_{1}g^{0} \rangle \leq 0 \\ & f^{\star} - f^{k} \leq 0 \ k \in \{0, 1\} \\ & f^{k} - f^{\star} - \frac{1}{2\mu_{p}} \|g^{k}\|^{2} \leq 0, \ k \in \{0, 1\}. \end{aligned}$$
(5.10)

By using the Gram matrix,

$$X = \begin{pmatrix} (g^{0})^{T} \\ (g^{1})^{T} \end{pmatrix} \begin{pmatrix} g^{0} & g^{1} \end{pmatrix} = \begin{pmatrix} \|g^{0}\|^{1} & \langle g^{0}, g^{1} \rangle \\ \langle g^{0}, g^{1} \rangle & \|g^{1}\|^{2} \end{pmatrix},$$

problem (5.10) can be relaxed as follows,

$$\max \frac{f^{1} - f^{\star}}{f^{0} - f^{\star}}$$
s.t. $tr(A_{1}X) - f^{1} + f^{0} \le 0$
 $tr(A_{2}X) - f^{0} + f^{1} \le 0$ (5.11)
 $f^{0} - f^{\star} + tr(A_{3}X) \le 0$
 $f^{1} - f^{\star} + tr(A_{4}X) \le 0$
 $f^{0}, f^{1} \ge f^{\star}, X \ge 0,$

where

$$\begin{split} A_1 &= \begin{pmatrix} \frac{1+\mu L t_1^2 - 2\mu t_1}{2(L-\mu)} - t_1 & \frac{\mu t_1 - 1}{2(L-\mu)} \\ \frac{\mu t_1 - 1}{2(L-\mu)} & \frac{1}{2(L-\mu)} \end{pmatrix} \qquad A_2 = \begin{pmatrix} \frac{1+\mu L t_1^2 - 2\mu t_1}{2(L-\mu)} & \frac{\mu t_1 - 1}{2(L-\mu)} + \frac{t_1}{2} \\ \frac{\mu t_1 - 1}{2(L-\mu)} + \frac{t_1}{2} & \frac{1}{2(L-\mu)} \end{pmatrix} \\ A_3 &= \begin{pmatrix} \frac{-1}{\mu_p^2} & 0 \\ 0 & 0 \end{pmatrix} \qquad A_4 = \begin{pmatrix} 0 & 0 \\ 0 & \frac{-1}{\mu_p^2} \end{pmatrix}. \end{split}$$

In addition, X, f^0 , f^1 are decision variables in this formulation. In the next theorem, we obtain an upper bound for problem (5.10) by using weak duality. This bound gives a new convergence rate for Algorithm 5.1 for a wide variety of functions.

Theorem 5.4. Let $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$ with $L \in (0, \infty)$, $\mu \in (-\infty, 0]$ and let f satisfy the PL inequality on $X = \{x : f(x) \leq f(x^0)\}$. Suppose that x^1 is generated by Algorithm 5.1.

i) If $t_1 \in (0, \frac{1}{L})$, then

$$\frac{f(x^{1}) - f^{\star}}{f(x^{0}) - f^{\star}} \leq \left(\frac{\mu_{p} (1 - Lt_{1}) + \sqrt{(L - \mu)(\mu - \mu_{p})(2 - Lt_{1})\mu_{p}t_{1} + (L - \mu)^{2}}}{L - \mu + \mu_{p}}\right)^{2}.$$

ii) If
$$t_1 \in \left[\frac{1}{L}, \frac{3}{\mu + L + \sqrt{\mu^2 - L\mu + L^2}}\right]$$
, then

$$\frac{f(x^1) - f^*}{f(x^0) - f^*} \leq \left(\frac{(Lt_1 - 2)(\mu t_1 - 2)\mu_p t_1}{(L + \mu - \mu_p)t_1 - 2} + 1\right).$$

iii) If
$$t_1 \in \left(\frac{3}{\mu + L + \sqrt{\mu^2 - L\mu + L^2}}, \frac{2}{L}\right)$$
, then
$$\frac{f(x^1) - f^*}{f(x^0) - f^*} \leq \frac{(Lt_1 - 1)^2}{(Lt_1 - 1)^2 + \mu_p t_1(2 - Lt_1)}.$$

In particular, if $t_1 = \frac{1}{L}$ and $\mu = -L$, we have

$$f(x^{1}) - f^{\star} \leq \left(\frac{2L - 2\mu_{p}}{2L + \mu_{p}}\right) \left(f(x^{0}) - f^{\star}\right).$$
(5.12)

Proof. First we consider $t_1 \in (0, \frac{1}{L})$. Let

$$b_1 = \frac{(L-\mu)\left(\alpha + \mu_p \left(1 - Lt_1\right)\right)}{\alpha \left(L - \mu + \mu_p\right)}$$
$$b_2 = b_1 - \left(\frac{\alpha}{L-\mu}b_1\right)^2,$$

where

$$\alpha = \sqrt{(L-\mu)(\mu_p t_1(\mu_p - \mu)(Lt_1 - 2) + (L-\mu))}.$$

It is readily seen that $b_1, b_2 \ge 0$. Furthermore,

$$\begin{split} & f^{1} - f^{\star} - (b_{1} - b_{2}) \left(f^{0} - f^{\star} \right) - b_{2} \left(-\frac{1}{2\mu_{p}} \left\| g^{0} \right\|^{2} + f^{0} - f^{\star} \right) \\ & - (1 - b_{1}) \left(-\frac{1}{2\mu_{p}} \left\| g^{1} \right\|^{2} + f^{1} - f^{\star} \right) - b_{1} \left(\frac{1}{2(L - \mu)} \left(\left\| g^{1} \right\|^{2} + (1 + \mu L t_{1}^{2} - 2\mu t_{1}) \left\| g^{0} \right\|^{2} + 2(\mu t_{1} - 1) \langle g^{0}, g^{1} \rangle \right) - f^{0} + f^{1} + \left\langle g^{1}, t_{1} g^{0} \right\rangle \right) = \\ & - \frac{1 - L t_{1}}{2\alpha} \left\| \frac{\alpha b_{1}}{L - \mu} g^{0} - g^{1} \right\|^{2} \leq 0. \end{split}$$

Therefore, for any feasible solution of problem (5.10), we have

$$\frac{f(x^{1}) - f^{\star}}{f(x^{0}) - f^{\star}} \leq \left(\frac{\mu_{p} (1 - Lt_{1}) + \sqrt{(L - \mu)(\mu - \mu_{p})(2 - Lt_{1})\mu_{p}t_{1} + (L - \mu)^{2}}}{L - \mu + \mu_{p}}\right)^{2},$$

and the proof of this part is complete. Now, we consider the case that $t_1 \in \left[\frac{1}{L}, \frac{3}{\mu+L+\sqrt{\mu^2-L\mu+L^2}}\right]$. Suppose that

$$a_{1} = \frac{\mu t_{1} - 1}{\left(L + \mu - \mu_{p}\right) t_{1} - 2}, \qquad a_{2} = \frac{1 - L t_{1}}{\left(L + \mu - \mu_{p}\right) t_{1} - 2}, \\ a_{3} = -\frac{\left((L t_{1} - 2)(\mu t_{1} - 2) - 1\right) \mu_{p} t_{1}}{\left(L + \mu - \mu_{p}\right) t_{1} - 2}, \qquad a_{4} = -\frac{\mu_{p} t_{1}}{\left(L + \mu - \mu_{p}\right) t_{1} - 2}.$$

It is readily seen that $a_1, a_2, a_3, a_4 \ge 0$. Furthermore,

$$\begin{split} f^{1} - f^{\star} - \left(1 - a_{3} - a_{4}\right) \left(f^{0} - f^{\star}\right) - a_{3} \left(-\frac{1}{2\mu_{p}} \left\|g^{0}\right\|^{2} + f^{0} - f^{\star}\right) - \\ a_{4} \left(-\frac{1}{2\mu_{p}} \left\|g^{1}\right\|^{2} + f^{1} - f^{\star}\right) - a_{1} \left(\frac{1}{2(L-\mu)} \left(\left\|g^{1}\right\|^{2} + (1 + \mu L t_{1}^{2} - 2\mu t_{1})\right) \left\|g^{0}\right\|^{2} + \\ 2(\mu t_{1} - 1) \langle g^{0}, g^{1} \rangle \right) - f^{0} + f^{1} + \left\langle g^{1}, t_{1} g^{0} \right\rangle \right) - a_{2} \left(\frac{1}{2(L-\mu)} \left(\left\|g^{1}\right\|^{2} + (1 + \mu L t_{1}^{2} - 2\mu t_{1})\right) \left\|g^{0}\right\|^{2} + 2(\mu t_{1} - 1) \langle g^{0}, g^{1} \rangle \right) - f^{1} + f^{0} - \left\langle g^{0}, t_{1} g^{0} \right\rangle \right) = 0. \end{split}$$

Therefore, for any feasible solution of problem (5.10), we have

$$f(x^{1}) - f^{\star} - \left(\frac{L\mu_{p}\mu t_{1}^{3} - 2\mu_{p}(L+\mu)t_{1}^{2} + 4\mu_{p}t_{1}}{\left(L+\mu-\mu_{p}\right)t_{1} - 2} + 1\right)\left(f(x^{0}) - f^{\star}\right) \le 0.$$

Now, we prove the last part. Assume that $t_1 \in \left(\frac{3}{\mu+L+\sqrt{\mu^2-L\mu+L^2}}, \frac{2}{L}\right)$. With some algebra, one can show

$$\begin{split} f^{1} - f^{\star} - \left(\frac{(Lt_{1} - 1)^{2}}{\beta}\right) & \left(f^{0} - f^{\star}\right) - \left(\frac{\mu_{p}t_{1}(2 - Lt_{1})}{\beta}\right) \left(-\frac{1}{2\mu_{p}} \left\|g^{1}\right\|^{2} + f^{1} - f^{\star}\right) - \\ & \left(\frac{(Lt_{1} - 1)(2 - Lt_{1})}{\beta}\right) \left(\frac{1}{2(L - \mu)} \left(\|g^{1}\|^{2} + (1 + \mu Lt_{1}^{2} - 2\mu t_{1})\|g^{0}\|^{2} + 2(\mu t_{1} - 1)\langle g^{0}, g^{1}\rangle\right) - \\ & f^{1} + f^{0} - \left\langle g^{0}, t_{1}g^{0}\right\rangle - \left(\frac{Lt_{1} - 1}{\beta}\right) \left(\frac{1}{2(L - \mu)} \left(\|g^{1}\|^{2} + (1 + \mu Lt_{1}^{2} - 2\mu t_{1})\|g^{0}\|^{2} + \\ & 2(\mu t_{1} - 1)\langle g^{0}, g^{1}\rangle\right) - f^{0} + f^{1} + \left\langle g^{1}, t_{1}g^{0}\right\rangle \right) = \\ & - \frac{(1 - Lt_{1})(L\mu t^{2} - 2(\mu + L)t + 3)}{2\beta(L - \mu)} \left\|\sqrt{Lt_{1} - 1}g^{0} + \frac{1}{\sqrt{Lt_{1} - 1}}g^{1}\right\|^{2} \leq 0, \end{split}$$

where

$$\beta = (Lt_1 - 1)^2 + \mu_p t_1 (2 - Lt_1).$$

The rest of the proof is similar to that of the former cases.

As the expressions provided in Theorem 5.4 are complicated here we provide an example for some parameters. Let L = 10, $\mu = -L = -10$ and $\mu_p = 1$. Then part i) of Theorem 5.4 is given by $t_1 \in (0, 0.1)$ and

$$\frac{f(x^1) - f^*}{f(x^0) - f^*} \le \frac{1}{441} \left(1 - 10t_1 + \sqrt{440t_1(5t_1 - 1) + 400} \right)^2.$$

One may wonder how we obtain Lagrange multipliers (dual variables) in Theorem 5.4. The multipliers are computed by solving the dual of problem (5.11) by hand; see Chapter 3 for more discussion on solving PEP. Furthermore, Theorem 5.4 provides a tighter bound in comparison with the convergence rate given in Theorem 5.3 for *L*-smooth functions with $t_1 \in (0, \frac{2}{L})$. To show this, we need investigate three subintervals:

i) Suppose that $t_1 \in (0, \frac{1}{L})$. As $1 - Lt_1 \le 0$,

$$\left(\frac{\mu_p (1-Lt_1) + \sqrt{2L(-L-\mu_p)(2-Lt_1)\mu_p t_1 + 4L^2}}{2L+\mu_p} \right)^2 \leq \frac{4L^2 + 2L\mu_p t_1 (L+\mu_p)(Lt_1-2) + (\mu_p - L\mu_p t_1)^2}{(2L+\mu_p)^2} \leq 1 - t_1 \mu_p (2-t_1L),$$

 \square

where the last inequality follows from non-positivity of the quadratic function $T_1(t_1) = -Lt_1^2(2L^2 + L\mu_p + \mu_p^2) + 2t_1(2L^2 + L\mu_p + \mu_p^2) - 4L$ on the given interval.

ii) Let $t_1 \in \left[\frac{1}{L}, \frac{\sqrt{3}}{L}\right]$. Since $\mu_p \le L$ and $(2 - Lt_1) > 0$, we have $1 \le \frac{Lt_1 + 2}{\mu_p t_1 + 2} \implies 1 - \frac{(2 - Lt_1)(Lt_1 + 2)\mu_p t_1}{\mu_p t_1 + 2} \le 1 - t_1 \mu_p (2 - Lt_1).$

iii) Assume that
$$t_1 \in (\frac{\sqrt{3}}{L}, \frac{2}{L})$$
. It is readily verified that the quadratic function $T_2(t_1) = (Lt_1 - 1)^2 + \mu_p t_1(2 - Lt_1) - 1$ is non-positive on the given interval.

$$\frac{(Lt_1-1)^2}{(Lt_1-1)^2+\mu_pt_1(2-Lt_1)} = 1 - \frac{\mu_pt_1(2-Lt_1)}{(Lt_1-1)^2+\mu_pt_1(2-Lt_1)} \le 1 - t_1\mu_p(2-Lt_1).$$

Therefore, for $t_1 \in (0, \frac{2}{L})$ the bound provided by Theorem 5.4 is tighter than that given by Theorem 5.3.

In most problems, the smoothness constant, *L*, is unknown. By using (5.2), any estimation of the smoothness constant *L*, say \tilde{L} , should satisfy the following inequality,

$$f\left(x-\frac{1}{\tilde{L}}\nabla f(x)\right) \leq f(x)-\frac{1}{2\tilde{L}}\|\nabla f(x)\|^{2}.$$

Thus one may try to obtain a suitable estimate by searching for a sufficiently large value of \tilde{L} that satisfies this inequality. This technique is due to Nesterov; see [Nes13, Section 3] for details.

The next proposition gives the optimal step length with respect to the worstcase convergence rate.

Proposition 5.5. Let $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$ with $L \in (0, \infty)$, $\mu \in (-\infty, 0]$ and let f satisfy the PL inequality on $X = \{x : f(x) \le f(x^0)\}$. Suppose that $r(t) = L\mu(L + \mu - \mu_p)t^3 - (L^2 - \mu_p(L + \mu) + 5L\mu + \mu^2)t^2 + 4(L + \mu)t - 4$ and \bar{t} is the unique root of r in $\left[\frac{1}{L}, \frac{3}{\mu + L + \sqrt{\mu^2 - L\mu + L^2}}\right]$ if it exists. Then t^* given by $t^* = \begin{cases} \bar{t} & \text{if } \bar{t} \text{ exists} \\ \frac{3}{\mu + L + \sqrt{\mu^2 - L\mu + L^2}} & \text{otherwise,} \end{cases}$

is the optimal step length for Algorithm 5.1 with respect to the worst-case convergence rate.

Proof. To obtain an optimal step length, we need to solve the optimization problem

$$\min_{t\in(0,\frac{2}{L})}h(t)$$

where h is given by

$$h(t) = \begin{cases} \left(\frac{\mu_p(1-Lt) + \sqrt{(L-\mu)(\mu-\mu_p)(2-Lt)\mu_p t + (L-\mu)^2}}{L-\mu+\mu_p}\right)^2 & t \in \left(0, \frac{1}{L}\right) \\ \frac{(Lt-2)(\mu t-2)\mu_p t}{(L+\mu-\mu_p)t-2} + 1 & t \in \left[\frac{1}{L}, \frac{3}{\mu+L+\sqrt{\mu^2-L\mu+L^2}}\right] \\ \frac{(Lt-1)^2}{(Lt-1)^2 + (2-Lt)\mu_p t} & t \in \left(\frac{3}{\mu+L+\sqrt{\mu^2-L\mu+L^2}}, \frac{2}{L}\right). \end{cases}$$

It is easily seen that *h* is decreasing on $\left(0, \frac{1}{L}\right)$ and is increasing on $\left(\frac{3}{\mu+L+\sqrt{\mu^2-L\mu+L^2}}, \frac{2}{L}\right)$. Hence, we need investigate the closed interval $\left[\frac{1}{L}, \frac{3}{\mu+L+\sqrt{\mu^2-L\mu+L^2}}\right]$. We will show that *h* is convex on the interval in question. First, we consider the case $L + \mu - \mu_p \leq 0$. Let $p(t) = \frac{\mu t - 2}{(L+\mu-\mu_p)t-2}$ and $q(t) = (Lt-2)\mu_p t$. By some algebra, one can show the following inequalities for $t \in \left[\frac{1}{L}, \frac{3}{\mu+L+\sqrt{\mu^2-L\mu+L^2}}\right]$:

$$p(t) \ge 0$$
 $q(t) \le 0$
 $p'(t) \ge 0$
 $q'(t) \ge 0$
 $p''(t) \le 0$
 $q''(t) \ge 0$

Hence, the convexity of *h* follows from h'' = p''q + 2p'q' + pq''. Now, we investigate the case that $L + \mu - \mu_p > 0$. Suppose that $p(t) = \frac{\mu_p t}{(L + \mu - \mu_p)t - 2}$ and $q(t) = (Lt - 2)(\mu t - 2)$. For these functions, we have the following inequalities

$p(t) \leq 0$	$q(t) \ge 0$
$p'(t) \leq 0$	$q'(t) \leq 0$
$p''(t) \ge 0$	$q''(t) \le 0,$

which analogous to the former case one can infer the convexity of *h* on the given interval. Hence, if *h* has a root in $\left[\frac{1}{L}, \frac{3}{\mu+L+\sqrt{\mu^2-L\mu+L^2}}\right]$, it will be the minimum. Otherwise, the point $t^* = \frac{3}{\mu+L+\sqrt{\mu^2-L\mu+L^2}}$ will be the minimum. This follows from the point that $h'(\frac{1}{L}) = \frac{2L\mu_p(\mu_p-L)}{(L+\mu_p-\mu)^2} \le 0$ and the convexity of *h* on the interval in question.

Thanks to Proposition 5.5, the following corollary gives the optimal step length for *L*-smooth convex functions satisfying the PŁ inequality.

Corollary 5.6. If f is an L-smooth convex function satisfying the PL inequality, then the optimal step length with respect to the worst-case convergence rate given by Theorem 5.4 is min $\left\{\frac{2}{L+\sqrt{L\mu_p}}, \frac{3}{2L}\right\}$.

The constant $\frac{2}{L+\sqrt{L\mu_p}}$ also appears in the the *fast gradient algorithm* introduced in [NNG19] for *L*-smooth convex functions which are $(1, \mu_s)$ -quasar-convex, see Definition 5.14. By Theorem 5.16, $(1, \mu_s)$ -quasar-convexity implies the PŁ inequality with the same constant. Algorithm 5.2 describes the method in question.

Algorithm 5.	2 Fast	gradient	method
--------------	---------------	----------	--------

Pick $x^0 \in \mathbb{R}^n$, set N and $y^0 = x^0$. For k = 1, 2, ..., N perform the following step: 1. $y^k = x^{k-1} - \frac{1}{L} \nabla f(x^{k-1})$ 2. $x^k = y^k + \frac{\sqrt{L} - \sqrt{\mu_p}}{\sqrt{L} + \sqrt{\mu_p}} (y^k - y^{k-1})$

One can verify that Algorithm 5.2, at the first iteration, generates $x^1 = x^0 - \frac{2}{L + \sqrt{L\mu_p}} \nabla f(x^0)$.

A more general form of the PL inequality, called the *Lojasiewicz inequality*, may be written as

$$(f(x) - f^*)^{2\theta} \le \frac{1}{2\mu_p} \|\nabla f(x)\|^2, \quad \forall x \in X,$$
 (5.13)

where $\theta \in (0, 1)$. As an example for a function with $\theta > \frac{1}{2}$ one can easily see that the function $f(x) = x^4$ has $\theta = \frac{3}{4}$ with $\mu_p = 8$. It is known that when $\theta \in (0, \frac{1}{2}]$ some algorithms, including Algorithm 5.1, are linearly convergent; see [AB09, ABRS10]. In the next theorem, we show that for functions with finite maximum and minimum curvature the Łojasiewicz inequality cannot hold for $\theta \in (0, \frac{1}{2})$.

Theorem 5.7. Let $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$ be a non-constant function. If f satisfies the Lojasiewicz inequality on $X = \{x : f(x) \le f(x^0)\}$, then $\theta \ge \frac{1}{2}$.

Proof. To the contrary, assume that $\theta \in (0, \frac{1}{2})$. Without loss of generality, we may assume that $\mu = -L$. It is known that Algorithm 5.1 generates a decreasing

sequence $\{f(x^k)\}$ and it is convergent, that is $\|\nabla f(x^k)\| \to 0$; see [Nes18, page 28]. Furthermore, (5.13) implies that $f(x^k) \to f^*$. Without loss of generality, we may assume that $f^* = 0$. First, we investigate the case that $f(x^0) = 1$. The semidefinite programming problem corresponding to performance estimation in this case may be formulated as follows,

$$\max f^{1}$$

s.t. $\operatorname{tr}(A_{1}X) - f^{1} + 1 \leq 0$
 $\operatorname{tr}(A_{2}X) - 1 + f^{1} \leq 0$ (5.14)
 $1 + \operatorname{tr}(A_{3}X) \leq 0$
 $(f^{1})^{2\theta} + \operatorname{tr}(A_{4}X) \leq 0$
 $f^{1} \geq 0, X \geq 0.$

Since Algorithm 5.1 is a monotone method, f^1 can take value in [0, 1]. In addition, we have $f^1 \le (f^1)^{2\theta}$ on this interval. Hence, by using Theorem 5.4, we get the following bound,

$$f^1 \leq \frac{2L - 2\mu_p}{2L + \mu_p}.$$

Now, suppose that $f(x^0) = f^0 > 0$. Consider the function $h : \mathbb{R}^n \to \mathbb{R}$ given by $h(x) = \frac{f(x)}{f^0}$. It is seen that *h* is $\frac{L}{f^0}$ -smooth and

$$h(x)^{2\theta} \leq \frac{1}{2\mu_p(f^0)^{2\theta-2}} \|\nabla h(x)\|^2, \quad \forall x \in X.$$

As Algorithm 5.1 generates the same x^1 for both functions, by using the first part, we obtain

$$\frac{f(x^1)}{f(x^0)} \le \frac{2L(f^0)^{-1} - 2\mu_p(f^0)^{2\theta - 2}}{2L(f^0)^{-1} + \mu_p(f^0)^{2\theta - 2}} = \frac{2L - 2\mu_p(f^0)^{2\theta - 1}}{2L + \mu_p(f^0)^{2\theta - 1}}.$$

For f^0 sufficiently small, we have $\frac{2L-2\mu_p(f^0)^{2\theta-1}}{2L+\mu_p(f^0)^{2\theta-1}} < 0$, which contradicts $f^* \ge 0$ and the proof is complete.

Necoara et al. gave necessary and sufficient conditions for linear convergence of the gradient method with constant step lengths when f is a smooth convex function; see [NNG19, Theorem 13]. Indeed, the theorem says that Algorithm 5.1 is linearly convergent if and only if f has a quadratic functional growth on $\{x : f(x) \le f(x^0)\}$; see Definition 5.11. However, this theorem does not hold necessarily for non-convex functions. The next theorem provides necessary and sufficient conditions for linear convergence of Algorithm 5.1. **Theorem 5.8.** Let $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$. Algorithm 5.1 is linearly convergent to the optimal value if and only if f satisfies PL inequality on $\{x : f(x) \leq f(x^0)\}$.

Proof. Let $\bar{x} \in \{x : f(x) \le f(x^0)\}$. Linear convergence implies the existence of $\gamma \in [0, 1)$ with

$$f(\hat{x}) - f^* \le \gamma (f(\bar{x}) - f^*), \tag{5.15}$$

where $\hat{x} = \bar{x} - \frac{1}{L} \nabla f(\bar{x})$. By (5.3), we have $f(\bar{x}) - f(\hat{x}) \le \frac{2L-\mu}{2L^2} \|\nabla f(\bar{x})\|^2$. By using this inequality with (5.15), we get

$$f(\bar{x}) - f^* \le \frac{1}{1 - \gamma} (f(\bar{x}) - f(\hat{x})) \le \frac{2L - \mu}{2L^2(1 - \gamma)} \|\nabla f(\bar{x})\|^2,$$

which shows that f satisfies PŁ inequality on $\{x : f(x) \le f(x^0)\}$. The other implication follows from Theorem 5.4.

5.3 The PŁ inequality: relation to some classes of functions

In this section, we study some classes of functions for which Algorithm 5.1 may be linearly convergent. We establish that these classes of functions satisfy the PŁ inequality under mild assumptions, and we infer the linear convergence by using Theorem 5.4. Moreover, one can get convergence rates by applying performance estimation.

Throughout the section, we denote the optimal solution set of problem (5.1) by X^* and we assume that X^* is non-empty. We denote the distance function to X^* by $d_{X^*}(x) := \inf_{y \in X^*} ||y - x||$. The set-valued mapping $\Pi_{X^*}(x)$ stands for the projection of x on X^* , that is, $\Pi_{X^*}(x) = \{y : ||y - x|| = d_{X^*}(x)\}$. Note that, as X^* is non-empty closed set, $\Pi_{X^*}(x)$ exists and is well-defined.

Definition 5.9. Let $\mu_g > 0$. A function f has a *quadratic gradient growth* on $X \subseteq \mathbb{R}^n$ if

$$\langle \nabla f(x), x - x^* \rangle \ge \mu_g d_{X^*}^2(x), \quad \forall x \in X,$$
(5.16)

for some $x^* \in \Pi_{X^*}(x)$.

Note that inequality (5.2) implies that $\mu_g \leq L$. Hu et al. [HSL21] investigated the convergence rate $\{x^k\}$ when f satisfies (5.16) and X^* is singleton. To the

best knowledge of me, there is no convergence rate result in terms of $\{f(x^k)\}$ for functions with a quadratic gradient growth. The next proposition states that quadratic gradient growth property implies the PL inequality.

Proposition 5.10. Let $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$. If f has a quadratic gradient growth on $X \subseteq \mathbb{R}^n$ with $\mu_g > 0$, then f satisfies the PL inequality with $\mu_p = \frac{\mu_g^2}{L}$.

Proof. Suppose that $x^* \in \Pi_{X^*}(x)$ satisfies (5.16). By the Cauchy-Schwarz inequality, we have

$$\mu_g \|x - x^*\| \le \|\nabla f(x)\|. \tag{5.17}$$

On the other hand, (5.2) implies that

$$f(x) \le f(x^*) + \frac{L}{2} ||x - x^*||^2.$$
(5.18)

The PŁ inequality follows from (5.17) and (5.18).

By Proposition 5.10 and Theorem 5.4, one can infer the linear convergence of Algorithm 5.1 when *f* has a quadratic gradient growth on $X = \{x : f(x) \le f(x^0)\}$. Indeed, one can derive the following bound if $t_1 = \frac{1}{L}$ and $\mu = -L$,

$$f(x^{1}) - f^{\star} \leq \left(\frac{2L^{2} - 2\mu_{g}^{2}}{2L^{2} + \mu_{g}^{2}}\right) \left(f(x^{0}) - f^{\star}\right).$$
(5.19)

Nevertheless, by using the performance estimation method, one can derive a better bound than the bound given by (5.19). The performance estimation problem for $t_1 = \frac{1}{L}$ in this case may be formulated as

$$\max \frac{f^{1} - f^{*}}{f^{0} - f^{*}}$$

s.t. $\{x^{k}, g^{k}, f^{k}\} \cup \{y^{k}, 0, f^{*}\}$ satisfy interpolation constraints (2.8) for $k \in \{0, 1\}$
 $x^{1} = x^{0} - \frac{1}{L}g^{0}$ (5.20)
 $f^{k} \ge f^{*} \ k \in \{0, 1\}$
 $\langle g^{k}, x^{k} - y^{k} \rangle \ge \mu_{g} \|y^{k} - x^{k}\|^{2}, \ k \in \{0, 1\}$
 $\|x^{0} - y^{0}\|^{2} \le \|x^{0} - y^{1}\|^{2}$
 $\|x^{1} - y^{1}\|^{2} \le \|x^{1} - y^{0}\|^{2}.$

Analogous to Section 5.2, one can obtain an upper bound for problem (5.20) by solving a semidefinite program. Our numerical results show that the bounds

 \square



Figure 5.1: Convergence rate computed by performance estimation (red line) and the bound given by (5.19) (blue line) for $\frac{\mu_g}{L} \in (0, 1)$.

generated by performance estimation are tighter than bound (5.19); see Figure 5.1. We do not have a closed-form bound on the optimal value of (5.20), though.

Definition 5.11. [NNG19, Definition 4], [Nes18, Definition 4.1.2] Let $\mu_q > 0$. A function *f* has a *quadratic functional growth* on $X \subseteq \mathbb{R}^n$ if

$$\frac{\mu_q}{2} d_{X^*}^2(x) \le f(x) - f^*, \quad \forall x \in X.$$
(5.21)

It is readily seen that, contrary to the previous situations, the quadratic functional growth property does not necessarily imply that each stationary point is a global optimal solution. The next theorem investigates the relationship between quadratic functional growth property and other notions.

Theorem 5.12. Let $f \in \mathcal{H}_{\mu,L}(\mathbb{R}^n)$ and let $X = \{x : f(x) \leq f(x^0)\}$. We have the following implications:

- i) (5.4) \Rightarrow (5.21) with $\mu_q = \mu_p$.
- ii) If $\mu_q > \frac{-\mu L}{L-\mu}$, then (5.21) \Rightarrow (5.16) with $\mu_g = \frac{\mu_q}{2}(1-\frac{\mu}{L}) + \frac{\mu}{2}$. iii) If

$$f(x) - f(x^*) \leq \langle \nabla f(x), x - x^* \rangle, \ \forall x \in X,$$

for some $x^* \in \Pi_{X^*}(x)$ then (5.21) \Rightarrow (5.16) with $\mu_g = \frac{\mu_q}{2}$.

Proof. One can establish *i*) similarly to the proof of [KNS16, Theorem 2]. Consider part *ii*). Let $x \in X$ and $x^* \in \Pi_{X^*}(x)$ with $d_{X^*}(x) = ||x - x^*||$. By (2.7), we have

$$f(x) - f(x^{\star}) \leq \frac{-1}{2(L-\mu)} \|\nabla f(x)\|^2 - \frac{\mu L}{2(L-\mu)} \|x - x^{\star}\|^2 + \frac{L}{L-\mu} \langle \nabla f(x), x - x^{\star} \rangle.$$

As $\frac{\mu_q}{2} ||x - x^*||^2 \le f(x) - f(x^*)$, we get

$$\left(\frac{\mu_q}{2}\left(1-\frac{\mu}{L}\right)+\frac{\mu}{2}\right)\|x-x^\star\|^2 \leq \langle \nabla f(x), x-x^\star \rangle,$$

which establishes the desired inequality. Part iii) is proved similarly to the former case.

By Theorem 5.4, it is clear that Algorithm 5.1 enjoys linear convergence rate if *f* has a quadratic gradient growth on $X = \{x : f(x) \le f(x^0)\}$ and if *f* satisfies assumptions *ii*) or *iii*) in Theorem 5.12. For instance, if $\mu = -L$ and $\mu_q \in (\frac{L}{2}, L)$, one can derive the following convergence rate for Algorithm 5.1 for fixed step length $t_k = \frac{1}{L}, k \in \{1, ..., N\}$,

$$f(x^{N}) - f(x^{0}) \le \left(\frac{2L^{2} - 2(\mu_{q} - \frac{L}{2})^{2}}{2L^{2} + (\mu_{q} - \frac{L}{2})^{2}}\right)^{N} \left(f(x^{0}) - f^{\star}\right).$$
(5.22)

It is interesting to compare the convergence rate (5.22) to the convergence rate obtained by using the performance estimation framework. In this case, the performance estimation problem may be cast as follows,

$$\max \frac{f^{N} - f^{\star}}{f^{0} - f^{\star}}$$
s.t. $\{x^{k}, g^{k}, f^{k}\} \cup \{y^{k}, 0, f^{\star}\}$ satisfy inequality (2.8) for $k \in \{0, ..., N\}$
 $x^{k} = x^{k-1} - \frac{1}{L}g^{k-1}, \ k \in \{1, ..., N\}$ (5.23)
 $f^{k} \ge f^{\star} \ k \in \{0, ..., N\}$
 $f^{k} - f^{\star} \ge \frac{\mu_{q}}{2} ||x^{k} - y^{k}||^{2}, \ k \in \{0, ..., N\}$
 $||x^{k} - y^{k}||^{2} \le ||x^{k} - y^{k'}||^{2}, \ k \in \{0, ..., N\}, k' \in \{0, ..., N\}.$

Since $x^k = x^{k-1} - \frac{1}{L}g^{k-1}$, we get $x^k = x^0 - \frac{1}{L}\sum_{l=0}^{k-1}g^l$. Hence, problem (5.23)

may be reformulated as follows,

$$\max \frac{f^{N} - f^{*}}{f^{0} - f^{*}}$$
s.t. $\{x^{0} - \frac{1}{L} \sum_{l=0}^{k-2} g^{l}, g^{k}, f^{k}\} \cup \{y^{k}, 0, f^{*}\}$ satisfy interpolation constraints (2.8)
 $f^{k} \ge f^{*} \ k \in \{1, ..., N\}$
(5.24)
 $f^{k} - f^{*} \ge \frac{\mu_{q}}{2} ||x^{0} - \frac{1}{L} \sum_{l=1}^{k-2} g^{l} - y^{k}||^{2}, \ k \in \{0, ..., N\}$
 $||x^{0} - \frac{1}{L} \sum_{l=0}^{k-2} g^{l} - y^{k}||^{2} \le ||x^{0} - \frac{1}{L} \sum_{l=0}^{k-2} g^{l} - y^{k'}||^{2}, \ k, k' \in \{0, ..., N\}.$

The next theorem provides an upper bound for problem (5.24) by using weak duality.

Theorem 5.13. Let $f \in \mathcal{H}_{-L,L}(\mathbb{R}^n)$ and let f have a quadratic functional growth on $X = \{x : f(x) \le f(x^0)\}$ with $\mu_q \in (\frac{L}{2}, L)$. If $t_k = \frac{1}{L}$, $k \in \{1, ..., N\}$, then we have the following convergence rate for Algorithm 5.1,

$$f(x^{N}) - f^{\star} \leq \frac{L}{\mu_{q}} \left(2 - \frac{2\mu_{q}}{L} \right)^{N} \left(f(x^{0}) - f^{\star} \right).$$
(5.25)

Proof. The proof is analogous to that of Theorem 5.4. Without loss of generality,

we may assume that $f^* = 0$. By some algebra, one can show that

ът

$$\begin{split} f^{N} - f^{\star} &- \frac{L}{\mu_{q}} \left(2 - \frac{2\mu_{q}}{L} \right)^{N} \left(f^{0} - f^{\star} \right) + \sum_{j=0}^{N} \left(2^{N-j} \left(1 - \frac{\mu_{q}}{L} \right)^{N-1} \right) \times \\ & \left(f^{\star} - f^{j} - \left\langle g^{j}, y^{0} - x^{0} + \frac{1}{L} \sum_{l=0}^{j-1} g^{l} \right\rangle - \frac{1}{2L} \left\| g^{j} \right\|^{2} + \frac{L}{4} \left\| y^{0} - x^{0} + \frac{1}{L} \sum_{l=0}^{j-1} g^{l} + \frac{1}{L} g^{j} \right\|^{2} \right) + \\ & \sum_{i=1}^{N-1} \sum_{j=i}^{N} \left(2^{N-j} \left(\frac{\mu_{q}}{L} \right) \left(1 - \frac{\mu_{q}}{L} \right)^{N-i-1} \right) \left(f^{\star} - f^{j} - \left\langle g^{j}, y^{i} - x^{0} + \frac{1}{L} \sum_{l=0}^{j-1} g^{l} \right\rangle - \\ & \frac{1}{2L} \left\| g^{j} \right\|^{2} + \frac{L}{4} \left\| y^{i} - x^{0} + \frac{1}{L} \sum_{l=0}^{j-1} g^{l} + \frac{1}{L} g^{j} \right\|^{2} \right) + \sum_{j=1}^{N-1} \left(2^{N-j} \left(1 - \frac{\mu_{q}}{L} \right)^{N-j-1} \right) \times \\ & \left(f^{j} - f^{\star} - \frac{\mu_{q}}{2} \left\| y^{j} - x^{0} + \frac{1}{L} \sum_{l=0}^{j-1} g^{l} \right\|^{2} \right) + \left(2^{N} \left(1 - \frac{\mu_{q}}{L} \right)^{N-1} + \frac{L}{\mu_{q}} \left(2 - \frac{2\mu_{q}}{L} \right)^{N} \right) \times \\ & \left(f^{0} - f^{\star} - \frac{\mu_{q}}{2} \left\| y^{0} - x^{0} \right\|^{2} \right) = - \left(\frac{L}{4} \left(1 - \frac{\mu_{q}}{L} \right)^{N-1} \left\| y^{0} - x^{0} + \frac{1}{L} \sum_{l=0}^{N} g^{l} \right\|^{2} \right) - \\ & \sum_{i=1}^{N-1} \left(\frac{\mu_{q}}{4} \left(1 - \frac{\mu_{q}}{L} \right)^{N-i-1} \left\| y^{i} - x^{0} + \frac{1}{L} \sum_{l=0}^{N} g^{l} \right\|^{2} \right) \le 0. \end{split}$$

By using the above inequality, we get

$$f^{N}-f^{\star} \leq \frac{L}{\mu_{q}} \left(2-\frac{2\mu_{q}}{L}\right)^{N} \left(f^{0}-f^{\star}\right),$$

for any feasible point of (5.24), and the proof is complete.

By doing some calculus, one can verify the following inequality

$$\frac{2L^2 - 2\left(\mu_q - \frac{L}{2}\right)^2}{2L^2 + \left(\mu_q - \frac{L}{2}\right)^2} \ge \left(2 - \frac{2\mu_q}{L}\right), \ \mu_q \in \left(\frac{L}{2}, L\right).$$

Hence, Theorem 5.13 provides a tighter bound than (5.22).

Definition 5.14. [HSS20, Definition 1] Let $\gamma \in (0, 1]$ and $\mu_s \ge 0$. A function f is called (γ, μ_s) -quasar-convex on $X \subseteq \mathbb{R}^n$ with respect to $x^* \in \operatorname{argmin}_{x \in \mathbb{R}^n} f(x)$ if

$$f(x) + \frac{1}{\gamma} \langle \nabla f(x), x^* - x \rangle + \frac{\mu_s}{2} \| x^* - x \|^2 \le f^*, \quad \forall x \in X.$$
(5.26)

The class of quasar-convex functions is large. For instance, non-negative homogeneous functions are (1,0)-quasar-convex on \mathbb{R}^n . (Recall that a function

 \square

 $f : \mathbb{R}^n \to \mathbb{R}$ is called homogeneous of degree k if $f(\alpha x) = \alpha^k f(x)$ for all $x \in \mathbb{R}^n, \alpha \in \mathbb{R}$.) Indeed, if f is non-negative homogeneous of degree $k \ge 1$, by the Euler identity, we have

$$f(x) + \langle \nabla f(x), x^* - x \rangle = (1 - k)f(x) \le 0, \ \forall x \in \mathbb{R}^n,$$

where $x^* = 0$. In what follows, we list some convergence results concerning quasar-convex functions for Algorithm 5.1.

Theorem 5.15. [BM20, Remark 4.3] Let f be L-smooth and let f be (γ, μ_s) -quasarconvex on $X = \{x : f(x) \le f(x^0)\}$. If $t_1 = \frac{1}{L}$ and if x^1 is from Algorithm 5.1, then

$$f(x^{1}) - f^{\star} \leq \left(1 - \frac{\gamma^{2} \mu_{s}}{L}\right) \left(f(x^{0}) - f^{\star}\right).$$
(5.27)

In the following theorem, we state the relationship between quasar-convexity and other concepts. Before we get to the theorem, we recall star convexity. A set X is called star convex at x^* if

$$\lambda x + (1 - \lambda)x^* \in X, \ \forall x \in X, \forall \lambda \in [0, 1].$$

Theorem 5.16. Let x^* be the unique solution of problem (5.1) and let $X = \{x : f(x) \le f(x^0)\}$. If X is star convex at x^* , then we have the following implications:

i) (5.26) \Rightarrow (5.16) with $\mu_g = \frac{\mu_s \gamma}{2} + \frac{\mu_s \gamma^2}{4}$.

ii) (5.16)
$$\Rightarrow$$
 (5.26) with $\mu_s = \ell - \frac{L}{2}$ and $\gamma = \frac{\mu_g}{\ell}$ for each $\ell \in (\max(\frac{L}{2}, \mu_g), \infty)$.

iii) (5.26)
$$\Rightarrow$$
 (5.4) with $\mu_p = \mu_s \gamma^2$.

Proof. The proof of *i*) is similar in spirit to the proof of Theorem 1 in [NNG19]. Let $x \in X$. By the fundamental theorem of calculus and (5.26),we have

$$\begin{split} f(x) - f(x^*) &= \int_0^1 \frac{1}{\lambda} \langle \nabla f(\lambda x + (1 - \lambda)x^*), \lambda x + (1 - \lambda)x^* - x^* \rangle d\lambda \\ &\geq \int_0^1 \frac{\gamma}{\lambda} \Big(f(\lambda x + (1 - \lambda)x^*) - f(x^*) + \frac{\mu_s \lambda^2}{2} \|x - x^*\|^2 \Big) d\lambda \\ &\geq \frac{\gamma \mu_s}{4} \|x - x^*\|^2, \end{split}$$

where the last inequality follows from the global optimality of x^* . By summing $f(x) - f(x^*) \ge \frac{\gamma \mu_s}{4} ||x - x^*||^2$ and (5.26), we get the desired inequality. Now, we prove part *ii*). Let $x \in \mathbb{R}^n$ and $\ell \in (\max(\frac{L}{2}, \mu_g), \infty)$. By (5.2), we have

$$f(x) \le f(x^*) + \frac{L}{2} ||x - x^*||^2.$$
(5.28)

By using (5.28) and (5.16), we get

$$f(x) + \left(\frac{\ell}{\mu_g}\right) \langle \nabla f(x), x^* - x \rangle + \left(\ell - \frac{L}{2}\right) \|x - x^*\|^2 \le f(x^*).$$

For the proof of *iii*), we refer the reader to [BM20, Lemma 3.2].

By combining Theorem 5.4 and Theorem 5.16, under the assumptions of Theorem 5.15, one can get the following convergence rate for Algorithm 5.1 with $t_1 = \frac{1}{L}$,

$$f(x^{1}) - f^{\star} \leq \left(\frac{2L - 2\mu_{s}\gamma^{2}}{2L + \mu_{s}\gamma^{2}}\right) (f(x^{0}) - f^{\star}),$$

which is tighter than the bound given in Theorem 5.15.

5.4 Concluding remarks

In this chapter we studied the convergence rate of the gradient method with fixed step lengths for smooth functions satisfying the PŁ inequality. We gave a new linear convergence rate, which is sharper than known bounds in the literature. One important question which remains to be addressed is the computation of the tightest convergence rate bound for Algorithm 5.1. Moreover, the performance analysis of fast gradient methods, like Algorithm 5.2, for these classes of functions that are discussed in this chapter may also be of interest.

We only studied the linear convergence in terms of the convergence of objective values. However, one can also infer the linear convergence in terms of distance to the solution set or the norm of the gradient by using our results. For instance, under the assumption of Theorem 5.4, we have

$$\frac{\mu_p}{2}d_{X^*}^2(x^k) \le f(x^k) - f^* \le \gamma^k \left(f(x^0) - f^*\right) \le \frac{L\gamma^k}{2}d_{X^*}^2(x^0),$$

where the first inequality follows from Theorem 5.12, γ is the linear convergence rate given in Theorem 5.4, and the last inequality resulted from (5.2). Hence,

$$d_{X^{\star}}^2(x^k) \leq \frac{L\gamma^k}{\mu_p} d_{X^{\star}}^2(x^0).$$

 \square

Moreover, the quadratic gradient growth is a necessary and sufficient conditions for the linear convergence in terms of distance to the solution set; see [ZAdK24, Theorem 3.4] and Theorem 7.10.

The object of mathematics is the honor of the human spirit. Carl Gustav Jacob Jacobi

Convergence rate analysis of the randomized and cyclic coordinate descent methods for convex optimization through semidefinite programming

Preamble

In line with the previous two chapters, in this chapter we study a derivative of the gradient descent method, known as the coordinate descent method. We study randomized and cyclic coordinate descent for convex unconstrained optimization problems. We improve the known convergence rates in some cases by using the numerical semidefinite programming performance estimation method. As a spin-off we provide a method to analyse the worst-case performance of the Gauss–Seidel iterative method for linear systems where the coefficient matrix is positive semidefinite with a positive diagonal. Moreover, we study weighted Jacobi method for solving quadratic programming problems and revisit some well-known results in the literature. This chapter is based on the paper [AdKZ23b], except for Section 6.3.2 which deals with the Jacobi method for solving linear system of

equations.

6.1 Introduction

We consider the unconstrained optimization problem

$$f^* = \inf_{x \in \mathbb{R}^n} f(x), \tag{6.1}$$

where $f : \mathbb{R}^n \to \mathbb{R}$ is convex. We assume that f attains its infimum and f^* denotes the optimal value. In addition, we assume that f is an *L*-smooth function, that is,

$$\|\nabla f(y) - \nabla f(x)\| \le L \|y - x\|, \quad \forall y, x \in \mathbb{R}^n.$$

Moreover, we denote the component Lipschitz constants by ℓ_i ($i \in \{1, ..., n\}$), i.e.,

$$|[\nabla f(x+te_i)]_i - [\nabla f(x)]_i| \le \ell_i |t|, \quad \forall x \in \mathbb{R}^n, t \in \mathbb{R},$$
(6.2)

where e_i is the *i*th standard unit vector. Let $\ell_{\max} := \max_{1 \le i \le n} \ell_i$, and note that $\ell_{\max} \le L \le n\ell_{\max}$ [Wri15].

Due to the simplicity and small per-iteration cost, *coordinate descent methods* have been employed extensively for large-scale optimization problems [Nes12, Wri15].

The generic coordinate descent method is shown in Algorithm 6.1.

Algorithm 6.1 Generic coordinate descent

Set N and $\{t_k\}_{k=0}^{N-1}$ (step lengths) and pick $x^0 \in \mathbb{R}^n$. For k = 0, 1, ..., N - 1 perform the following step: 1. Choose an index i_k from $\{1, 2, ..., n\}$. 2. $x^{k+1} = x^k - t_k [\nabla f(x^k)]_{i_k} e_{i_k}$.

In this chapter, we revisit the worst-case convergence rate analysis for Algorithm 6.1 for two of the best known variants, namely *randomized coordinate descent*, and *cyclic coordinate descent*. In the former, the index i_k is chosen uniformly at random from $\{1, 2, ..., n\}$, and in the latter, the cyclic ordering, (1, 2, ..., n, 1, 2, ..., n, ...), is used.

We will improve the best-known convergence rates from the literature for some specific values of the parameters n, L, N, t_k for $k \in \{0, 1, ..., N-1\}$ and ℓ_i for $i \in \{1, ..., n\}$. Finally, the Gauss–Seidel iterative method for positive semidefinite linear systems is a special case cyclic coordinate descent for convex quadratic

functions, and we will investigate the implications of our analysis for this classical method as well.

Recently, Taylor and Bach [TB19, Appendix I] and Kamri et al. [KHG23] studied the convergence of the coordinate descent algorithm using the semidefinite programming (SDP) performance estimation method. We will also use SDP performance estimation in our analysis, but in a different way than Taylor and Bach [TB19, Appendix I], and our main contribution may be seen as the extension and refinement of the approach by Kamri et al. [KHG23]. For general background information on SDP, see e.g. [WSV12].

6.2 Convergence rate of randomized coordinate descent

The *randomized coordinate descent method* is shown in Algorithm 6.2 for easy reference.

Algorithm 6.2 Randomized coordinate descent
Set N and $\{t_k\}_{k=0}^{N-1}$ (step lengths) and pick $x^0 \in \mathbb{R}^n$.
For $k = 0, 1,, N - 1$ perform the following step:
1. Choose index i_k with uniform probability from $\{1, 2,, n\}$.
2. $x^{k+1} = x^k - t_k [\nabla f(x^k)]_{i_k} e_{i_k}$.

We proceed to revisit its worst-case convergence rate for three classes for function, namely convex *L*-smooth functions, convex quadratic functions, and strongly convex, *L*-smooth functions.

6.2.1 The case of *L*-smooth functions

Regarding the convergence of Algorithm 6.2 for *L*-smooth convex functions, the following is known. (We state the result as in the survey [Wri15, Theorem 1], but it is originally due to Nesterov [Nes12]).

Theorem 6.1. [Wri15, Theorem 1] Let $f : \mathbb{R}^n \to \mathbb{R}$ is an L-smooth convex function for some L > 0. If $t_k = \frac{1}{\ell_{\max}}$ for all k, then, for each k > 0,

$$\mathbb{E}\left(f(x^k)\right) - f^* \le \left(\frac{2n\ell_{\max}}{k}\right) R_0^2,\tag{6.3}$$

where R_0 satisfies $\max_{x^* \in \mathbb{S}} \max_x \{ \|x - x^*\| : f(x) \le f(x^0) \} \le R_0$ and \mathbb{S} denotes the optimal solution set.

In this section, we study the behaviour of the randomized coordinate descent method for *L*-smooth convex functions. The worst-case convergence rate of the Algorithm 6.2 can be formulated as follows.

$$\max \mathbb{E}[f(x^{N})] - f(x^{*})$$

s.t. $f \in \mathcal{F}_{0,L}(\mathbb{R}^{n})$
f satisfies (6.2) for every $x \in \mathbb{R}^{n}$ for some $\ell_{i}, i \in \{1, ..., n\}$ (6.4)
 $||x^{0} - x^{*}||^{2} \leq \Delta$
 $x^{k}, k \in \{1, 2, ..., N\}$, are generated by Algorithm 6.2 with respect to x^{0}
and step length t_{k}
 $x^{0} \in \mathbb{R}^{n}, \nabla f(x^{*}) = 0$,

where f, x^k, x^* are decision variables and t, L, n and $\ell_i, i \in \{1, ..., n\}$, are the given parameters. Problem (6.4) in general is intractable. Moreover, note that x^k depends on the index i_k which is chosen uniformly at random from the set $\{1, ..., n\}$ therefore (6.4) is a stochastic programming problem. To deal with this we introduce a random variable d^k which depends on the index i_k and is defined by $d^k := [\nabla f(x^k)]_{i_k} e_{i_k}$. Note that d^k has the following properties:

$$\mathbb{E}\left[\|d^{k}\|^{2}\right] = \frac{1}{n} \mathbb{E}\left[\|\nabla f(x^{k})\|^{2}\right]$$
$$\mathbb{E}\left[\left\langle d^{k}, \nabla f(x^{k})\right\rangle\right] = \frac{1}{n} \mathbb{E}\left[\|\nabla f(x^{k})\|^{2}\right]$$
$$\mathbb{E}\left[\left\langle d^{k}, x^{k}\right\rangle\right] = \frac{1}{n} \mathbb{E}\left[\left\langle \nabla f(x^{k}), x^{k}\right\rangle\right],$$
(6.5)

where the expectation again refers to the joint distribution of all the random variables d^k for $k \in \{0, 1, ..., N\}$ and $x^k, \nabla f(x^k), f(x^k)$ for $k \in \{0, 1, ..., N\}$. By Taylor's theorem and (6.2), we have

$$f(x^{k+1}) \le f(x^k) + \left\langle \nabla f(x^k), x^{k+1} - x^k \right\rangle + \frac{\ell_{\max}}{2} ||x^{k+1} - x^k||^2$$

$$f(x^k) \le f(x^{k+1}) + \left\langle \nabla f(x^{k+1}), x^k - x^{k+1} \right\rangle + \frac{\ell_{\max}}{2} ||x^k - x^{k+1}||^2,$$
(6.6)

where $\ell_{\max} = \max_{i \in \{1,...,n\}} \ell_i$ as before. Therefore, the relaxation of problem (6.4)

is given by

$$\max \mathbb{E}[f(x^{N})] - f(x^{*})$$
s.t. $\{(x^{k}; \nabla f(x^{k}); f(x^{k}))\}$ satisfy (2.4) for $k \in \{0, 1, \dots, N, *\}$ w.r.t. $\mu = 0, L$
 $\{(x^{k}; \nabla f(x^{k}); f(x^{k}))\}$ satisfy (6.6) for $k \in \{0, 1, \dots, N\}$ w.r.t. ℓ_{\max}
 $\{x^{k}; \nabla f(x^{k}); d^{k}\}$ satisfies (6.5) $(k \in \{0, \dots, N\})$ (6.7)
 $||x^{0} - x^{*}||^{2} \le \Delta$
 $x^{k+1} = x^{k} - t_{k}d^{k}$
 $x^{0} \in \mathbb{R}^{n}, \nabla f(x^{*}) = 0,$

where $f(x^k), x^k, x^*, \nabla f(x^k)$ and d^k are decision variables. Note that because the problem (6.1) is invariant under translation, without loss of generality we may assume that x^* is the zero vector. Since $x^{k+1} = x^k - t_k d^k$ is a recursive relation, x^k can be written as linear combination of x^0 and d^i s. In this way, all the unknowns appear as entries in the following matrix:

$$G = \mathbb{E} \Big[\operatorname{Gram} \left(x^{0}, \nabla f(x^{0}), \cdots, \nabla f(x^{0}), d^{0}, \cdots, d^{N} \right) \Big]$$

$$= \begin{pmatrix} \mathbb{E} [\|x^{0}\|^{2}] & \mathbb{E} [\langle x^{0}, \nabla f(x^{0}) \rangle] & \cdots & \mathbb{E} [\langle x^{0}, d^{N} \rangle] \\ \mathbb{E} [\langle \nabla f(x^{0}), x^{0} \rangle] & \mathbb{E} [\|\nabla f(x^{0})\|^{2}] & \cdots & \mathbb{E} [\langle \nabla f(x^{0}), d^{N} \rangle] \\ \vdots & \vdots & \ddots & \vdots \\ \mathbb{E} [\langle \nabla f(x^{N}), x^{0} \rangle] & \mathbb{E} [\langle \nabla f(x^{N}), \nabla f(x^{0}) \rangle] & \cdots & \mathbb{E} [\langle \nabla f(x^{N}), d^{N} \rangle] \\ \mathbb{E} [\langle d^{0}, x^{0} \rangle] & \mathbb{E} [\langle d^{0}, \nabla f(x^{0}) \rangle] & \cdots & \mathbb{E} [\langle d^{0}, d^{N} \rangle] \\ \vdots & \vdots & \ddots & \vdots \\ \mathbb{E} [\langle d^{N}, x^{0} \rangle] & \mathbb{E} [\langle d^{N}, \nabla f(x^{0}) \rangle] & \cdots & \mathbb{E} [\|d^{N}\|^{2}] \end{pmatrix}$$

Note that *G* is the expectation of a random Gram matrix. Since every realization of this random matrix is positive semidefinite, and the expectation preserves positive semidefiniteness, it follows that *G* is positive semidefinite as well. Therefore, problem (6.7) can be written as an SDP problem, where the variables are *G* and $\mathbb{E}[f(x^i)]$.

In what follows we compare the convergence rate derived by solving the problem (6.7) and the bound by Wright (6.3) for some specific values of the parameters n, L, N, t_k for $k \in \{0, 1, \dots, N\}$ and ℓ_i for $i \in \{1, \dots, n\}$. All the figures in this chapter were obtained by solving the SDP problems with the solver Mosek [ApS19], using the Yalmip [Löf04] Matlab interface.



Figure 6.1: Convergence rate for Algorithm 6.2 computed by performance estimation problem (6.7) (dashed lines) and the bound given by (6.3) (thick lines) for L = 2, $l_{\text{max}} = 1$, t = 1, $\Delta = 1$ and different *n*



Figure 6.2: Convergence rate for Algorithm 6.2 computed by performance estimation problem (6.7) (dashed lines) and the bound given by (6.3) (thick lines) for L = 4, $l_{\text{max}} = 2$, t = 0.5, $\Delta = 1$ and different *n*

Note that the convergence rate provided by solving performance estimation is strictly better than the bound given by Wright. In other words, the bound (6.3) is not tight for the values of the parameters that we considered. Moreover, the bound given by performance estimation can also be calculated for different step lengths than the fixed step lengths $1/\ell_{\text{max}}$ in the bound (6.3).

6.2.2 The case of convex quadratic functions

In this section we study the convergence rate of the randomized coordinate descent method in case that the objective function is a *quadratic function* of the form

$$\min_{x \in \mathbb{R}^n} f(x) := \frac{1}{2} x^\top A x - b^\top x, \tag{6.8}$$

where *A* is a symmetric positive semidefinite matrix. To study this case we need to add an additional constraint to restrict our model to quadratic functions. The following necessary condition for f to be a quadratic function can be verified easily, and has been used in SDP performance analysis by Drori et al [DS20]

$$\frac{1}{2}\langle \nabla f(x) - \nabla f(y), x - y \rangle = f(x) - f(y) - \langle \nabla f(y), x - y \rangle.$$

Since this constraint holds for every point in the domain we just consider the relaxed constraint that only holds for the point generated by the method in addition to the initial point and the optimal point. In this case we add the following constraint to the problem (6.7):

$$\frac{1}{2}\langle \nabla f(x^i) - \nabla f(x^j), x^i - x^j \rangle = f(x^i) - f(x^j) - \langle \nabla f(x^j), x^i - x^j \rangle \quad \forall i, j \in \{0, \dots, N, *\}.$$
(6.9)

In what follows we compare the convergence rate of the randomized coordinate descent method for the general problem (6.7) to the convergence rate for quadratic problems.



Figure 6.3: Convergence rate for Algorithm 6.2 computed by performance estimation problem for quadratic functions (red line) and the bound given by (6.7) (blue line) for $n = 10, L = 2, l_{max} = 1, t = 1, \Delta = 1$.



Figure 6.4: Convergence rate for Algorithm 6.2 computed by performance estimation problem for quadratic functions (red line) and the bound given by (6.7) (blue line) for n = 2, L = 2, $l_{max} = 1$, t = 1, $\Delta = 1$.

Note that the convergence rate for the quadratic problem is sightly better than that of the general case.
6.2.3 The case of μ -strongly convex *L*-smooth functions

In this section, we study the convergence rate of the μ -strongly convex *L*-smooth functions. If $\mu > 0$ the optimal value of problem (6.7) for one iteration of Algorithm 6.2, i.e. N = 1, appears to be the same as the following bound (6.10) given by Wright [Wri15].

Theorem 6.2. [Wri15, Theorem 1] Let $f \in \mathcal{F}_{\mu,L}(\mathbb{R}^n)$. If $t_k = \frac{1}{\ell_{\max}}$ for each k, and $\mu > 0$, then, for all N > 0,

$$\mathbb{E}\left[f(x^N)\right] - f^* \le \left(1 - \frac{\mu}{n\ell_{\max}}\right)^N \left(f(x^0) - f^*\right).$$
(6.10)

The fact that the two bounds seem to coincide does not suggest that (6.10) is tight, since the SDP bound (6.7) is a relaxation, and not exact.

6.3 Cyclic coordinate descent

Cyclic coordinate descent is one of the most important coordinate descent algorithms due to its simplicity. The convergence rate of *cyclic coordinate descent method* for the class of *L*-smooth convex functions is studied by Kamri et al using the performance estimation method [KHG23]. This method is described in Algorithm 6.3.

Algorithm 6.3 Cyclic coordinate descent

Set number of cycles K, $\{t_k\}_{k=0}^{N-1}$ (step lengths), pick $x^0 \in \mathbb{R}^n$ and set N = nK. For k = 0, 1, 2, ..., N - 1 perform the following step: 1. Set $i = k \pmod{n} + 1$ 2. $x^{k+1} = x^k - t_k [\nabla f(x^k)]_i e_i$.

In each iteration the method updates the current point over one of the coordinates in cyclic order.

The following result is known about the rate of convergence. We present it as in [Wri15], but it is originally due to Beck and Tetruashvili [BT13].

Theorem 6.3. [Wri15, Theorem 3] Let $f : \mathbb{R}^n \to \mathbb{R}$ be an L-smooth convex function for some L > 0. If $t_k = \frac{1}{\ell_{\max}}$ for all k, then, for $N = n, 2n, 3n, \ldots$,

$$f(x^{N}) - f^{*} \leq \left(\frac{4nR_{0}^{2}\ell_{\max}(1 + nL^{2}/\ell_{\max}^{2})}{N + 8}\right),$$
(6.11)

where R_0 satisfies $\max_{x^* \in \mathbb{S}} \max_x \{ \|x - x^*\| : f(x) \le f(x^0) \} \le R_0$ and \mathbb{S} denotes the optimal solution set. If f is also strongly convex with parameter $\mu > 0$, then one has, for $k = n, 2n, 3n, \ldots$,

$$f(x^{k}) - f^{\star} \leq \left(1 - \frac{\mu}{2\ell_{\max}(1 + nL^{2}/\ell_{\max}^{2})}\right)^{k/n} \left(f(x^{0}) - f^{\star}\right).$$

For easy reference, we recall the interpolation conditions from Theorem 2.8 in the case that $\mu = 0$: The set $\{x^i, \nabla f(x^i), f(x^i)\}$ for $i \in \{0, 1, \dots, N, \star\}$ is $\mathcal{F}_{0,L}$ -interpolable if and only if

$$f(x^{i}) \ge f(x^{j}) + \langle \nabla f(x^{j}), x^{i} - x^{j} \rangle + \frac{1}{2L} \| \nabla f(x^{i}) - \nabla f(x^{j}) \|^{2}, \quad \forall i, j \in \{0, 1, \cdots, N, \star\}.$$
(6.12)

Using these conditions, we may formulate the worst-case convergence rate as performance estimation problem:

$$\max f(x^{N}) - f(x^{*})$$

s.t. { $(x^{i}; \nabla f(x^{i}); f(x^{i}))$ } satisfy (6.12) for $i \in \{0, 1, \dots, N, *\}$ w.r.t. L
 $||x^{0} - x^{*}||^{2} \le \Delta$ (6.13)
 x^{k} ($k \in \{1, 2, \dots, N\}$) are generated using Algorithm 6.3
 $x^{0} \in \mathbb{R}^{n}, \nabla f(x^{*}) = 0.$

Problem (6.13) can be formulated as a semidefinite programming problem, and this is precisely what was done by Kamri et al. [KHG23].

Since the univariate function $t \mapsto f(x^k + te_i)$ is convex and ℓ_i -smooth, it follows from (6.12) that, for every two consecutive points x^k and x^{k+1} generated by Algorithm 6.3, the following inequalities hold if $i = k \pmod{n} + 1$:

$$f(x^{k}) \ge f(x^{k+1}) + \nabla f(x^{k+1})_i (x_i^k - x_i^{k+1}) + \frac{1}{2\ell_i} (\nabla f(x^k)_i - \nabla f(x^{k+1})_i)^2 \quad (6.14)$$

$$f(x^{k+1}) \ge f(x^k) + \nabla f(x^k)_i (x_i^{k+1} - x_i^k) + \frac{1}{2\ell_i} (\nabla f(x^{k+1})_i - \nabla f(x^k)_i)^2.$$

By adding the above inequalities to (6.13) one can get a better upper bound for the worst-case convergence rate, i.e.

$$\max f(x^{N}) - f(x^{*})$$
s.t. $\{(x^{i}; \nabla f(x^{i}); f(x^{i}))\}$ satisfy (6.12) for $i \in \{1, \dots, N, *\}$ w.r.t. L
 $\{(x^{i}; \nabla f(x^{i}); f(x^{i}))\}$ satisfy (6.14) for $i \in \{1, \dots, N, *\}$ w.r.t. $\{\ell_{1}, \dots, \ell_{n}\}$
 $||x^{0} - x^{*}||^{2} \le \Delta$ (6.15)
 $x^{k} \ (k \in \{1, \dots, N\})$ are generated using Algorithm 6.3
 $x^{0} \in \mathbb{R}^{n}, \ \nabla f(x^{*}) = 0.$

In order to obtain an SDP relaxation to (6.15), we proceed in the same way as Kamri et al. [KHG23]. We view (6.15) as a quadratically constrained quadratic program (QCQP) in variables corresponding to the unknowns

$$x_i^k, \frac{\partial f(x^k)}{\partial x_i}, f(x^k) \quad k \in \{0, 1, 2, \cdots, N\}, \ i \in \{1, \dots, n\},$$

Next we use the following relations to eliminate variables:

$$x_i^{k+1} - x_i^k = \begin{cases} -t_k \frac{\partial f(x_i^k)}{\partial x_i} & \text{if } i = k \pmod{n} + 1\\ 0 & else \end{cases}$$

which hold for all $k \in \{0, 1, 2, \dots, N-1\}$, and $i \in \{1, \dots, n\}$. Subsequently we form the standard Shor SDP relaxation (see e.g. [WKK20]) of the resulting QCQP. Note that this is different to the approach we followed for randomized coordinate descent. In particular, the size of the SDP relaxation now depends on n, which was not the case before. This also limits the parameter values for which we may solve the SDP relaxations.

In Figure 6.5 we compare the SDP upper bounds from (6.13) and (6.15) for various parameter values.

The figure shows that the bound can be improved slightly by adding the set of constraints (6.14) to the model provided by Kamri et al. [KHG23]. Moreover, we add the constraint which correspond to the quadratic functions (6.9) to the model (6.15) which provides us with a better bound for quadratic functions. The computed values are much better that the theoretical bound (6.11), to the extent that we do not include this bound in the plot. Indeed, Kamri et al. [KHG23] already mentioned in their paper that the computed values for their model are much better than the theoretical bound (6.11).



Figure 6.5: Convergence rate Algorithm 6.3 computed by performance estimation problem (6.15) (blue), the bound given by (6.13) (red) and the bound for quadratic functions (green) for $n = 2, L = 2, \ell_1 = 1, \ell_2 = 1, t = 0.5, \Delta = 1$.



Figure 6.6: Convergence rate for Algorithm 6.3 computed by performance estimation problem of quadratic functions for $\ell_i = 1$ $i \in \{1, \dots, n\}, L = \sum_{i=1}^n \ell_i, t = 1, \Delta = 1$ and different *n*.

Note that our discussion for coordinate-wise cyclic coordinate descent in this section could be extended to block-wise cyclic coordinate descent in a similar way as was done by Kamri et al. [KHG23].

6.3.1 Relation to the Gauss–Seidel method

The minimization of the convex quadratic function in (6.8) is equivalent to the solution of the linear system Ax = b. Here, we may assume w.l.o.g. that *A* has a positive diagonal. Cyclic coordinate decent for problem (6.8) is closely related to the iterative Gauss–Seidel method for solving this linear system. For this reason, cyclic coordinate descent is sometimes also referred to as *nonlinear Gauss–Seidel*. It is therefore an interesting question whether the SDP performance estimation framework yields any new insights on the performance of the *Gauss–Seidel method*.

Denoting $A = (a_{ij})$, the iterative Gauss–Seidel method may be described as follows.

Algorithm 6.4 Gauss–Seidel method	
Set <i>N</i> and pick $x^0 \in \mathbb{R}^n$.	
For $k = 0, 1,, N - 1$ perform the following:	
$x_i^{k+1} = \frac{1}{a_{ii}} \left(b_i - \sum_{j=1}^{i-1} a_{ij} x_j^{k+1} - \sum_{j=i+1}^n a_{ij} x_j^k \right)$	$(i=1,\ldots,n).$

This is exactly cyclic coordinate descent with unit step lengths if the gradient at a point x is replaced by $D^{-1}\nabla f(x)$, where $f(x) = \frac{1}{2}x^{\top}Ax - b^{\top}x$ as before, and D is the diagonal matrix with the same diagonal entries as A. To see this, recall that the Fréchet derivative of a differentiable function $f : \mathbb{R}^n \to \mathbb{R}$ at a point $x \in \mathbb{R}^n$ is the unique linear operator, say $D_f(x) : \mathbb{R}^n \to \mathbb{R}$, such that

$$\lim_{\|h\|\to 0} \frac{f(x+h) - f(x) - D_f(x)h}{\|h\|} = 0.$$

Once an inner product on \mathbb{R}^n is fixed, say $\langle \cdot, \cdot \rangle$, one may, by the Riesz representation theorem, express $D_f(x)h = \langle g(x),h \rangle$, where g(x) is called the gradient vector of f at x with respect to $\langle \cdot, \cdot \rangle$. In particular, if $\langle \cdot, \cdot \rangle$ is the Euclidean dot product, then $g(x) = \nabla f(x)$. If one changes to the inner product $\langle \cdot, \cdot \rangle_D$ defined by

$$\langle u, v \rangle_D = \sum_{i=1}^n a_{ii} u_i v_i \qquad (u, v \in \mathbb{R}^n),$$
 (6.16)

then the gradient vector at x becomes $D^{-1}\nabla f(x)$, by the uniqueness of the Fréchet derivative.

It was shown in [DKGT20] that the interpolation condition in Theorem 2.37 holds for any reference inner product $\langle \cdot, \cdot \rangle$, provided that the gradient vector is interpreted accordingly.

In other words, the following SDP performance estimation problem gives a bound on the worst-case performance of the Gauss–Sidel method after N iterations, when A is a symmetric positive semidefinite matrix with a positive diagonal.

$$\max f(x^{N}) - f(x^{*})$$
s.t. { $(x^{i}; \nabla f(x^{i}); f(x^{i}))$ } satisfy (6.12) and (6.9) for $i \in \{0, 1, \dots, N, \star\}$
w.r.t. $L = \lambda_{\max}(D^{-1}A)$
{ $(x^{i}; \nabla f(x^{i}); f(x^{i}))$ } satisfy (6.14) for $i \in \{0, 1, \dots, N, \star\}$
w.r.t. $\ell_{1} = \dots = \ell_{n} = 1$
 $\|x^{0} - x^{*}\|^{2} \le \Delta$ (6.17)
 $x^{k} (k \in \{1, 2, \dots, N\})$ is generated using Algorithm 6.3
 $x^{0} \in \mathbb{R}^{n}, \nabla f(x^{*}) = 0,$

where the inner product is now understood to be the one in (6.16), and the norm the induced norm for this inner product, and $\lambda_{\max}(D^{-1}A)$ denotes the largest eigenvalue of $D^{-1}A$. (Note that the eigenvalues of $D^{-1}A$ are real.) Importantly, the reference inner product is not visible in the SDP performance estimation problem reformulation of (6.17), since only a Gram matrix for this inner product appears. It is therefore equally valid, for any inner product, provided that the inner product and norm are interpreted accordingly. Of course, the Lipschitz constants like (6.2) depend on the norm as well. It is easy to verify that, for the inner product (6.16), and $f(x) = \frac{1}{2}x^{T}Ax - b^{T}x$, one has $\ell_1 = \ldots = \ell_n = 1$ and $L = \lambda_{\max}(D^{-1}A)$ as is used in (6.17).

In summary, we have shown the following.

Theorem 6.4. Consider a solvable system of linear equations Ax = b where A is a symmetric positive semidefinite matrix with positive diagonal, and let x^* denote a solution. Letting $f(x) = \frac{1}{2}x^{\top}Ax - b^{\top}x$, after N iterations of the Gauss–Seidel method, an upper bound on $f(x^N) - f(x^*)$ is given by the optimal value of the SDP problem (6.17), provided that the starting point x^0 satisfies $||x^0 - x^*|| \le \Delta$ for a given Δ , where the norm is the induced norm of the inner product (6.16).

The Gauss–Seidel method is known to be convergent when *A* is symmetric positive-definite, e.g. [GVL13, Theorem 10.1.2], or strictly or irreducibly diagonally dominant, e.g. [Bag95]. The case when *A* is only positive semidefinite (with positive diagonal) seems to be less well-understood, and our approach sheds more

light on this case. In particular, numerical results of the type shown in Figure 6.6 apply here.

6.3.2 Weighted-Jacobi method

This section is dedicated to the studying of the quadratic optimization problem presented in (6.8), where the matrix *A* is both symmetric and positive definite. The challenge of solving (6.8) is inherently similar to the task of solving the subsequent linear equation system Ax = b. As a natural progression from our previous section where we delved into the Gauss–Seidel method, we now turn our attention to the analysis of an algorithm commonly employed for the general solution of linear equation systems.

The matrix $A = [a_{ij}]$ can be expressed as the sum of its components, $A = D + L + L^T$, with *D* representing the diagonal, and *L* denoting the lower triangular part of the matrix *A*. The Jacobi method is a well-known and widely-used approach for solving linear equation systems [Jac46]. In this section, we delve into a comprehensive examination of the Jacobi method and conduct a thorough review of its convergence analysis. The Jacobi method solves the following system in each iteration,

$$x^{k} = (I - D^{-1}A)x^{k-1} + D^{-1}b.$$
(6.18)

It is worth noting that there are certain similarities between the Gauss–Seidel and Jacobi methods. However, the key distinction between them lies in their updates of values from previous generated points. In the Jacobi method (6.18), values from the previous step are used, whereas the Gauss–Seidel method, as demonstrated in Algorithm 6.4, always utilizes the most recent updated values in every iteration, i.e. the Gauss–Seidel method runs the following updates at iteration k + 1,

$$x_i^{k+1} = \frac{1}{a_{ii}} \left(b_i - \sum_{j=1}^{i-1} a_{ij} x_j^{k+1} - \sum_{j=i+1}^n a_{ij} x_j^k \right) \quad \forall i,$$

whereas the Jacobi method performs the following updates at iteration k + 1,

$$x_i^{k+1} = \frac{1}{a_{ii}} \left(b_i - \sum_{j \neq i}^n a_{ij} x_j^k \right) \quad \forall i.$$

It is a well-known that the iteration $x^k = Rx^{k-1} + c$ converges if and only if the spectral radius $\rho(R) < 1$ [Dem97, Chapter 6.5]. In our specific case, where we

consider matrix *A* as symmetric and positive semidefinite, it is sufficient to ensure that $\lambda_{\max}(R) < 1$ since the spectral radius $\rho(R)$ is equivalent to $\lambda_{\max}(R)$. For the Jacobi method, this condition translates to $0 < \rho(D^{-1}A) < 2$, which is a somewhat limiting requirement. To address this issue, the weighted-Jacobi method is introduced, which is detailed in Algorithm 6.5, with the specific choice of $q^k = 0$ for $0 \le k \le N$ [Ric11].

In cases that each step of the *weighted-Jacobi method* is performed inexactly, an adaptation known as the *inexact weighted-Jacobi method* has been introduced; see [GO88, GO82] for more discussion. To be more specific, for a given parameter δ , the value of q_k is determined in each iteration as follows

$$\|q^{k}\| \le \delta \|b - Ax^{k}\|. \tag{6.19}$$

Al	gorithm	6.5	Inexact	Weighted-	Jacobi	method
----	---------	-----	---------	-----------	--------	--------

Set *N* and α (weight) and pick $x^0 \in \mathbb{R}^n$.

For k = 1, 2, ..., N perform the following step:

1. select q_k that satisfies (6.19)

2. $x^{k} = (I - \alpha D^{-1}A)x^{k-1} + \alpha D^{-1}q^{k-1} + \alpha D^{-1}b$

The convergence of the weighted-Jacobi method and its various adaptations has been extensively explored in the literature. You can find a concise overview in [Saa03, Chapter 4]. The following theorems represent well-established convergence results for the Jacobi method.

Theorem 6.5. If A is strictly diagonally dominant $(a_{ii} > \sum_{j=1}^{n} |a_{ij}|$ for all i), the Jacobi method converges.

Theorem 6.6. [E.g. Che] The weighted-Jacobi method converges if and only if $0 < \alpha < \frac{2}{\lambda_{\max}(D^{-1}A)}$. Moreover, by considering $\alpha^* = \frac{2}{\lambda_{\min}(D^{-1}A) + \lambda_{\max}(D^{-1}A)}$ one get the optimal convergence rate

$$\rho^{\star} = \frac{1 - \kappa_{D^{-1}A}}{1 + \kappa_{D^{-1}A}},$$

where $\kappa_{D^{-1}A} = \frac{\lambda_{\min}(D^{-1}A)}{\lambda_{\max}(D^{-1}A)}$

Now, let's delve into an examination of the convergence rate for the inexact weighted-Jacobi method.

Theorem 6.7. The inexact weighted-Jacobi method with $\delta \ge 0$ has the following convergent rate for a suitable choice of α

$$||e^{k+1}|| \le (||I - \alpha D^{-1}A|| + \alpha \delta ||D^{-1}|| ||A||) ||e^{k}||.$$

where $e^k = x^* - x^k$.

Proof. We define $r^k = b - Ax^k$. By some calculation one can show that $r^k = Ae^k$. By this definition we have

$$e^{k+1} = Ge^k + p^k,$$

where $p^k = -\alpha D^{-1}q^k$, and $G = I - \alpha D^{-1}A$. It can be easily seen that

$$\|p^{k}\| = \|\alpha D^{-1}q^{k}\| \le \alpha \delta \|D^{-1}\| \|b - Ax^{k}\| = \alpha \delta \|D^{-1}\| \|Ae^{k}\|.$$

Therefore,

$$||e^{k+1}|| = ||Ge^k + p^k|| \le (||G|| + \alpha\delta||D^{-1}|||A||)||e^k||,$$

which completes the proof.

It is evident that when $\delta = 0$, Theorem 6.7 simplifies to the weighted-Jacobi method, and the convergence rate mirrors that of Theorem 6.6. In other words, by maintaining the induced norm as two, you can establish $||I - \alpha D^{-1}A|| = 1 - \alpha \lambda_{\min}(D^{-1}A)$. Setting α as defined in Theorem 6.6 yields an equivalent convergence rate, while the performance estimation method offers the following convergence rate.

Theorem 6.8 (Based on Theorem 5.3. [DKGT20]). Consider the inexact Jacobi method with fixed step size α where $\mu = \lambda_{\min}(D^{-1}A)$ and $L = \lambda_{\max}(D^{-1}A)$. If $\delta \in [0, \frac{2\mu}{L+\mu}]$, and $\alpha \in [0, \frac{2\mu-\delta(L+\mu)}{(1-\delta)\mu(L+\mu)}]$, one has

$$\|x^{1} - x^{\star}\|_{D} \le (1 - (1 - \delta)\mu\alpha) \|x^{0} - x^{\star}\|_{D}.$$
(6.20)

Golub and Overton in [GO82] investigate the convergence of the inexact secondorder *Richardson method*. Notably, in their analysis by setting $\omega = 1$ and M = D, one can derive the inexact Jacobi method. In what follows, we aim to derive the convergence rate of the inexact Jacobi method based on their analysis. Note that here, $\|\cdot\|$ represents the Euclidean norm for vectors and the *spectral norm* for matrices.

$$||x|| = \langle x, x \rangle^{\frac{1}{2}}, \quad ||A|| = \max_{||x||=1} ||Ax||.$$

Let us introduce the following definitions $e^k = x^* - x^k$ and $r^k = b - Ax^k$. Just as in the previous case, we have $r^k = Ae^k$ and $e^{k+1} = Ge^k + p^k$, where $p^k = -\alpha D^{-1}q^k$ and $G = I - \alpha D^{-1}A$. The eigenvector decomposition (SVD-decomposition) of the

$$G = D^{-1/2} (I - \alpha D^{-1/2} A D^{-1/2}) D^{1/2} = D^{-1/2} V \Lambda V^T D^{1/2},$$

where Λ is a diagonal matrix with elements representing the eigenvalues λ_i of the matrix $I - \alpha D^{-1/2} A D^{-1/2}$, and *V* is an orthogonal matrix. Now we introduce

$$\hat{e}^{k} = V^{T} D^{1/2} e^{k}$$
 and $\hat{p}^{k} = -\alpha V^{T} D^{-1/2} q^{k}$.

These result in

matrix G is as follows

$$\hat{e}^{k+1} = \Lambda \hat{e}^k + \hat{p}^k.$$

By the Lemma 1 in [GO82], one gets

$$\hat{e}^k = S^k \hat{e}^1 + \sum_{l=1}^{k-1} S^{k-l} \hat{p}^l,$$

where S^k is diagonal matrix with diagonal elements

$$S_{jj}^k = \lambda_j^{k-1}$$

Define $\rho = \lambda_{\max}$ which is equal to $1 - \alpha \lambda_{\min}(D^{-1}A)$ if $\alpha \in [0, \frac{2}{\lambda_{\max}(D^{-1}A) + \lambda_{\min}(D^{-1}A)}]$. It is easily seen that $||S^k|| \le \rho^{k-1}$. Therefore,

$$\|\hat{e}^{k}\| \le \rho^{k-1} \|\hat{e}^{1}\| + \sum_{l=1}^{k-1} \rho^{k-l-1} \|\hat{p}^{l}\|.$$

By the fact that $\|\hat{p}^k\| \leq \epsilon \|\hat{e}^k\|$ where $\epsilon = \delta \alpha \|D^{-1/2}\| \|AD^{-1/2}\|$ and δ is given by (6.19), the above inequality yields to

$$\|\hat{e}^{k}\| \le \rho^{k-1} \|\hat{e}^{1}\| + \epsilon \sum_{l=1}^{k-1} \rho^{k-l-1} \|\hat{e}^{l}\|.$$

Now, we introduce a new variable, $\tau^k = \rho^{k-1} \|\hat{e}^1\| + \epsilon \sum_{l=1}^{k-1} \rho^{k-l-1} \tau^l$. Through some algebraic manipulation, we find that $\tau^1 = \|\hat{e}^1\|$ by definition. Furthermore, τ^k satisfies the following homogeneous equation

$$\tau^{k+1} = (\rho + \epsilon)\tau^k.$$

Therefore by Theorem 1 in [GO82] one may have the following convergence result.

Theorem 6.9. [GO82] The error norm $\|\hat{e}^k\|$ associated with the kth iterate of the inexact Jacobi method is bounded by

$$\|\hat{e}^{k}\| \le (\rho + \epsilon)^{k} \|\hat{e}^{0}\|.$$
(6.21)

Note that, based on norm properties, we have $\|\hat{e}^k\| = \|x^k - x^*\|_D$. To compare the result from the last theorem with Theorem 6.8, one can set k = 1 in the last theorem, yielding

$$\|\hat{e}^{1}\| \leq \left(\|I - \alpha D^{-1}A\| + \alpha \delta \|D^{-1/2}\| \|AD^{-1/2}\|\right) \|\hat{e}^{0}\|$$

As per the assumptions of Theorem 6.8, if we set $\delta = 0$, both bounds (6.20) and (6.21) become equal. However, when $\delta \neq 0$, one can demonstrate that

$$\alpha \delta \mu \le \alpha \delta \|AD^{-1}\| \le \alpha \delta \|D^{-\frac{1}{2}}\| \|AD^{-\frac{1}{2}}\|.$$
(6.22)

This inequality shows that the bound provided by Theorem 6.8 is better when compared to the bound given by Theorem 6.9. The example below illustrates that the inequality 6.22 can be strict in some cases.

Example 6.10. Let us define $A = \begin{pmatrix} 2 & 1 \\ 1 & 3 \end{pmatrix}$. It is easily seen that $\mu = 1.3820$ and $\|D^{-\frac{1}{2}}\| \|AD^{-\frac{1}{2}}\| = 1.5811$ which shows that for any positive α and δ inequality (6.22) is strict.

6.4 Conclusion

We have studied SDP performance estimation approaches to analyse randomized and cyclic coordinate descent, thereby complementing recent results in [KHG23]. For randomized coordinate descent, we have given the first known SDP performance estimation bound. For cyclic coordinate descent, we were able to improve slightly on the numerical values given in [KHG23]. Of course, to obtain new rates of convergence in general, it is necessary to solve the SDP performance estimation problems analytically, as opposed to numerically, but we have been unable to obtain analytic solutions for the SDP problems presented in this chapter. In this chapter, we also discussed the link with the Gauss–Seidel method in the case of convex quadratic functions. Moreover, we studied the weighted Jacobi method for solving a linear system of equations which is closely related to the Gauss–Seidel method. Obvious is the most dangerous word in mathematics.

E.T. Bell (1883–1960)

7

Convergence rate analysis of the gradient descent-ascent method for convex-concave saddle-point problems

Preamble

In this chapter, we study the gradient descent-ascent method for convex-concave saddle-point problems. We derive a new non-asymptotic global convergence rate in terms of distance to the solution set by using the semidefinite programming performance estimation method. The given convergence rate incorporates most parameters of the problem and it is exact for a large class of strongly convex-strongly concave saddle-point problems for one iteration. We also investigate the algorithm without strong convexity and we provide some necessary and sufficient conditions under which the gradient descent-ascent method enjoys linear convergence. This chapter is based on the paper [ZAdK24].

7.1 Introduction

We consider the convex-concave saddle point problem

$$\min_{x \in \mathbb{R}^n} \max_{y \in \mathbb{R}^m} F(x, y), \tag{7.1}$$

where $F : \mathbb{R}^n \times \mathbb{R}^m \to (-\infty, \infty)$, and $F(\cdot, y)$ and $F(x, \cdot)$ are convex and concave, respectively, for any fixed $x \in \mathbb{R}^n$ and $y \in \mathbb{R}^m$. We assume that problem (7.1) has some solution, that is, there exists a saddle point $(x^*, y^*) \in \mathbb{R}^n \times \mathbb{R}^m$ with

$$F(x^*, y) \le F(x^*, y^*) \le F(x, y^*), \quad \forall x \in \mathbb{R}^n, \forall y \in \mathbb{R}^m.$$

We denote the solution set of problem (7.1) with S^* . We call F smooth if for some L_x, L_y, L_{xy} , we have

i)
$$\|\nabla_x F(x_2, y) - \nabla_x F(x_1, y)\| \le L_x \|x_2 - x_1\|$$
 $\forall x_1, x_2, y$

ii)
$$\|\nabla_y F(x, y_2) - \nabla_y F(x, y_1)\| \le L_y \|y_2 - y_1\|$$
 $\forall x, y_1, y_2$

iii)
$$\|\nabla_x F(x, y_2) - \nabla_x F(x, y_1)\| \le L_{xy} \|y_2 - y_1\|$$
 $\forall x, y_1, y_2$

iv)
$$\|\nabla_{y}F(x_{2},y) - \nabla_{y}F(x_{1},y)\| \le L_{xy}\|x_{2} - x_{1}\|$$
 $\forall x_{1},x_{2},y.$

The function F is said to be strongly convex-strongly concave if

i) $F(\cdot, y) - \frac{\mu_x}{2} \| \cdot \|^2$ is convex for any fixed *y ii*) $F(x, \cdot) + \frac{\mu_y}{2} \| \cdot \|^2$ is concave for any fixed *x*,

for some $\mu_x, \mu_y > 0$. Note that strong convexitystrong concavity implies that problem (7.1) has a unique solution (x^*, y^*) . We denote the set of smooth strongly convex-strongly concave functions by $\mathcal{F}(L_x, L_y, L_{xy}, \mu_x, \mu_y)$.

Problem (7.1) has applications in game theory [BO98], robust optimization [BTEGN09], adversarial training [GPAM⁺20], and reinforcement learning [YND⁺20], to name but a few. In addition, various other algorithms have been developed for solving saddle point problems; see e.g. [HA21, JM22, LJJ20, NYZ21, SPPMD19, WL20, XZXL23].

One of the simplest approaches for handling problem (7.1) introduced in [AAH⁺58, Chapter 6] is the gradient-descent-ascent method, which may be regarded as a generalization of the gradient method to saddle point problems. The *gradient descent-ascent method* is described in Algorithm 7.1.

The local and global linear convergence of Algorithm 7.1 have been investigated in the literature; see [FOP20, LS19, ZWLG22] and the references therein. As

Algorithm 7.1 The gradient descent-ascent method

Set *N* and t > 0 (step length), pick x^0 and y^0 . For k = 1, 2, ..., N perform the following simultaneous steps: 1. $x^k = x^{k-1} - t \nabla_x F(x^{k-1}, y^{k-1})$. 2. $y^k = y^{k-1} + t \nabla_y F(x^{k-1}, y^{k-1})$.

we investigate the global linear convergence rate of Algorithm 7.1, we mention one known global convergence result, which is derived by using variational inequality techniques. Suppose that z = (x, y). Let the function $\phi : \mathbb{R}^{n+m} \to \mathbb{R}^{n+m}$ given by $\phi(z) = (\nabla_x F(z) - \nabla_y F(z))^T$. It is shown that, see e.g. [MOP20b],

$$\begin{aligned} \|\phi(\bar{z}) - \phi(\hat{z})\| &\leq 2L \|\bar{z} - \hat{z}\|, \\ \langle\phi(\bar{z}) - \phi(\hat{z}), \bar{z} - \hat{z}\rangle &\geq \mu \|\bar{z} - \hat{z}\|^2, \end{aligned}$$

where $L = \max\{L_x, L_y, L_{xy}\}$ and $\mu = \min\{\mu_x, \mu_y\}$. Indeed, ϕ is Lipschitz continuous and strongly monotone. By [FP03, Theorem 12.1.2], for $t \in (0, \frac{\mu}{2L^2})$, we have

$$\|x^{1} - x^{\star}\|^{2} + \|y^{1} - y^{\star}\|^{2} \le (1 + 4L^{2}t^{2} - 2\mu t) \left(\|x^{0} - x^{\star}\|^{2} + \|y^{0} - y^{\star}\|^{2}\right).$$
(7.2)

In this study, we revisit Algorithm 7.1 and improve the convergence rate (7.2). Indeed, we derive a new convergence rate involving most parameters of problem (7.1). It is worth noting that if one sets $L = \max\{L_x, L_y, L_{xy}\}$ and $\mu = \min\{\mu_x, \mu_y\}$, the new bound dominates the convergence rate (7.2) for any step length $t \in (0, \frac{\mu}{2L^2})$. Furthermore, by setting $t = \frac{\mu}{4L^2}$, one can infer that Algorithm 7.1 has a complexity of $\mathcal{O}\left(\frac{L^2}{\mu^2}\ln\left(\frac{1}{\epsilon}\right)\right)$ to compute iterates (x^k, y^k) such that $\|x^k - x^\star\|^2 + \|y^k - y^\star\|^2 \le \epsilon \left(\|x^0 - x^\star\|^2 + \|y^0 - y^\star\|^2\right)$, which is the known iteration complexity bound in the literature; see e.g. [BPG⁺23, ZBLG21]. In this study, thanks to the new convergence rate given in Theorem 7.2, the order of complexity of $\mathcal{O}\left(\left(\frac{L}{\mu} + \frac{L_{xy}^2}{\mu^2}\right)\ln\left(\frac{1}{\epsilon}\right)\right)$ is obtained when $L = \max\{L_x, L_y\}$ and $\mu = \min\{\mu_x, \mu_y\}$, which is more informative in comparison with the above-mentioned one. Moreover, by providing some example, we show that the given convergence rate is exact for one iteration.

The goal of this work is not to achieve the optimal algorithmic complexity for the class of saddle point problems introduced above. Rather, we have the more modest goal of giving the best possible worst-case complexity analysis of the gradient descent-ascent method (Algorithm 7.1). It is important to note that there are accelerated gradient descent-ascent methods with better worst-case complexity than Algorithm 7.1; see e.g. [LJJ20, WL20]. In particular, the accelerated methods may be shown to have a worst-case complexity $\mathcal{O}\left(\sqrt{\frac{L_x^2}{\mu_x^2} + \frac{L_{xy}^2}{\mu_x\mu_y} + \frac{L_y^2}{\mu_y^2}} \cdot \ln\left(\frac{1}{\epsilon}\right)\right)$, which may be compared to the best-known lower complexity bound $\mathcal{O}\left(\sqrt{\frac{L_x}{\mu_x} + \frac{L_{xy}^2}{\mu_x\mu_y} + \frac{L_y}{\mu_y}} \cdot \ln\left(\frac{1}{\epsilon}\right)\right)$ for the class of pure first-order algorithms [ZHZ22].

The chapter is organized as follows. First, we present basic definitions and preliminaries used to establish the results. Section 7.2 is devoted to the study of the linear convergence of Algorithm 7.1. In Section 7.3, we study the linear convergence of the gradient descent-ascent method without strong convexity. Indeed, we let $F \in \mathcal{F}(L_x, L_y, L_{xy}, 0, 0)$ and give some necessary and sufficient conditions for the linear convergence. Moreover, we derive a convergence rate under this setting.

Notation

Let $X \subseteq \mathbb{R}^n$. We denote the distance function to X by $d_X(x) := \inf_{\bar{x} \in X} ||x - \bar{x}||$ and the set-valued mapping $\Pi_X(x)$ stands for the projection of x on X, i.e., $\Pi_X(x) := \{y \in X : ||x - y|| = d_X(x)\}$; see also Definition 2.5.

It is worth mentioning that, under the assumptions of Theorem 2.37, the set $\{(x^i; g^i; f^i)\}_{i \in \mathcal{I}}$ is interpolable with an *L*-smooth μ -strongly concave function if and only if for any $i, j \in \mathcal{I}$, we have

$$\frac{1}{2(1-\frac{\mu}{L})} \left(\frac{1}{L} \left\| g^{i} - g^{j} \right\|^{2} + \mu \left\| x^{i} - x^{j} \right\|^{2} + \frac{2\mu}{L} \left\langle g^{j} - g^{i}, x^{j} - x^{i} \right\rangle \right) \leq -f^{i} + f^{j} + \left\langle g^{j}, x^{i} - x^{j} \right\rangle.$$
(7.3)

7.2 The gradient descent-ascent method

In this section, we study the convergence rate of gradient descent-ascent method when $F \in \mathcal{F}(L_x, L_y, L_{xy}, \mu_x, \mu_y)$ with $\min\{\mu_x, \mu_y\} > 0$. Indeed, we investigate the worst-case behavior of one step of Algorithm 7.1 in terms of distance to the unique saddle point (x^*, y^*) . Let (x^1, y^1) be generated by the algorithm using the starting point (x^0, y^0) . The worst-cast convergence rate of Algorithm 7.1 is given by the solution of the following abstract optimization problem:

$$\max \frac{\|x^{1} - x^{\star}\|^{2} + \|y^{1} - y^{\star}\|^{2}}{\|x^{0} - x^{\star}\|^{2} + \|y^{0} - y^{\star}\|^{2}}$$

s.t. (x^{1}, y^{1}) is generated by Algorithm 7.1 w.r.t. F, x^{0}, y^{0} (7.4)
 (x^{\star}, y^{\star}) is the unique saddle point of problem (7.1)
 $F \in \mathcal{F}(L_{x}, L_{y}, L_{xy}, \mu_{x}, \mu_{y})$
 $x^{0} \in \mathbb{R}^{n}, y^{0} \in \mathbb{R}^{m}.$

In problem (7.4), $F, x^0, x^1, x^*, y^0, y^1, y^*$ are decision variables and $\mu_x, L_x, \mu_y, L_y, L_{xy}, t$ are fixed parameters. As it is mentioned in Chapter 3, problem (7.4) seems completely intractable, but its solution may in fact be approximated using a suitable semidefinite programming (SDP) problem, as shown below. Suppose that

$$\begin{split} F^{i,j} &= F(x^{i}, y^{j}) & i, j \in \{0, 1, \star\}, \\ G^{i,j}_{x} &= \nabla_{x} F(x^{i}, y^{j}) & i, j \in \{0, 1, \star\}, \\ G^{i,j}_{y} &= \nabla_{y} F(x^{i}, y^{j}) & i, j \in \{0, 1, \star\}, \end{split}$$

where indices $\{0, 1, \star\}$ refers to the starting point, the point generated by Algorithm 7.1 and the saddle point of the problem, respectively. Note that due to the the necessary and sufficient conditions for convex-concave saddle point problems, we have

$$G_x^{\star,\star} = 0, \qquad G_y^{\star,\star} = 0.$$

By using Theorem 2.37, problem (7.4) may be relaxed as a finite dimensional optimization problem,

$$\max \frac{\|x^{1} - x^{\star}\|^{2} + \|y^{1} - y^{\star}\|^{2}}{\|x^{0} - x^{\star}\|^{2} + \|y^{0} - y^{\star}\|^{2}}$$
s.t. $\{(x^{0}; G_{x}^{0,k}; F^{0,k}), (x^{1}; G_{x}^{1,k}; F^{1,k}), (x^{\star}; G_{x}^{\star,k}; F^{\star,k})\}$ satisfy (2.4) for
 $k \in \{0, 1, \star\}$ w.r.t. μ_{x}, L_{x}
 $\{(y^{0}; G_{y}^{k,0}; F^{k,0}), (y^{1}; G_{y}^{k,1}; F^{k,1}), (y^{\star}; G_{y}^{k,\star}; F^{k,\star})\}$ satisfy (7.3) for
 $k \in \{0, 1, \star\}$ w.r.t. μ_{y}, L_{y} (7.5)
 $\|G_{x}^{k,i} - G_{x}^{k,j}\| \le L_{xy}\|y^{i} - y^{j}\|, \quad i, j, k \in \{0, 1, \star\}$
 $\|G_{y}^{i,k} - G_{y}^{j,k}\| \le L_{xy}\|x^{i} - x^{j}\|, \quad i, j, k \in \{0, 1, \star\}$
 $x^{1} = x^{0} - t G_{x}^{0,0}$
 $y^{1} = y^{0} + t G_{y}^{0,0},$
 $G_{x}^{\star,\star} = 0, \ G_{y}^{\star,\star} = 0.$

In problem (7.5), $\{(x^i; G_x^{i,j}; F^{i,j})\}$ and $\{(y^i; G_y^{j,i}; F^{j,i})\}$ $(i, j \in \{0, 1, \star\})$ are decision variables. We may assume that $x^* = 0$ and $y^* = 0$ as Algorithm 7.1 is invariant under translation. By elimination, problem (7.5) may be reformulated as follows,

$$\max \frac{\|x^0 - tG_x^{0,0}\|^2 + \|y^0 + tG_y^{0,0}\|^2}{\|x^0\|^2 + \|y^0\|^2}$$
s.t. $\{(x^0; G_x^{0,k}; F^{0,k}), (x^0 - tG_x^{0,0}; G_x^{1,k}; F^{1,k}), (0; G_x^{\star,k}; F^{\star,k})\}$ satisfy (2.4) for $k \in \{0, 1, \star\}$ w.r.t. μ_x, L_x
 $\{(y^0; G_y^{k,0}; F^{k,0}), (y^0 + tG_y^{0,0}; G_y^{k,1}; F^{k,1}), (0; G_y^{k,\star}; F^{k,\star})\}$ satisfy (7.3) for $k \in \{0, 1, \star\}$ w.r.t. μ_y, L_y (7.6)
 $\|G_x^{k,0} - G_x^{k,1}\| \le L_{xy} \|tG_y^{0,0}\|, \quad k \in \{0, 1, \star\}$
 $\|G_y^{0,k} - G_x^{1,k}\| \le L_{xy} \|tG_x^{0,0}\|, \quad k \in \{0, 1, \star\}$
 $\|G_x^{0,k} - G_x^{k,\star}\| \le L_{xy} \|y^0 - y^\star\|, \quad k \in \{0, 1, \star\}$
 $\|G_x^{0,k} - G_x^{k,\star}\| \le L_{xy} \|y^0 - x^\star\|, \quad k \in \{0, 1, \star\}$
 $\|G_x^{k,1} - G_x^{k,\star}\| \le L_{xy} \|y^0 + tG_y^{0,0} - y^\star\|, \quad k \in \{0, 1, \star\}$
 $\|G_x^{1,k} - G_x^{k,\star}\| \le L_{xy} \|x^0 - tG_x^{0,0} - x^\star\|, \quad k \in \{0, 1, \star\}$
 $\|G_y^{1,k} - G_y^{\star,k}\| \le L_{xy} \|x^0 - tG_x^{0,0} - x^\star\|, \quad k \in \{0, 1, \star\}$
 $\|G_x^{1,k} - G_y^{\star,k}\| \le L_{xy} \|x^0 - tG_x^{0,0} - x^\star\|, \quad k \in \{0, 1, \star\}$

To approximate the solution of problem (7.6), we reformulate it as a semidefinite program by using the Gram matrix of the unknown vectors in the problem. Indeed, we form the Gram matrices *X* and *Y* corresponding to $\{(x^i; G_x^{i,j})\}$ and $\{(y^i; G_y^{j,i})\}$ $(i, j \in \{0, 1, \star\})$, respectively. This results in an SDP problem, as long as we view the value $||x^0 - x^\star||^2 + ||y^0 - y^\star||^2$, that appears in the denominator of the objective of problem (7.6), as a fixed parameter. For this reason we may indeed view problem (7.6) as an SDP problem in the positive semidefinite matrix variables *X* and *Y*.

For the convenience of the analysis, we investigate the linear convergence of Algorithm 7.1 in terms of $L = \max\{L_x, L_y\}$ and $\mu = \min\{\mu_x, \mu_y\}$. Before we present the main theorem in this section, we need to present a lemma.

Lemma 7.1. Let $0 < \mu \le L$, $c \ge 0$ and let $I = \left(0, \frac{2\mu}{\mu L + c^2}\right)$. Suppose that the function $u: I \to \mathbb{R}$ given by

$$u(t) = \frac{1}{2} \left(L^2 + \mu^2 + 2c^2 \right) t^2 - (L+\mu)t + \frac{1}{2} (L-\mu)t \sqrt{(Lt+\mu t-2)^2 + 4c^2 t^2}.$$

Then u is convex on I and $u(I) \subseteq [-1, 0)$.

Proof. Consider the function $v : I \to \mathbb{R}$ given by

$$v(t) = (L^2 + \mu^2 + 2c^2)t + (L - \mu)\sqrt{(Lt + \mu t - 2)^2 + 4c^2t^2}$$

The function v is convex and positive on I. By elementary calculus, one can show that v'(0) > 0. So v is increasing on I due to the convexity. As the product of positive monotone convex functions is a convex function, the function $t \mapsto tv(t)$ is also convex, which implies the convexity of u. Indeed, u is strictly convex on I. Since strictly convex functions attain their maximum on endpoints of a given interval, $u(t) < \max\{u(0), u(\frac{2\mu}{\mu L + c^2})\} = 0$ for $t \in I$. It remains to show that $\min_{t \in I} u(t) \ge -1$. This follows from the point that

$$u(t) \ge \frac{1}{2} \left(L^2 + \mu^2 \right) t^2 - (L+\mu)t \ge \frac{-1}{2} \left(1 + \frac{2L\mu}{L^2 + \mu^2} \right) \ge -1,$$

and the proof is complete.

By the weak duality theorem for SDP, one may demonstrate an upper bound for the optimal value of the SDP problem (7.6), by constructing a feasible solution to its dual problem, i.e. feasible dual multipliers for the constraints of problem (7.6). This is done in the next theorem. In the proof, the correct value of the dual multipliers are simply given, and their correctness is verified. The correct values

were obtained by solving the SDP problem (7.6) repeatedly for different numerical values of the parameters, and noting the (numerical) optimal dual multiplier values. Based on these values, it was possible to deduce the analytical expressions for the multipliers. For this reason, the proof was found in a computer-assisted way, but it does not rely on any numerical calculations. Having said that, the proof involves a long identity, given in full in Appendix A.2 to this chapter, that is so long that it could only be obtained in a computer-assisted way.

Theorem 7.2. Let $F \in \mathcal{F}(L_x, L_y, L_{xy}, \mu_x, \mu_y)$. Suppose that $L = \max\{L_x, L_y\}$ and $\mu = \min\{\mu_x, \mu_y\} > 0$. If $t \in \left(0, \frac{2\mu}{\mu L + L_{xy}^2}\right)$, then Algorithm 7.1 generates (x^1, y^1) such that

$$\|x^{1} - x^{\star}\|^{2} + \|y^{1} - y^{\star}\|^{2} \le \alpha \left(\|x^{0} - x^{\star}\|^{2} + \|y^{0} - y^{\star}\|^{2}\right),$$
(7.7)

where

$$\alpha = 1 + \frac{1}{2} \left(L^2 + \mu^2 + 2L_{xy}^2 \right) t^2 - (L+\mu)t + \frac{1}{2} (L-\mu)t \sqrt{(Lt+\mu t-2)^2 + 4L_{xy}^2 t^2}.$$

Proof. As mentioned earlier, we may assume without loss of generality that $x^* = 0$ and $y^* = 0$. By assumption, $F(\cdot, y) \in \mathcal{F}_{\mu,L}(\mathbb{R}^n)$ and $F(x, \cdot) \in \mathcal{F}_{\mu,L}(\mathbb{R}^m)$ for any fixed x, y. Suppose that $L_{xy} \neq 0$. Without loss of generality, we may also assume that $L_{xy} = 1$, by replacing F by $\frac{1}{L_{xy}}F$. This follows from the observation that Algorithm 7.1 generates the same point (x^1, y^1) for the problem

$$\min_{x\in\mathbb{R}^n}\max_{y\in\mathbb{R}^m}\frac{1}{L_{xy}}F(x,y),$$

with the step length $L_{xy}t$. Moreover, one has $\frac{1}{L_{xy}}F \in \mathcal{F}(\frac{L_x}{L_{xy}}, \frac{L_y}{L_{xy}}, 1, \frac{\mu_x}{L_{xy}}, \frac{\mu_y}{L_{xy}})$ if and only if $F \in \mathcal{F}(L_x, L_y, L_{xy}, \mu_x, \mu_y)$. Now let $t \in (0, \frac{2\mu}{\mu L+1})$ and define (the multipliers):

$$\begin{split} \bar{\alpha} &= 1 + \frac{1}{2} \left(L^2 + \mu^2 + 2 \right) t^2 - (L+\mu)t + \frac{1}{2} (L-\mu)t \sqrt{(Lt+\mu t-2)^2 + 4t^2}, \\ \beta &= \sqrt{(Lt+\mu t-2)^2 + 4t^2}, \qquad \gamma_1 = \frac{t \left(t^2 \left(2 + L^2 + L\mu \right) - t \left(3L+\mu \right) + (Lt-1)\beta + 2 \right)}{\beta}, \\ \gamma_2 &= \frac{t \left(t^2 \left(2 + \mu^2 + L\mu \right) - t \left(3\mu + L \right) + (1-\mu t)\beta + 2 \right)}{\beta}, \quad \gamma_3 = \frac{t^2 (\beta + Lt-\mu t)}{2\beta}. \end{split}$$

It is readily verified that $\gamma_1, \gamma_2, \gamma_3 \ge 0$, but since this calculation is somewhat tedious we present it in Appendix A.1. Moreover, Lemma 7.1 implies that $\bar{\alpha} \in [0, 1)$.

The idea of the proof is now as follows: we first establish that, for any feasible solution of the SDP problem (7.6), it holds that

$$\left\|x^{0} - tG_{x}^{0,0}\right\|^{2} + \left\|y^{0} + tG_{y}^{0,0}\right\|^{2} - \bar{\alpha}\left(\left\|x^{0}\right\|^{2} + \left\|y^{0}\right\|^{2}\right) \le 0.$$
(7.8)

We do this by establishing an algebraic identity for the left-hand side of the inequality (7.8). The first and last terms of this identity (shown in full in Appendix A.2 to this chapter) are as follows:

$$\begin{split} & \left\| x^{0} - t G_{x}^{0,0} \right\|^{2} + \left\| y^{0} + t G_{y}^{0,0} \right\|^{2} - \bar{\alpha} \left(\left\| x^{0} \right\|^{2} + \left\| y^{0} \right\|^{2} \right) \\ &= -\gamma_{1} \left(F^{0,0} - F^{\star,0} - \left\langle G_{x}^{\star,0}, x^{0} \right\rangle - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \left\| G_{x}^{0,0} - G_{x}^{\star,0} \right\|^{2} + \mu \left\| x^{0} \right\|^{2} - \frac{2\mu}{L} \left\langle G_{x}^{\star,0} - G_{x}^{0,0}, -x^{0} \right\rangle \right) \right) \\ &\vdots \\ & - \frac{t(\beta + Lt - \mu t)^{2}}{4(L-\mu)\beta} \left\| G_{y}^{0,0} - G_{y}^{\star,0} - G_{y}^{0,\star} \right\|^{2}. \end{split}$$

Note that the first term on the right-hand-side is indeed nonpositive, since $\gamma_1 \ge 0$, and the expression in brackets is nonnegative at any feasible solution of the SDP problem (7.6), since it corresponds to one of the constraints in (7.6). The last term is nonpositive as well, since it is the product of a nonpositive multiplier with a squared expression. The remaining terms in the identity are similarly nonpositive (see Appendix A.2), proving the inequality (7.8). All that remains is to recognize that, in (7.8), $G_x^{0,0}$ corresponds to $\nabla_x F(x^0, y^0)$, so that $x^0 - t G_x^{0,0}$ corresponds to x^1 , etc. This yields the statement of the theorem, after rescaling to remove the assumption $L_{xy} = 1$.

One may wonder how we obtained the (analytical) expression for α in Theorem 7.2. Consider the optimization problem

$$\min_{x\in\mathbb{R}^n}f(x),\tag{7.9}$$

where $f \in \mathcal{F}_{\mu,L}$. It is known that the quadratic function $q(x) = x^T Q x$ with $\lambda_{\max}(Q) = L$ and $\lambda_{\min}(Q) = \mu$ attains the worst-case convergence rate for the gradient method; see e.g. [dKGT17]. We guessed that this property may hold for problem (7.1) and we investigated the bilinear saddle point problem

$$\min_{x \in \mathbb{R}^2} \max_{y \in \mathbb{R}^2} \frac{1}{2} x^T \begin{pmatrix} L_x & 0\\ 0 & \mu_x \end{pmatrix} x + x^T \begin{pmatrix} 0 & L_{xy}\\ L_{xy} & 0 \end{pmatrix} y - \frac{1}{2} y^T \begin{pmatrix} L_y & 0\\ 0 & \mu_y \end{pmatrix} y, \quad (7.10)$$

where $L_x \ge \mu_x > 0$, $L_y \ge \mu_y > 0$ and L_{xy} are fixed parameters and we derived the worst case convergence of Algorithm 7.1 with respect to this problem. Our numerical experiments showed that the derived convergence rate is the same as the optimal value of the semidefinite programming problem corresponding to problem (7.6). Moreover, as a by-product, we exhibit that the convergence rate (7.7) is exact for one iteration by using problem (7.10); see Proposition 7.4.

Theorem 7.2 provides some new information concerning Algorithm 7.1. Firstly, Theorem 7.2 improves the known convergence factor in the literature; see our discussion in Introduction. In addition, it investigates the convergence rate for a step length in a larger interval. Secondly, it does not assume the second order continuous differentiability of *F*, which is commonly used for deriving a local convergence rate; see [LS19, MNG17, ZWLG22]. Finally, the given convergence rate incorporates three parameter $\mu = \min\{\mu_x, \mu_y\}$, $L = \max\{L_x, L_y\}$ and L_{xy} , which is more informative in comparison with the results in the literature mostly given in terms of $\mu = \min\{\mu_x, \mu_y\}$ and $L = \max\{L_x, L_y, L_{xy}\}$; see [MOP20a, ZWLG22, ZHZ22] and references therein. Even if one considers $L = \max\{L_x, L_y, L_{xy}\}$ and $\mu = \min\{\mu_x, \mu_y\}$, convergence rate (7.7) dominates (7.2). This follows from that for $t \in (0, \frac{\mu}{2L^2})$, one has

$$\left(1 + 4L^{2}t^{2} - 2\mu t\right) - \left(1 + \frac{1}{2}\left(3L^{2} + \mu^{2}\right)t^{2} - (L + \mu)t + \frac{1}{2}(L - \mu)t\sqrt{(Lt + \mu t - 2)^{2} + 4L^{2}t^{2}}\right)$$

$$\ge (2L^{2} + L\mu - \mu^{2})t^{2} \ge 2L^{2}t^{2},$$

where the first inequality results from $\sqrt{(Lt + \mu t - 2)^2 + 4L^2t^2} \le (2 - Lt - \mu t) + 2Lt$. In addition, in this case, the step length can take value in a larger interval as $\left(0, \frac{\mu}{2L^2}\right) \le \left(0, \frac{2\mu}{L(L+\mu)}\right)$. Moreover, Conjecture 7.6 discusses the convergence rate in terms of $L_x, L_y, L_{xy}, \mu_x, \mu_y$.

The next proposition gives the optimal step length with respect to the worst case convergence rate.

Proposition 7.3. Let $F \in \mathcal{F}(L_x, L_y, L_{xy}, \mu_x, \mu_y)$. If $L = \max\{L_x, L_y\}$ and $\mu = \min\{\mu_x, \mu_y\} > 0$, then the optimal step length for Algorithm 7.1 with respect to the bound (7.7) is

$$t^{\star} = \frac{2\left((L+\mu)\sqrt{L_{xy}^{2}+L\mu}+L_{xy}(\mu-L)\right)}{\left(4L_{xy}^{2}+(L+\mu)^{2}\right)\sqrt{L_{xy}^{2}+L\mu}}.$$
(7.11)

Moreover, the convergence rate with respect to t* is

$$\alpha^{\star} = \frac{8L_{xy}(L^2 - \mu^2)\sqrt{L\mu + L_{xy}^2} + (L^2 - \mu^2)^2 + 16L_{xy}^2(L\mu + L_{xy}^2)}{\left((L + \mu)^2 + 4L_{xy}^2\right)^2}.$$
(7.12)

Proof. Let
$$\alpha : \left[0, \frac{2\mu}{\mu L + L_{xy}^2}\right] \to \mathbb{R}$$
 given by

$$\alpha(t) = 1 + \frac{1}{2} \left(L^2 + \mu^2 + 2L_{xy}^2 \right) t^2 - (L+\mu)t + \frac{1}{2} (L-\mu)t \sqrt{(Lt+\mu t-2)^2 + 4L_{xy}^2 t^2}.$$

By Lemma 7.1, α is a strictly convex function on its domain. By doing some algebra, one can verify that $\alpha'(t^*) = 0$, which implies that t^* is the minimum. \Box

If $L_{xy} = 0$, problem (7.1) reduces to a separable optimization problem. Indeed, the variables *x* and *y* are independent. Under this assumption, the optimal step length given by Proposition 7.3 is $t^* = \frac{2}{L+\mu}$, which is the well-known optimal step length for the optimization problem

$$\min_{x\in\mathbb{R}^n}f(x),$$

where $f \in \mathcal{F}_{\mu,L}$; see [Nes18, Theorem 2.1.15]. Moreover, the convergence rate corresponding to t^* is $\alpha^* = \left(\frac{L-\mu}{L+\mu}\right)^2$. By some algebra, one can show that under the assumptions of Proposition (7.3), Algorithm 7.1 has a complexity of $\mathcal{O}\left(\left(\frac{L}{\mu} + \frac{L_{xy}^2}{\mu^2}\right)\ln\left(\frac{1}{\epsilon}\right)\right)$. Note that the lower iteration complexity bound for first-order methods with $L = \max\{L_x, L_y\}$ and $\mu = \min\{\mu_x, \mu_y\}$ is $\Omega\left(\sqrt{\frac{L}{\mu} + \frac{L_{xy}^2}{\mu^2}}\ln\left(\frac{1}{\epsilon}\right)\right)$; see [ZHZ22].

As mentioned earlier, we calculated the convergence rate by using problem (7.10). The next proposition states that the bound (7.7) is tight for some class of bilinear saddle point problems.

Proposition 7.4. Let $F \in \mathcal{F}(L_x, L_y, L_{xy}, \mu_x, \mu_y)$. Suppose that $L_x = L_y$ and $\mu = \min\{\mu_x, \mu_y\} > 0$. If $t \in \left(0, \frac{2\mu}{\mu L + L_{xy}^2}\right)$, then convergence rate (7.7) is exact for one iteration.

Proof. To establish the proposition, it suffices to introduce a problem for which Algorithm 7.1 generates (x^1, y^1) with respect to the initial point (x^0, y^0) such that

$$\|x^{1} - x^{\star}\|^{2} + \|y^{1} - y^{\star}\|^{2} = \alpha \left(\|x^{0} - x^{\star}\|^{2} + \|y^{0} - y^{\star}\|^{2}\right),$$

where α is the convergence rate factor given in Theorem 7.2. Consider problem (7.10). Due to the symmetry of Algorithm 7.1 and the class of problems, we may assume $\mu_x \ge \mu_y$. Moreover, without loss of generality, we can take $L_{xy} = 1$; see our discussion in the proof of Theorem 7.2. Suppose $L = L_x$, $\mu = \mu_y$ and

 $\beta = \sqrt{(Lt + \mu t - 2)^2 + 4t^2}$. One can verify that Algorithm 7.1 with the initial point

$$\begin{split} x_1^0 &= 0, & x_2^0 = \sqrt{\frac{2 - t(L + \mu) + \beta}{2\beta}} \\ y_1^0 &= -t \sqrt{\frac{2}{\beta(2 - t(L + \mu) + \beta)}}, & y_2^0 = 0. \end{split}$$

generates (x^1, y^1) with the desired equality.

One may wonder why we stress on one iteration in Proposition 7.4. Based on our numerical results if $L_{xy} > 0$, under the setting of Theorem 7.2, we observed that

$$\|x^{k} - x^{\star}\|^{2} + \|y^{k} - y^{\star}\|^{2} < \alpha^{k} \left(\|x^{0} - x^{\star}\|^{2} + \|y^{0} - y^{\star}\|^{2}\right), \quad k \ge 2$$

for some $t \in \left(0, \frac{2\mu}{\mu L + L_{xy}^2}\right)$. The reason may be related to the fact that the vector field $\left(\nabla_x F(x, y) - \nabla_y F(x, y)\right)^T$ is not conservative.

It may be of interest whether inequality (7.7) may hold without strong convexity. By removing strong convexity, the solution set may not be singleton. Hence, we investigate distance to the solution set, that is, if there exists $0 \le \alpha < 1$ with

$$d_{S^{\star}}^{2}((x^{1}, y^{1})) \leq \alpha d_{S^{\star}}^{2}((x^{0}, y^{0})).$$

The next proposition says in general the answer is negative. Indeed, it gives an example with $\min\{\mu_x, \mu_y\} = 0$ and a unique saddle point for which

$$\|x^{1} - x^{\star}\|^{2} + \|y^{1} - y^{\star}\|^{2} \ge \alpha \left(\|x^{0} - x^{\star}\|^{2} + \|y^{0} - y^{\star}\|^{2}\right),$$

for some $\alpha \ge 1$, no matter how close (x^0, y^0) is to the unique saddle point and which positive step length *t* is taken. In the next proposition, we may assume without loss of generality $\mu_x = 0$ and make an example analogous to that given in Proposition 7.4.

Proposition 7.5. Let $L, L_{xy}, \mu_y, t, r > 0$ be given. Then there exist $\alpha \ge 1$ and a function $F \in \mathcal{F}(L, L, L_{xy}, 0, \mu_y)$ with the unique saddle point (x^*, y^*) and (x^1, y^1) such that, for (x^2, y^2) generated by Algorithm 7.1, we have

$$\|x^{2} - x^{\star}\|^{2} + \|y^{2} - y^{\star}\|^{2} \ge \alpha \left(\|x^{1} - x^{\star}\|^{2} + \|y^{1} - y^{\star}\|^{2}\right),$$

and $||x^1 - x^*||^2 + ||y^1 - y^*||^2 = r^2$.

Proof. As discussed before, we may assume $L_{xy} = 1$. Consider the bilinear saddle point problem,

$$\min_{x \in \mathbb{R}^2} \max_{y \in \mathbb{R}^2} F(x, y) = \frac{1}{2} x^T \begin{pmatrix} L & 0 \\ 0 & 0 \end{pmatrix} x + x^T \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} y - \frac{1}{2} y^T \begin{pmatrix} L & 0 \\ 0 & \mu_y \end{pmatrix} y.$$

It is clear that $F \in \mathcal{F}(L, L, L_{xy}, 0, \mu_y)$, and the unique saddle point is $(x^*, y^*) =$ (0,0). Suppose that

$$\begin{aligned} x_1^1 &= 0, & x_2^1 &= r \sqrt{\frac{2 - tL + \beta}{2\beta}}, \\ y_1^1 &= -rt \sqrt{\frac{2}{\beta(2 - tL + \beta)}}, & y_2^1 &= 0, \end{aligned}$$

where $\beta = \sqrt{(Lt-2)^2 + 4t^2}$. One can verify Algorithm 7.1 generates (x^2, y^2) with

can

$$\|x^{2} - x^{*}\|^{2} + \|y^{2} - y^{*}\|^{2} \ge \alpha \left(\|x^{1} - x^{*}\|^{2} + \|y^{1} - y^{*}\|^{2}\right) = \alpha r^{2},$$

where $\alpha = 1 + \frac{1}{2} \left(L^{2} + 2\right) t^{2} - Lt + \frac{1}{2} Lt \sqrt{(Lt - 2)^{2} + 4t^{2}}$. By Proposition 7.3, one can infer that $\alpha \ge 1$.

Note that in Proposition 7.5 r can take any positive value. By Proposition 7.4, one can infer that the convergence rate factor for bilinear saddle point problems may not be improved for one iteration since the given example is a bilinear saddle point problem. Furthermore, the given convergence rate factor is tight whether $L_x = L_y$. As discussed in [WL20], the function $H(x, y) = F\left(\sqrt[4]{\frac{L_y}{L_x}}x, \sqrt[4]{\frac{L_x}{L_y}}y\right)$ shares the same smoothness constants with respect to x and y, that is, $\nabla_x H(\cdot, y)$ and $\nabla_{v} H(x, \cdot)$ are Lipschitz continuous with the same modulus $\sqrt{L_{x}L_{y}}$. However, the gradient methods are not invariant under scaling; see [BV04, Chapter 9]. Hence, we may lose the generality of our discussion by assuming this condition.

Based on our numerical results and analysis of problem (7.10), we conjecture the following (exact) convergence rate of Algorithm 7.1 in terms of L_x , L_y , L_{xy} , μ_x , μ_{γ} . Due to the symmetry of Algorithm 7.1, we may assume that $L_{\chi} \ge L_{\gamma}$. Moreover, Proposition 7.4 implies that bound (7.7) is tight when $\mu_v \leq \mu_x$. Hence, we need only consider $\mu_v > \mu_x$.

Conjecture 7.6. Let $F \in \mathcal{F}(L_x, L_y, L_{xy}, \mu_x, \mu_y)$. Suppose that $\mu_y > \mu_x > 0$, $\max\{L_x, L_y\} = L_x$ and

$$\begin{split} c &= \frac{1}{2} (L_y^2 + \mu_x^2) t - (L_y + \mu_x) + \frac{1}{2} (L_y - \mu_x) \sqrt{(L_y t + \mu_x t - 2)^2 + 4L_{xy}^2 t^2}, \\ \bar{\mu} &= \frac{c + 2L_x - L_x^2 t + L_x L_{xy}^2 t^2 - (c + L_x (2 - L_x t)) \sqrt{1 + t(c + t L_{xy}^2)}}{t L_{xy} (c + t L_x^2 + L_x (2 - L_x t))}, \\ \alpha(\mu, L, L_{xy}, t) &= 1 + \frac{1}{2} \left(L^2 + \mu^2 + 2L_{xy}^2 \right) t^2 - (L + \mu) t + \frac{1}{2} (L - \mu) t \sqrt{(L t + \mu t - 2)^2 + 4L_{xy}^2 t^2}. \end{split}$$

Then, one of the following scenarios holds.

a) Assume that
$$\mu_{x}\mu_{y}(L_{x}-L_{y}) \ge L_{xy}^{2}(\mu_{y}-\mu_{x})$$
 and $t \in \left(0, \frac{2\mu_{y}}{L_{x}\mu_{y}+L_{xy}^{2}}\right)$.
i) If $\mu_{y} \le \bar{\mu}$, then
 $\|x^{1}-x^{*}\|^{2}+\|y^{1}-y^{*}\|^{2} \le \alpha(\mu_{y}, L_{x}, L_{xy}, t)(\|x^{0}-x^{*}\|^{2}+\|y^{0}-y^{*}\|^{2})$.
ii) If $\mu_{y} \ge \bar{\mu}$, then
 $\|x^{1}-x^{*}\|^{2}+\|y^{1}-y^{*}\|^{2} \le \alpha(\mu_{x}, L_{y}, L_{xy}, t)(\|x^{0}-x^{*}\|^{2}+\|y^{0}-y^{*}\|^{2})$.
b) Assume that $\mu_{x}\mu_{y}(L_{x}-L_{y}) \le L_{xy}^{2}(\mu_{y}-\mu_{x})$ and $t \in \left(0, \frac{2\mu_{x}}{L_{y}\mu_{x}+L_{xy}^{2}}\right)$.
i) If $\mu_{y} \le \bar{\mu}$, then
 $\|x^{1}-x^{*}\|^{2}+\|y^{1}-y^{*}\|^{2} \le \alpha(\mu_{y}, L_{x}, L_{xy}, t)(\|x^{0}-x^{*}\|^{2}+\|y^{0}-y^{*}\|^{2})$.
ii) If $\mu_{y} \ge \bar{\mu}$, then
 $\|x^{1}-x^{*}\|^{2}+\|y^{1}-y^{*}\|^{2} \le \alpha(\mu_{x}, L_{y}, L_{xy}, t)(\|x^{0}-x^{*}\|^{2}+\|y^{0}-y^{*}\|^{2})$.

Although we have extensive numerical evidence supporting Conjecture 7.6, we have been unable to prove either part a) or part b). To be more precise, we have verified numerically that the optimal value of the SDP problem (7.6) and problem (7.10) corresponds to the expressions in Conjecture 7.6 for many different numerical values of the parameters L_x , L_y , L_{xy} , μ_x , and μ_y , but we were unable to derive analytical expressions for the dual multipliers of the SDP problem (7.6) that would prove the conjecture.

7.2.1 Numerical illustration

In this section we provide randomly generated examples to compare the optimal step length (7.11) given in this chapter to the known step length $t = \mu/(4L^2)$ for the bilinear problem

$$\min_{x\in\mathbb{R}^5}\max_{y\in\mathbb{R}^4}\frac{1}{2}x^TA_xx + x^TA_{xy}y - \frac{1}{2}y^TA_yy,$$

where A_x and A_y are symmetric positive definite matrices. Moreover, the instances are constructed such that the spectra of A_x and A_y are contained in the interval [0.5, 5]. For this class of instances, one has $L = \max{\{\lambda_{\max}(A_x), \lambda_{\max}(A_y)\}} \in$ [0.5,5] and $\mu = \min\{\lambda_{\min}(A_x), \lambda_{\min}(A_y)\} \in [0.5, L]$. The matrix $A_{xy} \in \mathbb{R}^{5\times 4}$ has entries chosen uniformly at random from [0,1], and subsequently we set $L_{xy} = ||A_{xy}||_2$. By construction, the solution (saddle point) is $(x^*, y^*) = (0, 0)$. The starting points x^0 and y^0 are randomly drawn unit vectors so that the initial condition $||x^0 - x^*||^2 + ||y^0 - y^*||^2 = 2$ is satisfied.



Figure 7.1: Mean values of $||x^k - x^*||^2 + ||y^k - y^*||^2$ for 100 randomly generated instances for each iteration *k* using the two different step lengths *t*

In Figure 7.1 we show average values (over 100 randomly generated instances) for the convergence indicator $||x^k - x^*||^2 + ||y^k - y^*||^2$ after *k* iterations, for the two step lengths.¹ Note that our new step length (7.11) gives a clear improvement over the known step length $\mu/(4L^2)$.

7.3 Linear convergence without strong convexity

In this section, we study the linear convergence of Algorithm 7.1 without assuming strong convexity. Indeed, we suppose that $F \in \mathcal{F}(L_x, L_y, L_{xy}, 0, 0)$ and we propose some necessary and sufficient conditions for the linear convergence. This subject has received some attention in recent years and some sufficient conditions have

¹The 100 random instances and starting points that we generated to produce Figure 7.1 may be found on GitHub; see: https://github.com/molsemzamani/Bilinear-Minimax

been proposed in [DH19, ZWLG22] under which Algorithm 7.1 enjoys local linear convergence rate or it is linearly convergent for bilinear saddle point problems. This topic has been investigated extensively in the context of optimization. The interested reader can refer to [AdKZ23a, BNPS17, LT93, NNG19] alongside with Chapter 5 and references therein. In this study, we extend the quadratic gradient growth property introduced in [LT93] for saddle point problems.

Recall that we denote the nonempty solution set of problem (7.1) by S^* . As we do not assume the strong convexity (concavity), S^* may not be singleton. Note that S^* is a closed convex set under our assumptions. Recall that $\Pi_{S^*}((x, y))$ denotes the projection of (x, y) onto S^* .

Definition 7.7. Let $\mu_F > 0$. A function *F* has a quadratic gradient growth if for any $x \in \mathbb{R}^n$ and $y \in \mathbb{R}^m$,

$$\langle \nabla_x F(x,y), x - x^* \rangle - \langle \nabla_y F(x,y), y - y^* \rangle \ge \mu_F d_{S^*}^2((x,y)), \tag{7.13}$$

where $(x^*, y^*) = \prod_{S^*} ((x, y)).$

Note that if we set $y = y^*$ in (7.13), we have

$$\langle \nabla_x F(x, y^*), x - x^* \rangle \ge \mu_F ||x - x^*||^2.$$

Hence, L_x -smoothness implies that $\mu_F \leq L_x$. Consequently, due to the symmetry, we have $\mu_F \leq \min\{L_x, L_y\}$. The next proposition states that the quadratic gradient growth condition is weaker than the strong convexity-strong concavity. Indeed, the strong convexity-strong concavity implies the quadratic gradient growth property.

Proposition 7.8. Let $F \in \mathcal{F}(L_x, L_y, L_{xy}, \mu_x, \mu_y)$. If $\min\{\mu_x, \mu_y\} > 0$, then F has a quadratic gradient growth with $\mu_F = \min\{\mu_x, \mu_y\}$.

Proof. Under the assumptions, problem (7.1) has a unique solution (x^*, y^*) and $\nabla_x F(x^*, y^*) = 0$ and $\nabla_y F(x^*, y^*) = 0$. Let $\mu = \min\{\mu_x, \mu_y\}$ and $L = \max\{L_x, L_y\}$.

Suppose that $(x, y) \in \mathbb{R}^n \times \mathbb{R}^m$. By Theorem 2.37, we have

$$\begin{split} &0 \leq \left(F(x^*,y) - F(x,y) + \langle \nabla_x F(x,y), x - x^* \rangle - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \| \nabla_x F(x^*,y) - \nabla_x F(x,y) \|^2 \right. \\ &+ \mu \| x - x^* \|^2 - \frac{2\mu}{L} \left\langle \nabla_x F(x,y) - \nabla_x F(x^*,y), x - x^* \right\rangle \right) + \left(F(x,y^*) - F(x^*,y^*) - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \| \nabla_x F(x,y^*) \|^2 + \mu \| x - x^* \|^2 - \frac{2\mu}{L} \left\langle \nabla_x F(x,y^*), x - x^* \right\rangle \right) \right) + \left(F(x,y) - F(x^*,y) + \left(F(x,y), y - y^* \right) - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \| \nabla_y F(x,y^*) - \nabla_y F(x,y) \|^2 + \mu \| y - y^* \|^2 \right. \\ &- \frac{2\mu}{L} \left\langle \nabla_y F(x,y^*) - \nabla_y F(x,y), y - y^* \right\rangle \right) \right) + \left(F(x^*,y^*) - F(x^*,y) - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \| \nabla_y F(x^*,y) \|^2 + \mu \| y - y^* \|^2 - \frac{2\mu}{L} \left\langle \nabla_y F(x^*,y), y^* - y \right\rangle \right) \right) \\ &= \frac{-\mu^2}{L-\mu} \left\| (x - x^*) - \frac{1}{2\mu} \left(\nabla_x F(x,y) + \nabla_x F(x,y^*) - \nabla_x F(x^*,y) \right) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_x F(x,y) - \nabla_x F(x,y) - \nabla_y F(x^*,y) \right\|^2 - \frac{\mu^2}{L-\mu} \left\| (y - y^*) + \frac{1}{2\mu} \left(\nabla_y F(x,y) - \nabla_y F(x^*,y) \right) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y^*) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y^*) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y^*) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y^*) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y^*) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y^*) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y^*) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) - \nabla_y F(x^*,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) \right\|^2 + \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) \right\|^2 + \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) \right\|^2 - \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) \right\|^2 + \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) \right\|^2 + \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) - \nabla_y F(x,y) \right\|^2 + \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) \right\|^2 + \frac{1}{4(L-\mu)} \left\| \nabla_y F(x,y) \right\|^2 + \frac{1}{4(L-\mu)} \left\|$$

Hence,

$$\mu\left(\|x-x^{\star}\|^{2}+\|y-y^{\star}\|^{2}\right) \leq \langle \nabla_{x}F(x,y), x-x^{\star}\rangle - \left\langle \nabla_{y}F(x,y), y-y^{\star}\right\rangle,$$

and the proof is complete.

Note that the converse of Proposition 7.8 does not hold necessarily. Consider the following saddle point problem

$$\min_{x \in \mathbb{R}} \max_{y \in \mathbb{R}} F(x, y) := f(x + y) - 2y^2, \tag{7.14}$$

where

$$f(s) = \begin{cases} 0 & |s| \le 1\\ (s-1)^2 & s > 1\\ (s+1)^2 & s < -1. \end{cases}$$

It is seen that *F* is not strongly convex-strongly concave and the solution set of problem (7.14) is $\{(x,0) : |x| \le 1\}$. By doing some algebra, one can check that *F* has a quadratic gradient growth with $\mu_F = 1$ while it is not strongly convex with respect to the first component. For the case that $F(\cdot, y)$ is neither strongly

convex nor is $F(x, \cdot)$ strongly concave, one may consider uncoupled problem $\min_{x \in \mathbb{R}} \max_{y \in \mathbb{R}} f(x) - f(y)$.

In what follows, by using performance estimation, we establish that Algorithm 7.1 enjoys the linear convergence whether $F \in \mathcal{F}(L_x, L_y, L_{xy}, 0, 0)$ has a quadratic gradient growth. Without loss of generality, we may assume that $(0,0) = \prod_{S^*} ((x^0, y^0))$. To establish the linear convergence, it suffices to show that

$$d_{S^*}^2((x^1, y^1)) \le ||x^1||^2 + ||y^1||^2 \le \alpha d_{S^*}^2((x^0, y^0)),$$

for some $\alpha \in [0, 1)$. Similarly to Section 7.2, we formulate the following optimization problem

$$\max \frac{\|x^{0} - tG_{x}^{0,0}\|^{2} + \|y^{0} + tG_{y}^{0,0}\|^{2}}{\|x^{0}\|^{2} + \|y^{0}\|^{2}}$$
s.t. $\{(x^{0}; G_{x}^{0,k}; F^{0,k}), (x^{0} - tG_{x}^{0,0}; G_{x}^{1,k}; F^{1,k}), (0; G_{x}^{\star,k}; F^{\star,k})\}$ satisfy (2.4) for $k \in \{0, 1, \star\}$ w.r.t. $\mu_{x} = 0, L_{x}$
 $\{(y^{0}; G_{y}^{k,0}; F^{k,0}), (y^{0} + tG_{y}^{0,0}; G_{y}^{k,1}; F^{k,1}), (0; G_{y}^{k,\star}; F^{k,\star})\}$ satisfy (7.3) for $k \in \{0, 1, \star, \star\}$ w.r.t. $\mu_{y} = 0, L_{y}$
 (7.15)
 $\|G_{x}^{k,i} - G_{x}^{k,j}\| \le L_{xy}\|y^{i} - y^{j}\|, \quad i, j, k \in \{0, 1, \star\}$
 $\|G_{y}^{i,k} - G_{y}^{j,k}\| \le L_{xy}\|x^{i} - x^{j}\|, \quad i, j, k \in \{0, 1, \star\}$
 $\mu_{F}\left(\|x^{0}\|^{2} + \|y^{0}\|^{2}\right) \le \langle G_{x}^{0,0}, x^{0} \rangle - \langle G_{y}^{0,0}, y^{0} \rangle,$
 $G_{x}^{\star,\star} = 0, \ G_{y}^{\star,\star} = 0.$

Note that in the formulation (7.15), we only use a subset of constraints for the performance estimation. In the next theorem, we prove the linear convergence of Algorithm 7.1 when F has a quadratic gradient growth.

Theorem 7.9. Let $F \in \mathcal{F}(L_x, L_y, L_{xy}, 0, 0)$ and $L = \max\{L_x, L_y\}$. Assume that F has a quadratic gradient growth with $\mu_F > 0$. If $t \in \left(0, \frac{2\mu_F}{L\mu_F + 2L_{xy}\sqrt{\mu_F(L-\mu_F)} + L_{xy}^2}\right)$, then Algorithm 7.1 generates (x^1, y^1) such that $d_{S^*}^2((x^1, y^1)) \leq \alpha d_{S^*}^2((x^0, y^0))$, (7.16)

where

$$\alpha = t \left(2t L_{xy} \sqrt{\mu_F (L - \mu_F)} + \mu_F (Lt - 2) + t L_{xy}^2 \right) + 1.$$

Proof. The argument is similar to that of Theorem 7.2. It is seen that for any step length *t* in the given interval, $\alpha \in [0, 1)$. We may assume without loss of generality $L_{xy} = 1$. By the assumptions, $F(\cdot, y) \in \mathcal{F}_{0,L}(\mathbb{R}^n)$ and $F(x, \cdot) \in \mathcal{F}_{0,L}(\mathbb{R}^m)$ for any fixed *x*, *y*. Suppose that

$$\begin{split} \bar{\alpha} &= t \left(2t \sqrt{\mu_F (L - \mu_F)} + \mu_F (Lt - 2) + t \right) + 1, \quad \beta = t^2 \left(\mu_F \sqrt{L - \mu_F} + \sqrt{\mu_F} \right), \\ \gamma_1 &= t^2 \left(\frac{\mu_F}{\sqrt{\mu_F (L - \mu_F)}} + \mu_F \right), \qquad \gamma_2 = \frac{t^2 \left(\mu_F (L - \mu_F) + \sqrt{\mu_F (L - \mu_F)} \right)}{\mu_F}, \\ \gamma_3 &= -\frac{t^2 \left(\mu_F (L + \mu_F) + \sqrt{\mu_F (L - \mu_F)} \right)}{\mu_F} + \frac{\beta}{\sqrt{L - \mu_F}} + 2t, \qquad \gamma_4 = \frac{1}{2} t^2 \left(\sqrt{\mu_F (L - \mu_F)} + 1 \right). \end{split}$$

One may readily verify that $\gamma_1, \gamma_2, \gamma_3, \gamma_4 \ge 0$. By doing some algebra, one can show that

$$\begin{split} \left\| x^{0} - t G_{x}^{0,0} \right\|^{2} + \left\| y^{0} + t G_{y}^{0,0} \right\|^{2} - \bar{\alpha} \left(\left\| x^{0} \right\|^{2} + \left\| y^{0} \right\|^{2} \right) + \gamma_{1} \left(F^{0,0} - F^{\star,0} - \left\langle G_{x}^{\star,0}, x^{0} \right\rangle - \frac{1}{2L} \left\| G_{x}^{0,0} - G_{x}^{0,0} \right\|^{2} \right) + \gamma_{2} \left(F^{0,\star} - F^{0,0} + \left\langle G_{x}^{0,0}, x^{0} \right\rangle - \frac{1}{2L} \left\| G_{x}^{\star,0} - G_{x}^{0,0} \right\|^{2} \right) + \gamma_{2} \left(F^{0,\star} - F^{\star,\star} - \frac{1}{2L} \left\| G_{x}^{0,\star} \right\|^{2} \right) + \gamma_{1} \left(F^{\star,\star} - F^{0,\star} + \left\langle G_{x}^{0,\star}, x^{0} \right\rangle - \frac{1}{2L} \left\| G_{x}^{0,\star} \right\|^{2} \right) + \gamma_{1} \left(F^{0,\star} - F^{0,0} + \left\langle G_{y}^{0,\star}, y^{0} \right\rangle - \frac{1}{2L} \left\| G_{y}^{0,0} - G_{y}^{0,0} \right\|^{2} \right) + \gamma_{2} \left(F^{0,0} - F^{0,\star} - \left\langle G_{y}^{0,0}, y^{0} \right\rangle - \frac{1}{2L} \left\| G_{y}^{0,0} - G_{y}^{0,0} \right\|^{2} \right) + \gamma_{2} \left(-F^{\star,0} + F^{\star,\star} - \frac{1}{2L} \left\| G_{y}^{\star,0} \right\|^{2} \right) + \gamma_{1} \left(-F^{\star,\star} + F^{\star,0} + \left\langle G_{y}^{\star,0}, -y^{0} \right\rangle - \frac{1}{2L} \left\| G_{y}^{\star,0} \right\|^{2} \right) + \gamma_{3} \left(\left\langle G_{x}^{0,0}, x^{0} \right\rangle - \left\langle G_{y}^{0,0}, y^{0} \right\rangle - \mu_{F} \left(\left\| x^{0} \right\|^{2} + \left\| y^{0} \right\|^{2} \right) \right) + \gamma_{4} \left(\left\| x^{0} \right\|^{2} - \left\| G_{y}^{0,0} - G_{y}^{\star,0} \right\|^{2} \right) + \gamma_{4} \left(\left\| x^{0} \right\|^{2} - \left\| G_{y}^{0,0} - G_{y}^{\star,0} \right\|^{2} \right) + \gamma_{4} \left(\left\| x^{0} \right\|^{2} - \left\| G_{y}^{0,0} - G_{y}^{\star,0} \right\|^{2} \right) \right\|^{2} - \zeta_{4} \left\| G_{x}^{0,0} - G_{x}^{0,0} - G_{x}^{0,0} - G_{y}^{0,0} \right\|^{2} - \zeta_{6} \right\|^{2} - \zeta_{6} \right\|^{2} \right\|^{2} = 0,$$

where the multipliers $\zeta_1, \zeta_2, \zeta_3, \zeta_4$ are given as follows

$$\begin{aligned} \zeta_1 &= \mu_F \left(\frac{\beta}{\sqrt{L - \mu_F}} - \mu_F t^2 \right), \ \zeta_2 &= \frac{\beta}{2\mu_F \sqrt{\mu_F} t^2} - \frac{1}{\mu_F}, \ \zeta_3 &= \frac{t^2 \left(\sqrt{\mu_F (L - \mu_F)} + 1 \right)}{2\mu_F t^2}, \\ \zeta_4 &= \frac{1}{4} \left(\frac{2t^2 (\mu_F (L - \mu_F) + 1)}{\sqrt{\mu_F (L - \mu_F)}} - \frac{\left(2\mu_F t^2 (\mu_F - L) + \beta \sqrt{L - \mu_F} \right)^2}{\mu_F (L - \mu_F) t^2 \sqrt{\mu_F (L - \mu_F)}} \right). \end{aligned}$$

One can show by some algebra that $\zeta_1, \zeta_4 \ge 0$. Hence, for any feasible solution

of problem (7.15), we have

$$\frac{\left\|x^{0} - tG_{x}^{0,0}\right\|^{2} + \left\|y^{0} + tG_{y}^{0,0}\right\|^{2}}{\left\|x^{0}\right\|^{2} + \left\|y^{0}\right\|^{2}} \le \bar{\alpha},$$

and the proof is complete.

We obtained the linear convergence by using quadratic gradient growth in Theorem 7.9. The next theorem states that quadratic gradient growth property is also a necessary condition for the linear convergence.

Theorem 7.10. If Algorithm 7.1 is linearly convergent for any initial point, then F has a quadratic gradient growth for some $\mu_F > 0$.

Proof. Let $(x^0, y^0) \in \mathbb{R}^n \times \mathbb{R}^m$ and (x^1, y^1) be generated by Algorithm 7.1. Suppose that $(x^*, y^*) = \prod_{S^*} ((x^1, y^1))$. As Algorithm 7.1 is linearly convergent, there exist $\alpha \in [0, 1)$ with

$$d_{S^{\star}}^{2}((x^{1}, y^{1})) \leq \alpha d_{S^{\star}}^{2}((x^{0}, y^{0})) \leq \alpha \left(\|x^{0} - x^{\star}\|^{2} + \|y^{0} - y^{\star}\|^{2} \right).$$
(7.17)

By setting $x^1 = x^0 - t \nabla_x F(x^0, y^0)$ and $y^1 = y^0 + t \nabla_y F(x^0, y^0)$ in inequality (7.17), we get

$$\frac{1-\alpha}{2t} \left(\|x^0 - x^\star\|^2 + \|y^0 - y^\star\|^2 \right) \le \left\langle \nabla_x F(x^0, y^0), x^0 - x^\star \right\rangle - \left\langle \nabla_y F(x^0, y^0), y^0 - y^\star \right\rangle,$$

which implies that

$$\mu_F d_{S^*}^2(x^0, y^0) \le \left\langle \nabla_x F(x^0, y^0), x^0 - x^* \right\rangle - \left\langle \nabla_y F(x^0, y^0), y^0 - y^* \right\rangle,$$

for $\mu_F = \frac{1-\alpha}{2t}$ and the proof is complete.

7.4 Concluding remarks

In this chapter, we provided a new convergence rate for the gradient descentascent method for saddle point problems. Furthermore, we gave some necessary and sufficient conditions for the linear convergence without strong convexity. We employed the performance estimation method for proving the results. For future work, it would be interesting to consider the case where the variables x and yin the saddle point problem are constrained to lie in given, compact convex sets, since many saddle point problems fall in this category. In this case, one could use the performance estimation framework to analyze other methods, e.g. proximal type algorithms.

Be less curious about people and more curious about ideas.

Marie Curie

On the convergence rate of the difference-of-convex algorithm (DCA)

Preamble

In this chapter, we study the non-asymptotic convergence rate of the DCA (differenceof-convex algorithm), also known as the convex–concave procedure, with two different termination criteria that are suitable for smooth and non-smooth decompositions, respectively. The DCA is a popular algorithm for difference-of-convex (DC) problems and known to converge to a stationary point of the objective under some assumptions. We derive a worst-case convergence rate of $O(1/\sqrt{N})$ after *N* iterations of the objective gradient norm for certain classes of DC problems, without assuming strong convexity in the DC decomposition and give an example which shows the convergence rate is exact. We also provide a new convergence rate of O(1/N) for the DCA with the second termination criterion. Furthermore, we study the impact of using regularization in DCA. Moreover, we derive a new linear convergence rate result for the DCA under the assumption of the Polyak–Lojasiewicz inequality. The novel aspect of our analysis is that it employs semidefinite programming performance estimation. This chapter is based on the paper [AdKZ23c], except for Section 8.2.2, 8.3.3, 8.3.4 and 8.5 that deals with an example for DCA, studying the gradient descent method, studying the proximal gradient method, and DCA with regularization, respectively.

8.1 Introduction

We consider the general difference-of-convex (DC) optimization problem,

$$\inf f(x) := f_1(x) - f_2(x) \tag{8.1}$$

s.t. $x \in \mathbb{R}^n$,

where $f_1, f_2 : \mathbb{R}^n \to (-\infty, \infty]$ are convex functions and f is an lowersemicontinuous function on \mathbb{R}^n to $(-\infty, \infty]$. Throughout the chapter, we assume that the infimum in problem (8.1) is finite, and denote by f^* a lower bound of f on \mathbb{R}^n .

DC problems appear naturally in many applications, e.g. power allocation in digital communication systems [ASP14], production-transportation planning [HT99a], location planning [CHJT98], image processing [LZOX15], sparse signal recovering [GRC09], cluster analysis [BU18, BTU16], and supervised data classification [AFG12, LTN17], to name but a few.

This wide range of applications is to be expected, since some important classes of nonconvex functions may be represented as DC functions. For instance, twice continuously differentiable functions on any convex subset of \mathbb{R}^n [Har59], and continuous piece-wise linear functions [Mel86] may be written as DC functions. Furthermore, every continuous function on a compact and convex set can be approximated by a DC function [HT99b, THH⁺98]. We refer the interested reader to [HU85, THH⁺98] for more information on DC representable functions.

The celebrated Difference-of-Convex Algorithm (DCA), also known as the convex-concave procedure, has been applied extensively to problem (8.1); see [LTPD18, LB16, TA97] and the references therein. Algorithm 8.1 presents the basic form of the *DCA*.

In the description of the DCA in Algorithm 8.1, (sub)gradients of f_1 and f_2 are assumed to be available at given points, the so-called black-box formulation. The DCA is sometimes also presented as a primal-dual method, where a dual subproblem is solved to obtain the required (sub)gradients; see [LTPD18, LB16] for further discussions of this topic. In recent years, some scholars have also extended

Algorithm 8.1 DCA

Pick $x^0 \in \mathbb{R}^n$.

For k = 0, 1, ... perform the following steps:

1. Choose $g_2^k \in \partial f_2(x^k)$.

2. Choose

$$x^{k+1} \in \operatorname{argmin}_{x \in \mathbb{R}^n} f_1(x) - f_2(x^k) - \langle g_2^k, x - x^k \rangle.$$
(8.2)

3. If the termination criteria are satisfied, then stop.

the DCA and proposed some new variations; see [GTT18, LZ19, LZS19, PRA17, SS22].

The first convergence results for Algorithm 8.1 were given in [TA97, Theorem 3(iv)]. The authors showed that, if the sequence of iterates $\{x^k\}$ is bounded, then each accumulation point of this sequence is a critical point of f.

Le Thi et al. [LTHPD18] established an asymptotic linear convergence rate of $\{x^k\}$ under some conditions, in particular under the assumption that f satisfies the *Lojasiewicz gradient inequality* at all stationary points. Recall that a differentiable function f is said to satisfy this inequality at a stationary point a ($\nabla f(a) = 0$), if there exist constants $\theta \in (0, 1)$, C > 0 and $\epsilon > 0$ such that

$$|f(x) - f(a)|^{\theta} \le C \|\nabla f(x)\| \text{ if } \|x - a\| \le \epsilon, \tag{8.3}$$

where the constant θ is called the *Lojasiewicz exponent*. This inequality is known to hold, for example, for real analytic functions, but has been extended to include classes of non-smooth functions as well by considering general sub-differentials instead of gradients; see [BDL07, BST14], and the references therein.

The convergence rates established by Le Thi et al. [LTHPD18] depend on the value of the Łojasiewicz exponent, as the following theorem shows. The theorem stated here is a special case of Theorems 3.4 and 3.5 in [LTHPD18], to give a flavor of the convergence results in [LTHPD18].

Theorem 8.1 (Theorems 3.4 and 3.5 in Le Thi et al. [LTHPD18]). Let f_1 and f_2 be proper convex functions and let the domain of f be closed. Also assume that at least one of f_1 and f_2 is strongly convex, and f_1 or f_2 is differentiable with locally Lipschitz gradient in every critical point of the DC problem. Finally, assume the sequence $\{x^k\}$

is bounded, and let x^{∞} be a limit point of $\{x^k\}$. Then x^{∞} is also a stationary point. Moreover, if f satisfies the Łojasiewicz gradient inequality (8.3) at all stationary points, then

1. if
$$\theta \in (1/2, 1)$$
, then $||x^k - x^{\infty}|| \le ck^{\frac{1-\theta}{1-2\theta}}$ for some $c > 0$.
2. if $\theta \in (0, 1/2]$, then $||x^k - x^{\infty}|| \le cq^k$ for some $c > 0$ and $q \in (0, 1)$

In particular, item 2 shows a linear convergence rate when $\theta \in (0, 1/2]$. Yen et al. [YPWL12] had already shown linear convergence earlier for a much smaller class of DC functions. We will present a complementary result to this theorem (see Theorem 8.20 below), for the case $\theta = 1/2$, where we show linear convergence of the objective function values, and give explicit expressions for the constants that determine the linear convergence rate. Moreover, we will relax the assumption of a bounded sequence of iterates, and the assumption of strong convexity.

In the absence of conditions like the Łojasiewicz gradient inequality (8.3), only weaker convergence rates are known for the DCA. In particular, Tao and An [TA97, Proposition 2] and Le Thi et al. [LTPD21, Corollary 1] have shown an $\mathcal{O}\left(\frac{1}{\sqrt{N}}\right)$ convergence rate after *N* iterations under suitable assumptions, as given in the next theorem.

Theorem 8.2 (Corollary 1 in [LTPD21], Proposition 2 in [TA97]). If x^{∞} is a limit point of the iteration sequence generated by the DCA, and at least one of f_1 and f_2 is strongly convex, i.e., for some $\mu_1, \mu_2 \ge 0$ such that $\mu_1 + \mu_2 > 0$,

$$x \mapsto f_i(x) - \frac{\mu_i}{2} ||x||^2$$
 is convex for $i \in \{1, 2\}$,

then the series $||x^{k} - x^{k-1}||$ converges, and, after N + 1 iterations,

$$\sum_{k=0}^{N} \|x^{k} - x^{k-1}\|^{2} \leq \frac{2(f(x^{0}) - f(x^{N}))}{\mu_{1} + \mu_{2}},$$

and, consequently,

$$\min_{0 \le k \le N} \|x^k - x^{k-1}\| \le \sqrt{\frac{2(f(x^0) - f^*)}{(\mu_1 + \mu_2)N}} = \mathcal{O}\left(\frac{1}{\sqrt{N}}\right).$$

We will derive some variants on this $\mathcal{O}\left(\frac{1}{\sqrt{N}}\right)$ convergence result in Corollary 8.7 and in Section 8.3.2, where we improve the constants in the $\mathcal{O}\left(\frac{1}{\sqrt{N}}\right)$ bounds. We also show that we obtain the best possible constants, by demonstrating an example where our bound in Corollary 8.7 is tight.
Outline and further contributions of this chapter

The novel aspect of the analysis in this chapter is that we will apply performance estimation to derive convergence rates. This chapter is organized as follows. In Section 8.2 we review some definitions and notions from convex analysis, which will be used in the following sections. We continue with a simple example to show how DCA works in Section 8.2.2. We study the DCA for sufficiently smooth DC decompositions in Section 8.3. By using performance estimation, we give a convergence rate of $\mathcal{O}(1/\sqrt{N})$ in Corollary 8.7, without any strong convexity assumption, thus extending and complementing Le Thi et al. [LTPD21, Corollary 1]. We construct an example that shows this $\mathcal{O}(1/\sqrt{N})$ bound is tight. Since the first termination criterion is not suitable for the analysis of nonsmooth DC compositions, we investigate the DCA with another stopping criterion in Section 8.4, and we show a convergence rate of $\mathcal{O}(1/N)$. This result is completely new to the best of our knowledge. Furthermore, we investigate DCA with regularization and derive a convergence rate for this version. Moreover, we discuss the best choice of regularization parameter from worst-case complexity perspective. In Section 8.6 we study the DCA when the objective function satisfies the Polyak-Lojasiewicz inequality, and we derive a linear convergence rate in Theorem 8.20, thereby refining some linear convergence results in Le Thi et al. [LTHPD18] as described above.

8.2 Basic Definitions and Preliminaries

 $I_{\mathbb{R}_+}$ stands for the indicator function on $\mathbb{R}_+ \cup \{\infty\}$, i.e.,

$$I_{\mathbb{R}_{+}}(x) = \begin{cases} 1 & x \ge 0 \cup \{\infty\} \\ 0 & x < 0 \cup \{-\infty\}. \end{cases}$$
(8.4)

We denote the *convex hull* of $X \subseteq \mathbb{R}^n$ by co(X). We adopt the conventions that, for $a, b, c, d \in \mathbb{R}$ with $c \neq d$ and $a \neq 0$, $\frac{b}{\infty} = 0, 0 \times \infty = 0$ and $\frac{a \infty + b}{c \infty - d \infty} = \frac{a}{c - d}$.

Let $L \in (0, \infty]$ and $\mu \in (0, \infty)$. We call an extended convex function $f : \mathbb{R}^n \to (-\infty, \infty]$ *L*-smooth if for any $x_1, x_2 \in \mathbb{R}^n$,

$$\|g_1 - g_2\| \le L \|x_1 - x_2\| \quad \forall g_1 \in \partial f(x_1), \ g_2 \in \partial f(x_2).$$

Note that if $L < \infty$, then f must be differentiable on \mathbb{R}^n . In addition, any extended convex function is ∞ -smooth.

Lemma 8.3. If $-\eta \leq \mu$, then $\mathcal{F}_{\mu,L}(\mathbb{R}^n) + \frac{\eta}{2} \|\cdot\|^2 = \mathcal{F}_{\mu+\eta,L+\eta}(\mathbb{R}^n)$.

Proof. First, we show the inclusion $\mathcal{F}_{\mu,L}(\mathbb{R}^n) + \frac{\eta}{2} \|\cdot\|^2 \subseteq \mathcal{F}_{\mu+\eta,L+\eta}(\mathbb{R}^n)$. Let $f \in \mathcal{F}_{\mu,L}$. The $(\mu + \eta)$ -strong convexity of the function $f + \frac{\eta}{2} \|\cdot\|^2$ is immediate from the definition. It is easily seen the inclusion holds when $L = \infty$. Hence, we investigate $L < \infty$. By *L*-smoothness, we have

$$f(y) \le f(x) + \langle \nabla f(x), y - x \rangle + \frac{L}{2} ||y - x||^2.$$

By adding $\frac{\eta}{2} ||y||^2 = \frac{\eta}{2} ||x||^2 + \langle \eta x, y - x \rangle + \frac{\eta}{2} ||y - x||^2$ to the above inequality, we get

$$f(y) + \frac{\eta}{2} \|y\|^2 \le f(x) + \frac{\eta}{2} \|x\|^2 + \langle \nabla f(x) + \eta x, y - x \rangle + \frac{L + \eta}{2} \|y - x\|^2,$$

which establishes $(L + \eta)$ -smoothness $f + \frac{\eta}{2} \|\cdot\|^2$; see Theorem 2.15 in [Nes18]. Now, we establish the converse inclusion. Suppose that $f \in \mathcal{F}_{\mu+\eta,L+\eta}(\mathbb{R}^n)$. By the definition, it follows that $f - \frac{\eta}{2} \|\cdot\|^2$ is μ -strongly convex. The *L*-smoothness of *f* is proved similar to the former case and the proof is complete.

In the next lemma, we extend the descent lemma for DCA when L_1 or L_2 is finite.

Lemma 8.4. Let $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$ and let $f = f_1 - f_2$. If $g_1 \in \partial f_1(x)$ and $g_2 \in \partial f_2(x)$, then

$$f^{\star} \leq f(x) - \frac{1}{2(L_1 - \mu_2)} ||g_1 - g_2||^2.$$

Proof. If $L_1 = \infty$, the proof is immediate. Let $L_1 < \infty$. By *L*-smoothness and strong convexity, we have

$$\begin{split} f_1(y) &\leq f_1(x) + \langle g_1, y - x \rangle + \frac{L_1}{2} \|y - x\|^2, \\ f_2(y) &\geq f_2(x) + \langle g_2, y - x \rangle + \frac{\mu_2}{2} \|y - x\|^2, \end{split}$$

for $y \in \mathbb{R}^n$. By the above inequalities, we get

$$f(y) \le f(x) + \langle g_1 - g_2, y - x \rangle + \frac{L_1 - \mu_2}{2} ||y - x||^2.$$

Hence, by taking minimum on both sides of the last inequality with respect to y for fixed x, we get

$$f^* \leq f(x) - \frac{1}{2(L_1 - \mu_2)} ||g_1 - g_2||^2.$$

Since the DC optimization problem (8.1) may have a non-convex and nonsmooth objective function f, we will also need a more general notion of subgradients than in the convex case which is defined in Section 2.1.6.

Definition 8.5. Let f_1, f_2 be closed proper convex functions, and let f be lower semi-continuous.

• The point $\bar{x} \in \text{dom}(f)$ is called a *critical point* of problem (8.1) if

$$\partial f_1(\bar{x}) \cap \partial f_2(\bar{x}) \neq \emptyset.$$
 (8.5)

• The point $\bar{x} \in \text{dom}(f)$ is called a *stationary point* of problem (8.1) if

$$0 \in \partial_L f(\bar{x}), \tag{8.6}$$

where ∂_L stands for general subdifferential, see Section 2.1.6.

Obviously, the stationarity condition is stronger than criticality. We recall that a convex function will be locally Lipschitz around \bar{x} providing it takes finite values in a neighborhood of \bar{x} ; see Theorem 35.1 in [Roc97]. Consequently, if f_1 or f_2 takes finite values around a neighborhood of a stationary point \bar{x} , then \bar{x} is a critical point; see Corollary 10.9 in [RW09]. However, its converse does not hold in general. For instance, consider $f : \mathbb{R} \to \mathbb{R}$ given as f(x) = x. The function fmay be written as $f = f_1 - f_2$ where $f_1(x) = \max(x, 0)$ and $f_2(x) = \max(-x, 0)$. Suppose that $\bar{x} = 0$. It is readily seen that $\partial f_1(\bar{x}) \cap \partial f_2(\bar{x}) \neq \emptyset$, but $\bar{x} = 0$ is not a stationary point of f. It is worth noting that, if f_2 is strictly differentiable at \bar{x} , these definitions are equivalent; see Example 10.10 in [RW09]. Recall that function f is strictly differentiable at \bar{x} , if

$$\lim_{\substack{(x,x')\to(\bar{x},\bar{x})\\x\neq x'}}\frac{f(x)-f(x')-\langle \nabla f(\bar{x}), x-x'\rangle}{\|x-x'\|}=0.$$

We refer the interested reader to [AT05, JBK⁺18, PRA17] and references therein for more discussions on optimality conditions for DC problems.

8.2.1 The DC problem

In this section, we consider

$$\min f(x) = f_1(x) - f_2(x)$$
s.t. $x \in \mathbb{R}^n$,
$$(8.7)$$

where $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$. Here, we assume that $L_1, L_2 \in (0, \infty)$ and $\mu_1, \mu_2 \in [0, \infty)$, and consequently f may be non-differentiable. We may assume without loss of generality that f_1 and f_2 satisfy the following assumptions:

$$L_1 > \mu_2, \quad L_2 > \mu_1.$$
 (8.8)

Indeed, if $L_1 \leq \mu_2$, then for $x, y \in \mathbb{R}^n$ and $\lambda \in [0, 1]$, we have

$$\begin{aligned} \lambda f_1(x) + (1-\lambda)f_1(y) &\leq f_1(\lambda x + (1-\lambda)y) + \lambda(1-\lambda)\frac{L_1}{2} ||x-y||^2 \\ &-\lambda f_2(x) - (1-\lambda)f_2(y) \leq -f_2(\lambda x + (1-\lambda)y) - \lambda(1-\lambda)\frac{\mu_2}{2} ||x-y||^2; \end{aligned}$$

see Theorem 2.15 and Theorem 2.19 in [Nes18]. By summing the above inequalities, we obtain

$$\lambda f(x) + (1-\lambda)f(y) \le f(\lambda x + (1-\lambda)y) + \lambda(1-\lambda)\frac{L_1-\mu_2}{2} ||x-y||^2,$$

which implies concavity of f on \mathbb{R}^n . In this case, problem (8.7) will be unbounded from below. This follows from the fact that a concave function on \mathbb{R}^n is unbounded from below unless it is constant. Likewise, one can show that problem (8.7) will be convex providing $L_2 \leq \mu_1$.

The Toland dual [Tol79] of problem (8.7) may be written as

$$\min f_{2}^{*}(x) - f_{1}^{*}(x)$$
s.t. $x \in \mathbb{R}^{n}$.
(8.9)

It is known that problems (8.7) and (8.9) share the same optimal value [Tol79].

In what follows, we investigate the convergence rate of Algorithm 8.1 with the termination criterion $||g_1^k - g_2^k|| \le \epsilon$. As a motivation of this criterion, recall that $||g_1^k - g_2^k|| = 0$ implies that x^k is a critical point of (8.1) in the non-smooth case, and a stationary point of f if f_2 is differentiable; see our discussion following Definition 8.5. In Section 8.3 we will derive results for the case that at least one of f_1 or f_2 is differentiable, and we will consider the more general situation in Section 8.4.

For well-definedness of the DCA (Algorithm 8.1), throughout the chapter, we assume that

$$x^k \in \operatorname{dom}(\partial f_1) \cap \operatorname{dom}(\partial f_2) \ k = 0, 1, \dots,$$

where dom $(\partial f_1) = \{x : \partial f_1(x) \neq \emptyset\}$. It is worth noting that similar algorithm has been developed for the dual problem in [LTPD18], and (8.2) is equivalent to $x^{k+1} \in \partial f_1^*(g_2^k)$.

8.2.2 One iteration of DCA on an example

Consider the following unconstrained optimization problem (see Figure 8.1).

 $\min_{x \in \mathbb{R}} f(x) := \frac{1}{4}x^4 - \frac{2}{3}x^3 - \frac{1}{2}x^2 + 2x, \quad x^0 = 3.$



Figure 8.1: Illustration of the function (8.10)

It is easily seen that the problem can be written in the following form.

$$f(x) = \underbrace{\left(\frac{1}{4}x^4 - \frac{2}{3}x^3 + 2x^2\right)}_{f_1} - \underbrace{\left(\frac{1}{2}x^2 - 2x + 2x^2\right)}_{f_2},\tag{8.11}$$

where f_1 and f_2 are convex functions (see Figure 8.2).



Figure 8.2: Illustration of the functions f_1 and f_2 of problem (8.11)

(8.10)

By taking $x^0 = 3$ as the starting point of the algorithm, the following subproblem of the algorithm should be solved (see Figure 8.3).



Figure 8.3: Illustration of the subproblem (8.12)

By solving subproblem (8.12), DCA generates $x^1 = 2.5$ which is illustrated in Figure 8.4.





For this example, in the next iteration DCA generates $x^2 \approx 2.2728$, $x^3 \approx 2.1576$ and so on that converges to the local minimizer at $\bar{x} = 2$.

8.3 Performance analysis of the DCA for smooth f_1 or f_2

In this subsection, we apply performance estimation for the analysis of Algorithm 8.1 for the case that at least one of f_1 or f_2 is *L*-smooth for some finite L > 0. The worst-case convergence rate of Algorithm 8.1 can be obtained by solving the following abstract optimization problem:

$$\max\left(\min_{0\leq k\leq N} \left\|g_{1}^{k}-g_{2}^{k}\right\|^{2}\right)$$

$$g_{1}^{N}, g_{2}^{N}, x^{N}, \dots, x^{1} \text{ are generated by Algorithm 8.1 w.r.t. } f_{1}, f_{2}, x^{0}$$

$$f(x) \geq f^{\star} \quad \forall x \in \mathbb{R}^{n}$$

$$f_{1} \in \mathcal{F}_{\mu_{1}, L_{1}}(\mathbb{R}^{n}), f_{2} \in \mathcal{F}_{\mu_{2}, L_{2}}(\mathbb{R}^{n})$$

$$f_{1}(x^{0}) - f_{2}(x^{0}) - f^{\star} \leq \Delta$$

$$x^{0} \in \mathbb{R}^{n},$$

$$(8.13)$$

where $\Delta \ge 0$ denote the difference between the optimal value and the value of f at the starting point. Here, f_1, f_2 and x^k , g_1^k and g_2^k ($k \in \{0, ..., N\}$) are decision variables, and $\Delta, \mu_1, L_1, \mu_2, L_2$ and N are fixed parameters.

Problem (8.13) is an intractable *infinite-dimensional optimization problem* with an infinite number of constraints. In what follows, we provide a *semidefinite programming* relaxation of the problem.

By Theorem 2.37, problem (8.13) can be written as,

$$\max\left(\min_{0 \le k \le N} \left\|g_{1}^{k} - g_{2}^{k}\right\|^{2}\right)$$
s.t.
$$\frac{1}{2(1 - \frac{\mu_{1}}{L_{1}})} \left(\frac{1}{L_{1}} \left\|g_{1}^{i} - g_{1}^{j}\right\|^{2} + \mu_{1} \left\|x^{i} - x^{j}\right\|^{2} - \frac{2\mu_{1}}{L_{1}} \left\langle g_{1}^{j} - g_{1}^{i}, x^{j} - x^{i} \right\rangle \right) \le$$

$$f_{1}^{i} - f_{1}^{j} - \left\langle g_{1}^{j}, x^{i} - x^{j} \right\rangle \quad i, j \in \{0, \dots, N\}$$

$$\frac{1}{2(1 - \frac{\mu_{2}}{L_{2}})} \left(\frac{1}{L_{2}} \left\|g_{2}^{i} - g_{2}^{j}\right\|^{2} + \mu_{2} \left\|x^{i} - x^{j}\right\|^{2} - \frac{2\mu_{2}}{L_{2}} \left\langle g_{2}^{j} - g_{2}^{i}, x^{j} - x^{i} \right\rangle \right) \le$$

$$f_{2}^{i} - f_{2}^{j} - \left\langle g_{2}^{j}, x^{i} - x^{j} \right\rangle \quad i, j \in \{0, \dots, N\}$$

$$g_{1}^{k+1} = g_{2}^{k} \quad k \in \{0, \dots, N-1\}$$

$$f_{1}^{k} - f_{2}^{k} - \frac{1}{2(L_{1} - \mu_{2})} \left\|g_{1}^{k} - g_{2}^{k}\right\|^{2} \ge f^{\star} \quad k \in \{0, \dots, N\}$$

$$f_{1}^{0} - f_{2}^{0} - f^{\star} \le \Delta.$$

$$(8.14)$$

In problem (8.14), f^* and x^k , g_1^k , g_2^k , f_1^k , f_2^k , $k \in \{0, ..., N\}$, are decision variables. By virtue of Lemma 8.4, constraints $f(x) \ge f^*$ for each $x \in \mathbb{R}^n$

is replaced by $f_1^k - f_2^k - \frac{1}{2(L_1 - \mu_2)} ||g_1^k - g_2^k||^2 \ge f^*$, $k \in \{0, \dots, N\}$. Due to the necessary and sufficient optimality conditions for convex problems, $x^{k+1} \in \arg\min_{x \in \mathbb{R}^n} f_1(x) - f_2(x^k) - \langle g_2^k, x - x^k \rangle$, $k \in \{0, \dots, N-1\}$ implies $g_1^{k+1} = g_2^k$ for some $g_1^{k+1} \in \partial f(x^{k+1})$; see Theorem 3.63 in [Bec17]. By substituting $g_2^k = g_1^{k+1}$, $k \in \{0, \dots, N-1\}$, the above formulation may be written as:

$$\begin{split} \max \ell \\ \text{s.t. } & \left\| g_{1}^{i} - g_{1}^{i+1} \right\|^{2} \geq \ell \quad i \in \{0, \dots, N-1\} \\ & \left\| g_{1}^{N} - g_{2}^{N} \right\|^{2} \geq \ell \\ & \frac{1}{2(1 - \frac{\mu_{1}}{L_{1}})} \left(\frac{1}{L_{1}} \left\| g_{1}^{i} - g_{1}^{j} \right\|^{2} + \mu_{1} \left\| x^{i} - x^{j} \right\|^{2} - \frac{2\mu_{1}}{L_{1}} \left\langle g_{1}^{j} - g_{1}^{i}, x^{j} - x^{i} \right\rangle \right) \leq \\ & f_{1}^{i} - f_{1}^{j} - \left\langle g_{1}^{j}, x^{i} - x^{j} \right\rangle \quad i, j \in \{0, \dots, N\} \\ & \frac{1}{2(1 - \frac{\mu_{2}}{L_{2}})} \left(\frac{1}{L_{2}} \left\| g_{1}^{i+1} - g_{1}^{j+1} \right\|^{2} + \mu_{2} \left\| x^{i} - x^{j} \right\|^{2} - \frac{2\mu_{2}}{L_{2}} \left\langle g_{1}^{j+1} - g_{1}^{i+1}, x^{j} - x^{i} \right\rangle \right) \leq \\ & f_{2}^{i} - f_{2}^{j} - \left\langle g_{1}^{j+1}, x^{i} - x^{j} \right\rangle \quad i, j \in \{0, \dots, N-1\} \\ & \frac{1}{2(1 - \frac{\mu_{2}}{L_{2}})} \left(\frac{1}{L_{2}} \left\| g_{2}^{N} - g_{1}^{j+1} \right\|^{2} + \mu_{2} \left\| x^{N} - x^{j} \right\|^{2} - \frac{2\mu_{2}}{L_{2}} \left\langle g_{1}^{j+1} - g_{2}^{N}, x^{j} - x^{N} \right\rangle \right) \\ & \leq f_{2}^{N} - f_{2}^{j} - \left\langle g_{1}^{j+1}, x^{N} - x^{j} \right\rangle \quad j \in \{0, \dots, N-1\} \\ & \frac{1}{2(1 - \frac{\mu_{2}}{L_{2}})} \left(\frac{1}{L_{2}} \left\| g_{1}^{i+1} - g_{2}^{N} \right\|^{2} + \mu_{2} \left\| x^{i} - x^{N} \right\|^{2} - \frac{2\mu_{2}}{L_{2}} \left\langle g_{2}^{N} - g_{1}^{i+1}, x^{N} - x^{i} \right\rangle \right) \\ & \leq f_{2}^{N} - f_{2}^{j} - \left\langle g_{1}^{j+1}, x^{N} - x^{j} \right\rangle \quad j \in \{0, \dots, N-1\} \\ & \frac{1}{2(1 - \frac{\mu_{2}}{L_{2}})} \left(\frac{1}{L_{2}} \left\| g_{1}^{i+1} - g_{2}^{N} \right\|^{2} + \mu_{2} \left\| x^{i} - x^{N} \right\|^{2} - \frac{2\mu_{2}}{L_{2}} \left\langle g_{2}^{N} - g_{1}^{i+1}, x^{N} - x^{i} \right\rangle \right) \\ & \leq f_{2}^{1} - f_{2}^{N} - \left\langle g_{2}^{N}, x^{i} - x^{N} \right\rangle \quad i \in \{0, \dots, N-1\} \\ & f_{1}^{N} - f_{2}^{N} - \frac{1}{2(L_{1} - \mu_{2})} \left\| g_{1}^{N} - g_{2}^{N} \right\|^{2} \geq f^{\star} \quad k \in \{0, \dots, N-1\} \\ & f_{1}^{N} - f_{2}^{N} - \frac{1}{2(L_{1} - \mu_{2})} \left\| g_{1}^{N} - g_{2}^{N} \right\|^{2} \geq f^{\star} \\ & f_{1}^{0} - f_{2}^{0} - f^{\star} \leq \Delta. \end{split}$$

By using this formulation, the next result (Theorem 8.6) provides a convergence rate for Algorithm 8.1. Since the proof is quite technical, a few remarks are in order. The proof uses the performance estimation technique of Drori and Teboulle [DT14], that consists of the following steps:

- 1. Observe that problem (8.15) may be rewritten as a semidefinite programming (SDP) problem (for sufficiently large *N*) by replacing all inner products by the entries of an unknown Gram matrix.
- 2. Use weak duality of SDP to bound the optimal value of (8.15) by construct-

ing a dual feasible solution.

- 3. The dual feasible solution is constructed empirically, by first doing numerical experiments with fixed values of the parameters Δ , N, μ_1 , L_1 , μ_2 , L_2 , and noting the dual multipliers.
- 4. Subsequently, the analytical expressions of the dual multipliers are guessed, based on the numerical values, and the guess is verified analytically.
- 5. In the proof of Theorem 8.6, the conjectured dual multipliers are simply stated, and then shown to provide the required bound on the optimal value of (8.15) through the corresponding aggregation of the constraints of (8.15).

Theorem 8.6. Let $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$ and let $f(x^0) - f^* = \Delta$. Suppose that L_1 or L_2 is finite. Then after N iterations of Algorithm 8.1, one has:

$$\min_{0 \le k \le N} \left\| g_1^k - g_2^k \right\| \le \sqrt{\frac{\mathcal{A}\Delta}{\mathcal{B}N + \mathcal{C}}},\tag{8.16}$$

where

$$\begin{aligned} \mathcal{A} &= 2 \left(L_1 L_2 - \mu_1 L_2 I_{\mathbb{R}_+} (L_1 - L_2) - \mu_2 L_1 I_{\mathbb{R}_+} (L_2 - L_1) \right), \\ \mathcal{B} &= L_1 + L_2 + \mu_1 \left(\frac{L_1}{L_2} - 3 \right) I_{\mathbb{R}_+} (L_1 - L_2) + \mu_2 \left(\frac{L_2}{L_1} - 3 \right) I_{\mathbb{R}_+} (L_2 - L_1), \end{aligned}$$

and

$$C = \frac{L_1 L_2 - \mu_1 L_2 I_{\mathbb{R}_+} (L_1 - L_2) - \mu_2 L_1 I_{\mathbb{R}_+} (L_2 - L_1)}{L_1 - \mu_2}$$

where $I_{\mathbb{R}_+}$ stands for indicator function defined by (8.4).

Proof. We investigate two cases $L_1 \ge L_2$ and $L_1 < L_2$. Suppose that *U* denotes the square of the right-side of inequality (8.16) and let $B = \frac{U}{\Delta}$. To prove this bound, we show that *U* is an upper bound for problem (8.15). First, we consider $L_1 \ge L_2$.

Let

$$\begin{split} \bar{\lambda} &= \frac{2\left(L_{1}L_{2} - \mu_{1}(2L_{2} - L_{1})\right)}{N\left(L_{1} + L_{2} + \mu_{1}\left(\frac{L_{1}}{L_{2}} - 3\right)\right) + \frac{L_{2}(L_{1} - \mu_{1})}{L_{1} - \mu_{2}}} \\ \bar{\eta}_{1} &= \frac{L_{2} - \mu_{1}}{\left(L_{1} + L_{2} + \mu_{1}\left(\frac{L_{1}}{L_{2}} - 3\right)\right)N + \frac{L_{2}(L_{1} - \mu_{1})}{L_{1} - \mu_{2}}} \\ \bar{\eta}_{k} &= \frac{\frac{L_{1}\mu_{1}}{L_{2}} + \left(L_{1} + L_{2} - 3\mu_{1}\right)}{\left(L_{1} + L_{2} + \mu_{1}\left(\frac{L_{1}}{L_{2}} - 3\right)\right)N + \frac{L_{2}(L_{1} - \mu_{1})}{L_{1} - \mu_{2}}}, \quad k \in \{2, \dots, N\} \\ \bar{\eta}_{N+1} &= 1 - \bar{\eta}_{1} - \sum_{k=2}^{N} \bar{\eta}_{k} = \frac{\frac{L_{1}\mu_{1}}{L_{2}} + L_{1} - 2\mu_{1} + \frac{L_{2}(L_{1} - \mu_{1})}{L_{1} - \mu_{2}}}{\left(L_{1} + L_{2} + \mu_{1}\left(\frac{L_{1}}{L_{2}} - 3\right)\right)N + \frac{L_{2}(L_{1} - \mu_{1})}{L_{1} - \mu_{2}}}. \end{split}$$

By direct calculation, one can verify that

$$\begin{split} \ell &-U + \bar{\eta}_1 \left(\left\| g_1^0 - g_1^1 \right\|^2 - \ell \right) + \sum_{k=2}^N \bar{\eta}_k \left(\left\| g_1^{k-1} - g_1^k \right\|^2 - \ell \right) + \bar{\eta}_{N+1} \left(\left\| g_1^N - g_2^N \right\|^2 - \ell \right) \\ &+ B \left(f^* - f_1^0 + f_2^0 + \Delta \right) + B \left(f_1^N - f_2^N - \frac{1}{2(L_1 - \mu_2)} \| g_1^N - g_2^N \|^2 - f^* \right) + \\ B \sum_{k=1}^N \left(f_1^{k-1} - f_1^k - \left\langle g_1^k, x^{k-1} - x^k \right\rangle - \frac{1}{2(1 - \frac{\mu_1}{L_1})} \left(\frac{1}{L_1} \| g_1^{k-1} - g_1^k \|^2 + \mu_1 \| x^{k-1} - x^k \|^2 - \frac{2\mu_1}{L_1} \left\langle g_1^k - g_1^{k-1}, x^k - x^{k-1} \right\rangle \right) \right) \\ &+ \bar{\lambda} \sum_{k=1}^{N-1} \left(f_2^k - f_2^{k-1} - \left\langle g_1^k, x^k - x^{k-1} \right\rangle - \frac{1}{2(1 - \frac{\mu_2}{L_2})} \left(\frac{1}{L_2} \| g_1^k - g_1^{k+1} \|^2 + \mu_2 \| x^{k-1} - x^k \|^2 - \frac{2\mu_2}{L_2} \left\langle g_1^{k+1} - g_1^k, x^k - x^{k-1} \right\rangle \right) \right) \\ &+ (\bar{\lambda} - B) \sum_{k=1}^{N-1} \left(f_2^{k-1} - f_2^k - \left\langle g_1^{k+1}, x^{k-1} - x^k \right\rangle - \frac{1}{2(1 - \frac{\mu_2}{L_2})} \left(\frac{1}{L_2} \| g_1^k - g_1^{k+1} \|^2 + \mu_2 \| x^{N-1} - x^k \|^2 - \frac{2\mu_2}{L_2} \left\langle g_2^N - g_1^N, x^N - x^{N-1} \right\rangle \right) \right) \\ &+ (\bar{\lambda} - B) \sum_{k=1}^{N-1} \left(f_2^{k-1} - g_2^k \|^2 + \mu_2 \| x^{N-1} - x^k \|^2 - \frac{2\mu_2}{2L_2} \left\langle g_2^N - g_1^N, x^N - x^{N-1} \right\rangle \right) \\ &- \frac{1}{2(1 - \frac{\mu_2}{L_2})} \left(\frac{1}{L_2} \| g_1^N - g_2^N \|^2 + \mu_2 \| x^{N-1} - x^N \|^2 - \frac{2\mu_2}{2L_2} \left\langle g_2^N - g_1^N, x^N - x^{N-1} \right\rangle \right) \right) \\ &+ \bar{\lambda} \left(f_2^N - f_2^{N-1} - \left\langle g_1^N, x^N - x^{N-1} \right\rangle - \frac{1}{2(1 - \frac{\mu_2}{L_2})} \left(\frac{1}{L_2} \| g_1^N - g_2^N \|^2 + \mu_2 \| x^{N-1} - x^N \|^2 - \frac{2\mu_2}{L_2} \left\langle g_2^N - g_1^N, x^N - x^{N-1} \right\rangle \right) \right) \\ &- \bar{\beta}_1^{-1} \sum_{i=1}^N \left\| \bar{\beta}_1 g_1^{i-1} - \bar{\beta}_1 g_1^i - \bar{\alpha}_1 x^{i-1} + \bar{\alpha}_1 x^i \|^2 - \bar{\alpha}_2^{-1} \sum_{i=1}^{N-1} \| \bar{\alpha}_2 x^{i-1} - \bar{\alpha}_2 x^i - \bar{\beta}_2 g_1^{i+1} \|^2 \\ &- \bar{\alpha}_2^{-1} \| \| \bar{\alpha}_2 x^{N-1} - \bar{\alpha}_2 x^N - \bar{\beta}_2 g_1^N + \bar{\beta}_2 g_2^N \|^2 \leq 0, \end{split}$$

where

$$\begin{split} \bar{\alpha}_1 &= \frac{\mu_1 B}{2(L_1 - \mu_1)}, \quad \bar{\beta}_1 = \frac{\mu_1 B}{2L_2(L_1 - \mu_1)}, \\ \bar{\alpha}_2 &= \frac{(-\mu_1 L_2^2 - 2\mu_1 \mu_2 L_2 + \mu_1 L_1 L_2 + \mu_1 \mu_2 L_1 + \mu_2 L_1 L_2) B}{2(L_1 - \mu_1)(L_2 - \mu_2)} \\ \bar{\beta}_2 &= \frac{(L_1 L_2 \mu_2 - 2\mu_1 \mu_2 L_2 + \mu_1 \mu_2 L_1 - \mu_1 L_2^2 + \mu_1 L_1 L_2) B}{2L_2(L_1 - \mu_1)(L_2 - \mu_2)}. \end{split}$$

It is readily seen that $\bar{\lambda}, \bar{\eta}_k$ ($k \in \{1, ..., N+1\}$), $\bar{\lambda} - B, \bar{\beta}_1, \bar{\alpha}_2 \ge 0$. Thus we have $\ell \le U$ for any feasible point of problem (8.15). Now, we consider $L_1 < L_2$. In this case, because bound (8.16) does not depend on μ_1 , we may assume $\mu_1 = 0$ in problem (8.15). Let

$$\begin{split} \hat{\lambda} &= \frac{2\left(L_{1}L_{2} - \mu_{2}(2L_{1} - L_{2})\right)}{\left(L_{1} + L_{2} + \mu_{2}\left(\frac{L_{2}}{L_{1}} - 3\right)\right)N + \frac{L_{1}(L_{2} - \mu_{2})}{L_{1} - \mu_{2}}} \\ \hat{\eta}_{1} &= \frac{\frac{L_{2}(L_{1} + \mu_{2})}{L_{1}} - 2\mu_{2}}{\left(L_{1} + L_{2} + \mu_{2}\left(\frac{L_{2}}{L_{1}} - 3\right)\right)N + \frac{L_{1}(L_{2} - \mu_{2})}{L_{1} - \mu_{2}}} \\ \hat{\eta}_{k} &= \frac{\frac{L_{2}(L_{1} + \mu_{2})}{L_{1}} + (L_{1} - 3\mu_{2})}{\left(L_{1} + L_{2} + \mu_{2}\left(\frac{L_{2}}{L_{1}} - 3\right)\right)N + \frac{L_{1}(L_{2} - \mu_{2})}{L_{1} - \mu_{2}}}, \quad k \in \{2, \dots, N\} \\ \hat{\eta}_{N+1} &= 1 - \hat{\eta}_{1} - \sum_{k=2}^{N} \hat{\eta}_{k} = \frac{\frac{L_{1}(L_{2} - \mu_{2})}{L_{1} - \mu_{2}} + L_{1} - \mu_{2}}{\left(L_{1} + L_{2} + \mu_{2}\left(\frac{L_{2}}{L_{1}} - 3\right)\right)N + \frac{L_{1}(L_{2} - \mu_{2})}{L_{1} - \mu_{2}}} \end{split}$$

,

With some calculation, one can establish that

$$\begin{split} \ell - U + \hat{\eta}_1 \left(\left\| g_1^0 - g_1^1 \right\|^2 - \ell \right) + \sum_{k=2}^N \hat{\eta}_k \left(\left\| g_1^{k-1} - g_1^k \right\|^2 - \ell \right) + \hat{\eta}_{N+1} \left(\left\| g_1^N - g_2^N \right\|^2 - \ell \right) \\ + B \left(f^* - f_1^0 + f_2^0 + \Delta \right) + B \left(f_1^N - f_2^N - \frac{1}{2(L_1 - \mu_2)} \| g_1^N - g_2^N \|^2 - f^* \right) \\ + (\hat{\lambda} - B) \sum_{k=1}^N \left(f_1^k - f_1^{k-1} - \left\langle g_1^{k-1}, x^k - x^{k-1} \right\rangle - \frac{1}{2L_1} \| g_1^k - g_1^{k-1} \|^2 \right) \\ + \hat{\lambda} \sum_{k=1}^N \left(f_1^{k-1} - f_1^k - \left\langle g_1^k, x^{k-1} - x^k \right\rangle - \frac{1}{2L_1} \| g_1^{k-1} - g_1^k \|^2 \right) \\ + B \sum_{k=1}^{N-1} \left(f_2^k - f_2^{k-1} - \left\langle g_1^k, x^k - x^{k-1} \right\rangle - \frac{1}{2L_2} \left\| g_1^k - g_1^{k-1} - g_1^k \right\|^2 + \mu_2 \left\| x^{k-1} - x^k \right\|^2 - \frac{2\mu_2}{L_2} \left\langle g_1^{k+1} - g_1^k, x^k - x^{k-1} \right\rangle \right) \right) \\ + B \left(f_2^N - f_2^{N-1} - \left\langle g_1^N, x^N - x^{N-1} \right\rangle - \frac{1}{2(1 - \frac{\mu_2}{L_2})} \left(\frac{1}{L_2} \left\| g_1^N - g_2^N \right\|^2 + \mu_2 \left\| x^{N-1} - x^N \right\|^2 - \frac{2\mu_2}{L_2} \left\langle g_2^N - g_1^N, x^N - x^{N-1} \right\rangle \right) \right) \\ = -\hat{\beta}_1^{-1} \sum_{i=1}^N \left\| \hat{\beta}_1 g_1^{i-1} - \hat{\beta}_1 g_1^i - \hat{\alpha}_1 x_1^{i-1} + \hat{\alpha}_1 x^i \right\|^2 - \hat{\alpha}_2^{-1} \sum_{i=1}^{N-1} \left\| \hat{\alpha}_2 x^{i-1} - \hat{\alpha}_2 x^i - \hat{\beta}_2 g_1^i + \hat{\beta}_2 g_1^{i+1} \right\|^2 \\ - \hat{\alpha}_2^{-1} \left\| \hat{\alpha}_2 x^{N-1} - \hat{\alpha}_2 x^N - \hat{\beta}_2 g_1^N + \hat{\beta}_2 g_2^N \right\|^2 \le 0, \end{split}$$

where

$$\hat{\alpha}_1 = \frac{\mu_2 B (1 - \frac{L_1}{L_2})}{2L_1 (1 - \frac{\mu_2}{L_2})}, \quad \hat{\alpha}_2 = \frac{\mu_2 L_1 B}{2(L_2 - \mu_2)}, \quad \hat{\beta}_1 = \frac{\mu_2 B (1 - \frac{L_1}{L_2})}{2L_1^2 (1 - \frac{\mu_2}{L_2})}, \quad \hat{\beta}_2 = \frac{\mu_2 B}{2(L_2 - \mu_2)}$$

It is readily seen that $\hat{\lambda}, \hat{\eta}_k$ ($k \in \{1, ..., N+1\}$), $\hat{\lambda}-B, \hat{\beta}_1, \hat{\alpha}_2 \ge 0$. The rest of proof is similar to that of the former case, and the proof is complete.

The theorem implies that Algorithm 8.1 is convergent when at least one of the Lipschitz constants is finite. In the following corollary, we simplify the inequality (8.16) for some special cases of L_1 , L_2 , μ_1 , and μ_2 .

Corollary 8.7. Suppose that $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$. Then, after N iterations of Algorithm 8.1, one has:

i) If $L_1 = \infty$, $L_2 < \infty$, then

$$\min_{0 \le k \le N} \left\| g_1^k - g_2^k \right\| \le \sqrt{\frac{2L_2^2(f(x^0) - f^*)}{N(L_2 + \mu_1)}}$$

ii) If $L_2 = \infty$, $L_1 < \infty$, then

$$\min_{0 \le k \le N} \left\| g_1^k - g_2^k \right\| \le \sqrt{\frac{2L_1^2(L_1 - \mu_2)(f(x^0) - f^*)}{(L_1^2 - \mu_2^2)N + L_1^2}}.$$
(8.17)

iii) If $L_1, L_2 < \infty$, and $\mu_1 = \mu_2 = 0$ then

$$\min_{0 \le k \le N} \left\| g_1^k - g_2^k \right\| \le \sqrt{\frac{2L_1 L_2 (f(x^0) - f^*)}{(L_1 + L_2)N + L_2}}.$$

One can compare the results in Corollary 8.7 to that of Le Thi et al. [LTPD21] as reviewed earlier in Theorem 8.2. First of all, Corollary 8.7 part *iii*) does not assume strict convexity of f_1 or f_2 , and in this sense it is more general than the result in Theorem 8.2. If we do assume $\mu_1 + \mu_2 > 0$, then, for example, if $L_1 < \infty$, Theorem 8.2 implies,

$$\min_{0 \le k \le N} \left\| g_1^k - g_2^k \right\| \le L_1 \sqrt{\frac{2(f(x^0) - f^*)}{(\mu_1 + \mu_2)N}},$$

which is weaker than our bound (8.17) since $\mu_1 \leq L_1$, although the $O(1/\sqrt{N})$ dependence on *N* is the same. We will do a further, more direct, comparison of Theorem 8.2 and Corollary 8.7 in Section 8.3.2, where we consider the convergence rate of the sequence $||x^{k+1} - x^k||$.

8.3.1 An example to prove tightness

In what follows, we give a class of functions for which the bound in Corollary 8.7, part *ii*), is attained, implying that the $O(1/\sqrt{N})$ convergence rate is tight. This result is new to the best of our knowledge.

Example 8.8. Let $L_1 \in (0, \infty)$. Suppose that N is selected such that $U := \sqrt{\frac{2}{L_1(N+1)}} < 1$. Let $f_1 : \mathbb{R} \to \mathbb{R}$ be given as follows,

$$f_{1}(x) = \begin{cases} \frac{L_{1}}{2} (x - i(1 - U))^{2} + \frac{L_{1}Ui(i - 1)(1 - U)}{2} & x \in [\alpha_{i}, \beta_{i + 1}) \\ L_{1}U\beta_{i}(x - \beta_{i}) + \frac{\beta_{i}L_{1}U^{2}}{2} + \frac{\beta_{i}(\beta_{i} - 1)L_{1}U}{2} & x \in [\beta_{i}, \alpha_{i}) \\ \frac{L_{1}}{2}x^{2} & x \in (-\infty, 0), \end{cases}$$

where, for $i \in \{1, ..., N+1\}$, $\alpha_i = i - U$, $\beta_i = i - 1$, and $\beta_{N+2} = \infty$. Note that $f_1 \in \mathcal{F}_{0,L_1}(\mathbb{R})$. Suppose that $f_2 : \mathbb{R} \to \mathbb{R}$ is given by

$$f_2(x) = \max_{1 \le i \le N+1} \left\{ L_1 U(i-1)(x-i) + \frac{i(i-1)L_1 U}{2} \right\}.$$

An easy computation shows that

$$\begin{cases} \partial f_2(i) = [L_1 U(i-1), L_1 Ui] \quad i \in \{1, \dots, N, \} \\ \partial f_2(N+1) = L_1 UN. \end{cases}$$

Note that $f_2 \in \mathcal{F}_{0,\infty}(\mathbb{R})$. One can check that, at $x^0 = N + 1$, one has $f_1(x^0) - f_2(x^0) = 1$, $\min_{x \in \mathbb{R}} f_1(x) - f_2(x) = 0$ and $\operatorname{argmin}_{x \in \mathbb{R}} f_1(x) - f_2(x) = [0, 1 - U]$. By taking x^0 as a starting point, Algorithm 8.1 can generate the following iterates:

$$x^{k} = N + 1 - k, \quad k \in \{0, \dots, N\}.$$

Here at iteration, $k \in \{0, ..., N\}$, we set $g_2^k = L_1 U(N-k)$. It follows that $|\nabla f_1(x_k) - g_2^k| = \sqrt{\frac{2L_1}{N+1}}$, $k \in \{0, ..., N\}$. Hence,

$$\min_{0 \le k \le N} \left\| g_1^k - g_2^k \right\| = \sqrt{\frac{2L_1}{N+1}},$$

which shows bound (8.17) in Corollary 8.7 is exact for this example.

8.3.2 Convergence rates for the iterates

In this section we investigate the implications of our results so far on convergence rates of the iterates $\{x^k\}$.

Proposition 8.9. Let $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$ and let $f(x^0) - f^* \leq \Delta$. If μ_1 or μ_2 is strictly positive, then after N iterations of Algorithm 8.1, one has:

$$\min_{0 \le k \le N-1} \left\| x^{k+1} - x^k \right\| \le \left(\frac{\mathcal{A}}{\mathcal{B}N + \mathcal{C}} \cdot \Delta \right)^{\frac{1}{2}},$$

where

$$\begin{split} \mathcal{A} &= 2 \left(\mu_2^{-1} \mu_1^{-1} - L_2^{-1} \mu_1^{-1} I_{\mathbb{R}_+} (\mu_2^{-1} - \mu_1^{-1}) - L_1^{-1} \mu_2^{-1} I_{\mathbb{R}_+} (\mu_1^{-1} - \mu_2^{-1}) \right), \\ \mathcal{B} &= \mu_2^{-1} + \mu_1^{-1} + L_2^{-1} \left(\frac{\mu_1}{\mu_2} - 3 \right) I_{\mathbb{R}_+} \left(\mu_2^{-1} - \mu_1^{-1} \right) + L_1^{-1} \left(\frac{\mu_2}{\mu_1} - 3 \right) I_{\mathbb{R}_+} \left(\mu_1^{-1} - \mu_2^{-1} \right), \\ and \\ \mathcal{C} &= \frac{\mu_2^{-1} \mu_1^{-1} - L_2^{-1} \mu_1^{-1} I_{\mathbb{R}_+} \left(\mu_2^{-1} - \mu_1^{-1} \right) - L_1^{-1} \mu_2^{-1} I_{\mathbb{R}_+} \left(\mu_1^{-1} - \mu_2^{-1} \right)}{\mu_2^{-1} - L_1^{-1}}. \end{split}$$

Proof. The proof is based on the computation of the worst case convergence rate of DCA for problem (8.9) by applying Theorem 8.6. By Toland duality, f^* is also

a lower bound of problem (8.9). By virtue of conjugate function properties, it follows that $f_2^*(g_2^0) - f_1^*(g_2^0) - f^* \leq \Delta$ and $f_2^* \in \mathcal{F}_{L_2^{-1}, \mu_2^{-1}}(\mathbb{R}^n)$ and $f_1^* \in \mathcal{F}_{L_1^{-1}, \mu_1^{-1}}(\mathbb{R}^n)$. In addition, $x^{k+1} \in \partial f_1^*(g_2^k)$ and $x^k \in \partial f_2^*(g_2^k)$ for $k \in \{0, \dots, N-1\}$. Hence, all assumptions of Theorem 8.6 hold, and subsequently the bound follows from Theorem 8.6.

Recall the known result from Theorem 8.2:

$$\min_{0 \le k \le N-1} \left\| x^{k+1} - x^k \right\| \le \left(\frac{2(f(x^0) - f^*)}{N(\mu_1 + \mu_2)} \right)^{\frac{1}{2}}.$$
(8.18)

By employing Theorem 8.9, we get

$$\min_{0 \le k \le N-1} \left\| x^{k+1} - x^k \right\| \le \left(\frac{2(f(x^0) - f^*)}{N(\mu_1 + \mu_2) + \mu_1} \right)^{\frac{1}{2}},$$

which is tighter than the bound (8.18). Moreover, the bound given in Proposition 8.9 provides more information concerning the worst-case convergence rate of the DCA when $L_1 < \infty$ or $L_2 < \infty$.

8.3.3 The gradient descent method

In this section, we study the relationship between the gradient descent method and DCA given by Algorithm 4.1 and 8.1, respectively. Using the convergence rate provided by Theorem 8.6 we derive the same convergence rate for the gradient descent given by Theorem 4.3 when the step length $t_k = t$ lies in $(0, \frac{1}{L}]$.

Consider the following optimization problem

$$\inf_{x\in\mathbb{R}^n}f(x)$$

where *f* is lower-bounded by f^* and is an *L*-smooth function on \mathbb{R}^n with $L < \infty$, that is $f \in \mathcal{F}_{-L,L}(\mathbb{R}^n)$. It is easily seen that the function *f* can be written as

$$f(x) := \frac{1}{2t} \|x\|^2 - \left(\frac{1}{2t} \|x\|^2 - f(x)\right),$$

where t is in $(0, \frac{1}{L}]$. We define $f_1 := \frac{1}{2t} ||x||^2$ and $f_2 := \frac{1}{2t} ||x||^2 - f(x)$. Both functions f_1 and f_2 are convex since t is in $(0, \frac{1}{L}]$. If we solve the subproblem of the DCA at iteration k given by

$$x^{k+1} = \operatorname{argmin}_{x} \frac{1}{2t} \|x\|^{2} - \left(\frac{1}{2t} \|x^{k}\|^{2} - f(x^{k})\right) - \left\langle\frac{1}{t}x^{k} - \nabla f(x^{k}), x - x^{k}\right\rangle,$$

$$x^{k+1} = x^k - t\nabla f(x^k),$$

which is exactly the steps of the gradient descent method; see Algorithm 4.1. Note that f_1 and f_2 are $\frac{1}{t}$ -smooth $\frac{1}{t}$ -strongly convex and $(\frac{1}{t} + L)$ -smooth $(\frac{1}{t} - L)$ -strongly convex functions, respectively. Using the Theorem 8.6, one can derive the following convergence rate for the gradient descent method.

Proposition 8.10. Suppose that f(x) is an L-smooth function. If N iterations of the gradient descent method runs with $t \in (0, \frac{1}{L}]$, then

$$\min_{0 \le k \le N} \left\| g^k \right\| \le \sqrt{\frac{4(f(x^0) - f^*)}{(4t - Lt^2)N + \frac{2}{L}}}.$$

As the given bound with Proposition 8.10 matches the bound given by Theorem 4.3 when $t_k = t \in (0, \frac{1}{L}]$, the given bound by Theorem 8.6 is tight for some cases.

8.3.4 Proximal gradient method

As a by-product of our analysis, we conclude the section by giving a convergence rate for the *proximal gradient method* (known as forward-backward algorithm) with constant step length for non-convex problems. Consider the non-convex optimization problem,

$$\inf \phi(x) := g(x) + h(x)$$
s.t. $x \in \mathbb{R}^n$,
$$(8.19)$$

where *g* is an *L*-smooth function on \mathbb{R}^n and $h \in \mathcal{F}_{0,\infty}(\mathbb{R}^n)$. Algorithm 8.2 below describes the proximal gradient method with constant step.

Remark that Algorithm 8.2 reduces to the gradient method when h = 0. To the best knowledge of the authors, the proximal gradient method for non-convex problems, first is developed in [FM81].

We use the first-order optimality condition for computing the convergence rate. Note that \bar{x} satisfies the first order optimality condition with accuracy $\epsilon \ge 0$ if

$$\min_{\xi \in \partial_L \phi(\bar{x})} \|\xi\| \le \epsilon.$$
(8.21)

Algorithm 8.2 Proximal gradient method with constant step length

Pick $x^0 \in \mathbb{R}^n$ and the step length t > 0. For k = 0, 1, ... perform the following steps:

1. Compute the *proximal operator* of *th* at $x^k - t\nabla g(x^k)$, that is,

$$x^{k+1} = \operatorname{argmin}_{x \in \mathbb{R}^n} th(x) + \frac{1}{2} \left\| x - x^k + t \nabla g(x^k) \right\|^2.$$
(8.20)

2. If the termination criterion is satisfied, then stop.

It is worth noting that (8.21) implies that

$$\phi'(\bar{x};d) \ge -\epsilon, \ \|d\| = 1,$$

where $\phi'(\bar{x}; d)$ denotes the directional derivative of ϕ at \bar{x} in the direction of d; see [Nes13].

Suppose that $t \leq \frac{1}{L}$ and the sequence $\{x^k\}$ is generated by Algorithm 8.2. We define $f_1 = h + \frac{1}{2t} ||.||^2$ and $f_2 = \frac{1}{2t} ||.||^2 - g$. Clearly, $f_1 \in \mathcal{F}_{\frac{1}{t},\infty}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\frac{1}{t}-L,\frac{1}{t}+L}(\mathbb{R}^n)$. Consider the functions f_1 and f_2 as just defined above. The subproblem (8.2) is

$$x^{k+1} = \operatorname{argmin}_{x \in \mathbb{R}^n} h(x) + \frac{1}{2t} \|x\|^2 - \frac{1}{2t} \|x^k\|^2 + g(x^k) - \langle \frac{1}{t} x^k - \nabla g(x^k), x - x^k \rangle.$$

The optimality conditions imply

$$x^{k+1} = x^k - t \left(\nabla h(x^{k+1} + \nabla g(x^k)) \right).$$

The last expression is the same as x^{k+1} generated by considering optimality conditions of subproblem (8.20). This shows that Algorithm 8.2 and Algorithm 8.1 share the same sequence $\{x^k\}$. Hence, one can obtain the following convergence rate by using Corollary 8.7.

Proposition 8.11. Let g be a L-smooth function on \mathbb{R}^n and $h \in \mathcal{F}_{0,\infty}(\mathbb{R}^n)$ and let $-\infty < \phi^* = \min_{x \in \mathbb{R}^n} \phi(x)$. If $t \leq \frac{1}{L}$, then after N iterations of Algorithm 8.2, we have

$$\min_{0\leq k\leq N}\left(\min_{\xi\in\partial_L\phi(x^k)}\|\xi\|\right)\leq \sqrt{\frac{2(L+\frac{1}{t})^2(\phi(x^1)-\phi^*)}{(L+\frac{2}{t})N}}.$$

If h = 0, in the same line by using Theorem 8.6, one can infer the following convergence rate for the gradient descent method with fixed step length $\frac{1}{L}$,

$$\min_{0\leq k\leq N} \|\nabla\phi(x^k)\| \leq \sqrt{\frac{4L(\phi(x^0)-\phi^*)}{4N+3}},$$

which is established by authors recently; see [AdKZ22, Theorem 2] and Theorem 4.3.

8.4 Performance estimation using a convergence criterion for critical points in the nonsmooth case

Theorem 8.6 addresses the case that f_1 or f_2 is *L*-smooth with $L < \infty$. In what follows, we investigate the case that f_1 and f_2 are proper convex functions and where both may be non-smooth. For this general case, we need to adopt a different termination criterion to obtain results, since the termination criterion $||g_1^k - g_2^k|| \le \epsilon$ may be of no use in this case. For example, suppose that a DC function $f : \mathbb{R} \to \mathbb{R} \cup \{\infty\}$ is given by

$$f(x) = \begin{cases} f_1(x) - f_2(x) & x \ge 0 \\ \infty & x < 0, \end{cases}$$

where

$$f_1(x) = \max_{n \in \mathbb{N} \cup \{0\}} \{-n(x-2^{-n}) + 2 - 2^{1-n} - n2^{-n}\},$$

$$f_2(x) = \max_{n \in \mathbb{N} \cup \{0\}} \{-(n+1)(x-2^{-n}) + 2 - 3(2^{-n}) - n2^{-n}\}.$$

With $x^0 = 1$ and the given DC decomposition, Algorithm 8.1 may generate

$$x^{k} = 2^{-k}, \quad g_{1}^{k} = -(k-1), \quad g_{2}^{k} = -k, \quad k \in \{1, 2, ...\}.$$

As $|g_1^k - g_2^k| = 1$, Algorithm 8.1 never stops by employing the given termination criterion while it is convergent to global minimum $\bar{x} = 0$. We therefore will use the termination criterion of the following value being sufficiently small:

$$T(x^{k+1}) := f_1(x^k) - f_2(x^k) - \min_{x \in \mathbb{R}^n} (f_1(x) - f_2(x^k) - \langle g_2^k, x - x^k \rangle)$$

= $f_1(x^k) - f_1(x^{k+1}) - \langle g_2^k, x^k - x^{k+1} \rangle.$ (8.22)

Note that $T(x^{k+1}) \ge 0$. It follows that if $T(x^{k+1}) = 0$ then $f(x^k) = f(x^{k+1})$, and $x^k \in \operatorname{argmin}_{x \in \mathbb{R}^n} f_1(x) - f_2(x^k) - \langle g_2^k, x - x^k \rangle$. Indeed, by the optimality conditions for convex problems, we have $\partial f_1(x^k) \cap \partial f_2(x^k) \neq \emptyset$. Consequently, $T(x^{k+1}) = 0$ implies that x^k is a critical point of problem (8.7). The aforementioned stopping criterion has also been employed for the analysis of the Frank-Wolfe method for nonconvex problems; see equation (2.6) in [Gha19].

In what follows, we investigate Algorithm 8.1 with the termination criterion $T(x^{k+1}) < \epsilon$ for the given accuracy $\epsilon > 0$. The performance estimation problem with termination criterion (8.22) may be written as follows,

$$\begin{aligned} \max \ell \\ \text{s.t. } f_{1}(x^{k}) - f_{1}(x^{k+1}) - \left\langle g_{1}^{k+1}, x^{k} - x^{k+1} \right\rangle \geq \ell \quad i \in \{0, \dots, N-1\} \\ \frac{1}{2(1-\frac{\mu_{1}}{L_{1}})} \left(\frac{1}{L_{1}} \left\| g_{1}^{i} - g_{1}^{j} \right\|^{2} + \mu_{1} \left\| x^{i} - x^{j} \right\|^{2} - \frac{2\mu_{1}}{L_{1}} \left\langle g_{1}^{j} - g_{1}^{i}, x^{j} - x^{i} \right\rangle \right) \\ \leq f_{1}^{i} - f_{1}^{j} - \left\langle g_{1}^{j}, x^{i} - x^{j} \right\rangle \quad i, j \in \{0, \dots, N\} \\ \frac{1}{2(1-\frac{\mu_{2}}{L_{2}})} \left(\frac{1}{L_{2}} \left\| g_{1}^{i+1} - g_{1}^{j+1} \right\|^{2} + \mu_{2} \left\| x^{i} - x^{j} \right\|^{2} - \frac{2\mu_{2}}{L_{2}} \left\langle g_{1}^{j+1} - g_{1}^{i+1}, x^{j} - x^{i} \right\rangle \right) \\ \leq f_{2}^{i} - f_{2}^{j} - \left\langle g_{1}^{j+1}, x^{i} - x^{j} \right\rangle \quad i, j \in \{0, \dots, N-1\} \\ \frac{1}{2(1-\frac{\mu_{2}}{L_{2}})} \left(\frac{1}{L_{2}} \left\| g_{2}^{N} - g_{1}^{j+1} \right\|^{2} + \mu_{2} \left\| x^{N} - x^{j} \right\|^{2} - \frac{2\mu_{2}}{L_{2}} \left\langle g_{1}^{j+1} - g_{2}^{N}, x^{j} - x^{N} \right\rangle \right) \\ \leq f_{2}^{N} - f_{2}^{j} - \left\langle g_{1}^{j+1}, x^{N} - x^{j} \right\rangle \quad j \in \{0, \dots, N-1\} \\ \frac{1}{2(1-\frac{\mu_{2}}{L_{2}})} \left(\frac{1}{L_{2}} \left\| g_{1}^{i+1} - g_{2}^{N} \right\|^{2} + \mu_{2} \left\| x^{i} - x^{N} \right\|^{2} - \frac{2\mu_{2}}{L_{2}} \left\langle g_{2}^{N} - g_{1}^{j+1}, x^{N} - x^{i} \right\rangle \right) \\ \leq f_{2}^{N} - f_{2}^{j} - \left\langle g_{1}^{j,1}, x^{N} - x^{j} \right\rangle \quad i \in \{0, \dots, N-1\} \\ f_{1}^{k} - f_{2}^{k} \geq f^{k} \quad k \in \{0, \dots, N\} \\ f_{1}^{0} - f_{2}^{0} - f^{k} \leq \Delta. \end{aligned}$$

Note that we do not employ Lemma 8.4 in this formulation because we consider a general DC problem. Using the performance estimation procedure as described before the proof of Theorem 8.6 once more, we obtain the following result.

Theorem 8.12. Let $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$. Then, after N iterations of Algorithm 8.1, one has

$$\min_{0 \le k \le N-1} f_1(x^k) - f_1(x^{k+1}) - \langle g_2^k, x^k - x^{k+1} \rangle \le$$

$$\min\left\{\frac{L_1}{N(L_1 + \mu_2)}, \frac{L_2}{N(L_2 + \mu_1) - \mu_1}\right\} (f(x^0) - f^*).$$
(8.24)

Proof. We show separately that $\frac{L_1(f(x^0)-f^*)}{N(L_1+\mu_2)}$ and $\frac{L_2(f(x^0)-f^*)}{N(L_2+\mu_1)-\mu_1}$ are upper bounds for problem (8.23). The proof is analogous to that of Theorem 8.6. First, consider the bound $\frac{L_1(f(x^0)-f^*)}{N(L_1+\mu_2)}$. Since the given bound does not depend on μ_1 and L_2 , we may assume without loss of generality that $L_2 = \infty$ and $\mu_1 = 0$. Suppose that $B_1 = \frac{L_1}{N(L_1+\mu_2)}$. With some algebra, one can show that

$$\begin{split} \ell &-B_{1}\Delta + \frac{1}{N}\sum_{k=1}^{N} \left(f_{1}^{k-1} - f_{1}^{k} - \langle g_{1}^{k}, x^{k-1} - x^{k} \rangle - \ell\right) + B_{1}\left(f_{1}^{N} - f_{2}^{N} - f^{\star}\right) + \\ &B_{1}\left(f^{\star} - f_{1}^{0} + f_{2}^{0} + \Delta\right) + \left(\frac{1}{N} - B_{1}\right)\sum_{k=1}^{N} \left(f_{1}^{k} - f_{1}^{k-1} - \langle g_{1}^{k-1}, x^{k} - x^{k-1} \rangle - \frac{1}{2L_{1}} \left\|g_{1}^{k} - g_{1}^{k-1}\right\|^{2}\right) \\ &+ B_{1}\sum_{k=1}^{N} \left(f_{2}^{k} - f_{2}^{k-1} - \langle g_{1}^{k}, x^{k} - x^{k-1} \rangle - \frac{\mu_{2}}{2} \left\|x^{k} - x^{k-1}\right\|^{2}\right) \\ &= -\frac{B_{1}\mu_{2}}{2}\sum_{k=1}^{N} \left\|x^{k-1} - x^{k} - \frac{1}{L_{1}}\left(g_{1}^{k-1} - g_{1}^{k}\right)\right\|^{2} \leq 0. \end{split}$$

The rest of proof is similar to that of Theorem 8.6. Now, we consider the bound $\frac{L_2(f(x^0)-f^*)}{N(L_2+\mu_1)-\mu_1}$. Without loss generality, we may assume that $L_1 = \infty$ and $\mu_2 = 0$. By doing some calculus, one can show that

$$\begin{split} \ell &-B_{2}\Delta + B_{2}\left(f_{1}^{0} - f_{1}^{1} - \left\langle g_{1}^{1}, x^{0} - x^{1} \right\rangle - \ell\right) + B_{2}\left(f_{1}^{N} - f_{2}^{N} - f^{*}\right) \\ &+ B_{2}\left(f^{*} - f_{1}^{0} + f_{2}^{0} + \Delta\right) + \frac{1 - B_{2}}{N - 1} \sum_{k=2}^{N} \left(f_{1}^{k-1} - f_{1}^{k} - \left\langle g_{1}^{k}, x^{k-1} - x^{k} \right\rangle - \ell\right) \\ &+ \alpha \sum_{k=2}^{N} \left(f_{1}^{k} - f_{1}^{k-1} - \left\langle g_{1}^{k-1}, x^{k} - x^{k-1} \right\rangle - \frac{\mu_{1}}{2} \left\| x^{k} - x^{k-1} \right\|^{2} \right) \\ &+ B_{2} \sum_{k=1}^{N} \left(f_{2}^{k} - f_{2}^{k-1} - \left\langle g_{1}^{k}, x^{k} - x^{k-1} \right\rangle - \frac{1}{2L_{2}} \left\| g_{1}^{k+1} - g_{1}^{k} \right\|^{2} \right) \\ &+ B_{2} \left(f_{2}^{N} - f_{2}^{N-1} - \left\langle g_{1}^{N}, x^{N} - x^{N-1} \right\rangle - \frac{1}{2L_{2}} \left\| g_{2}^{N} - g_{1}^{N} \right\|^{2} \right) \\ &= - \frac{B_{2}}{2L_{2}} \left\| g_{2}^{N} - g_{1}^{N} \right\|^{2} - \frac{B_{2}}{2L_{2}} \sum_{k=2}^{N} \left\| g_{1}^{k-1} - g_{1}^{k} - \frac{\alpha L_{2}}{B_{2}} (x^{k-1} - x^{k}) \right\|^{2} \le 0, \end{split}$$

where $B_2 = \frac{L_2}{N(L_2 + \mu_1) - \mu_1}$ and $\alpha = \frac{1 - B_2}{N - 1} - B_2$. Since we assume $L_2 > \mu_1$, we have B_2 , $\alpha \ge 0$. The rest of the proof runs as before.

The important point is that the last result provides a rate of convergence even if neither L_1 nor L_2 is finite, and we therefore state it as a corollary.

Corollary 8.13. Let $f_1 \in \mathcal{F}_{\mu_1,\infty}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,\infty}(\mathbb{R}^n)$, i.e. consider any DC decomposition in problem (8.1). Then, after N iterations of Algorithm 8.1, one has

$$\min_{0 \le k \le N-1} f_1(x^k) - f_1(x^{k+1}) - \langle g_2^k, x^k - x^{k+1} \rangle \le \frac{1}{N} \left(f(x^0) - f^* \right)$$

This result is new to the best of our knowledge.

8.5 Performance analysis of DCA with regularization

Pang et al. [PRA17] developed a version of Algorithm 8.1 with regularization. The DCA with a *regularization term* is described in Algorithm 8.3. As $r \ge -\min(\mu_1, \mu_2)$,

Algorithm 8.3 DCA with regularization Pick $x^0 \in \mathbb{R}^n$, $N \in \mathbb{N}$, $r \ge -\min(\mu_1, \mu_2)$ and $\epsilon > 0$. For k = 0, 1, ..., N - 1 perform the following steps:

- 1. Choose $g_1^k \in \partial f_1(x^k)$ and $g_2^k \in \partial f_2(x^k)$. If $\|g_1^k g_2^k\| \le \epsilon$, then stop.
- 2. Choose

$$x^{k+1} \in \operatorname{argmin}_{x \in \mathbb{R}^n} f_1(x) - f_2(x^k) - \langle g_2^k, x - x^k \rangle + \frac{r}{2} \|x - x^k\|^2.$$
 (8.25)

the regularization parameter may take negative values. Our aim for the investigation of regularization parameter on this large interval is to have a comprehensive analysis. One can check that subproblem (8.25) is convex; see Lemma 8.3. It is worth noting that they [PRA17] consider only the fixed regularization parameter r = 1. Subproblem (8.25) may be reformulated as follows:

$$x^{k+1} \in \operatorname{argmin}_{x \in \mathbb{R}^n} f_1(x) + \frac{r}{2} ||x||^2 - f_2(x^k) - \frac{r}{2} ||x_k||^2 - \left\langle g_2^k + rx^k, x - x^k \right\rangle.$$

Corollary 8.14. Let $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$, $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$ and $f = f_1 - f_2$. If $r \ge -\min(\mu_1,\mu_2)$, then f has a DC decomposition of form

$$f=\bar{f}_1-\bar{f}_2,$$

where $\bar{f}_1 \in \mathcal{F}_{\mu_1+r,L_1+r}(\mathbb{R}^n)$ and $\bar{f}_2 \in \mathcal{F}_{\mu_2+r,L_2+r}(\mathbb{R}^n)$ given by $\bar{f}_1(x) = f_1(x) + \frac{r}{2} ||x||^2$ and $\bar{f}_2(x) = f_2(x) + \frac{r}{2} ||x||^2$. *Proof.* The proof is immediate from Lemma 8.3.

By Corollary 8.14, it is readily seen that Algorithm 8.3 is the same as Algorithm 8.1, but with a different decomposition of f. It is clear that $\partial f_1(x) = \partial \bar{f}_1(x) + rx$ and $\partial f_2(x) = \partial \bar{f}_2(x) + rx$. Hence, by using Theorem 8.6, the following corollary gives a convergence rate for Algorithm 8.3.

Corollary 8.15. Let $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$ and let L_1 or L_2 be finite. Suppose that $f(x^0) - f^* \leq \Delta$ and $r \geq -\min(\mu_1, \mu_2)$. Then after N iterations of Algorithm 8.3, one has:

$$\min_{0 \le k \le N} \left\| g_1^k - g_2^k \right\| \le \left(\frac{\mathcal{A}}{\mathcal{B}N + \mathcal{C}} \cdot \Delta \right)^{\frac{1}{2}},$$

where

$$\mathcal{A} = 2\left((L_1+r)(L_2+r) - (\mu_1+r)(L_2+r)I_{\mathbb{R}_+}(L_1-L_2) - (\mu_2+r)(L_1+r)I_{\mathbb{R}_+}(L_2-L_1)\right),$$

 $\mathcal{B} = (L_1 + r) + (L_2 + r) + (\mu_1 + r) \left(\frac{L_1 + r}{L_2 + r} - 3\right) I_{\mathbb{R}_+} (L_1 - L_2) + (\mu_2 + r) \left(\frac{L_2 + r}{L_1 + r} - 3\right) I_{\mathbb{R}_+} (L_2 - L_1),$

and

$$C = \frac{(L_1+r)(L_2+r) - I_{\mathbb{R}_+}(L_1-L_2)(\mu_1+r)(L_2+r) - I_{\mathbb{R}_+}(L_2-L_1)(\mu_2+r)(L_1+r)}{L_1-\mu_2}.$$

One interesting question concerning Algorithm 8.3 is the choice of suitable regularization parameter, r. By minimizing the right hand side of bound given in Corollary 8.15 with respect to r, the next proposition provides the optimal regularization parameter in terms of the worst-case convergence rate.

Proposition 8.16. Let the assumptions of Corollary 8.15 hold. Then, the optimal regularization parameter is $\bar{r} = -\min(\mu_1, \mu_2)$.

Proof. By Corollary 8.15, the following optimization problem gives the optimal regularization parameter,

$$\min_{r \ge -\min(\mu_1, \mu_2)} h(r) := \frac{(L_1 + r)(L_2 + r) - (\mu_1 + r)(L_2 + r)}{\left((L_1 + r) + (L_2 + r) + (\mu_1 + r)(\frac{L_1 + r}{L_2 + r} - 3)\right)N + L_1 - \mu_1}.$$

It is readily seen that *h* is positive on $[-\min(\mu_1, \mu_2), \infty)$. In addition, we have

$$\frac{dh(r)}{dr} = \frac{(L_1 - \mu_1)(L_2 + r)\left(2L_2^2N - L_2(\mu_1 + 4\mu_1N) - \mu_1(1 + 2N)r + L_1(L_2 + 2\mu_1N + r + 2Nr)\right)}{\left(L_2^2N - L_2(\mu_1 + 3\mu_1N) - \mu_1(1 + 2N)r + L_1(L_2 + L_2N + \mu_1N + r + 2Nr)\right)^2} \\ = \frac{(L_1 - \mu_1)(L_2 + r)(r(2N + 1)(L_1 - \mu_1) + 2N(\mu_1(L_1 - L_2) + L_2(L_2 - \mu_1)) + L_2(L_1 - \mu_1))}{\left(L_2^2N - L_2(\mu_1 + 3\mu_1N) - \mu_1(1 + 2N)r + L_1(L_2 + L_2N + \mu_1N + r + 2Nr)\right)^2}.$$

For $r \ge -\min(\mu_1, \mu_2) \ge -\mu_1$, we have

$$r(L_1 - \mu_1) + L_2^2 + L_1\mu_1 - 2L_2\mu_1 \ge -\mu_1(L_1 - \mu_1) + L_2^2 + L_1\mu_1 - 2L_2\mu_1 = (L_2 - \mu_1)^2 \ge 0,$$

which implies positivity of $\frac{dh(r)}{dr}$ on the given interval. Hence, the aforementioned problem attains its minimum at $\bar{r} = -\min(\mu_1, \mu_2)$.

The result of Proposition 8.16 is unexpected. Due to Lemma 8.3, the Lipschitz modulus and strongly convex constant alter equally by the change of r. The underlying reason for this result may be that Lipschitz modulus plays a more important role than the strongly convex constant in the worst-case convergence rate.

Remark 8.17. For undominated D.C. decompositions, one has $\min(\mu_1, \mu_2) = 0$. (Recall that a D.C. decomposition $f = f_1 - f_2$ is undominated if there is no other D.C. decomposition, say $f = \hat{f}_1 - \hat{f}_2$, such that $f_1 - \hat{f}_1$ is convex but not affine; see e.g. [AH18] for more details on undominated D.C. decompositions, including their construction for polynomials.)

Thus, if we only consider given instances with undominated decompositions $f = f_1 - f_2$, the optimal regularization parameter (with respect to the worst-case convergence rate) is zero.

In the last part of the section, we investigate the nonsmooth case with regularization. First, we need to adopt an appropriate termination criterion. Suppose that $r \ge -\min(\mu_1, \mu_2)$. Similar to the termination criterion (8.22), we stop the algorithm if the following value is sufficiently small:

$$T_{r}(x^{k+1}) = f_{1}(x^{k}) - f_{2}(x^{k}) - \min_{x \in \mathbb{R}^{n}} \left(f_{1}(x) - f_{2}(x^{k}) - \left\langle g_{2}^{k}, x - x^{k} \right\rangle + \frac{r}{2} \left\| x - x^{k} \right\|^{2} \right)$$

= $f_{1}(x^{k}) - f_{1}(x^{k+1}) - \left\langle g_{2}^{k}, x^{k} - x^{k+1} \right\rangle - \frac{r}{2} \left\| x^{k} - x^{k+1} \right\|^{2}.$ (8.26)

One has $T_r(x^{k+1}) \ge 0$ and one can show that $\partial f_1(x^k) \cap \partial f_2(x^k) \neq \emptyset$ providing $T(x^{k+1}) = 0$.

By defining convex functions $\bar{f}_1(x) = f_1(x) + \frac{r}{2} ||x||^2$ and $\bar{f}_2(x) = f_2(x) + \frac{r}{2} ||x||^2$, it is readily seen that

$$T_r(x^{k+1}) = \bar{f}_1(x^k) - \bar{f}_1(x^{k+1}) - \left\langle g_2^k + rx^k, x^k - x^{k+1} \right\rangle.$$

By analysis similar to that of Corollary 8.15, the next remark provides a convergence rate.

Algorithm 8.4

Pick $x^0 \in \mathbb{R}^n$, $N \in \mathbb{N}$, and $\epsilon > 0$. For k = 0, 1, ..., N - 1 perform the following steps:

1. Choose
$$g_2^k \in \partial f_2(x^k)$$
 and

$$x^{k+1} \in \operatorname{argmin}_{x \in \mathbb{R}^n} f_1(x) - f_2(x^k) - \langle g_2^k, x - x^k \rangle + \frac{r}{2} \|x - x^k\|^2.$$
 (8.27)

2. If $f_1(x^k) - f_1(x^{k+1}) - \langle g_2^k, x^k - x^{k+1} \rangle - \frac{r}{2} \| x^k - x^{k+1} \|^2 \le \epsilon$, then stop.

Remark 8.18. Let $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$. Then, after N iterations of Algorithm 8.4, one has

$$\min_{0 \le k \le N-1} f_1(x^k) - f_1(x^{k+1}) - \left\langle g_2^k, x^k - x^{k+1} \right\rangle - \frac{r}{2} \left\| x^k - x^{k+1} \right\|^2 \le (8.28)$$

$$\min\left\{ \frac{L_1 + r}{N(L_1 + \mu_2 + 2r)}, \frac{L_2 + r}{N(L_2 + \mu_1 + 2r) - \mu_1 - r} \right\} \left(f(x^0) - f^* \right).$$

Note that this result holds even if L_1 and L_2 are not finite. In this case one has the bound

$$\min_{0 \le k \le N-1} f_1(x^k) - f_1(x^{k+1}) - \left\langle g_2^k, x^k - x^{k+1} \right\rangle - \frac{r}{2} \left\| x^k - x^{k+1} \right\|^2 \le \frac{1}{N} \left(f(x^0) - f^* \right).$$

8.6 Linear convergence of the DCA under the Polyak-Łojasiewicz inequality

In the section, we provide some sufficient conditions under which the DCA is linearly convergent. Similar to the former sections, we employ the performance estimation for obtaining convergence rate.

In recent years, the linear convergence of some optimization methods for nonconvex problems have been investigated under the Polyak-Łojasiewicz (PŁ) inequality; see [AdKZ23a, BNPS17, KNS16] alongside with Chapter 5 and the reference therein. We say that f satisfies PŁ inequality on X if there exists $\eta > 0$ such that

$$f(x) - f^* \le \frac{1}{2\eta} \|\xi\|^2, \quad \forall x \in X, \forall \xi \in \operatorname{co}(\partial_L f(x)).$$
(8.29)

Note that when f is differentiable inequality (8.29) is a special case of (8.3) with $\theta = \frac{1}{2}$ and different ground set. If f_2 is *strictly differentiable*, we have have

 $co(\partial_L f) = \partial f_1 - \partial f_2$; see Example 10.10 in [RW09]. Hence, the performance estimation problem with the PL inequality may be formulated as follows:

$$\begin{aligned} \max \frac{(f_1^1 - f_2^1) - f^*}{(f_1^0 - f_2^0) - f^*} \\ \text{s.t.} & \frac{1}{2(1 - \frac{\mu_1}{L_1})} \left(\frac{1}{L_1} \left\| g_1^i - g_1^j \right\|^2 + \mu_1 \left\| x^i - x^j \right\|^2 - \frac{2\mu_1}{L_1} \left\langle g_1^j - g_1^i, x^j - x^i \right\rangle \right) \\ & \leq f_1^i - f_1^j - \left\langle g_1^j, x^i - x^j \right\rangle \ i, j \in \{0, 1\} \\ & \frac{1}{2(1 - \frac{\mu_2}{L_2})} \left(\frac{1}{L_2} \left\| g_2^j - g_2^j \right\|^2 + \mu_2 \left\| x^i - x^j \right\|^2 - \frac{2\mu_2}{L_2} \left\langle g_2^j - g_2^i, x^j - x^i \right\rangle \right) \\ & \leq f_2^i - f_2^j - \left\langle g_2^j, x^i - x^j \right\rangle \ i, j \in \{0, 1\} \\ & f_1^k - f_2^k \ge f^* \ k \in \{0, 1\} \\ & g_2^0 = g_1^1 \\ & \left(f_1^k - f_2^k \right) - f^* \le \frac{1}{2\eta} \| g_1^k - g_2^k \|^2, \ k \in \{0, 1\}. \end{aligned}$$

The following lemma shows that $\eta \leq L_1$.

Lemma 8.19. Let $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$, where L_1 is finite. If f satisfies the PŁ inequality on $X = \{x : f(x) \leq f(x^0)\}$ with modulus $\eta > 0$, then $\eta \leq L_1$.

Proof. Without loss of generality, assume that $\mu_1 = \mu_2 = 0$. Let $g_1 \in \partial f_1$ and $g_2 \in \partial f_2$. According to the PL inequality,

$$f(x) - f^* \le \frac{1}{2\eta} ||g_1 - g_2||^2, \quad \forall x \in X,$$

and using Lemma 8.4, we obtain

$$\frac{1}{2L_1} \|g_1 - g_2\|^2 \le f(x) - f^*, \quad \forall x \in X,$$

which implies $\eta \leq L_1$.

By doing constraint aggregation in problem (8.30) as before (i.e. demonstrating a dual feasible solution and using weak duality), we obtain the following linear convergence rate for the DCA under the PŁ inequality.

Theorem 8.20. Let $f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n)$ and $f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$. If L_1 is finite and if f satisfies PL inequality on $X = \{x : f(x) \leq f(x^0)\}$ with modulus $0 < \eta \leq L_1$, then for x^1 from Algorithm 8.1, we have

$$\frac{f(x^{1}) - f^{\star}}{f(x^{0}) - f^{\star}} \le \left(\frac{1 - \frac{\eta}{L_{1}}}{1 + \frac{\eta}{L_{2}}}\right).$$
(8.31)

Proof. Since the given bound is independent of μ_1 and μ_2 , without loss of generality, we assume that $\mu_1 = \mu_2 = 0$. In addition, we assume that $f^* = 0$. Direct calculation shows that

$$\begin{split} & \left(f_{1}^{1}-f_{2}^{1}\right)-f^{\star}-\left(\frac{1-\frac{\eta}{L_{1}}}{1+\frac{\eta}{L_{2}}}\right)\left(\left(f_{1}^{0}-f_{2}^{0}\right)-f^{\star}\right)+\left(\frac{1}{1+\frac{\eta}{L_{2}}}\right)\times\\ & \left(f_{1}^{0}-f_{1}^{1}-\left\langle g_{1}^{1},x^{0}-x^{1}\right\rangle-\frac{1}{2L_{1}}\left\|g_{1}^{0}-g_{1}^{1}\right\|^{2}\right)\\ & +\left(\frac{1}{1+\frac{\eta}{L_{2}}}\right)\left(f_{2}^{1}-f_{2}^{0}-\left\langle g_{1}^{1},x^{1}-x^{0}\right\rangle-\frac{1}{2L_{2}}\left\|g_{1}^{1}-g_{2}^{1}\right\|^{2}\right)+\left(\frac{\frac{\eta}{L_{1}}}{1+\frac{\eta}{L_{2}}}\right)\times\\ & \left(\frac{1}{2\eta}\left\|g_{1}^{0}-g_{1}^{1}\right\|^{2}-f_{1}^{0}+f_{2}^{0}\right)+\left(\frac{\frac{\eta}{L_{2}}}{1+\frac{\eta}{L_{2}}}\right)\left(\frac{1}{2\eta}\left\|g_{1}^{1}-g_{2}^{1}\right\|^{2}-f_{1}^{1}+f_{2}^{1}\right)=0. \end{split}$$

As all the multipliers in the last expression are non-negative, for any feasible solution of problem (8.15), we have

$$f(x^{1}) - f^{\star} - \left(\frac{1 - \frac{\eta}{L_{1}}}{1 + \frac{\eta}{L_{2}}}\right) \left(f(x^{0}) - f^{\star}\right) \le 0,$$

completing the proof.

Note that Theorem 8.1 by Le Thi et al. [LTHPD18] does not imply Theorem 8.20 if inequality (8.3) holds on $\{x : f(x) \le f(x^0)\}$ with $\theta = \frac{1}{2}$, since we assume neither strong convexity of f_1 or f_2 , nor boundedness of the sequence of iterates. Moreover, we give explicit expressions for the constants that determine the linear convergence rate of the sequence of objective values.

8.7 Conclusion

We have shown that the performance estimation framework of Drori and Teboulle [DT14] yields new insights into the convergence behavior of the Difference-ofconvex algorithm (DCA). As future work, one may also consider the convergence

of the DCA on more restricted classes of DC problems, e.g. where f_1 and f_2 are convex polynomials, as studied in [AH18]. For constrained problems, even the case where f_1 and f_2 are quadratic polynomials is of interest, e.g. in the study of (extended) trust region problems.

Nothing contributes so much to tranquilize the mind as a steady purpose - a point on which the soul may fix its intellectual eyes.

Mary W Shelley



The exact worst-case convergence rate of the alternating direction method of multipliers

Preamble

Recently, semidefinite programming performance estimation has been employed as a strong tool for the worst-case performance analysis of first-order methods. In this chapter, we derive new non-ergodic convergence rates for the alternating direction method of multipliers (ADMM) by using performance estimation. We give some examples which show the exactness of the given bounds. We also study the linear and R-linear convergence of ADMM under some assumptions. We establish that ADMM enjoys a global linear convergence rate if and only if the dual objective satisfies the Polyak-Łojasiewicz (PŁ) inequality in the presence of strong convexity. In addition, we give an explicit formula for the linear convergence rate factor. Moreover, we study the *R*-linear convergence of ADMM under two new scenarios. This chapter is based on the paper [ZAdK23].

9.1 Introduction

We consider the optimization problem

$$\min_{\substack{(x,z)\in\mathbb{R}^n\times\mathbb{R}^m}} f(x) + g(z),$$
s.t. $Ax + Bz = b,$
(9.1)

where $f : \mathbb{R}^n \to \mathbb{R} \cup \{\infty\}$ and $g : \mathbb{R}^m \to \mathbb{R} \cup \{\infty\}$ are closed proper convex functions, $0 \neq A \in \mathbb{R}^{r \times n}$, $0 \neq B \in \mathbb{R}^{r \times m}$ and $b \in \mathbb{R}^r$. Moreover, we assume that (x^*, z^*) is an optimal solution of problem (9.1) and λ^* is its corresponding Lagrange multipliers. Moreover, we denote the value of f and g at x^* and z^* with f^* and g^* , respectively.

Problem (9.1) appears naturally (or after variable splitting) in many applications in statistics, machine learning and image processing to name but a few [BPC⁺11, ROF92, HTW15, LLF22]. The most common method for solving problem (9.1) is the *alternating direction method of multipliers* (ADMM). ADMM is a dual based approach that exploits separable structure and it may be described as follows.

Algorithm 9.1 ADMM

Set *N* and t > 0 (step length), pick λ^0, z^0 . For k = 1, 2, ..., N perform the following step: 1. $x^k \in \operatorname{argmin} f(x) + \langle \lambda^{k-1}, Ax \rangle + \frac{t}{2} ||Ax + Bz^{k-1} - b||^2$ 2. $z^k \in \operatorname{argmin} g(z) + \langle \lambda^{k-1}, Bz \rangle + \frac{t}{2} ||Ax^k + Bz - b||^2$ 3. $\lambda^k = \lambda^{k-1} + t(Ax^k + Bz^k - b)$.

ADMM was first proposed in [GM76, GM75] for solving nonlinear variational problems. We refer the interested reader to [GOY17] for a historical review of ADMM. The popularity of ADMM is due to its capability to be implemented parallelly and hence can handle large-scale problems [BPC⁺11, Han22, MKL15, SBG⁺20]. For example, it is used for solving inverse problems governed by partial differential equation forward models [IV21], and distributed energy resource coordinations [LSW⁺22], to mention but a few.

The convergence of ADMM has been investigated extensively in the literature and there exist many convergence results. However, different performance measures have been used for the computation of convergence rate; see [FRV18, GOSB14, ST22, GMS13, HY12, MS13, LLF22, LL19]. In this chapter, we consider the dual objective value as a performance measure.

Throughout the chapter, we assume that each subproblem in steps 1 and 2 of Algorithm 9.1 attains its minimum. The Lagrangian function of problem (9.1) may be written as

$$L(x,z,\lambda) = f(x) + g(z) + \langle \lambda, Ax + Bz - b \rangle, \qquad (9.2)$$

and the dual objective of problem (9.1) is also defined as

$$D(\lambda) = \min_{(x,z) \in \mathbb{R}^n \times \mathbb{R}^m} f(x) + g(z) + \langle \lambda, Ax + Bz - b \rangle.$$

We assume throughout the chapter that strong duality holds for problem (9.1), that is

$$\max_{\lambda \in \mathbb{R}^r} D(\lambda) = \min_{Ax+Bz=b} f(x) + g(z).$$

Note that we have strong duality when both functions f and g are real-valued. For extended convex functions, strong duality holds under some mild conditions; see e.g. [Bec17, Chapter 15].

Some common performance measures for the analysis of ADMM are as follows,

- Objective value: $|f(x^N) + g(z^N) f^* g^*|;$
- Primal and dual feasibility: $||Ax^N + Bz^N b||$ and $||A^TB(z^N z^{N-1})||$;
- Dual objective value: $D(\lambda^*) D(\lambda^N)$;
- Distance between (x^N, z^N, λ^N) and a saddle point of problem (9.2).

Note that the mathematical expressions are written in a non-ergodic sense for convenience. Each measure is useful in monitoring the progress and convergence of ADMM. The objective value is the most commonly used performance measure for the analysis of algorithms in convex optimization [Ber15, Bec17, Nes03]. As mentioned earlier, ADMM is a dual based method and it may be interpreted as a proximal method applied to the dual problem; see [Ber15, LLF22] for further discussions and insights. Thus, a natural performance measure for ADMM would be dual objective value. In this study, we investigate the convergence rate of ADMM in terms of dual objective value and feasibility. It worth noting that most performance measures may be analyzed through the framework developed in Section 9.2.

Regarding the dual objective value, the following convergence rate is known in the literature. This theorem holds for strongly convex functions f and g; recall that f is called strongly convex with modulus $\mu \ge 0$ if the function $f - \frac{\mu}{2} \|\cdot\|^2$ is convex.

Theorem 9.1. [GOSB14, Theorem 1] Let f and g be strongly convex with moduli $\mu_1 > 0$ and $\mu_2 > 0$, respectively. If $t \leq \sqrt[3]{\frac{\mu_1 \mu_2^2}{\lambda_{\max}(A^T A)\lambda_{\max}^2(B^T B)}}$, then $D(\lambda^*) - D(\lambda^N) \leq \frac{\|\lambda^1 - \lambda^*\|^2}{2t(N-1)}.$ (9.3)

In this study we establish that Algorithm 9.1 has the convergence rate of $\mathcal{O}(\frac{1}{N})$ in terms of dual objective value without assuming the strong convexity of g. Under this setting, we also prove that Algorithm 9.1 has the convergence rate of $\mathcal{O}(\frac{1}{N})$ in terms of primal and dual residuals. Moreover, we show that the given bounds are exact. Furthermore, we study the linear and R-linear convergence.

Outline of the chapter

The chapter is structured as follows. We present the semidefinite programming (SDP) performance estimation method of ADMM in Section 9.2, and we develop the performance estimation to handle dual based methods including ADMM. In Section 9.3, we derive some new non-asymptotic convergence rates by using performance estimation for ADMM in terms of dual function, primal and dual residuals. Furthermore, we show that the given bounds are tight by providing some examples. In Section 9.4 we proceed with the study of the linear convergence of ADMM. We establish that ADMM enjoys a linear convergence if and only if the dual function satisfies the PŁ inequality when the objective function is strongly convex. Furthermore, we investigate the relation between the PŁ inequality and common conditions used by scholars to prove the linear convergence. Section 9.5 is devoted to the R-linear convergence. We prove that ADMM is R-linear convergent under two new scenarios which are weaker than the existing ones in the literature.

Terminology and notation

We have the following identity

$$\xi \in \partial f(x) \iff x \in \partial f^*(\xi).$$
(9.4)

By using conjugate functions, the dual of problem (9.1) may be written as

$$D(\lambda) = \min_{(x,z) \in \mathbb{R}^n \times \mathbb{R}^m} f(x) + g(z) + \langle \lambda, Ax + Bz - b \rangle$$

= $-\langle \lambda, b \rangle - f^*(-A^T \lambda) - g^*(-B^T \lambda).$ (9.5)

By the optimality conditions for the dual problem, we get

$$b - Ax^* - Bz^* = 0, (9.6)$$

for some $x^* \in \partial f^*(-A^T \lambda^*)$ and $z^* \in \partial g^*(-B^T \lambda^*)$. Equation (9.6) with (9.4) imply that (x^*, z^*) is an optimal solution to problem (9.1).

The optimality conditions for the subproblems of Algorithm 9.1 may be written as

$$0 \in \partial f(x^{k}) + A^{T}\lambda^{k-1} + tA^{T}\left(Ax^{k} + Bz^{k-1} - b\right),$$

$$0 \in \partial g(z^{k}) + B^{T}\lambda^{k-1} + tB^{T}\left(Ax^{k} + Bz^{k} - b\right).$$
(9.7)

As $\lambda^k = \lambda^{k-1} + t(Ax^k + Bz^k - b)$, we get

$$0 \in \partial f(x^k) + A^T \lambda^k + t A^T B(z^{k-1} - z^k), \quad 0 \in \partial g(z^k) + B^T \lambda^k.$$
(9.8)

So, (x^k, z^k) is optimal for dual objective at λ^k if and only if $A^T B(z^{k-1}-z^k) = 0$. We call $A^T B(z^{k-1}-z^k)$ dual residual.

9.2 Performance estimation

In this section, we develop the performance estimation for ADMM. Gu and Yang [GY20] employed performance estimation to study the extension of the dual step length for ADMM. Note that while there are some similarities between our work and [GY20] in using performance estimation, the formulations and results are different.

The worst-case convergence rate of Algorithm 9.1 with respect to dual objective value may be cast as the following abstract optimization problem,

$$\max D(\lambda^{*}) - D(\lambda^{N})$$
s.t. $\{x^{k}, z^{k}, \lambda^{k}\}_{1}^{N}$ is generated by Algorithm 9.1 w.r.t. $f, g, A, B, b, \lambda^{0}, z^{0}, t$
 (x^{*}, z^{*}) is an optimal solution with Lagrangian multipliers λ^{*}
 $\|\lambda^{0} - \lambda^{*}\|^{2} + t^{2} \|z^{0} - z^{*}\|_{B}^{2} = \Delta$
 $f \in \mathcal{F}_{c_{1},\infty}^{A}(\mathbb{R}^{n}), g \in \mathcal{F}_{c_{2},\infty}^{B}(\mathbb{R}^{m})$
 $\lambda^{0} \in \mathbb{R}^{r}, z^{0} \in \mathbb{R}^{m}, A \in \mathbb{R}^{r \times n}, B \in \mathbb{R}^{r \times m}, b \in \mathbb{R}^{r},$
(9.9)

where $f, g, A, B, b, z^0, \lambda^0, x^*, z^*, \lambda^*$ are decision variables and N, t, c_1, c_2, Δ are the given parameters. Note that problem (9.9) will be unbounded unless we impose some initial condition. We regard boundedness of $\|\lambda^0 - \lambda^*\|^2 + t^2 \|z^0 - z^*\|_B^2$ as an initial condition. The boundedness of $t^{-1}\|\lambda^0 - \lambda^*\|^2 + t \|z^0 - z^*\|_B^2$ is commonly used for the convergence analysis of ADMM; see e.g. [BPC⁺11, LLF22]. We opt to utilize the positive multiplication of this criterion for notational convenience as *t* is a fixed positive constant in Algorithm 9.1. Moreover, we use this measure to establish R-linear convergence in terms of dual objective; see Section 9.5 for more discussion.

Note that $D(\lambda^*) = f^* + g^*$ and $(\tilde{x}, \tilde{z}) \in \operatorname{argmin} f(x) + g(z) + \langle \lambda^N, Ax + Bz - b \rangle$ if and only if

$$\tilde{\xi} + A^T \lambda^N = 0, \quad \tilde{\eta} + B^T \lambda^N = 0, \tag{9.10}$$

for some $\tilde{\xi} \in \partial f(\tilde{x})$ and $\tilde{\eta} \in \partial g(\tilde{z})$. It is worth noting that a point \tilde{x} satisfying these conditions exists, as function f is strongly convex relative to A. In addition, one may consider $\tilde{z} = z^N$ by virtue of (9.8). For the sake of notational convenience, we introduce $x^{N+1} = \tilde{x}$ and $\xi^{N+1} = \tilde{\xi}$. The reader should bear in mind that x^{N+1} is not generated by Algorithm 9.1. Therefore,

$$D(\lambda^N) = f(x^{N+1}) + g(z^N) + \left\langle \lambda^N, Ax^{N+1} + Bz^N - b \right\rangle$$

for some x^{N+1} with $-A^T \lambda^N \in \partial f(x^{N+1})$.

By using Theorem 2.40 to replace the conditions $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$, and $g \in \mathcal{F}^{B}_{c_{2},\infty}(\mathbb{R}^{m})$ by finite interpolation conditions, and by using the optimality conditions (9.7), problem (9.9) may be reformulated as a finite dimensional optimiza-

tion problem, through the performance estimation technique:

$$\max f^{*} + g^{*} - (f^{N+1} + g^{N} + \langle \lambda^{N}, Ax^{N+1} + Bz^{N} - b \rangle)$$
s.t. $\{(x^{k}; \xi^{k}; f^{k})\}_{1}^{N+1} \cup \{(x^{*}; \xi^{*}; f^{*})\}$ satisfy interpolation constraints (2.5)
 $\{(z^{k}; \eta^{k}; g^{k})\}_{0}^{N} \cup \{(z^{*}; \eta^{*}; g^{*})\}$ satisfy interpolation constraints (2.5)
 (x^{*}, z^{*}) is an optimal solution with Lagrangian multipliers λ^{*}
 $\|\lambda^{0} - \lambda^{*}\|^{2} + t^{2} \|z^{0} - z^{*}\|_{B}^{2} = \Delta$ (9.11)
 $\xi^{k} = tA^{T}b - tA^{T}Ax^{k} - tA^{T}Bz^{k-1} - A^{T}\lambda^{k-1}, \ k \in \{1, ..., N\}$
 $\eta^{k} = tB^{T}b - tB^{T}Ax^{k} - tB^{T}Bz^{k} - B^{T}\lambda^{k-1}, \ k \in \{1, ..., N\}$
 $\lambda^{k} = \lambda^{k-1} + t(Ax^{k} + Bz^{k} - b), \ k \in \{1, ..., N\}$
 $\xi^{N+1} + A^{T}\lambda^{N} = 0$
 $\lambda^{0} \in \mathbb{R}^{r}, z^{0} \in \mathbb{R}^{m}, A \in \mathbb{R}^{r \times n}, B \in \mathbb{R}^{r \times m}, b \in \mathbb{R}^{r}.$

In problem (9.11), $A, B, \{x^k; \xi^k; f^k\}_1^{N+1}, \{(x^*; \xi^*; f^*)\}, \{\lambda^k\}_0^N, \{z^k; \eta^k; g^k\}_0^N, \{(z^*; \eta^*; g^*)\}, \lambda^*, b \text{ are decision variables. To handle problem (9.11), without loss of generality, we assume that the matrix <math>\begin{pmatrix} A & B \end{pmatrix}$ has full row rank. Note this assumption does not appear in our arguments in the following sections. In addition, we introduce some new variables. As problem (9.1) is invariant under translation of (x, z), we may assume without loss of generality that b = 0 and $(x^*, z^*) = (0, 0)$. In addition, due to the full row rank of the matrix $\begin{pmatrix} A & B \end{pmatrix}$, we may assume that

$$\lambda^{0} = \begin{pmatrix} A & B \end{pmatrix} \begin{pmatrix} x^{\dagger} \\ z^{\dagger} \end{pmatrix} \text{ and } \lambda^{\star} = \begin{pmatrix} A & B \end{pmatrix} \begin{pmatrix} \bar{x} \\ \bar{z} \end{pmatrix} \text{ for some } \bar{x}, x^{\dagger}, \bar{z}, z^{\dagger}. \text{ So,}$$
$$\xi^{\star} = -A^{T}A\bar{x} - A^{T}B\bar{z} \in \partial f(0), \quad \eta^{\star} = -B^{T}A\bar{x} - B^{T}B\bar{z} \in \partial g(0),$$

and $D(\lambda^*) = f^* + g^*$.

By using equality constraints of problem (9.11) and the newly introduced variables, we have for $k \in \{1, ..., N\}$

$$\lambda^{k} = (Ax^{\dagger} + Bz^{\dagger}) + \sum_{i=1}^{k} t(Ax^{i} + Bz^{i}),$$

$$-(A^{T}Ax^{\dagger} + A^{T}Bz^{\dagger}) - \sum_{i=1}^{k-1} t(A^{T}Ax^{i} + A^{T}Bz^{i}) - tA^{T}Ax^{k} - tA^{T}Bz^{k-1} \in \partial f(x^{k}),$$

$$-(B^{T}Ax^{\dagger} + B^{T}Bz^{\dagger}) - \sum_{i=1}^{k} t(B^{T}Ax^{i} + B^{T}Bz^{i}) \in \partial g(z^{k}).$$
(9.12)

Hence, problem (9.11) may be written as

$$\begin{split} \max f^{*} + g^{*} - f^{N+1} - g^{N} - \left\langle Ax^{\dagger} + Bz^{\dagger} + \sum_{i=1}^{N} t(Ax^{i} + Bz^{i}), Ax^{N+1} + Bz^{N} \right\rangle \\ \text{s.t.} \quad \frac{c_{i}}{2} \left\| x^{k} - x^{j} \right\|_{A}^{2} \leq \left\langle Ax^{\dagger} + Bz^{\dagger} + \sum_{i=1}^{k-1} t(Ax^{i} + Bz^{i}) + tAx^{k} + tBz^{k-1}, A(x^{j} - x^{k}) \right\rangle + \\ f^{j} - f^{k}, \ k \in \{1, \dots, N\}, \ j \in \{1, \dots, N+1\}, \\ \frac{c_{i}}{2} \left\| x^{N+1} - x^{j} \right\|_{A}^{2} \leq \left\langle Ax^{\dagger} + Bz^{\dagger} + \sum_{i=1}^{N} t(Ax^{i} + Bz^{i}), A(x^{j} - x^{N+1}) \right\rangle + \\ f^{j} - f^{N+1}, \ j \in \{1, \dots, N\}, \\ \frac{c_{i}}{2} \left\| z^{k} - z^{j} \right\|_{B}^{2} \leq \left\langle Ax^{\dagger} + Bz^{\dagger} + \sum_{i=1}^{k} t(Ax^{i} + Bz^{i}), B(z^{j} - z^{k}) \right\rangle + \\ g^{j} - g^{k}, \ j, k \in \{1, \dots, N\}, \\ \frac{c_{i}}{2} \left\| x^{k} \right\|_{A}^{2} \leq f^{k} - f^{*} + \left\langle A\bar{x} + B\bar{z}, Ax^{k} \right\rangle, \ k \in \{1, \dots, N+1\}, \\ \frac{c_{i}}{2} \left\| x^{k} \right\|_{A}^{2} \leq - \left\langle Ax^{\dagger} + Bz^{\dagger} + \sum_{i=1}^{k-1} t(Ax^{i} + Bz^{i}) + tAx^{k} + tBz^{k-1}, Ax^{k} \right\rangle + \\ f^{*} - f^{k}, \ k \in \{1, \dots, N\}, \\ \frac{c_{i}}{2} \left\| x^{N+1} \right\|_{A}^{2} \leq f^{*} - f^{N+1} - \left\langle Ax^{\dagger} + Bz^{\dagger} + \sum_{i=1}^{N} t(Ax^{i} + Bz^{i}), Ax^{N+1} \right\rangle, \\ \frac{c_{i}}{2} \left\| x^{N+1} \right\|_{A}^{2} \leq g^{k} - g^{*} + \left\langle A\bar{x} + B\bar{z}, Bz^{k} \right\rangle, \ k \in \{1, \dots, N\}, \\ \frac{c_{i}}{2} \left\| z^{k} \right\|_{B}^{2} \leq g^{k} - g^{*} - \left\langle Ax^{\dagger} + Bz^{\dagger} + \sum_{i=1}^{k} t(Ax^{i} + Bz^{i}), Bz^{k} \right\rangle, \ k \in \{1, \dots, N\}, \\ \frac{c_{i}}{2} \left\| z^{k} \right\|_{B}^{2} \leq g^{k} - g^{k} - \left\langle Ax^{\dagger} + Bz^{\dagger} + \sum_{i=1}^{k} t(Ax^{i} + Bz^{i}), Bz^{k} \right\rangle, \ k \in \{1, \dots, N\}, \\ \frac{(i_{i} Ax^{\dagger} + Bz^{\dagger} - (A\bar{x} + B\bar{z}) \right\|_{a}^{2} + t^{2} \left\| z^{0} \right\|_{B}^{2} = \Delta, \\ x^{\dagger} \in \mathbb{R}^{n}, z^{0}, z^{\dagger} \in \mathbb{R}^{m}, A \in \mathbb{R}^{r \times n}, B \in \mathbb{R}^{r \times m}. \end{split}$$

In problem (9.13), $A, B, \{x^k, f^k\}_1^{N+1}, \{z^k, g^k\}_1^N, x^{\dagger}, z^{\dagger}, \bar{x}, f^{\star}, \bar{z}, g^{\star}, z^0$ are decision variables. By using the Gram matrix method, problem (9.13) may be relaxed as a semidefinite program as follows. Let

$$U = \begin{pmatrix} x^{\dagger} & x^1 & \dots & x^{N+1} & \bar{x} \end{pmatrix}, \qquad V = \begin{pmatrix} z^{\dagger} & z^0 & \dots & z^N & \bar{z} \end{pmatrix}.$$

By introducing matrix variable

$$Y = \begin{pmatrix} AU & BV \end{pmatrix}^T \begin{pmatrix} AU & BV \end{pmatrix},$$
problem (9.13) may be relaxed as the following SDP,

$$\max f^{*} + g^{*} - f^{N+1} - g^{N} - \operatorname{tr}(L_{o}Y)$$

s.t. $\operatorname{tr}(L_{i,j}^{f}Y) \leq f^{i} - f^{j}, \quad i, j \in \{1, ..., N + 1, \star\}$
 $\operatorname{tr}(L_{i,j}^{g}Y) \leq g^{i} - g^{j}, \quad i, j \in \{1, ..., N, \star\}$
 $\operatorname{tr}(L_{0}Y) = \Delta$ (9.14)
 $Y \geq 0,$

where the constant matrices $L_{i,j}^f, L_{i,j}^g, L_o, L_0$ are determined according to the constraints of problem (9.13). In the following sections, we present some new convergence results that are derived by solving this kind of formulation.

9.3 Worst-case convergence rate

In this section, we provide new convergence rates for ADMM with respect to some performance measures. Before we get to the theorems we need to present some lemmas.

Lemma 9.2. Let $N \ge 4$ and $t, c \in \mathbb{R}$. Let E(t, c) be $(N + 1) \times (N + 1)$ symmetric matrix given by

where

$$\alpha_{k} = \begin{cases} 6c - 5t, & k = 2\\ 2(2k^{2} - 3k + 1)c - (4k - 1)t, & 3 \le k \le N - 1\\ 2N(N - 1)c - (2N + 1)t, & k = N\\ 2Nc - (N + 1)t, & k = N + 1, \end{cases}$$
$$\beta_{k} = \begin{cases} 2kt - (2k^{2} - k - 1)c, & 2 \le k \le N - 1\\ 3t - 2(N - 1)c, & k = N, \end{cases}$$

and k denotes row number. If c > 0 is given, then

$$[0,c] \subseteq \{t : E(t,c) \succeq 0\}.$$

Proof. As $\{t : E(t,c) \geq 0\}$ is a convex set, it suffices to prove the positive semidefiniteness of E(0,c) and E(c,c). Since E(0,c) is diagonally dominant, it is positive semidefinite. Now, we establish that the matrix K = E(1,1) is positive definite. To this end, we show that all leading principal minors of K are positive. To compute the leading principal minors, we perform the following elementary row operations on K:

- i) Add the second row to the third row;
- ii) Add the second row to the last row;
- iii) Add the third row to the forth row;

iv) For
$$i = 4: N - 1$$

- Add i th row to (i + 1) th row;
- Add $\frac{3-i}{2i^2-3i-1}$ times of i th row to the last row;
- v) Add $\frac{N-1}{3N-5}$ times of N th row to (N + 1) th row.

It is seen that $K_{k-1,k} + K_{k,k} = -K_{k+1,k}$ for $2 \le k \le N - 1$. Hence, by performing these operations, we get an upper triangular matrix *J* with diagonal

$$J_{k,k} = \begin{cases} 2, & k = 1 \\ 2k^2 - 3k - 1, & 2 \le k \le N - 1 \\ 3N - 5, & k = N \\ N - 2 - \frac{(N-1)^2}{3N - 5} - \sum_{i=4}^{N-1} \frac{(i-3)^2}{2i^2 - 3i - 1}, & k = N + 1. \end{cases}$$

It is seen all first *N* diagonal elements of *J* are positive. We show that $J_{N+1,N+1}$ is also positive. For $i \ge 4$ we have

$$\frac{(i-3)^2}{2i^2-3i-1} \le \frac{(i-1)^2+4}{2(i-1)^2} \le \frac{1}{2} + \frac{2}{(i-1)(i-2)}.$$
(9.15)

So,

$$\frac{2N^2 - 9N + 9}{3N - 5} - \sum_{i=4}^{N-1} \frac{(i-3)^2}{2i^2 - 3i - 1} \ge \frac{(N-2)(N^2 - 5N + 10)}{2N(3N - 5)} > 0,$$

which implies $J_{N+1,N+1} > 0$. Since we add a factor of i - th row to j - th row with i < j, all leading principal minors of matrices K and J are the same. Hence K is positive definite. As E(c,c) = cK, one can infer the positive definiteness of E(c,c) and the proof is complete.

In the upcoming lemma, we establish a valid inequality for ADMM that will be utilized in all the subsequent results presented in this section.

Lemma 9.3. Let $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$, $g \in \mathcal{F}_{0,\infty}(\mathbb{R}^{m})$ and $x^{*} = 0$, $z^{*} = 0$. Suppose that ADMM with the starting points λ^{0} and z^{0} generates $\{(x^{k}; z^{k}; \lambda^{k})\}$. If $N \geq 4$ and $\nu \in \mathbb{R}^{r}$, then

$$N\langle\lambda^{N}, Ax^{N} + Bz^{N}\rangle - \langle\lambda^{N} + tAx^{N} + tBz^{N-1}, Ax^{N} - \nu\rangle + \langle\lambda^{0} + tAx^{1} + tBz^{0}, Ax^{1} - \nu\rangle + \frac{1}{2t} \|\lambda^{0} - \lambda^{\star}\|^{2} - \frac{1}{2t} \|\lambda^{N} - \lambda^{\star}\|^{2} + \frac{t}{2} \|z^{0}\|_{B}^{2} - t \langle Ax^{1} - Ax^{2} + (N+1)Ax^{N} + Bz^{N}, \nu\rangle - t \sum_{k=3}^{N} \langle Ax^{k}, \nu\rangle + \frac{t(N-1)}{2} \|\nu\|^{2} - \frac{c_{1}}{2} \|x^{1}\|_{A}^{2} + \sum_{k=2}^{N} \frac{\alpha_{k}}{2} \|x^{k}\|_{A}^{2} + \sum_{k=2}^{N-1} \beta_{k} \langle Ax^{k}, Ax^{k+1}\rangle + tN \langle Bz^{N-1}, Ax^{N} - \nu\rangle + t \langle Ax^{N}, Bz^{N}\rangle - \frac{t(N-1)^{2}}{2} \|z^{N} - z^{N-1}\|_{B}^{2} - \frac{tN^{2}}{2} \|Ax^{N} + Bz^{N}\|^{2} - t \|x^{2}\|_{A}^{2} + f(x^{1}) - f(x^{N}) + N (f(x^{N}) - f^{\star} + g(x^{N}) - g^{\star}) \ge 0,$$

$$(9.16)$$

where

$$\begin{aligned} \alpha_k &= \begin{cases} (4k-1)\,t-2\,\big(2k^2-3k+1\big)\,c_1, & 2\leq k\leq N-1,\\ (4N+1)\,t-\big(2N^2-5N+3\big)\,c_1, & k=N, \end{cases} \\ \beta_k &= \big(2k^2-k-1\big)\,c_1-2kt. \end{aligned}$$

Proof. To establish the desired inequality, we demonstrate its validity by summing a series of valid inequalities. To simplify the notation, let $f^k = f(x^k)$ and $g^k = g(z^k)$ for $k \in \{1, ..., N\}$. Note that b = 0 because $x^* = 0, z^* = 0$. By (2.5) and (9.7), we get the following inequality

$$\begin{split} &\sum_{k=1}^{N-1} (k^2 - 1) \left(f^{k+1} - f^k + \left\langle \lambda^{k-1} + tAx^k + tBz^{k-1}, A(x^{k+1} - x^k) \right\rangle - \frac{c_1}{2} \left\| x^{k+1} - x^k \right\|_A^2 \right) \\ &+ \sum_{k=1}^{N-1} (k^2 - k) \left(f^k - f^{k+1} + \left\langle \lambda^k + tAx^{k+1} + tBz^k, A(x^k - x^{k+1}) \right\rangle - \frac{c_1}{2} \left\| x^{k+1} - x^k \right\|_A^2 \right) \\ &+ \sum_{k=1}^N \left(f^k - f^\star + \left\langle \lambda^\star, Ax^k \right\rangle - \frac{c_1}{2} \left\| x^k \right\|_A^2 \right) + \sum_{k=1}^{N-1} k^2 \left(g^k - g^{k+1} + \left\langle \lambda^{k+1}, B(z^k - z^{k+1}) \right\rangle \right) \\ &+ \sum_{k=1}^{N-1} (k^2 + k) \left(g^{k+1} - g^k + \left\langle \lambda^k, B(z^{k+1} - z^k) \right\rangle \right) + \sum_{k=1}^N \left(g^k - g^\star + \left\langle \lambda^\star, Bz^k \right\rangle \right) \\ &+ \frac{t}{2} \left\| Ax^1 + Bz^0 - v \right\|^2 \ge 0. \end{split}$$

As $\lambda^k = \lambda^{k-1} + tAx^k + tBz^k$, the inequality can be expressed as

$$\begin{split} &\sum_{k=1}^{N-1} (k^2 - 1) \left(\left\langle tAx^k + tBz^{k-1}, A(x^{k+1} - x^k) \right\rangle - \frac{c_1}{2} \left\| x^{k+1} - x^k \right\|_A^2 \right) + \\ &\sum_{k=1}^{N-1} (k^2 - 1) \left(\left\langle \lambda^k, Ax^{k+1} \right\rangle - \left\langle \lambda^{k-1}, Ax^k \right\rangle - \left\langle tAx^k + tBz^k, Ax^{k+1} \right\rangle \right) + \\ &\sum_{k=1}^{N-1} (k^2 - k) \left(\left\langle tAx^{k+1} + tBz^k, A(x^k - x^{k+1}) \right\rangle - \frac{c_1}{2} \left\| x^{k+1} - x^k \right\|_A^2 \right) + \\ &\sum_{k=1}^{N-1} (k^2 - k) \left(\left\langle \lambda^{k-1}, Ax^k \right\rangle - \left\langle \lambda^k, Ax^{k+1} \right\rangle + \left\langle tAx^k + tBz^k, Ax^k \right\rangle \right) + \\ &\sum_{k=1}^{N-1} (k^2 + k) \left(\left\langle \lambda^k, Bz^{k+1} \right\rangle - \left\langle \lambda^{k-1}, Bz^k \right\rangle - \left\langle tAx^k + tBz^k, Bz^k \right\rangle \right) + \\ &\sum_{k=1}^{N-1} k^2 \left(\left\langle \lambda^{k-1}, Bz^k \right\rangle - \left\langle \lambda^k, Bz^{k+1} \right\rangle + \left\langle tAx^k + tBz^k + tAx^{k+1} + tBz^{k+1}, Bz^k \right\rangle - \\ &\left\langle tAx^{k+1} + tBz^{k+1}, Bz^{k+1} \right\rangle \right) + \\ &\sum_{k=1}^{N} \left(\left\langle \lambda^*, Ax^k + Bz^k \right\rangle - \frac{c_1}{2} \left\| x^k \right\|_A^2 \right) + \frac{t}{2} \left\| Bz^0 \right\|^2 + \\ &\frac{t}{2} \left\| Ax^1 - v \right\|^2 + t \left\langle Ax^1 - v, Bz^0 \right\rangle + f^1 - f^N + N(f^N - f^* + g^N - g^*) \ge 0. \end{split}$$

After performing some algebraic manipulations, we obtain

$$\begin{split} N\langle\lambda^{N-1}, Ax^{N} + Bz^{N}\rangle &- \langle\lambda^{N-1}, Ax^{N}\rangle + \langle\lambda^{0}, Ax^{1}\rangle - \sum_{k=0}^{N-1} \langle\lambda^{k} - \lambda^{\star}, Ax^{k+1} + Bz^{k+1}\rangle + \\ \frac{t}{2} \left\| Ax^{1} - v \right\|^{2} + \frac{t}{2} \left\| Bz^{0} \right\|^{2} + t \left\langle Ax^{1} - v, Bz^{0} \right\rangle - t(N^{2} - 3N + 1) \langle Ax^{N}, Bz^{N-1} \rangle - \\ t \sum_{k=1}^{N-1} \left((k-1)^{2} \| Ax^{k} \|^{2} - (k^{2} - k) \langle Ax^{k}, Ax^{k+1} \rangle - (k^{2} - 1) \langle Ax^{k+1}, Bz^{k-1} \rangle \right) - \\ t \sum_{k=1}^{N-1} \left((k^{2} - k + 1) \| Bz^{k} \|^{2} + (-k^{2} + k + 1) \langle Ax^{k}, Bz^{k} \rangle - k^{2} \langle Bz^{k}, Bz^{k+1} \rangle \right) - \\ t \sum_{k=2}^{N-1} \left((2k^{2} - 3k) \langle Ax^{k}, Bz^{k-1} \rangle \right) - t(N-1)^{2} \| Bz^{N} \|^{2} - t(N^{2} - 3N + 2) \| Ax^{N} \|^{2} - \\ t(N-1)^{2} \langle Ax^{N}, Bz^{N} \rangle - \sum_{k=1}^{N-1} \left((2k^{2} - k - 1) \frac{c_{1}}{2} \| x^{k+1} - x^{k} \|_{A}^{2} + \frac{c_{1}}{2} \| x^{k+1} \|_{A}^{2} \right) - \\ \frac{c_{1}}{2} \| x^{1} \|_{A}^{2} + f^{1} - f^{N} + N(f^{N} - f^{*} + g^{N} - g^{*}) \ge 0. \end{split}$$

By using $\lambda^{N-1} = \lambda^N - tAx^N - tBz^N$ and

$$2\langle \lambda^{k} - \lambda^{\star}, Ax^{k+1} + Bz^{k+1} \rangle = \frac{1}{t} \|\lambda^{k+1} - \lambda^{\star}\|^{2} - \frac{1}{t} \|\lambda^{k} - \lambda^{\star}\|^{2} - t \|Ax^{k+1} + Bz^{k+1}\|^{2},$$

we get

$$\begin{split} & N\langle\lambda^{N}, Ax^{N} + Bz^{N}\rangle - \langle\lambda^{N} + tAx^{N} + tBz^{N-1}, Ax^{N} - \nu\rangle + \langle\lambda^{0} + tAx^{1} + tBz^{0}, Ax^{1} - \nu\rangle \\ & + \frac{1}{2t} \left\|\lambda^{0} - \lambda^{\star}\right\|^{2} - \frac{1}{2t} \left\|\lambda^{N} - \lambda^{\star}\right\|^{2} + \frac{t}{2} \left\|z^{0}\right\|_{B}^{2} - t \left\langle Ax^{1} - Ax^{2} + (N+1)Ax^{N} + Bz^{N}, \nu \right\rangle \\ & - t \sum_{k=3}^{N} \left\langle Ax^{k}, \nu \right\rangle - \frac{t}{2} \sum_{k=2}^{N-1} \left\| (k-1)Bz^{k-1} - (k-1)Bz^{k} + kAx^{k} - (k+1)Ax^{k+1} + \nu \right\|^{2} \\ & + \frac{t(N-1)}{2} \left\|\nu\right\|^{2} - \frac{c_{1}}{2} \left\|x^{1}\right\|_{A}^{2} - 2t \left\|x^{2}\right\|_{A}^{2} + \frac{1}{2} \sum_{k=2}^{N-1} \left((4k-1)t - 2\left(2k^{2} - 3k + 1\right)c_{1} \right) \left\|x^{k}\right\|_{A}^{2} \\ & + \sum_{k=2}^{N-1} \left(\left(2k^{2} - k - 1\right)c_{1} - 2kt \right) \left\langle Ax^{k}, Ax^{k+1} \right\rangle + \left(\left(2N + \frac{1}{2}\right)t - \left(N^{2} - \frac{5}{2}N + \frac{3}{2}\right)c_{1} \right) \left\|x^{N}\right\|_{A}^{2} \\ & + tN \left\langle Bz^{N-1}, Ax^{N} - \nu \right\rangle + t \left\langle Ax^{N}, Bz^{N} \right\rangle - \frac{t(N-1)^{2}}{2} \left\|z^{N} - z^{N-1}\right\|_{B}^{2} \\ & - \frac{tN^{2}}{2} \left\|Ax^{N} + Bz^{N}\right\|^{2} + f^{1} - f^{N} + N \left(f^{N} - f^{*} + g^{N} - g^{*} \right) \ge 0, \end{split}$$

which implies the desired inequality.

We may now prove the main result of this section.

Theorem 9.4. Let $f \in \mathcal{F}^A_{c_1,\infty}(\mathbb{R}^n)$ and $g \in \mathcal{F}_{0,\infty}(\mathbb{R}^m)$ with $c_1 > 0$. If $t \leq c_1$ and $N \geq 4$, then

$$D(\lambda^{\star}) - D(\lambda^{N}) \le \frac{\|\lambda^{0} - \lambda^{\star}\|^{2} + t^{2} \|z^{0} - z^{\star}\|_{B}^{2}}{4Nt}.$$
(9.17)

Proof. As discussed in Section 9.2, we may assume that $x^* = 0$ and $z^* = 0$. By (9.10), we have $D(\lambda^N) = f(\hat{x}^N) + g(z^N) + \langle \lambda^N, A\hat{x}^N + Bz^N \rangle$ for some \hat{x}^N with $-A^T \lambda^N \in \partial f(\hat{x}^N)$. By employing (2.5) and (9.7), we obtain

$$N\left(g(x^{N}) - g^{\star} + \langle \lambda^{\star}, Bz^{N} \rangle\right) + (N-1)\left(f(x^{N}) - f^{\star} + \langle \lambda^{\star}, Ax^{N} \rangle - \frac{c_{1}}{2} \|x^{N}\|_{A}^{2}\right) + \left(f(\hat{x}^{N}) - f(x^{1}) + \langle \lambda^{0} + tAx^{1} + tBz^{0}, A\hat{x}^{N} - Ax^{1} \rangle - \frac{c_{1}}{2} \|\hat{x}^{N} - x^{1}\|_{A}^{2}\right) + (2N-2)\left(f(\hat{x}^{N}) - f(x^{N}) + \langle \lambda^{N} - tBz^{N} + tBz^{N-1}, A\hat{x}^{N} - Ax^{N} \rangle - \frac{c_{1}}{2} \|\hat{x}^{N} - x^{N}\|_{A}^{2}\right) + \left(f(\hat{x}^{N}) - f^{\star} + \langle \lambda^{\star}, A\hat{x}^{N} \rangle - \frac{c_{1}}{2} \|\hat{x}^{N}\|_{A}^{2}\right) \geq 0.$$

$$(9.18)$$

By substituting v with $A\hat{x}^N$ in inequality (9.16) and summing it with (9.18), we get the following inequality after performing some algebraic manipulations

$$2N\left(f\left(\hat{x}^{N}\right) + g\left(x^{N}\right) + \left\langle\lambda^{N}, A\hat{x}^{N} + Bz^{N}\right\rangle - f^{\star} - g^{\star}\right) + \frac{1}{2t}\left\|\lambda^{0} - \lambda^{\star}\right\|^{2} + \frac{t}{2}\left\|z^{0}\right\|_{B}^{2} - \frac{1}{2t}\left\|\lambda^{N} - \lambda^{\star} + t(N-1)Ax^{N} + tA\hat{x}^{N} + tNBz^{N}\right\|^{2} - \frac{1}{2t}\left\|(N-1)(Bz^{N-1} - Bz^{N}) + tAx^{N} - tA\hat{x}^{N}\right\|^{2} - \frac{1}{2}\operatorname{tr}\left(E(t, c_{1})\left(Ax^{1} \dots A\hat{x}^{N}\right)^{T}\left(Ax^{1} \dots A\hat{x}^{N}\right)\right) \geq 0,$$
(9.19)

where the positive semidefinite matrix $E(t, c_1)$ is given in Lemma 9.2. As the inner product of positive semidefinite matrices is non-negative, inequality (9.19) implies that

$$2N\left(D(\lambda^{\star})-D(\lambda^{N})\right) \leq \frac{1}{2t} \left\|\lambda^{0}-\lambda^{\star}\right\|^{2}+\frac{t}{2} \left\|z^{0}\right\|_{B}^{2},$$

and the proof is complete.

In comparison with Theorem 9.1, we get a new convergence rate when only f is strongly convex, i.e. g does not need to be strongly convex. Also, the constant does not depend on λ^1 . One important question concerning bound (9.17) is its

tightness, that is, if there is an optimization problem which attains the given convergence rate. It turns out that the bound (9.17) is exact. The following example demonstrates this point.

Example 9.5. Suppose that $c_1 > 0$, $N \ge 4$ and $t \in (0, c_1]$. Let $f, g : \mathbb{R} \to \mathbb{R}$ be given as follows,

$$f(x) = \frac{1}{2}|x| + \frac{c_1}{2}x^2, \ g(z) = \frac{1}{2}\max\{\frac{N-1}{N}(z-\frac{1}{2Nt}) - \frac{1}{2Nt}, -z\}.$$

Consider the optimization problem

$$\min_{\substack{(x,z)\in\mathbb{R}\times\mathbb{R}}} f(x) + g(z),$$

s.t. $x + z = 0$,

It is seen that A = B = I in this problem. Note that $(x^*, z^*) = (0, 0)$ with Lagrangian multiplier $\lambda^* = \frac{1}{2}$ is an optimal solution and the optimal value is zero. One can check that Algorithm 9.1 with initial point $\lambda^0 = \frac{-1}{2}$ and $z^0 = 0$ generates the following points,

$$\begin{aligned} x^{k} &= 0 & k \in \{1, \dots, N\} \\ z^{k} &= \frac{1}{2Nt} & k \in \{1, \dots, N\} \\ \lambda^{k} &= \frac{-1}{2} + \frac{k}{2N} & k \in \{1, \dots, N\}. \end{aligned}$$

At λ^N , we have $D(\lambda^N) = \frac{-1}{4Nt} = -\frac{\|\lambda^0 - \lambda^\star\|^2 + t^2\|_B^2}{4Nt}$, which shows the tightness of bound (9.17).

One important factor concerning dual-based methods that determines the efficiency of an algorithm is primal and dual feasibility (residual) convergence rates. In what follows, we study this subject under the setting of Theorem 9.4. The next theorem gives a convergence rate in terms of primal residual under the setting of Theorem 9.4.

Theorem 9.6. Let $f \in \mathcal{F}^A_{c_1,\infty}(\mathbb{R}^n)$ and $g \in \mathcal{F}_{0,\infty}(\mathbb{R}^m)$ with $c_1 > 0$. If $t \leq c_1$ and $N \geq 4$, then

$$\left\|Ax^{N} + Bz^{N} - b\right\| \le \frac{\sqrt{\|\lambda^{0} - \lambda^{\star}\|^{2} + t^{2} \|z^{0} - z^{\star}\|_{B}^{2}}}{tN}.$$
(9.20)

Proof. The argument is similar to that used in the proof of Theorem 9.4. By setting $v = Ax^N$ in (9.16), one can infer the following inequality

$$N\left\langle\lambda^{N}, Ax^{N} + Bz^{N}\right\rangle + \left\langle\lambda^{0} + tAx^{1} + tBz^{0}, Ax^{1} - Ax^{N}\right\rangle + \frac{1}{2t}\left\|\lambda^{0} - \lambda^{*}\right\|^{2} + \frac{t}{2}\left\|z^{0}\right\|_{B}^{2} - t\left\langleAx^{1} - Ax^{2}, Ax^{N}\right\rangle + \frac{t(N-1)}{2}\left\|Ax^{N}\right\|^{2} - t\sum_{k=3}^{N}\left\langleAx^{k}, Ax^{N}\right\rangle - \frac{c_{1}}{2}\left\|x^{1}\right\|_{A}^{2} - t\left\|x^{2}\right\|_{A}^{2} + \sum_{k=2}^{N-1}\left(\left(2k - \frac{1}{2}\right)t - \left(2k^{2} - 3k + 1\right)c_{1}\right)\left\|x^{k}\right\|_{A}^{2} + \left(\left(\frac{3}{2}N - \frac{3}{2}\right)t - \left(N^{2} - \frac{5}{2}N + \frac{3}{2}\right)c_{1}\right)\left\|x^{N}\right\|_{A}^{2} + \sum_{k=2}^{N-1}\left(\left(2k^{2} - k - 1\right)c_{1} - 2kt\right)\left\langleAx^{k}, Ax^{k+1}\right\rangle - \frac{t(N-1)^{2}}{2}\left\|z^{N} - z^{N-1}\right\|_{B}^{2} - \frac{tN^{2}}{2}\left\|Ax^{N} + Bz^{N}\right\|^{2} + f(x^{1}) - f(x^{N}) + N\left(f(x^{N}) - f^{*} + g(x^{N}) - g^{*}\right) \ge 0.$$

$$(9.21)$$

By employing (2.5) and (9.7), we have

$$N\left(f^{\star} - f(x^{N}) - \langle \lambda^{N} + Bz^{N-1} - Bz^{N}, Ax^{N} \rangle - \frac{c_{1}}{2} \left\| x^{N} \right\|_{A}^{2} \right) + \left(f(x^{N}) - f^{1} + \langle \lambda^{0} + tAx^{1} + tBz^{0}, Ax^{N} - Ax^{1} \rangle - \frac{c_{1}}{2} \left\| x^{N} - x^{1} \right\|_{A}^{2} \right) + N\left(g^{\star} - g(x^{N}) - \langle \lambda^{N}, Bz^{N} \rangle\right) \ge 0.$$
(9.22)

By summing (9.21) and (9.22), we obtain

$$\frac{1}{2t} \left\| \lambda^{0} - \lambda^{\star} \right\|^{2} + \frac{t}{2} \left\| z^{0} \right\|_{B}^{2} - \frac{t(N-1)^{2}}{2} \left\| z^{N-1} - z^{N} + \frac{N}{(N-1)^{2}} x^{N} \right\|_{B}^{2} - \frac{tN^{2}}{2} \left\| Ax^{N} + Bz^{N} \right\|^{2} - \frac{1}{2} \operatorname{tr} \left(D(t,c_{1}) \left(Ax^{1} \dots Ax^{N} \right)^{T} \left(Ax^{1} \dots Ax^{N} \right) \right) \ge 0, \quad (9.23)$$

where the matrix $D(t, c_1)$ is as follows,

and

$$\begin{split} \alpha_k &= \begin{cases} 6c_1 - 5t, & k = 2 \\ 2\left(2k^2 - 3k + 1\right)c_1 - \left(4k - 1\right)t, & 3 \leq k \leq N - 1, \\ \left(2N^2 - 4N + 4\right)c_1 - \left(3N - 5 + \frac{N^2}{\left(N - 1\right)^2}\right)t, & k = N, \end{cases} \\ \beta_k &= 2kt - \left(2k^2 - k - 1\right)c_1, \quad 2 \leq k \leq N - 1 \end{split}$$

As the matrix $D(t, c_1)$ is positive semidefinite, see Appendix A.3 Lemma A.1, inequality (9.23) implies that

$$\frac{tN^{2}}{2} \left\| Ax^{N} + Bz^{N} \right\|^{2} \leq \frac{1}{2t} \left\| \lambda^{0} - \lambda^{\star} \right\|^{2} + \frac{t}{2} \left\| z^{0} \right\|_{B}^{2},$$

and the proof is complete.

The following example shows the exactness of bound (9.20).

Example 9.7. Let $c_1 > 0$, $N \ge 4$ and $t \in (0, c_1]$. Consider functions $f, g : \mathbb{R} \to \mathbb{R}$ given by the formulae follows,

$$f(x) = \frac{1}{2}|x| + \frac{c_1}{2}x^2,$$

$$g(z) = \max\{\left(\frac{1}{2} - \frac{1}{N}\right)\left(z - \frac{1}{Nt}\right), \frac{1}{2}\left(\frac{1}{Nt} - z\right)\}.$$

We formulate the following optimization problem,

$$\min_{\substack{(x,z)\in\mathbb{R}\times\mathbb{R}}} f(x) + g(z),$$

s.t. $Ax + Bz = 0,$

where A = B = I. One can verify that $(x^*, z^*) = (0, 0)$ with Lagrangian multiplier $\lambda^* = \frac{1}{2}$ is an optimal solution. Algorithm 9.1 with initial point $\lambda^0 = \frac{-1}{2}$ and $z^0 = 0$ generates the following points,

$$\begin{aligned} x^k &= 0 & k \in \{1, \dots, N\} \\ z^k &= \frac{1}{Nt} & k \in \{1, \dots, N\} \\ \lambda^k &= \frac{2k-N}{2N} & k \in \{1, \dots, N\}. \end{aligned}$$

At iteration N, we have $||Ax^N + Bz^N|| = \frac{1}{tN} = \frac{\sqrt{||\lambda^0 - \lambda^*||^2 + t^2||z^0 - z^*||_B^2}}{tN}$, which shows the tightness of bound (9.20).

In what follows, we study the convergence rate of ADMM in terms of residual dual. To this end, we investigate the convergence rate of $\{B(z^{k-1}-z^k)\}$ as $\|A^TB(z^{k-1}-z^k)\| \leq \|A\| \|z^{k-1}-z^k\|_B$. The next theorem provides a convergence rate for the aforementioned sequence.

Theorem 9.8. Let $f \in \mathcal{F}^A_{c_1,\infty}(\mathbb{R}^n)$ and $g \in \mathcal{F}_{0,\infty}(\mathbb{R}^m)$ with $c_1 > 0$. If $t \leq c_1$ and $N \geq 4$, then

$$\left\| z^{N} - z^{N-1} \right\|_{B} \le \frac{\sqrt{\|\lambda^{0} - \lambda^{\star}\|^{2} + t^{2} \|z^{0} - z^{\star}\|_{B}^{2}}}{(N-1)t}.$$
(9.24)

Proof. Similar to the proof of Theorem 9.4, by setting $v = Ax^N$ in (9.16) for N-1 iterations, one can infer the following inequality

$$\begin{split} &(N-1)\langle\lambda^{N-1},Ax^{N-1}+Bz^{N-1}\rangle+\frac{1}{2t}\|\lambda^{0}-\lambda^{\star}\|^{2}-\frac{1}{2t}\|\lambda^{N-1}-\lambda^{\star}\|^{2}+\\ &\frac{t}{2}\|z^{0}\|_{B}^{2}-\langle\lambda^{N-1}+tAx^{N-1}+tBz^{N-2},Ax^{N-1}-Ax^{N}\rangle+\frac{t(N-2)}{2}\|x^{N}\|_{A}^{2}+\\ &\langle\lambda^{0}+tAx^{1}+tBz^{0},Ax^{1}-Ax^{N}\rangle-t\langle Ax^{1}-Ax^{2}+NAx^{N-1}+Bz^{N-1},Ax^{N}\rangle\\ &+\frac{1}{2}\sum_{k=2}^{N-2}\left((4k-1)t-2\left(2k^{2}-3k+1\right)c_{1}\right)\|x^{k}\|_{A}^{2}+t\langle Ax^{N-1},Bz^{N-1}\rangle+\\ &\sum_{k=2}^{N-2}\left(\left(2k^{2}-k-1\right)c_{1}-2kt\right)\langle Ax^{k},Ax^{k+1}\rangle+t(N-1)\langle Bz^{N-2},Ax^{N-1}-Ax^{N}\rangle\\ &+\frac{1}{2}\left((4N-3)t-\left(2N^{2}-9N+10\right)c_{1}\right)\|x^{N-1}\|_{A}^{2}-t\|x^{2}\|_{A}^{2}-\frac{c_{1}}{2}\|x^{1}\|_{A}^{2}-\\ &\frac{t(N-2)^{2}}{2}\|z^{N-1}-z^{N-2}\|_{B}^{2}-\frac{t(N-1)^{2}}{2}\|Ax^{N-1}+Bz^{N-1}\|^{2}-t\sum_{k=3}^{N-1}\langle Ax^{k},Ax^{N}\rangle+\\ &f(x^{1})-f(x^{N-1})+(N-1)(f(x^{N-1})-f^{\star}+g(x^{N-1})-g^{\star})\geq 0. \end{split}$$

By using (2.5) and (9.7), we have

$$(N^{2} - 3N + 2) \Big(f(x^{N-1}) - f(x^{N}) + \langle \lambda^{N-1} + tAx^{N} + tBz^{N-1}, A(x^{N-1} - x^{N}) \rangle - \frac{c_{1}}{2} \|x^{N} - x^{N-1}\|_{A}^{2} \Big) + \Big(f(x^{N}) - f(x^{1}) + \langle \lambda^{0} + tAx^{1} + tBz^{0}, A(x^{N} - x^{1}) \rangle - \frac{c_{1}}{2} \|x^{N} - x^{1}\|_{A}^{2} \Big) + N(N - 1) \Big(g(z^{N}) - g(z^{N-1}) + \langle \lambda^{N-1}, B(z^{N} - z^{N-1}) \rangle \Big) +$$
(9.26)

$$(N^{2} - 3N + 1) \Big(f(x^{N}) - f(x^{N-1}) + \langle \lambda^{N-1} - tBz^{N-1} + tBz^{N-2}, A(x^{N} - x^{N-1}) \rangle - \frac{c_{1}}{2} \|x^{N} - x^{N-1}\|_{A}^{2} \Big) + (N - 1) \Big(g^{\star} - g(z^{N}) - \langle \lambda^{N-1} + tAx^{N} + tBz^{N}, Bz^{N} \rangle \Big) +$$
(N - 1)
$$\Big(f^{\star} - f(x^{N-1}) - \langle \lambda^{N-1} - tBz^{N-1} + tBz^{N-2}, Ax^{N-1} \rangle - \frac{c_{1}}{2} \|x^{N-1}\|_{A}^{2} \Big) + (N - 1)^{2} \Big(g(z^{N-1}) - g(z^{N}) + \langle \lambda^{N-1} + tAx^{N} + Bz^{N}, B(z^{N-1} - z^{N}) \rangle \Big) \ge 0.$$

By summing (9.25) and (9.26), we obtain

$$\frac{1}{2t} \left\| \lambda^{0} - \lambda^{\star} \right\|^{2} + \frac{t}{2} \left\| z^{0} \right\|_{B}^{2} - \frac{(N^{2} - 1)t}{2} \left\| \frac{N}{N + 1} A x^{N} + B z^{N} \right\|^{2} - \frac{t(N - 1)^{2}}{2} \left\| z^{N} - z^{N - 1} \right\|_{B}^{2} - \frac{(N - 2)^{2}t}{2} \left\| B z^{N - 2} - B z^{N - 1} + \frac{N - 1}{N - 2} A x^{N - 1} - \left(1 - \frac{1}{(N - 2)^{2}} \right) A x^{N} \right\|^{2} - \frac{1}{2} \operatorname{tr} \left(F(t, c_{1}) \left(A x^{1} \dots A x^{N} \right)^{T} \left(A x^{1} \dots A x^{N} \right) \right) \geq 0,$$

where the matrix $F(t, c_1)$ is as follows,

and

$$\begin{split} \alpha_k &= \begin{cases} 6c_1 - 5t, & k = 2\\ 2\left(2k^2 - 3k + 1\right)c_1 - \left(4k - 1\right)t, & 3 \le k \le N - 1, \\ \left(2N^2 - 6N + 4\right)c_1 - 2\left(N + \frac{1}{(N-2)^2} - \frac{2}{N+1} - 3\right)t, & k = N, \end{cases} \\ \beta_k &= \begin{cases} 2kt - \left(2k^2 - k - 1\right)c_1, & 2 \le k \le N - 2, \\ \left(N + \frac{1}{2-N} - 1\right)t - \left(2N^2 - 6N + 3\right)c_1, & k = N - 1, \end{cases} \end{split}$$

The rest of the proof proceeds analogously to the proof of Theorem 9.6. \Box

The following example shows the tightness of this bound.

Example 9.9. Assume that $c_1 > 0$, $N \ge 4$ and $t \in (0, c_1]$ are given, and $f, g : \mathbb{R} \to \mathbb{R}$ are defined by,

$$f(x) = \frac{1}{2} \max\left\{-\frac{N+1}{N-1}x, x\right\} + \frac{c_1}{2}x^2,$$

$$g(z) = \frac{1}{2} \max\left\{\frac{1}{t(N-1)} - z, \frac{N-3}{N-1}\left(z - \frac{1}{t(N-1)}\right)\right\}.$$

Consider the optimization problem

$$\min_{(x,z)\in\mathbb{R}\times\mathbb{R}} f(x) + g(z),$$

s.t. $Ax + Bz = 0.$

where A = B = I. The point $(x^*, z^*) = (0, 0)$ with Lagrangian multiplier $\lambda^* = \frac{1}{2}$ is an optimal solution. After performing N iterations of Algorithm 9.1 with setting $\lambda^0 = \frac{-1}{2}$ and $z^0 = 0$, we have

$$\begin{aligned} x^{k} &= 0, \qquad k \in \{1, \dots, N\}, \\ z^{k} &= \begin{cases} \frac{1}{t(N-1)}, & k \in \{1, \dots, N-1\}, \\ 0, & k = N, \end{cases} \\ \lambda^{k} &= \begin{cases} \frac{2k+1-N}{2(N-1)}, & k \in \{1, \dots, N-1\}, \\ \frac{1}{2}, & k = N. \end{cases} \end{aligned}$$

It can be seen that $\left\|A^T B\left(z^N - z^{N-1}\right)\right\| = \frac{1}{(N-1)t} = \frac{\sqrt{\|\lambda^0 - \lambda^\star\|^2 + t^2\|z^0 - z^\star\|_B^2}}{(N-1)t}$, which shows that the bound is tight.

Theorem 9.4 and 9.6 address the case that f is strongly convex relative to $\|.\|_A$ and g is convex. Based on numerical results by solving performance estimation problems including (9.13) we conjecture, under the assumptions of Theorem 9.4, that if g is c_2 -strongly convex relative to $\|.\|_B$, Algorithm 9.1 enjoys the following convergence rates

$$D(\lambda^{\star}) - D(\lambda^{N}) \leq \frac{\|\lambda^{0} - \lambda^{\star}\|^{2} + t^{2} \|z^{0} - z^{\star}\|_{B}^{2}}{4Nt + \frac{2c_{1}c_{2}}{c_{1}+c_{2}}},$$
$$\|Ax^{N} + Bz^{N} - b\| \leq \frac{\sqrt{\|\lambda^{0} - \lambda^{\star}\|^{2} + t^{2} \|z^{0} - z^{\star}\|_{B}^{2}}}{Nt + \frac{c_{1}c_{2}}{c_{1}+c_{2}}}.$$

We have verified these conjectures numerically for many specific values of the parameters. Nevertheless, we could not manage to guess a closed-form formula for the residual dual in this case.

9.4 Linear convergence of ADMM

In this section we study the linear convergence of ADMM. The linear convergence of ADMM has been addressed by some authors and some conditions for linear convergence have been proposed, see [DY16, LYZZ18, Han22, HSZ18, HL17, NLR⁺15, YZZ20]. Two common types of assumptions employed for proving the linear convergence of ADMM are error bound property and *L*-smoothness. To the best knowledge of authors, most scholars investigated the linear convergence of the sequence $\{(x^k, z^k, \lambda^k)\}$ to a saddle point and there is no result in terms of dual objective value for ADMM. In line with the previous section, we study the linear convergence in terms of dual objective value and we derive some formulas for linear convergence rate by using performance estimation. It is noteworthy to mention that the term "Q-linear convergence" is also employed to describe the linear convergence in the literature.

As mentioned earlier, the *error bound property* is used by scholars for establishing the linear convergence; see e.g. [LYZZ18, HSZ18, HL17, PVZ21, YZZ20]. Let

$$D^{a}(\lambda) := \min f(x) + g(z) + \langle \lambda, Ax + Bz - b \rangle + \frac{a}{2} ||Ax + Bz - b||^{2}, \qquad (9.27)$$

where stands for augmented dual objective for the given a > 0 and Λ^* denotes the optimal solution set of the dual problem. Note that the function D^a is an $\frac{1}{a}$ smooth function on its domain without assuming strong convexity; see [HL17, Lemma 2.2].

Definition 9.10. The function D^a is said to satisfy the error bound property if we have

$$d_{\Lambda^*}(\lambda) \le \tau \|\nabla D^a(\lambda)\|, \ \lambda \in \mathbb{R}^r, \tag{9.28}$$

for some $\tau > 0$.

Hong et al. [HL17] established the linear convergence by employing the *error bound* property (9.28).

Recently, some scholars established the linear convergence of gradient methods for *L*-smooth convex functions by replacing strong convexity with some mild conditions, see [NNG19, AdKZ23a, BNPS17] and references therein. Inspired by these results, we prove the linear convergence of ADMM by using the so-called PŁ inequality. It is worth noting that we employ the nonsmooth version of the PŁ inequality introduced in [BDL07]. Concerning differentiability of dual objective, by (9.5), we have

$$b - A\partial f^*(-A^T\lambda) - B\partial g^*(-B^T\lambda) \subseteq \partial (-D(\lambda)).$$
(9.29)

Note that the inclusion in (9.29) holds as an equality under some mild conditions, see e.g. [Bec17, Chapter 3].

Definition 9.11. The function *D* is said to satisfy the PL inequality if there exists an $L_p > 0$ such that for any $\lambda \in \mathbb{R}^r$ we have

$$D(\lambda^{\star}) - D(\lambda) \le \frac{1}{2L_p} \|\xi\|^2, \quad \xi \in \partial \left(-D(\lambda)\right).$$
(9.30)

Note that if f and g are strongly convex, then -D is an L-smooth convex function with $L \leq \frac{\lambda_{\max}(A^TA)}{\mu_1} + \frac{\lambda_{\max}(B^TB)}{\mu_2}$. In this setting, we have $L_p \leq \frac{\lambda_{\max}(A^TA)}{\mu_1} + \frac{\lambda_{\max}(B^TB)}{\mu_2}$. This follows from the duality between smoothness and strong convexity and

$$\begin{aligned} \|\nabla D(\lambda) - \nabla D(\nu)\| &\leq \left\|\nabla f^*(-A^T\lambda) - \nabla f^*(-A^T\nu)\right\|_A + \left\|\nabla g^*(-B^T\lambda) - \nabla g^*(-B^T\nu)\right\|_B \\ &\leq \frac{1}{\mu_1} \left\|A^T\lambda - A^T\nu\right\|_A + \frac{1}{\mu_2} \left\|B^T\lambda - B^T\nu\right\|_B \leq \left(\frac{\lambda_{\max}(A^TA)}{\mu_1} + \frac{\lambda_{\max}(B^TB)}{\mu_2}\right) \|\lambda - \nu\|.\end{aligned}$$

In the next proposition, we show that definitions (9.28) and (9.30) are equivalent.

Proposition 9.12. Let $L_a = \frac{1}{a}$ denote the Lipschitz constant of ∇D^a , where D^a is given in (9.27). Suppose that (9.29) holds as equality.

- i) If D^a satisfies the error bound (9.28), then D satisfies the PL inequality (9.30) with $L_p = \frac{1}{L_a \tau^2}$.
- ii) If D satisfies the PL inequality (9.30), then D^a satisfies the error bound (9.28) with $\tau = \frac{L_p}{1+aL_p}$.

Proof. First we prove *i*). Suppose $\lambda \in \mathbb{R}^r$ and $\xi \in b - A\partial f^*(-A^T\lambda) - B\partial g^*(-B^T\lambda)$. By the identity (9.4), we have $\xi = b - A\bar{x} - B\bar{z}$ for some $(\bar{x}, \bar{z}) \in \operatorname{argmin} f(x) + g(z) + \langle \lambda, Ax + Bz - b \rangle$. Due to the smoothness of D^a and (9.28), we get

$$D^{a}(\lambda^{\star}) - D^{a}(\nu) \leq \frac{L_{a}\tau^{2}}{2} \|\nabla D^{a}(\nu)\|^{2}, \quad \nu \in \mathbb{R}^{r},$$
 (9.31)

where $\lambda^* \in \Lambda^*$ with $d_{\Lambda^*} = ||v - \lambda^*||$. Suppose that $\bar{v} = \lambda - a(A\bar{x} + B\bar{z} - b)$. As we assume strong duality, we have $D^a(\lambda^*) = D(\lambda^*)$. By the definitions of \bar{x}, \bar{y} , we get

 $(\bar{x},\bar{z}) \in \operatorname{argmin} f(x) + g(z) + \langle \bar{\nu}, Ax + Bz - b \rangle + \frac{a}{2} ||Ax + Bz - b||^2.$

By [HL17, Lemma 2.1], we have $\nabla D^a(\bar{\nu}) = A\bar{x} + B\bar{z} - b$. This equality with (9.31) imply

$$D(\lambda^{\star}) - D(\lambda) \le D^{a}(\lambda^{\star}) - D^{a}(\bar{\nu}) \le \frac{L_{a}\tau^{2}}{2} \|A\bar{x} + B\bar{z} - b\|^{2},$$

and the proof of i) is complete.

Now we establish *ii*). Let λ be in the domain of ∇D^a . By [HL17, Lemma 2.1], we have $\nabla D^a(\lambda) = A\bar{x} + B\bar{z} - b$ for some $(\bar{x}, \bar{z}) \in \operatorname{argmin} f(x) + g(z) + \langle \lambda, Ax + Bz - b \rangle + \frac{a}{2} ||Ax + Bz - b||^2$, which implies that

$$0 \in \partial f(\bar{x}) + A^T \left(\lambda + a(A\bar{x} + B\bar{z} - b)\right), 0 \in \partial g(\bar{z}) + B^T \left(\lambda + a(A\bar{x} + B\bar{z} - b)\right).$$
(9.32)

Supposing $v = \lambda + a(A\bar{x} + B\bar{z} - b)$. By (9.32), one can infer that $D(v) = f(\bar{x}) + g(\bar{z}) + \langle v, A\bar{x} + B\bar{z} - b \rangle$. In addition, (9.4) implies that $b - A\bar{x} - B\bar{z} \in b - A\partial f^*(-A^T v) - B\partial g^*(-B^T v)$. By the PL inequality, we have

$$\frac{1}{2L_p} \|A\bar{x} + B\bar{z} - b\|^2 \ge D(\lambda^*) - D(\nu) = D^a(\lambda^*) - D^a(\lambda) - \frac{a}{2} \|A\bar{x} + B\bar{z} - b\|^2,$$

where the equality follows from $D(\nu) = D^a(\lambda) + \frac{a}{2} ||A\bar{x} + B\bar{z} - b||^2$ and $D^a(\lambda^*) = D(\lambda^*)$. Hence,

$$D^{a}(\lambda^{\star}) - D^{a}(\lambda) \leq \left(\frac{1}{2L_{p}} + \frac{a}{2}\right) \|\nabla D^{a}(\lambda)\|^{2}.$$

This inequality says that D^a satisfies the PL inequality. On the other hand, the PL inequality implies the error bound with the same constant, see [BNPS17], and the proof is complete.

In what follows, we employ performance estimation to derive a linear convergence rate for ADMM in terms of dual objective when the PŁ inequality holds. To this end, we compare the value of dual problem in two consecutive iterations, that is, $\frac{D(\lambda^*)-D(\lambda^2)}{D(\lambda^*)-D(\lambda^1)}$. The following optimization problem gives the worst-case convergence rate,

$$\max \frac{D(\lambda^{*})-D(\lambda^{2})}{D(\lambda^{*})-D(\lambda^{1})}$$
s.t. $\{x^{2}, z^{2}, \lambda^{2}\}$ is generated by Algorithm 9.1 w.r.t. $f, g, A, B, b, \lambda^{1}, z^{1}$ (9.33)
 (x^{*}, z^{*}) is an optimal solution and its Lagrangian multipliers is λ^{*}
 D satisfies the PL inequality
 $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n}), g \in \mathcal{F}^{B}_{c_{2},\infty}(\mathbb{R}^{n})$
 $\lambda^{1} \in \mathbb{R}^{r}, z^{1} \in \mathbb{R}^{m}, A \in \mathbb{R}^{r \times n}, B \in \mathbb{R}^{r \times m}, b \in \mathbb{R}^{r}.$

Analogous to our discussion in Section 9.2, we may assume without loss of generality b = 0, $\lambda^1 = \begin{pmatrix} A & B \end{pmatrix} \begin{pmatrix} x^{\dagger} \\ z^{\dagger} \end{pmatrix}$ and $\lambda^* = \begin{pmatrix} A & B \end{pmatrix} \begin{pmatrix} \bar{x} \\ \bar{z} \end{pmatrix}$ for some $\bar{x}, x^{\dagger}, \bar{z}, z^{\dagger}$. In addition, we assume that $\hat{x}^1 \in \operatorname{argmin} f(x) + \langle \lambda^1, Ax \rangle$ and $\hat{x}^2 \in \operatorname{argmin} f(x) + \langle \lambda^2, Ax \rangle$. Hence,

$$D(\lambda^{1}) = f(\hat{x}^{1}) + g(z^{1}) + \langle \lambda^{1}, A\hat{x}^{1} + Bz^{1} \rangle, \ D(\lambda^{2}) = f(\hat{x}^{2}) + g(z^{2}) + \langle \lambda^{2}, A\hat{x}^{2} + Bz^{2} \rangle,$$

and

$$-A^{T}\lambda^{1} \in \partial f(\hat{x}^{1}), \qquad -B^{T}\lambda^{1} \in \partial g(z^{1}), \qquad (9.34)$$
$$-A^{T}\lambda^{2} \in \partial f(\hat{x}^{2}), \qquad -B^{T}\lambda^{2} \in \partial g(z^{2}).$$

Moreover, by (9.34) and (9.29), we get

$$-A\hat{x}^{1}-Bz^{1}\in\partial\left(-D(\lambda^{1})\right), \quad -A\hat{x}^{2}-Bz^{2}\in\partial\left(-D(\lambda^{2})\right).$$

On the other hand, $\lambda^2 = \lambda^1 + tAx^2 + tBz^2$. Therefore, by using Theorem 2.40, problem (9.33) may be relaxed as follows,

$$\max \frac{f^{*} + g^{*} - \hat{f}^{2} - g^{2} - \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{2}, A\hat{x}^{2} + Bz^{2} \rangle}{f^{*} + g^{*} - \hat{f}^{1} - g^{1} - \langle Ax^{\dagger} + Bz^{\dagger}, A\hat{x}^{1} + Bz^{1} \rangle}$$
s.t. $\left\{ (\hat{x}^{1}, -A^{T}Ax^{\dagger} - A^{T}Bz^{\dagger}, \hat{f}^{1}), (x^{2}, -A^{T}Ax^{\dagger} - A^{T}Bz^{\dagger} - tA^{T}Ax^{2} - tA^{T}Bz^{1}, f^{2}), (\hat{x}^{2}, -A^{T}Ax^{\dagger} - A^{T}Bz^{\dagger} - tA^{T}Ax^{2} - tA^{T}Bz^{2}, \hat{f}^{2}), (0, -A^{T}A\bar{x} - A^{T}B\bar{z}, f^{*}) \right\}$
satisfy interpolation constraints (2.5)
 $\left\{ (z^{1}, -B^{T}Ax^{\dagger} - B^{T}Bz^{\dagger}, g^{1}), (z^{2}, -B^{T}Ax^{\dagger} - B^{T}Bz^{\dagger} - tB^{T}Ax^{2} - tB^{T}Bz^{2}, g^{2}), (0, -B^{T}A\bar{z} - B^{T}B\bar{z}, g^{*}) \right\}$ satisfy interpolation constraints (2.5)
 $f^{*} + g^{*} - \hat{f}^{1} - g^{1} - \langle Ax^{\dagger} + Bz^{\dagger}, A\hat{x}^{1} + Bz^{1} \rangle \leq \frac{1}{2L_{p}} \left\| A\hat{x}^{1} + Bz^{1} \right\|^{2}$ (9.35)
 $f^{*} + g^{*} - \hat{f}^{2} - g^{2} - \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{2}, A\hat{x}^{2} + Bz^{2} \rangle \leq \frac{1}{2L_{p}} \left\| A\hat{x}^{2} + Bz^{2} \right\|^{2}$
 $A \in \mathbb{R}^{r \times n}, B \in \mathbb{R}^{r \times m}.$

By deriving an upper bound for the optimal value of problem (9.35) in the next theorem, we establish the linear convergence of ADMM in the presence of the PŁ inequality.

Theorem 9.13. Let $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$ and $g \in \mathcal{F}^{B}_{c_{2},\infty}(\mathbb{R}^{m})$ with $c_{1}, c_{2} > 0$, and let D satisfies the PL inequality with L_{p} . Suppose that $t \leq \sqrt{c_{1}c_{2}}$.

(i) If $c_1 \ge c_2$, then

$$\frac{D(\lambda^*) - D(\lambda^2)}{D(\lambda^*) - D(\lambda^1)} \le \frac{2c_1c_2 - t^2}{2c_1c_2 - t^2 + L_pt(4c_1c_2 - c_2t - 2t^2)},$$
(9.36)

in particular, if $t = \sqrt{c_1 c_2}$,

$$\frac{D(\lambda^{\star}) - D(\lambda^2)}{D(\lambda^{\star}) - D(\lambda^1)} \leq \frac{1}{1 + L_p \left(2\sqrt{c_1 c_2} - c_2\right)}$$

(*ii*) If $c_1 < c_2$, then

$$\frac{D(\lambda^{\star}) - D(\lambda^{2})}{D(\lambda^{\star}) - D(\lambda^{1})} \leq (9.37)$$

$$\frac{4c_{2}^{2} - 2c_{2}\sqrt{c_{1}c_{2}} - t^{2}}{4c_{2}^{2} - 2c_{2}\sqrt{c_{1}c_{2}} - t^{2} + L_{p}t\left(8c_{2}^{2} + 5c_{2}t - 2\sqrt{c_{1}c_{2}}\left(1 + \frac{t}{c_{1}}\right)(2c_{2} + t)\right)}.$$

Proof. The argument is based on weak duality. Indeed, by introducing suitable Lagrangian multipliers, we establish that the given convergence rates are upper

bounds for problem (9.35). First, we prove (*i*). Assume that α denotes the right hand side of inequality (9.36). As $2c_1c_2 - t^2 > 0$ and $4c_1c_2 - c_2t - 2t^2 > 0$, we have $0 < \alpha < 1$. With some algebra, one can show that

$$\begin{split} f^{\star} + g^{\star} - \hat{f}^{2} - g^{2} - \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{2}, A\hat{x}^{2} + Bz^{2} \rangle - \\ \alpha \left(f^{\star} + g^{\star} - \hat{f}^{1} - g^{1} - \langle Ax^{\dagger} + Bz^{\dagger}, A\hat{x}^{1} + Bz^{1} \rangle \right) + \\ \alpha \left(\hat{f}^{2} - \hat{f}^{1} + \langle Ax^{\dagger} + Bz^{\dagger}, A\hat{x}^{2} - A\hat{x}^{1} \rangle - \frac{c_{1}}{2} \left\| \hat{x}^{2} - \hat{x}^{1} \right\|_{A}^{2} \right) + \\ \alpha \left(f^{2} - \hat{f}^{2} + \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{2}, Ax^{2} - A\hat{x}^{2} \rangle - \frac{c_{1}}{2} \left\| x^{2} - \hat{x}^{2} \right\|_{A}^{2} \right) + \\ \alpha \left(\hat{f}^{2} - f^{2} + \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{1}, A\hat{x}^{2} - Ax^{2} \rangle - \frac{c_{1}}{2} \left\| \hat{x}^{2} - x^{2} \right\|_{A}^{2} \right) + \\ \alpha \left(\hat{f}^{2} - f^{2} + \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{1}, A\hat{x}^{2} - Ax^{2} \rangle - \frac{c_{1}}{2} \left\| \hat{x}^{2} - x^{2} \right\|_{A}^{2} \right) + \\ \alpha \left(g^{2} - g^{1} + \langle Ax^{\dagger} + Bz^{\dagger}, Bz^{2} - Bz^{1} \rangle - \frac{c_{2}}{2} \left\| z^{2} - z^{1} \right\|_{B}^{2} \right) + \\ (1 - \alpha) \left(-f^{\star} - g^{\star} + \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{2}, A\hat{x}^{2} + Bz^{2} \rangle + \hat{f}^{2} + g^{2} + \right) \\ \frac{1}{2L_{p}} \left\| A\hat{x}^{2} + Bz^{2} \right\|^{2} \right) = \frac{-c_{1}\alpha}{2} \left\| \hat{x}^{1} - \hat{x}^{2} \right\|_{A}^{2} - \frac{c_{2}\alpha}{2} \left\| Bz^{1} - Bz^{2} + \frac{t}{c_{2}}Ax^{2} - \frac{t}{c_{2}}A\hat{x}^{2} \right\|^{2} - \\ \alpha (c_{1} - \frac{t^{2}}{2c_{2}}) \left\| Ax^{2} + \frac{tc_{2}}{2c_{1}c_{2}-t^{2}}Bz^{2} - \frac{tc_{2}-2c_{1}c_{2}+t^{2}}{t^{2}-2c_{1}c_{2}}A\hat{x}^{2} \right\|^{2}. \end{split}$$

Hence, we get

$$\begin{split} &f^{\star}+g^{\star}-\hat{f}^2-g^2-\langle Ax^{\dagger}+Bz^{\dagger}+tAx^2+tBz^2,A\hat{x}^2+Bz^2\rangle \leq \\ &\alpha\left(f^{\star}+g^{\star}-\hat{f}^1-g^1-\langle Ax^{\dagger}+Bz^{\dagger},A\hat{x}^1+Bz^1\rangle\right) \end{split}$$

for any feasible point of problem (9.33) and the proof of the first part is complete. For (*ii*), we proceed analogously to the proof of (*i*), but with different Lagrange multipliers. Let β denote the right hand side of inequality (9.37), i.e.

$$\beta = \frac{4c_2^2 - 2c_2\sqrt{c_1c_2} - t^2}{4c_2^2 - 2c_2\sqrt{c_1c_2} - t^2 + L_p t \left(8c_2^2 + 5c_2t - 2\sqrt{c_1c_2}\left(1 + \frac{t}{c_1}\right)(2c_2 + t)\right)}$$

It is seen that $0 < \beta < 1$. By doing some calculations, we have

$$\begin{split} f^{*} + g^{*} - \hat{f}^{2} - g^{2} - \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{2}, A\hat{x}^{2} + Bz^{2} \rangle - \\ \beta \left(f^{*} + g^{*} - \hat{f}^{1} - g^{1} - \langle Ax^{\dagger} + Bz^{\dagger}, A\hat{x}^{1} + Bz^{1} \rangle \right) + \\ \beta \left(\hat{f}^{2} - \hat{f}^{1} + \langle Ax^{\dagger} + Bz^{\dagger}, A\hat{x}^{2} - A\hat{x}^{1} \rangle - \frac{c_{1}}{2} \left\| \hat{x}^{2} - \hat{x}^{1} \right\|_{A}^{2} \right) + \\ \sqrt{\frac{c_{2}}{c_{1}}} \beta \left(\hat{f}^{2} - \hat{f}^{2} + \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{2}, Ax^{2} - A\hat{x}^{2} \rangle - \frac{c_{1}}{2} \left\| \hat{x}^{2} - x^{2} \right\|_{A}^{2} \right) + \\ \sqrt{\frac{c_{2}}{c_{1}}} \beta \left(\hat{f}^{2} - f^{2} + \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{1}, A\hat{x}^{2} - Ax^{2} \rangle - \frac{c_{1}}{2} \left\| \hat{x}^{2} - x^{2} \right\|_{A}^{2} \right) + \\ \sqrt{\frac{c_{2}}{c_{1}}} \beta \left(g^{2} - g^{1} + \langle Ax^{\dagger} + Bz^{\dagger}, Bz^{2} - Bz^{1} \rangle - \frac{c_{2}}{2} \left\| z^{2} - z^{1} \right\|_{B}^{2} \right) + \\ \left(\sqrt{\frac{c_{2}}{c_{1}}} - 1 \right) \beta \left(g^{1} - g^{2} + \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{2}, Bz^{1} - Bz^{2} \rangle - \\ \frac{c_{2}}{2} \left\| z^{1} - z^{2} \right\|_{B}^{2} \right) + (1 - \beta) \left(-f^{*} - g^{*} + \langle Ax^{\dagger} + Bz^{\dagger} + tAx^{2} + tBz^{2}, A\hat{x}^{2} + Bz^{2} \rangle + \\ \hat{f}^{2} + g^{2} + \frac{1}{2L_{p}} \left\| A\hat{x}^{2} + Bz^{2} \right\|^{2} \right) \\ = -\frac{c_{1}\beta}{2} \left\| \hat{x}^{1} - \hat{x}^{2} \right\|_{A}^{2} - \left(\sqrt{c_{1}c_{2}}\beta \right) \right\| Ax^{2} - \left(\frac{1 - \frac{t}{2\sqrt{c_{1}c_{2}}} \right) A\hat{x}^{2} + \frac{t}{2\sqrt{c_{1}c_{2}}} Bz^{1} \right\|^{2} - \\ \left(\frac{\beta - 1}{2L_{p}} + \beta t \left(1 - \frac{t}{4\sqrt{c_{1}c_{2}}} \right) \right) \left\| A\hat{x}^{2} - \left(\frac{\beta L_{p} \left(-2c_{2}\sqrt{c_{1}c_{2}} + 4c_{2}^{2} - t^{2} \right)}{-\beta L_{p}t^{2} + 2\sqrt{c_{1}c_{2}} \left(2\beta L_{p} t + \beta - 1 \right)} \right)^{\frac{1}{2}} Bz^{1} + \\ \left(\frac{2 \left(2\beta c_{2}L_{p} \left(t + c_{2} \right) + \sqrt{c_{1}c_{2}} \left(\beta - \beta L_{p}c_{2} - 1 \right) \right)}{-\beta L_{p}t^{2} + 2\sqrt{c_{1}c_{2}} \left(2\beta L_{p} t + \beta - 1 \right)} \right)^{\frac{1}{2}} Bz^{2} \right\|^{2}. \end{split}$$

The rest of the proof is similar to that of the former case.

We computed the bounds in Theorem 9.13 by selecting suitable Lagrangian multipliers and solving the semidefinite formulation of problem (9.35) by hand. The semidefinite formulation is formed analogously to problem (9.14). Note that the optimal value of problem (9.35) may be smaller than the bounds introduced in Theorem 9.13. Indeed, our aim was to provide a concrete mathematical proof for the linear convergence rate. However, the linear convergence rate factor is not necessarily tight. Needless to say that the optimal value of problem (9.35) also does not necessarily give the tight convergence factor as it is just a relaxation of problem (9.33).

Recently the authors showed that the PŁ inequality is necessary and sufficient conditions for the linear convergence of the gradient method with constant step lengths for *L*-smooth function; see[AdKZ23a, Theorem 5]. In what follows, we establish that the PŁ inequality is a necessary condition for the linear convergence of ADMM. Firstly, we present a lemma that is very useful for our proof.

Lemma 9.14. Let $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$ and $g \in \mathcal{F}^{B}_{c_{2},\infty}(\mathbb{R}^{m})$. Consider Algorithm 9.1. If $(\hat{x}^{1}, z^{1}) \in \operatorname{argmin} f(x) + g(z) + \langle \lambda^{1}, Ax + Bz - b \rangle$, then

$$\langle A\hat{x}^{1} + Bz^{1} - b, Ax^{2} + Bz^{2} - b \rangle \le \left\| A\hat{x}^{1} + Bz^{1} - b \right\|^{2}.$$
 (9.38)

Proof. Without loss of generality we assume that $c_1 = c_2 = 0$. By optimality conditions, we have

$$\begin{split} f(\hat{x}^{1}) - \langle \lambda^{1}, Ax^{2} - A\hat{x}^{1} \rangle &\leq f(x^{2}), \quad g(z^{1}) - \langle \lambda^{1}, Bz^{2} - Bz^{1} \rangle \leq g(z^{2}), \\ f(x^{2}) - \langle \lambda^{1} + t(Ax^{2} + Bz^{1} - b), A\hat{x}^{1} - Ax^{2} \rangle &\leq f(\hat{x}^{1}), \\ g(z^{2}) - \langle \lambda^{1} + t(Ax^{2} + Bz^{2} - b), Bz^{1} - Bz^{2} \rangle \leq g(z^{1}). \end{split}$$

By using these inequities, we get

$$\begin{split} &0 \leq \frac{1}{t} \left(f\left(x^{2}\right) - f\left(\hat{x}^{1}\right) + \left\langle \lambda^{1}, Ax^{2} - A\hat{x}^{1} \right\rangle \right) + \frac{1}{t} \left(g(z^{2}) - g(z^{1}) + \left\langle \lambda^{1}, Bz^{2} - Bz^{1} \right\rangle \right) + \\ & \frac{1}{t} \left(f\left(\hat{x}^{1}\right) - f\left(x^{2}\right) + \left\langle \lambda^{1} + t\left(Ax^{2} + Bz^{1} - b\right), A\hat{x}^{1} - Ax^{2} \right\rangle \right) + \\ & \frac{1}{t} \left(g(z^{1}) - g(z^{2}) + \left\langle \lambda^{1} + t\left(Ax^{2} + Bz^{2} - b\right), Bz^{1} - Bz^{2} \right\rangle \right) \\ &= \left\| A\hat{x}^{1} + Bz^{1} - b \right\|^{2} - \left\langle A\hat{x}^{1} + Bz^{1} - b, Ax^{2} + Bz^{2} - b \right\rangle - \frac{3}{4} \left\| B\left(z^{1} - z^{2}\right) \right\|^{2} - \\ & \left\| A\left(\hat{x}^{1} - x^{2}\right) + \frac{1}{2}B\left(z^{1} - z^{2}\right) \right\|^{2}. \end{split}$$

Hence, we have

$$\frac{\langle A \hat{x}^1 + B z^1 - b, A x^2 + B z^2 - b \rangle}{\|A \hat{x}^1 + B z^1 - b\|^2} \le 1,$$

which completes the proof.

The next theorem establishes that the PŁ inequality is a necessary condition for the linear convergence of ADMM.

Theorem 9.15. Let $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$ and $g \in \mathcal{F}^{B}_{c_{2},\infty}(\mathbb{R}^{m})$. If Algorithm 9.1 is linearly convergent with respect to the dual objective value, then D satisfies the PL inequality.

Proof. Consider $\lambda^1 \in \mathbb{R}^r$ and $\xi \in b - A\partial f^*(-A^T\lambda^1) - B\partial g^*(-B^T\lambda^1)$. Hence, $\xi = b - A\hat{x}^1 - Bz^1$ for some $(\hat{x}^1, z^1) \in \operatorname{argmin} f(x) + g(z) + \langle \lambda, Ax + Bz - b \rangle$. If one sets $z^0 = z^1$ and $\lambda^0 = \lambda^1 - t(A\hat{x}^1 + Bz^1 - b)$ in Algorithm 9.1, the algorithm may generate λ^1 . As Algorithm 9.1 is linearly convergent, there exist $\gamma \in [0, 1)$ with

$$D(\lambda^*) - D(\lambda^2) \leq \gamma \left(D(\lambda^*) - D(\lambda^1) \right).$$

So, we have

$$(1-\gamma)\left(D(\lambda^{\star})-D(\lambda^{1})\right) \leq D(\lambda^{2})-D(\lambda^{1}) \leq \left\langle A\hat{x}^{1}+Bz^{1}-b,\lambda^{2}-\lambda^{1}\right\rangle,$$

where the last inequality follows from the concavity of the function *D*. Since $\lambda^2 - \lambda^1 = t(Ax^2 + Bz^2 - b)$, Lemma 9.14 implies that

$$D(\lambda^*) - D(\lambda^1) \leq \frac{t}{1-\gamma} \|\xi\|^2,$$

so D satisfies the PŁ inequality.

Another assumption used in the literature for establishing linear convergence is *L*-smoothness; see for example [NLR⁺15, DY16, GB16, DY17]. Deng et al. [DY16] show that the sequence $\{(x^k, z^k, \lambda^k)\}$ is convergent linearly to a saddle point under Scenario 1 and 2 given in Table 9.1.

Table 9.1: Scenarios leading to linear convergence rates

Scenario	Strong convexity	Lipschitz continuity	Full row rank
1	<i>f</i> , g	∇f	Α
2	<i>f</i> , <i>g</i>	$\nabla f, \nabla g$	-
3	f	$\nabla f, \nabla g$	B^T

It is worth mentioning that Scenario 1 or Scenario 2 implies strong convexity of the dual objective function and therefore the PŁ inequality is implied, see [AdKZ23a]. Hence, Theorem 9.13 implies the linear convergence in terms of dual value under Scenario 1 or Scenario 2. Deng et al. [DY16] studied the linear convergence under Scenario 3, but they just proved the linear convergence of the sequence { (x^k, Bz^k, λ^k) }. In the next section, we investigate the R-linear convergence without assuming *L*-smoothness of *f*. Indeed, we establish the R-linear convergence when *f* is strongly convex, *g* is *L*-smooth and *B* has full row rank.

Note that the PŁ inequality does not imply necessarily Scenario 1 or Scenario 2. Indeed, consider the following optimization problem,

min
$$f(x) + g(z)$$
,
s.t. $x + z = 0$,
 $x, z \in \mathbb{R}^n$,

where $f(x) = \frac{1}{2} ||x||^2 + ||x||_1$ and $g(z) = \frac{1}{2} ||z||^2 + ||z||_1$. With some algebra, one may show that $D(\lambda) = \sum_{i=1}^n h(\lambda_i)$ with

$$h(s) = \begin{cases} -(s-1)^2, & s > 1\\ 0, & |s| \le 1\\ -(s+1)^2, & s < -1 \end{cases}$$

 \square

Hence, the PŁ inequality holds for $L_p = \frac{1}{2}$ while neither *f* nor *g* is *L*-smooth.

As mentioned earlier the performance estimation problem including the PŁ inequality at finite set of points is a relaxation for computing the worst-case convergence rate. Contrary to Theorem 9.13, we could not manage to prove the linear convergence of primal and dual residuals under the assumptions of Theorem 9.13 by employing performance estimation.

9.5 *R*-linear convergence of ADMM

This section focuses on examining the linear convergence rate for ADMM from a weaker convergence rate perspective than *Q*-linear which is already studied in Section 9.4. This concept is known as *R*-linear convergence where *R* stands for root [NW06]. Recall that ADMM enjoys *R*-linear convergence in terms of dual objective value if there exists sequence $\{s_k\} \subseteq \mathbb{R}_+$ such that

$$D(\lambda^{\star}) - D(\lambda^N) \leq s_k,$$

and s_k tends *Q*-linearly to zero. It is easily seen that the linear convergence implies *R*-linear convergence. For an extensive discussion of convergence rates see [NW06, Section A.2] or [BGLS06, Section 1.5] and Section 1.4.

We investigate the *R*-linear convergence under the following scenarios:

- (S1): $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$ is *L*-smooth with $c_{1} > 0$ and *A* has full row rank;
- (S2): $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$ with $c_{1} > 0$, g is *L*-smooth and *B* has full row rank.

Under these scenarios, we could not manage to find a value of q within the range [0, 1) that satisfies the inequality:

$$D(\lambda^{\star}) - D(\lambda^{N+1}) \le q \left(D(\lambda^{\star}) - D(\lambda^{N}) \right).$$

As a result, we turn our attention towards studying the *R*-linear convergence.

Our technique for proving the *R*-linear convergence is based on establishing the linear convergence of the sequence $\{V^k\}$ given by

$$V^{k} = \|\lambda^{k} - \lambda^{\star}\|^{2} + t^{2} \|z^{k} - z^{\star}\|_{B}^{2}.$$
(9.39)

Note that V^k is called *Lyapunov function* for ADMM and it decreases in each iteration; see [BPC⁺11]. It is worth noting *Q*-linear and *R*-linear convergence of

ADMM have been studied under similar scenarios for some performance measures, see e.g. [DY17, GB16, NLR⁺15]. However, to the best of knowledge, no existing results in the literature address the dual objective and V^k under Scenario (S1) and (S2).

First we consider the case that f is *L*-smooth and c_1 -strongly convex relative to *A*. The following proposition establishes the linear convergence of $\{V^k\}$.

Proposition 9.16. Let $f \in \mathcal{F}_{c_1,\infty}^A(\mathbb{R}^n)$ be L-smooth with $c_1 > 0$, $g \in \mathcal{F}_{0,\infty}(\mathbb{R}^m)$ and let A has full row rank. If $t < \sqrt{\frac{c_1L}{\lambda_{\min}(AA^T)}}$, then

$$V^{k+1} \le \left(1 - \frac{2c_1 t}{c_1 d + 2c_1 t + t^2}\right) V^k, \tag{9.40}$$

where $d = \frac{L}{\lambda_{\min}(AA^T)}$.

Proof. We may assume without loss of generality that x^*, z^* and *b* are zero; see our discussion in Section 9.2. By optimality conditions, we have

$$\nabla f(x^{k+1}) = -A^T \left(\lambda^k + tAx^{k+1} + tBz^k \right), \qquad \eta^k = -B^T \lambda^{k+1},$$

$$\nabla f(x^*) = -A^T \lambda^*, \qquad \eta^* = -B^T \lambda^*,$$

for some $\eta^k \in \partial g(z^{k+1})$ and $\eta^* \in \partial g(z^*)$. Let $\alpha = \frac{2t}{c_1^2 d^2 + 2c_1 dt^2 - 4c_1^2 t^2 + t^4}$. By Theorem 2.5, we get

$$\begin{split} &\alpha\left(t^{2}+c_{1}d\right)^{2}\left(f(x^{k+1})-f^{*}+\left\langle\lambda^{*},Ax^{k+1}\right\rangle-\frac{1}{2L}\left\|A^{T}\left(\lambda^{k}+tAx^{k+1}+tBz^{k}-\lambda^{*}\right)\right\|^{2}\right)+\\ &2\alpha t^{2}(c_{1}d+t^{2})\left(f^{*}-f(x^{k+1})-\frac{c_{1}}{2}\left\|x^{k+1}\right\|_{A}^{2}-\left\langle\lambda^{k}+tAx^{k+1}+tBz^{k},Ax^{k+1}\right\rangle\right)+\\ &2t\left(g(z^{k+1})-g^{*}+\left\langle\lambda^{*},Bz^{k+1}\right\rangle\right)+2t\left(g^{*}-g(z^{k+1})-\left\langle\lambda^{k+1},Bz^{k+1}\right\rangle\right)+\\ &\alpha\left(c_{1}^{2}d^{2}-t^{4}\right)\left(f^{*}-f(x^{k+1})-\left\langle\lambda^{k}+tAx^{k+1}+tBz^{k},Ax^{k+1}\right\rangle-\\ &\frac{1}{2L}\left\|A^{T}\left(\lambda^{k}+tAx^{k+1}+tBz^{k}-\lambda^{*}\right)\right\|^{2}\right)\geq0. \end{split}$$

As $||A^T \lambda||^2 \ge \frac{L}{d} ||\lambda||^2$ and $\lambda^{k+1} = \lambda^k + tAx^{k+1} + tBz^{k+1}$, we obtain the following inequality after performing some algebraic manipulations

$$(1 - \frac{2ct}{cd + 2ct + t^2}) \Big(\left\| \lambda^k - \lambda^* \right\|^2 + t^2 \left\| Bz^k \right\|^2 \Big) - \Big(\left\| \lambda^{k+1} - \lambda^* \right\|^2 + t^2 \left\| Bz^{k+1} \right\|^2 \Big) - 2\alpha c_1^2 t \left\| \lambda^k - \lambda^* + \frac{t^2 + 2c_1 t + c_1 d}{2c_1} Ax^{k+1} + \frac{t^2 + c_1 d}{2c_1} Bz^k \right\|^2 \ge 0.$$

The above inequality implies that

$$V^{k+1} \le \left(1 - \frac{2c_1t}{c_1d + 2c_1t + t^2}\right)V^k,$$

and the proof is complete.

Note that one can improve bound (9.40) under the assumptions of Proposition 9.16 and the μ -strong convexity of f by employing the following known inequality

$$\frac{1}{2(1-\frac{\mu}{L})} \left(\frac{1}{L} \|\nabla f(x) - \nabla f(y)\|^2 + \mu \|x - y\|^2 - \frac{2\mu}{L} \langle \nabla f(x) - \nabla f(y), x - y \rangle \right)$$

$$\leq f(y) - f(x) - \langle \nabla f(x), y - x \rangle.$$

Indeed, we employed the given inequality but we could not manage to obtain a closed form formula for the convergence rate. The next theorem establishes the *R*-linear convergence of ADMM in terms of dual objective value under the assumptions of Proposition 9.16.

Theorem 9.17. Let $N \ge 4$ and let A has full row rank. Suppose that $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$ is L-smooth with $c_{1} > 0$ and $g \in \mathcal{F}_{0,\infty}(\mathbb{R}^{m})$. If $t < \min\{c_{1}, \sqrt{\frac{c_{1}L}{\lambda_{\min}(AA^{T})}}\}$, then

$$D(\lambda^*) - D(\lambda^N) \leq \rho \left(1 - \frac{2c_1t}{c_1d + 2c_1t + t^2}\right)^N,$$

where $d = \frac{L}{\lambda_{\min}(AA^T)}$ and $\rho = \frac{V^0}{16t} \left(1 - \frac{2c_1 t}{c_1 d + 2c_1 t + t^2} \right)^{-4}$.

Proof. By Theorem 9.4 and Proposition 9.16, one can infer the following inequalities,

$$D(\lambda^{\star}) - D(\lambda^{N}) \leq \frac{V^{N-4}}{16t} \leq \frac{V^{0}}{16t} \left(1 - \frac{2c_{1}t}{c_{1}d + 2c_{1}t + t^{2}}\right)^{N-4},$$

which shows the desired inequality.

In the sequel, we investigate the *R*-linear convergence under the hypotheses of scenario (S2). The next proposition shows the linear convergence of $\{V^k\}$.

Proposition 9.18. Let $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$ with $c_{1} > 0$ and let $g \in \mathcal{F}_{0,\infty}(\mathbb{R}^{m})$ be L-smooth. Suppose that B has full row rank and $k \geq 1$. If $t \leq \min\{\frac{c_{1}}{2}, \frac{L}{2\lambda_{\min}(BB^{T})}\}$, then

$$V^{k+1} \le \left(\frac{L}{L+t\lambda_{\min}(BB^T)}\right)^2 V^k.$$
(9.41)

Proof. Analogous to the proof of Proposition 9.16, we assume that $x^* = 0$, $z^* = 0$ and b = 0. Due to the optimality conditions, we have

$$\xi^{k+1} = -A^T \left(\lambda^k + tAx^{k+1} + tBz^k \right), \qquad \xi^* = -A^T \lambda^*,$$

$$\nabla g(z^k) = -B^T \lambda^k, \qquad \nabla g(z^{k+1}) = -B^T \lambda^{k+1}, \qquad \nabla g(z^*) = -B^T \lambda^*,$$

_	

for some $\xi^{k+1} \in \partial f(x^{k+1})$ and $\xi^* \in \partial f(x^*)$. Suppose that $d = \frac{L}{\lambda_{\min}(BB^T)}$ and $\alpha = \frac{2dt}{d+t}$. By Theorem 2.5, we obtain

$$\begin{aligned} &\frac{a\left(d^{2}+t^{2}\right)}{d^{2}-t^{2}}\left(f^{*}-f\left(x^{k+1}\right)-\left\langle\lambda^{k}+tAx^{k+1}+tBz^{k},Ax^{k+1}\right\rangle-\frac{c_{1}}{2}\left\|x^{k+1}\right\|_{A}^{2}\right)+\\ &\frac{a\left(d^{2}+t^{2}\right)}{d^{2}-t^{2}}\left(f\left(x^{k+1}\right)-f\left(x^{*}\right)+\left\langle\lambda^{*},Ax^{k+1}\right\rangle-\frac{c_{1}}{2}\left\|x^{k+1}\right\|_{A}^{2}\right)+\\ &\alpha\left(g\left(z^{k+1}\right)-g^{*}+\left\langle\lambda^{*},Bz^{k+1}\right\rangle-\frac{1}{2L}\left\|B^{T}\left(\lambda^{*}-\lambda^{k+1}\right)\right\|^{2}\right)+\\ &\alpha\left(g^{*}-g\left(z^{k+1}\right)-\left\langle\lambda^{k+1},Bz^{k+1}\right\rangle-\frac{1}{2L}\left\|B^{T}\left(\lambda^{*}-\lambda^{k+1}\right)\right\|^{2}\right)+\\ &\alpha\left(g\left(z^{k}\right)-g\left(z^{k+1}\right)+\left\langle\lambda^{k+1},Bz^{k}-Bz^{k+1}\right\rangle-\frac{1}{2L}\left\|B^{T}\left(\lambda^{k+1}-\lambda^{k}\right)\right\|^{2}\right)+\\ &\alpha\left(g\left(z^{k+1}\right)-g\left(z^{k}\right)+\left\langle\lambda^{k},Bz^{k+1}-Bz^{k}\right\rangle-\frac{1}{2L}\left\|B^{T}\left(\lambda^{k+1}-\lambda^{k}\right)\right\|^{2}\right)=0.\end{aligned}$$

By employing $||B^T \lambda||^2 \ge \frac{L}{d} ||\lambda||^2$ and $\lambda^{k+1} = \lambda^k + tAx^{k+1} + tBz^{k+1}$, the aforementioned inequality can be expressed as follows after some algebraic manipulation,

$$\begin{split} & \frac{-a^2}{4} \left\| \left(\frac{2t^2}{d^2 - dt} \right) Ax^{k+1} + Bz^k - \left(1 + \frac{t}{d} \right) Bz^{k+1} \right\|^2 - \frac{2t \left(d^2 + t^2 \right) \left(cd^2 - dt(c+t) - t^3 \right)}{(d^2 - t^2)^2} \\ & \left\| Ax^{k+1} \right\|^2 - \frac{a^2}{4d^2} \left\| \lambda^k - \lambda^\star + \left(\frac{2d^2 - (d-t)^2}{d-t} \right) Ax^{k+1} + (d+t) Bz^{k+1} \right\|^2 \\ & + \left(\frac{d}{d+t} \right)^2 \left(\left\| \lambda^k - \lambda^\star \right\|^2 + t^2 \left\| Bz^k \right\|^2 \right) - \left(\left\| \lambda^{k+1} - \lambda^\star \right\|^2 + t^2 \left\| Bz^{k+1} \right\|^2 \right) \ge 0. \end{split}$$

Hence, we have

$$V^{k+1} \le \left(\frac{d}{d+t}\right)^2 V^k,$$

and the proof is complete.

As the sequence $\{V^k\}$ is not increasing [BPC⁺11, Convergence Proof], we have $V^1 \leq V^0$. Thus, by using Theorem 9.4 and Proposition 9.18, one can infer the following theorem.

Theorem 9.19. Let $f \in \mathcal{F}^{A}_{c_{1},\infty}(\mathbb{R}^{n})$ with $c_{1} > 0$ and let $g \in \mathcal{F}_{0,\infty}(\mathbb{R}^{m})$ be L-smooth. Assume that $N \geq 5$ and B has full row rank. If $t < \min\{\frac{c_{1}}{2}, \frac{L}{2\lambda_{\min}(BB^{T})}\}$, then

$$D(\lambda^*) - D(\lambda^N) \le \rho \left(\frac{L}{L + t\lambda_{\min}(BB^T)}\right)^{2N},$$
 (9.42)

where $\rho = \frac{V^0}{16t} \left(\frac{L}{L + t\lambda_{\min}(BB^T)} \right)^{-10}$.

In the same line, one can infer the R-linear convergence in terms of primal and dual residuals under the assumptions of Theorem 9.17 and Theorem 9.19. In this

section, we proved the linear convergence of $\{V^k\}$ under two scenarios (S1) and (S2). By (9.5), it is readily seen that function -D is strongly convex under the hypotheses of both scenarios (S1) and (S2). Therefore, both scenarios imply the PL inequality. One may wonder that if the PL inequality and the strong convexity of f imply the linear of $\{V^k\}$. By using performance estimation, we could not establish such an implication.

As mentioned above, function -D under both scenarios are μ -strongly convex. Hence, the optimal solution set of the dual problem is unique and one can infer the R-linear convergence of λ^N by using Theorem 9.17 (Theorem 9.19) and the known inequality,

$$\frac{\mu}{2} \left\| \lambda^N - \lambda^* \right\|^2 \le D(\lambda^*) - D(\lambda^N).$$

9.6 Concluding remarks

In this chapter we developed a performance estimation framework to handle dualbased methods. Thanks to this framework, we could obtain some tight convergence rates for ADMM. This framework may be exploited for the analysis of other variants of ADMM in the ergodic and non-ergodic sense. Moreover, similarly to [KF16], one can apply this framework for introducing and analyzing new accelerated ADMM variants. Moreover, most results hold for any arbitrary positive step length, t, but we managed to get closed form formulas for some interval of positive numbers.

It is worth mentioning that $\mathcal{F}_{c_1,\infty}^A = \mathcal{F}_{\tau^{-2}c_1,\infty}^{\tau A}$. However, strong convexity relative to $\|.\|_A$ is assumed throughout the chapter when *A* is the first block of the constraint matrix, which is fixed for a given problem. If one considers strong convexity relative to $\|.\|_{\tau A}$, they need to modify constraints accordingly. In what follows, we investigate two possible cases, and show that these transformations have no influence.

Case 1. Consider the following problem

$$\min f(x) + g(z)$$

s.t. $\tau A(\tau^{-1}x) + Bz = b$.

To be consistent with problem (1), we define new variable $y = \tau^{-1}x$, and the

problem is formulated as

$$\min h(y) + g(z)$$

s.t. $\tau Ay + Bz = b$,

where *h* is given by $h(y) = f(\tau y)$. It is seen that *f* is *c*-strongly convex relative to $\|.\|_A$ if and only if *h* is *c*-strongly convex relative to $\|.\|_{\tau A}$. In addition, we have the same initial condition as $(\tau^{-1}x^*, z^*, \lambda^*)$ is a saddle point for the new problem. Hence, we get same results for both formulations.

Case 2. Consider the following case,

$$\min f(x) + g(z)$$

s.t. $\tau A x + \tau B z = \tau b$.

For this problem $(x^*, z^*, \tau^{-1}\lambda^*)$ is a saddle point and $f \in \mathcal{F}_{\tau^{-2}c_{1,\infty}}^{\tau A}$. Let $\{(x^k, z^k, \lambda^k)\}$ be the generated points via ADMM for solving Problem (9.1) with initial point (z^0, λ^0) and step length t. It is seen that ADMM can generate $\{(x^k, z^k, \tau^{-1}\lambda^k)\}$ for solving the new problem with initial point $(z^0, \tau^{-1}\lambda^0)$ and step length $\tau^{-2}t$. In this case, we also get the same results as

$$\frac{1}{t} \|\lambda^0 - \lambda^\star\|^2 + t \left\| z^0 - z^\star \right\|_B^2 = \frac{1}{t\tau^{-2}} \|\tau^{-1}\lambda^0 - \tau^{-1}\lambda^\star\|^2 + t\tau^{-2} \left\| z^0 - z^\star \right\|_{\tau B}^2$$

Not what we have, but what we enjoy, constitutes our abundance.

Epicurus

10 Conclusion and outlook

The primary focus of this thesis has been the complexity analysis of first-order methods. This line of investigation has far-reaching implications for researchers and practitioners across various fields, particularly in the realm of machine learning, as first-order methods are used extensively to solve the optimization problems that arise in real-world applications. The use of performance estimation methods has allowed us to study the worst-case behavior of some first-order methods.

Using this methodology, we started with one of the most famous first-order methods, the gradient descent method with fixed step length. We studied the complexity of this algorithm over *L*-smooth functions and by means of Proposition 4.4 showed that the presented bound is in fact tight for some step lengths. We show that in this case the bound is tight for step-lengths in the interval $(0, \frac{1}{L}]$. It is known that the gradient descent method is convergent if the step-length lies in $(0, \frac{2}{L})$.

Open problem 10.1. Consider the class of *L*-smooth functions. Given the norm of the gradient as the stopping criterion for the gradient descent method, what is the tight bound when all step-lengths are within the interval $(\frac{1}{L}, \frac{2}{L})$?

We continued to study the gradient method by providing necessary and sufficient conditions so that the algorithm enjoys linear convergence, namely if the function class under the study satisfies PŁ inequality; see Theorem 5.8. As the given bound under the PŁ inequality is not tight, an important question to be answered is the following.

Open problem 10.2. When may a function that satisfies the PL inequality at a given, finite set of points be extended to a function that satisfies the same PL inequality on some open set containing these points?

We also studied the relation between this class of functions with other class of functions; see Section 5.3.

We continued by studying the coordinate descent method, a variant of the gradient descent method. We showed that the worst-case bounds given in the literature are not tight for some class of functions. As the bounds we obtained via SDP are only numerical, see Sections 6.2 and 6.3, it is important to provide a mathematical proof as well as a closed-form formula for the given bounds. We also studied worst case behavior of the non-linear Gauss–Seidel method as well as the related weighed Jacobi method; see Sections 6.3.1 and 6.3.2, respectively. Another important question might be to find interpolation constraints to find a better bound for the randomized coordinate descent method. These constraints not only contribute to refining the bounds for the randomized coordinate method but also pave the way for effectively addressing other stochastic methods, such as the widely used stochastic gradient descent method in the machine learning community, and answers several open questions in this area.

Proceeding with the examination of gradient-based algorithms, we studied the gradient descent-ascent method which is mainly used to find saddle points of minimax problems. In this study, we presented a tight convergence rate for one iteration of the algorithm for some class of functions as well as an optimal step length based on the given bound; see Section 7.2. Moreover, we studied necessary and sufficient conditions that the algorithm enjoys linear convergence under those conditions; see Section 7.3. As it is mentioned, the given bound is tight only for one iteration, this might be because of the lack of an interpolation constraint for minimax problems.

Open problem 10.3. What are the condition(s) that a saddle-point function F(x, y), with properties provided in Section 7.1, must satisfy for a finite set of given points to be extended to a function with the same properties on some open set containing these points? In other words, what is the interpolation constraint for this function class?

In Chapter 8, we delved into the Difference of Convex Functions Algorithm

(DCA). A large class of functions may be written as the difference of two convex functions. In our study, we provided some worst-case convergence rate for some class of functions. We provided the worst-case convergence rate for the case that at least one of the two convex functions is *L*-smooth. We showed that the order of the given bound is tight by providing an example; see Section 8.3.1. Moreover, we studied the convergence of DCA when both functions are non-smooth. Also, we studied the relation between DCA and the gradient descent method in Section 8.3.3, and DCA with the proximal gradient method in Section 8.3.4. In addition, we provided necessary conditions that DCA enjoys linear convergence.

Open problem 10.4. *Is the PL inequality a sufficient condition for the DCA to have linear convergence?*

The convergence of DCA with regularization also is studied in this chapter. As a side result of our analysis, we derived a convergence rate for proximal gradient method. Regarding this algorithm the following important questions arise.

Open problem 10.5. What conditions must a function satisfy in order to belong to the function class $\{f \mid f := f_1 - f_2, f_1 \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n), f_2 \in \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)\}$, for some $0 \le \mu_1 < L_1$ and $0 \le \mu_2 < L_2$? In the words, what is a (easily verifiable) sufficient condition for $f \in \mathcal{F}_{\mu_1,L_1}(\mathbb{R}^n) - \mathcal{F}_{\mu_2,L_2}(\mathbb{R}^n)$?

Another open problem would be the following.

Open problem 10.6. Assume a function f with its gradient vector is only specified on a finite set of points in \mathbb{R}^n . Under what condition(s) can the function f be decomposed as $f_1 - f_2$ on some open convex set containing the points, where f_1 and f_2 are convex? In other words, what is the interpolation theorem for the class of difference of two convex functions? In particular, are the usual interpolation conditions for f_1 and f_2 enough?

In the last chapter of the thesis, we studied convergence rate for the alternating direction method of multipliers (ADMM). We studied the convergence rate of the algorithm for different class of functions. We derive new non-ergodic convergence rates for ADMM and showed that the given bound are tight by providing a worst-case example; see Section 9.3. In a similar vein, we also examined the conditions that the algorithm enjoys linear convergence under those conditions, namely PL inequality that is shown as necessary and sufficient condition for linear convergence of the algorithm; see Section 9.4. Moreover, we study the *R*-linear convergence of ADMM under two new scenarios; see Section 9.5. It is important for future research to study the convergence rate of ADMM for long-step sizes and find the optimal step-length by the provided bound.

Open problem 10.7. *Is ADMM convergent for long step-sizes? If so, what are the optimal step-sizes for ADMM?*

To the best of our knowledge, the accelerated version of ADMM is not studied very deeply. Most of the researchers just use the same step-length as the steplength given for the accelerated gradient method.

Open problem 10.8. What is the optimal step-length for the accelerated ADMM?

It is worth mentioning that, as our analysis does not depend on the dimension of the problem n, the most of our results hold on any Hilbert space equipped with an appropriate inner-product.

Likewise, to offer a viewpoint for subsequent investigations, besides the possible research topics that are mentioned earlier in this chapter or referenced in the final segments of each chapter for the algorithm under the study, additional areas of study that may hold potential for future exploration. As we derived convergence rate of the proximal gradient method from the convergence rate of the DCA, studying the relation of the algorithms and deriving convergence rate of algorithms using other algorithms is very interesting research topic. As we studied the convergence rate of the algorithms with fixed step length for a bounded step length, it is shown in practice and, in some cases in theory, that the convergence rate can be improved for considering a periodic long step regime. Therefore, for algorithms like the descent ascent method it can be very interesting topic as well as studying accelerated version of this algorithm. Moreover, there are some algorithms that are not studied using performance estimation method or the known bound is not tight, e.g., the Chambolle-Pock algorithm [CP11], studying their convergence rate also can be interesting. At the end, in Table 10.1, we present a summary of the main results presented in the thesis.

Table 10.1: The summary of the main results on the convergence rates presented in the thesis

Method	Function class	Analytical	Rate of convergence	Proven tightness	
Cuediant descent method	L-smooth	Yes (Thm. 4.3)	Sub-linear	For some cases (Prop. 4.4)	
Gradient descent method	L-smooth satisfying	Vac (Then 5.4)	Timore	N	
	PŁ inequality	ies (11111. 5.4)	Linear	INO	
Coordinate descent method	L-smooth	No	-	No	
	Quadratic	No	-	No	
Randomized coordinate descent method	L-smooth	No	-	No	
	μ -strongly convex L-smooth	No	-	No	
	Convex quadratic	No	-	No	
Gradient descent-ascent	Strongly convex-strongly concave smooth	Yes (Thm. 7.2)	Linear	For some cases (Prop. 7.4)	
method	Convex-concave smooth satisfying	Vac (Then 7.0)	Timona	No	
	quadratic gradient growth	ies (11111. 7.9)	Linear	INO	
Difference of convex algorithm (DCA)	smooth	Yes (Thm. 8.6)	Sub-linear	For some cases (Ex. 8.8)	
	Nonsmooth	Yes (Thm. 8.12)	Sub-linear	No	
	Smooth satisfying PŁ inequality	Yes (Thm. 8.20)	Linear	No	
ADMM	Strongly convex	Yes (Thm. 9.4, 9.6, 9.8)	Sub-linear	Yes (Ex. 9.5, 9.7, 9.9)	
	Strongly convex with dual satisfying PŁ	Yes (Thm. 9.13)	Linear	No	
	One function is smooth and strongly	Vec (Thur 0.17)	R-linear	No	
	convex the other one is convex	ies (11111. 9.17)			
	One function is strongly convex the other	Vac (Thur 0.10)	DU	NT.	
	one is smooth	ies (111111, 9,19)	n-iiiieai	110	



A.1 Nonnegativity of multipliers in Theorem 7.2

Recall that, in the proof of Theorem 7.2, γ_1 is defined by

$$\gamma_1 = \frac{t \left(t^2 \left(2 + L^2 + L\mu \right) - t \left(3L + \mu \right) + \left(Lt - 1 \right) \sqrt{\left(Lt + \mu t - 2 \right)^2 + 4t^2} + 2 \right)}{\sqrt{\left(Lt + \mu t - 2 \right)^2 + 4t^2}}$$

Since *t* is nonnegative, we only need to prove that

$$\hat{\gamma}_1 := 2t^2 - 3Lt - \mu t + (Lt - 1)\sqrt{(Lt + \mu t - 2)^2 + 4t^2} + L^2t^2 + L\mu t^2 + 2t^2$$

is nonnegative. We show that the following optimization problem is lower bounded by zero,

$$\min_{L,t,\mu} \hat{\gamma}_1$$
s.t. $L \ge \mu, \mu \ge 0, t \ge 0,$

where L, t, μ are decision variables. First we consider the case that $Lt - 1 \le 0$. We have the following optimization problem

$$\min_{L,t} \left(\min_{0 \le \mu \le L} 2t^2 + L^2 t^2 + L\mu t^2 - 3Lt - \mu t + (Lt - 1)\sqrt{(Lt + \mu t - 2)^2 + 4t^2} + 2 \right)$$
s.t. $Lt \le 1, L \ge \mu, t \ge 0.$
(A.1)

The function $\hat{\gamma}_1$ is concave in μ , therefore, we just consider $\mu = 0$ and $\mu = L$. First we consider the case that $\mu = 0$. By substituting $\mu = 0$ in $\hat{\gamma}_1$ we have

$$\hat{\gamma}_1 = 2t^2 + (Lt - 1)(Lt - 2 + \sqrt{(Lt - 2)^2 + 4t^2}).$$

We argue that the above function is nonnegative on the feasible set of problem (A.1). By a conjugate multiplication of $Lt - 2 + \sqrt{(Lt - 2)^2 + 4t^2}$ one has

$$\hat{\gamma}_1 = 2t^2 \left(1 - \frac{2(1-Lt)}{(2-Lt) + \sqrt{(Lt-2)^2 + 4t^2}} \right),$$

since $(2 - Lt) + \sqrt{(Lt - 2)^2 + 4t^2} \ge 2(2 - Lt)$ we conclude that $0 \le \frac{2(Lt - 1)}{(Lt - 2) + \sqrt{(Lt - 2)^2 + 4t^2}} \le 1$ which proves $\hat{\gamma}_1$ is nonnegative.

Now we consider the case that $\mu = L$. By substituting $\mu = L$ we have

$$\hat{\gamma}_1 = 2t^2 + 2L^2t^2 - 4Lt + 2(Lt - 1)\sqrt{(Lt - 1)^2 + t^2} + 2.$$

Now we show that $\frac{1}{2}\hat{\gamma}_1 = t^2 + (Lt-1)((Lt-1) + \sqrt{(Lt-1)^2 + t^2})$ is nonnegative on the given set. Note that, again by conjugate multiplication,

$$\frac{1}{2}\hat{\gamma}_1 = t^2 \left(1 - \frac{(1-Lt)}{(1-Lt) + \sqrt{(Lt-1)^2 + t^2}} \right),$$

which always is nonnegative due to the nonnegativity of (1 - Lt).

Now we consider the case that tL - 1 > 0. We have

$$\begin{split} \hat{\gamma}_1 = & 2t^2 + L^2t^2 + L\mu t^2 - 3Lt - \mu t + (Lt - 1)\sqrt{(Lt + \mu t - 2)^2 + 4t^2} + 2 \\ \geq & 2t^2 + L^2t^2 + L\mu t^2 - 3Lt - \mu t + (Lt - 1)|Lt + \mu t - 2| + 2. \end{split}$$

Here, we need to consider two sub-cases. Firstly, when $2 - Lt - \mu t \ge 0$, we have

$$2t^{2} + L^{2}t^{2} + L\mu t^{2} - 3Lt - \mu t + (Lt - 1)(2 - Lt - \mu t) + 2 - 2t^{2} \ge 0.$$

If $Lt + \mu t - 2 \ge 0$, we have

$$2t^{2} + L^{2}t^{2} + L\mu t^{2} - 3Lt - \mu t + (Lt - 1)(Lt\mu t - 2) + 2$$

=(Lt - 2)² + (L - \mu)t + t² + L\mu t² \ge 0,

which completes the proof.

To show that γ_2 is nonnegative we follow the same procedure. Recall the definition of γ_2

$$\gamma_2 = \frac{t \left(t^2 \left(2 + \mu^2 + L\mu \right) - t \left(3\mu + L \right) + \left(1 - \mu t \right) \sqrt{\left(Lt + \mu t - 2 \right)^2 + 4t^2} + 2 \right)}{\sqrt{\left(Lt + \mu t - 2 \right)^2 + 4t^2}}$$
We define $\hat{\gamma}_2$ as

$$\hat{\gamma}_2 = t^2 \left(2 + \mu^2 + L\mu \right) - t \left(3\mu + L \right) + (1 - \mu t) \sqrt{\left(Lt + \mu t - 2 \right)^2 + 4t^2} + 2.$$

Due to $t \ge 0$, we only need to show that $\hat{\gamma}_2$ is nonnegative. To this end, we show that the following optimization problem is lower bounded by zero.

$$\min_{L,t,\mu} \hat{\gamma}_2 = t^2 \left(2 + \mu^2 + L\mu \right) - t \left(3\mu + L \right) + (1 - \mu t) \sqrt{(Lt + \mu t - 2)^2 + 4t^2 + 2}$$

s.t. $L \ge \mu, \mu \ge 0, t \ge 0.$

First we consider the case that $1 - \mu t \ge 0$. We have

$$\hat{\gamma}_2 \ge t^2 \left(2 + \mu^2 + L\mu\right) - t \left(3\mu + L\right) + (1 - \mu t) |Lt + \mu t - 2| + 2.$$

We consider two sub-cases. Firstly, $Lt + \mu t - 2 \ge 0$:

$$\hat{\gamma}_{2} = t^{2} \left(2 + \mu^{2} + L\mu \right) - t \left(3\mu + L \right) + (1 - \mu t) \sqrt{(Lt + \mu t - 2)^{2} + 4t^{2}} + 2$$

$$\geq t^{2} \left(2 + \mu^{2} + L\mu \right) - t \left(3\mu + L \right) + (1 - \mu t) (Lt + \mu t - 2) + 2 = 2t^{2} \geq 0.$$

Now assume that $Lt + \mu t - 2 \le 0$.

$$\begin{split} \hat{\gamma}_2 &= t^2 \left(2 + \mu^2 + L\mu \right) - t \left(3\mu + L \right) + (1 - \mu t) \sqrt{(Lt + \mu t - 2)^2 + 4t^2 + 2t^2} \\ &\geq t^2 \left(2 + \mu^2 + L\mu \right) - t \left(3\mu + L \right) + (1 - \mu t) \left(2 - Lt - \mu t \right) + 2t^2 \\ &= 2 \left((\mu t - 1)^2 + (2 - \mu t - Lt) + \mu Lt^2 + t^2 \right) \geq 0. \end{split}$$

Now we consider the case that $1 - \mu t \leq 0$.

$$\min_{t,\mu} \left(\min_{\mu \le L \le \frac{2\mu - t}{\mu t}} \hat{\gamma}_2 = t^2 \left(2 + \mu^2 + L\mu \right) - t \left(3\mu + L \right) + (1 - \mu t) \sqrt{\left(Lt + \mu t - 2 \right)^2 + 4t^2} + 2 \right)$$
s.t. $\mu t \ge 1, \mu \ge 0, t \ge 0.$

Note that $\hat{\gamma}_2$ is concave with respect to the variable *L*. Therefore, we should study the boundaries of *L*. If we set $L = \mu$ we have

$$\hat{\gamma}_2 = 2(t^2 + \mu^2 t^2 - 2\mu t + (1 - \mu t)\sqrt{(\mu t - 1)^2 + t^2} + 1)$$

= 2t² + 2(\mu t - 1)((\mu t - 1) - \sqrt{(\mu t - 1)^2 + t^2}).

By conjugate multiplication, we have

$$\hat{\gamma}_2 = 2t^2 \left(1 - \frac{(\mu t - 1)}{\mu t - 1 + \sqrt{(\mu t - 1)^2 + t^2}} \right) \ge 0.$$

By $L \leq \frac{2\mu - t}{\mu t}$ one can see that $L \leq \frac{2}{t}$. Setting $L = \frac{2}{t}$:

$$\begin{split} \hat{\gamma}_2 &= -\mu t + 2t^2 + \mu^2 t^2 + (1 - \mu t) \sqrt{4t^2 + \mu^2 t^2} \\ &= 2t^2 \left(1 - \frac{2(\mu t - 1)}{\mu t + \sqrt{\mu^2 t^2 + 4t^2}} \right) \ge 0. \end{split}$$

This completes the proof.

A.2 Identity used in the proof of Theorem 7.2

The proof of Theorem 7.2 requires the following identity, that may be verified through direct (symbolic) calculation:

$$\begin{split} \left\| x^{0} - t G_{x}^{0,0} \right\|^{2} + \left\| y^{0} + t G_{y}^{0,0} \right\|^{2} - \bar{a} \left(\left\| x^{0} \right\|^{2} + \left\| y^{0} \right\|^{2} \right) + \gamma_{1} \left(F^{0,0} - F^{*,0} - \langle G_{x}^{*,0}, x^{0} \rangle - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \left\| G_{x}^{0,0} - G_{x}^{*,0} \right\|^{2} + \mu \left\| x^{0} \right\|^{2} - \frac{2\mu}{L} \left\langle G_{x}^{*,0} - G_{x}^{0,0}, -x^{0} \right\rangle \right) \right) + \gamma_{2} \left(F^{*,0} - F^{0,0} + \left\langle G_{x}^{0,0}, x^{0} \right\rangle - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \left\| G_{x}^{*,0} - G_{x}^{0,0} \right\|^{2} + \mu \left\| x^{0} \right\|^{2} - \frac{2\mu}{L} \left\langle G_{x}^{0,0}, -G_{x}^{*,0}, x^{0} \right\rangle \right) \right) + \gamma_{2} \left(F^{0,*} - F^{*,*} - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \left\| G_{x}^{0,*} \right\|^{2} + \mu \left\| x^{0} \right\|^{2} - \frac{2\mu}{L} \left\langle G_{x}^{0,*}, x^{0} \right\rangle \right) \right) + \gamma_{1} \left(F^{*,*} - F^{0,*} + \left\langle G_{x}^{0,*}, x^{0} \right\rangle - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \left\| G_{x}^{0,*} \right\|^{2} + \mu \left\| x^{0} \right\|^{2} - \frac{2\mu}{L} \left\langle G_{x}^{0,*}, x^{0} \right\rangle \right) \right) + \gamma_{1} \left(F^{0,*} - F^{0,0} + \left\langle G_{y}^{0,*}, y^{0} \right\rangle - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \left\| G_{y}^{0,0} - G_{y}^{0,*} \right\|^{2} + \mu \left\| y^{0} \right\|^{2} - \frac{2\mu}{L} \left\langle G_{y}^{0,0} - G_{y}^{0,0}, y^{0} \right\rangle \right) \right) + \gamma_{2} \left(F^{0,0} - F^{0,*} - \left\langle G_{y}^{0,0}, y^{0} \right\rangle - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \left\| G_{y}^{0,0} - G_{y}^{0,0} \right\|^{2} + \mu \left\| y^{0} \right\|^{2} - \frac{2\mu}{L} \left\langle G_{y}^{0,0} - G_{y}^{0,0}, y^{0} \right\rangle \right) \right) + \gamma_{2} \left(F^{0,0} - F^{0,*} - \left\langle G_{y}^{0,0}, y^{0} \right\rangle - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \left\| G_{y}^{0,0} - G_{y}^{0,0} \right\|^{2} + \mu \left\| y^{0} \right\|^{2} - \frac{2\mu}{L} \left\langle G_{y}^{0,0} - G_{y}^{0,0}, y^{0} \right\rangle \right) \right) + \gamma_{1} \left(- F^{*,*} + F^{*,0} + \left\langle G_{y}^{*,0}, - y^{0} \right\rangle - \frac{L}{2(L-\mu)} \left(\frac{1}{L} \left\| G_{y}^{*,0} \right\|^{2} + \mu \left\| y^{0} \right\|^{2} - \frac{2\mu}{L} \left\langle G_{y}^{*,0}, - y^{0} \right\rangle \right) \right) + \gamma_{3} \left(\left\| x^{0} \right\|^{2} - \left\| G_{y}^{0,0} - G_{y}^{*,0} \right\|^{2} \right) + \gamma_{3} \left(\left\| x^{0} \right\|^{2} - \left\| G_{y}^{0,0} - G_{y}^{*,0} \right\|^{2} \right) \right) \right) + \gamma_{3} \left(\left\| x^{0} \right\|^{2} - \left\| G_{y}^{0,0} - G_{y}^{*,0} \right\|^{2} \right) \right) \right) \right\|_{1}^{2} + 2\mu \left\| y^{0} \right\|_{1}^{2} - \frac{2\mu}{L} \left\langle G_{y}^{0,0} - G_{y}^{0,0} \right\|_{1}^{2} \right) \right) \right\|_{1}^{2} + 2\mu \left\| y^{0} \right\|_{1}^{2} - \frac{2\mu}{L} \left\langle G_{y}^{0,0} - G_{y}^{0,0} \right) \right) + \gamma_{3} \left(\left\| x^{0} \right\|^{2} - \left\| G_{y}^{0,0} - G_{y}^{0,0} \right\|^{2} \right) \right\|_{1}^{2} + \mu \left\| y^{0}$$

where $\zeta_1, \zeta_2, \zeta_3, \zeta_4$ are given by

$$\begin{split} \zeta_{1} &= \frac{1}{2}t\left(\frac{(L^{2}+\mu^{2})\beta}{L-\mu} - \frac{2t^{2}(L-\mu)}{\beta} + (L+\mu)(t(L+\mu)-2)\right),\\ \zeta_{2} &= -\frac{(L^{2}t-L-\mu^{2}t+\mu)\beta-L^{2}t(Lt+\mu t-3)-(L+\mu)(\mu^{2}t^{2}-2\mu t+2t^{2}+2)+\mu^{2}t}{2t^{2}(L+\mu)^{2}(L\mu+1)-8L\mu t(L+\mu)+8L\mu},\\ \zeta_{3} &= -\frac{t(L^{2}+6L\mu+\mu^{2})-2t^{2}(L+\mu)(L\mu+1)-(L-\mu)\beta-2(L+\mu)}{2t^{2}(L+\mu)^{2}(L\mu+1)-8L\mu t(L+\mu)+8L\mu},\\ \zeta_{4} &= \frac{t(\beta+Lt-\mu t)^{2}}{4(L-\mu)\beta}. \end{split}$$

Note that $\zeta_1, \zeta_4 \ge 0$, as required.

A.3 Proof for positive semidefniteness of the matrix presented in Theorem 9.6

Lemma A.1. Let $N \ge 4$ and $t, c_1 \in \mathbb{R}$. Let $D(t, c_1)$ be $N \times N$ symmetric matrix given in Theorem 9.6. If $c_1 > 0$ is given, then

$$[0, c_1] \subseteq \{t : D(t, c_1) \succeq 0\}.$$

Proof. The argument proceeds in the same manner as in Lemma 9.2. Due to the convexity of $\{t : D(t, c_1) \geq 0\}$, is sufficient to establish the positive semidefiniteness of $D(0, c_1)$ and $D(c_1, c_1)$. As $D(0, c_1)$ is diagonally dominant, it is positive semidefinite. Next, we proceed to demonstrate the positive definiteness of the matrix K = D(1, 1) by computing its leading principal minors. One can show that the claim holds for N = 4. So we investigate $N \geq 5$. To accomplish this, we perform the following elementary row operations on matrix D:

- i) Add the second row to the third row;
- ii) Add the second row to the last row;
- iii) Add the third row to the forth row;
- iv) For i = 4: N 2
 - Add i th row to (i + 1) th row;
 - Add $\frac{3-i}{2i^2-3i-1}$ times of i th row to the last row;
- v) Add $\frac{2N^2-8N+9}{2N^2-7N+4}$ times of (N-1)-th row to N-th row.

By executing these operations, we transform K into an upper triangular matirx J with diagonal

$$J_{k,k} = \begin{cases} 2, & k = 1\\ 2k^2 - 3k - 1, & 2 \le k \le N - 1\\ 2N^2 - 7N + 8 - \frac{N^2}{(N-1)^2} - \frac{(2N^2 - 8N + 9)^2}{2N^2 - 7N + 4} - \sum_{i=4}^{N-2} \frac{(i-3)^2}{2i^2 - 3i - 1}, & k = N. \end{cases}$$

It is seen all first (N-1) diagonal elements of *J* are positive. We show that $J_{N,N}$ is also positive. By using inequality (9.15), we get

$$2N^{2} - 7N + 8 - \frac{N^{2}}{(N-1)^{2}} - \frac{(2N^{2} - 8N + 9)^{2}}{2N^{2} - 7N + 4} - \sum_{i=4}^{N-2} \frac{(i-3)^{2}}{2i^{2} - 3i - 1} \ge 2N^{2} - 7N + 8 - \frac{25}{16} - (2N^{2} - 8N + 9) - \frac{N-5}{2} - 1 + \frac{2}{N-3} \ge \frac{N}{2} - \frac{17}{16} > 0,$$

for $N \ge 5$, which implies $J_{N,N} > 0$. Hence, $D(c_1, c_1) \ge 0$ and the proof is complete.

Summary

Academic Summary

This thesis presents a comprehensive analysis of the convergence rates and performance of iterative optimization algorithms used in machine learning. Focusing on gradient descent and other first-order methods, the study introduces novel theoretical insights on convergence rates of several methods by utilizing Performance Estimation Problems (PEPs) formulated as semidefinite programming problems. Key contributions include the establishment of exact worst-case convergence rates and conditions under which these methods achieve linear convergence. This work not only deepens the theoretical understanding of several optimization algorithms, but also guides practical applications in machine learning where efficient data processing is crucial.

Summary for Non-Experts

This thesis investigates how certain mathematical methods, known as iterative optimization algorithms, where the best option must be chosen among feasible alternatives. Specifically, it explores how quickly these methods can reach reliable conclusions. The study introduces and improves mathematical tools and theories that help predict the performance of these algorithms, focusing on those that use a technique called gradient descent. This research is important because it helps improve the efficiency of algorithms that process vast amounts of data, making them faster and more effective in applications like image recognition, voice recognition, and other AI technologies.

Academische Samenvatting

Dit proefschrift presenteert een uitgebreide analyse van de convergentiesnelheden en prestaties van iteratieve optimalisatie-algoritmen die gebruikt worden in machine learning. De studie richt zich op gradiëntafdaling en andere methoden van de eerste orde, en introduceert nieuwe theoretische inzichten in de convergentietempo van verschillende methoden door gebruik te maken van prestatieschattingsproblemen (PSP's), geformuleerd als semidefiniete optimalisatieproblemen. Belangrijke bijdragen omvatten het vaststellen van exacte convergentietempo's in het slechtste geval en de omstandigheden waaronder deze methoden lineaire convergentie bereiken. Dit werk verdiept niet alleen het theoretische begrip van verschillende optimalisatie-algoritmen, maar faciliteert ook praktische toepassingen in machine learning waarbij efficiënte dataverwerking cruciaal is.

Samenvatting voor Niet-Experts

Dit proefschrift onderzoekt bepaalde wiskundige methoden, bekend als iteratieve optimalisatie-algoritmen, die zich bezighouden met het kiezen van de beste optie uit haalbare alternatieven. Het onderzoekt wijst hoe snel deze methoden betrouwbare conclusies kunnen bereiken. De studie introduceert en verbetert wiskundige hulpmiddelen en theorieën die de prestaties van deze algoritmen helpen voorspellen, met een nadruk op technieken zoals gradiëntafdaling. Dit onderzoek is belangrijk omdat het de efficiëntie van algoritmen verbetert die grote hoeveelheden gegevens verwerken, waardoor ze sneller en effectiever worden in toepassingen zoals beeldherkenning, spraakherkenning en andere AI-technologieën.

Index

 (γ, μ_s) -quasar-convex, **79** *L*-smooth function, **20** *L*-smooth functions, 2 *L*-smoothness, 128 *Q*-convergence, 4 *Q*-linearly convergent, 4 *Q*-super-linearly convergent, 4 *R*-convergence, 4 *R*-linear convergence, 184 μ -strongly convex *L*-smooth functions, 3 μ -strongly convex function, 3, **22** Lojasiewicz inequality, **72**, 125 Lojasiewicz exponent, 125

Alternating direction method of multipliers, 156

Biconjugate, **19** Bilinear problem, 116 Black-box first-order methods, **35**

Closed function, complementary slackness, Conjugate of function, Convergence, 4 Convex function, **15**, Convex hull, 127 Convex interpolation, 28 Convex set, 14 convex-concave saddle point problem, 104 Coordinate descent methods, 84 Critical point, 129 Cyclic coordinate descent method, 91 DCA, 124 Descent lemma, 21, 60 Difference-of-convex (DC) optimization problem, 124 Distance function, 15 Dual norm, 16 Dual residual, 159 Dual space, 16 Duality gap, 25 Epigraph, 16 Error bound property, 175 Euclidean space, 14

Fast gradient algorithm, 72 Fenchel's duality theorem, Fenchel's inequality, Fermat's optimality condition, First-order methods, 3 First-order oracle,

Fixed-step iterative methods. 36 Frobenius inner product, 25 Gauss-Seidel method, 95 General subgradient, 24 Gradient descent method with fixed step-size, 37 Gradient descent-ascent method, 104 Gradient dominated, 63 Gram matrix, 37, 87, 162 Induced norm, 14 Inexact weighted-Jacobi method, 98 Infinite-dimensional optimization problem, 133 Inner product, 14 Jacobian matrix, 63 Lagrangian dual, 40 Lagrangian multipliers, 179 Linear matrix inequality, 25 Lower semi-continuous function, 16 Lyapunov function, 184 Maximum curvature, 62 Minimum curvature, 30, 62 Nonlinear least squares problem, 63 Nonlinear programming, 2 Oracle, 34 Polyak-Łojasiewicz inequality, 63, 150, 176 Projection, 15 Proper function, 15 Proximal gradient method, 142 Proximal operator, 143

Q-linear convergence, 175Quadratic function, 89Quadratic functional growth, 76Quadratic gradient growth, 74, 118

Randomized coordinate descent method, 85 Regular subgradient, **24** Regularization term, 147 Relative interior, **15** Relative smoothness, **23** Relative strong convexity, **23** Richardson method, 99

Semidefinite programming, **24**, 38, 73, 107, 133 Seminorm, **14** Stationary point, 129 Strictly convex, **16** Strictly diagonally dominant matrix, **98** Strictly differentiable, 150 Strictly feasible, 25 Strong duality, **25** Sub-linear, 4 Subdifferential, **17** Subgradient, **16**

Toland dual, 130

Undominated D.C. decompositions, 149

Weak duality, **25**, 53, 67, 109, 134, 179 Weighted-Jacobi method, 98 Whitney extension theorem, 57

Bibliography

- [AAH⁺58] Kenneth Joseph Arrow, Hirofumi Azawa, Leonid Hurwicz, Hirofumi Uzawa, Hollis Burnley Chenery, Selmer M Johnson, and Samuel Karlin. Studies in linear and non-linear programming, volume 2. Stanford University Press, 1958.
 - [AB09] Hedy Attouch and Jérôme Bolte. On the convergence of the proximal algorithm for nonsmooth functions involving analytic features. *Mathematical Programming*, 116:5–16, 2009.
- [ABRS10] Hédy Attouch, Jérôme Bolte, Patrick Redont, and Antoine Soubeyran. Proximal alternating minimization and projection methods for nonconvex problems: An approach based on the kurdyka-Łojasiewicz inequality. *Mathematics of Operations Research*, 35(2):438–457, 2010.
- [AdKZ22] Hadi Abbaszadehpeivasti, Etienne de Klerk, and Moslem Zamani. The exact worst-case convergence rate of the gradient method with fixed step lengths for L-smooth functions. *Optimization Letters*, 16(6):1649–1661, 2022.
- [AdKZ23a] Hadi Abbaszadehpeivasti, Etienne de Klerk, and Moslem Zamani. Conditions for linear convergence of the gradient method for nonconvex optimization. Optimization Letters, 17(5):1105–1125, 2023.
- [AdKZ23b] Hadi Abbaszadehpeivasti, Etienne de Klerk, and Moslem Zamani. Convergence rate analysis of randomized and cyclic coordinate descent for convex optimization through semidefinite programming. *Applied Set-Valued Analysis and Optimization*, 5(2):141–153, 2023.

[AdKZ23c] Hadi Abbaszadehpeivasti, Etienne de Klerk, and Moslem Zamani. On

the rate of convergence of the difference-of-convex algorithm (DCA). *Journal of Optimization Theory and Applications*, pages 1–22, 2023.

- [AFG12] Annabella Astorino, Antonio Fuduli, and Manlio Gaudioso. Margin maximization in spherical separation. *Computational Optimization and Applications*, 53:301–322, 2012.
 - [AH18] Amir Ali Ahmadi and Georgina Hall. DC decomposition of nonconvex polynomials with algebraic techniques. *Mathematical Programming*, 169:69–94, 2018.
- [AP23a] Jason M Altschuler and Pablo A Parrilo. Acceleration by stepsize hedging i: Multi-step descent and the silver stepsize schedule. *arXiv* preprint arXiv:2309.07879, 2023.
- [AP23b] Jason M Altschuler and Pablo A Parrilo. Acceleration by stepsize hedging ii: Silver stepsize schedule for smooth convex optimization. *arXiv preprint arXiv:2309.16530*, 2023.
- [ApS19] MOSEK ApS. The MOSEK optimization toolbox for MATLAB manual. Version 9.0., 2019.
- [ASP14] Alberth Alvarado, Gesualdo Scutari, and Jong-Shi Pang. A new decomposition method for multiuser DC-programming and its applications. *IEEE Transactions on Signal Processing*, 62(11):2984–2998, 2014.
- [AT05] Le Thi Hoai An and Pham Dinh Tao. The DC (difference of convex functions) programming and DCA revisited with DC models of real world nonconvex optimization problems. *Annals of operations research*, 133:23–46, 2005.
- [Bag95] Roberto Bagnara. A unified proof for the convergence of Jacobi and Gauss–Seidel methods. *SIAM review*, 37(1):93–97, 1995.
- [BB24] Christopher M. Bishop and Hugh Bishop. *Gradient Descent*, pages 209–232. Springer International Publishing, Cham, 2024.
- [BBT17] Heinz H Bauschke, Jérôme Bolte, and Marc Teboulle. A descent lemma beyond lipschitz gradient continuity: first-order methods revisited and applications. *Mathematics of Operations Research*, 42(2):330–348, 2017.

- [BCN18] Léon Bottou, Frank E Curtis, and Jorge Nocedal. Optimization methods for large-scale machine learning. SIAM review, 60(2):223–311, 2018.
- [BDL07] Jérôme Bolte, Aris Daniilidis, and Adrian Lewis. The Łojasiewicz inequality for nonsmooth subanalytic functions with applications to subgradient dynamical systems. SIAM Journal on Optimization, 17(4):1205–1223, 2007.
- [Bec17] Amir Beck. First-order methods in optimization. SIAM, 2017.
- [Ber15] Dimitri Bertsekas. *Convex optimization algorithms*. Athena Scientific, 2015.
- [BGLS06] Joseph-Frédéric Bonnans, Jean Charles Gilbert, Claude Lemaréchal, and Claudia A Sagastizábal. *Numerical optimization: theoretical and practical aspects*. Springer Science & Business Media, 2006.
- [BHG22] Nizar Bousselmi, Julien Hendrickx, and François Glineur. Performance estimation of first-order methods on quadratic functions. In *MTNS 2022*, 2022.
- [BHG23] Nizar Bousselmi, Julien M Hendrickx, and François Glineur. Interpolation conditions for linear operators and applications to performance estimation problems. *arXiv preprint arXiv:2302.08781*, 2023.
- [BM20] Jingjing Bu and Mehran Mesbahi. A note on Nesterov's accelerated method in nonconvex optimization: a weak estimate sequence approach. *arXiv preprint arXiv:2006.08548*, 2020.
- [BNPS17] Jérôme Bolte, Trong Phong Nguyen, Juan Peypouquet, and Bruce W Suter. From error bounds to the complexity of first-order descent methods for convex functions. *Mathematical Programming*, 165:471–507, 2017.
 - [BO98] Tamer Başar and Geert Jan Olsder. *Dynamic noncooperative game theory*. SIAM, 1998.
- [BPC⁺11] Stephen Boyd, Neal Parikh, Eric Chu, Borja Peleato, Jonathan Eckstein, et al. Distributed optimization and statistical learning via the alternating direction method of multipliers. *Foundations and Trends*® *in Machine Learning*, 3(1):1–122, 2011.

- [BPG⁺23] Aleksandr Beznosikov, Boris Polyak, Eduard Gorbunov, Dmitry Kovalev, and Alexander Gasnikov. Smooth monotone stochastic variational inequalities and saddle point problems: A survey. European Mathematical Society Magazine, (127):15–28, 2023.
 - [BST14] Jérôme Bolte, Shoham Sabach, and Marc Teboulle. Proximal alternating linearized minimization for nonconvex and nonsmooth problems. *Mathematical Programming*, 146(1-2):459–494, 2014.
 - [BT09] Amir Beck and Marc Teboulle. Fast gradient-based algorithms for constrained total variation image denoising and deblurring problems. *IEEE Transactions on Image Processing*, 18(11):2419–2434, 2009.
 - [BT13] Amir Beck and Luba Tetruashvili. On the convergence of block coordinate descent type methods. *SIAM journal on Optimization*, 23(4):2037–2060, 2013.
- [BTEGN09] Aharon Ben-Tal, Laurent El Ghaoui, and Arkadi Nemirovski. *Robust optimization*, volume 28. Princeton University Press, 2009.
 - [BTU16] Adil M Bagirov, Sona Taheri, and Julien Ugon. Nonsmooth DC programming approach to the minimum sum-of-squares clustering problems. *Pattern Recognition*, 53:12–24, 2016.
 - [BU18] Adil M Bagirov and Julien Ugon. Nonsmooth DC programming approach to clusterwise linear regression: optimality conditions and algorithms. *Optimization methods and software*, 33(1):194–219, 2018.
 - [BV04] Stephen P Boyd and Lieven Vandenberghe. *Convex optimization*. Cambridge University Press, 2004.
 - [Cau47] Augustin Cauchy. Méthode générale pour la résolution des systemes d'équations simultanées. Comp. Rend. Sci. Paris, 25(1847):536–538, 1847.
 - [CDHS20] Yair Carmon, John C Duchi, Oliver Hinder, and Aaron Sidford. Lower bounds for finding stationary points I. *Mathematical Programming*, 184(1-2):71–120, 2020.

- [CGT10] Coralia Cartis, Nicholas IM Gould, and Ph L Toint. On the complexity of steepest descent, Newton's and regularized Newton's methods for nonconvex unconstrained optimization problems. SIAM journal on Optimization, 20(6):2833–2852, 2010.
 - [Che] Long Chen. Classical iterative methods. https://www.math.uci. edu/~chenlong/226/Ch6IterativeMethod.pdf.
- [CHJT98] Pey-Chun Chen, Pierre Hansen, Brigitte Jaumard, and Hoang Tuy. Solution of the multisource weber and conditional weber problems by D.-C. programming. *Operations Research*, 46(4):548–562, 1998.
 - [CP11] Antonin Chambolle and Thomas Pock. A first-order primal-dual algorithm for convex problems with applications to imaging. *Journal of mathematical imaging and vision*, 40:120–145, 2011.
 - [DD19] Damek Davis and Dmitriy Drusvyatskiy. Stochastic model-based minimization of weakly convex functions. *SIAM Journal on Optimization*, 29(1):207–239, 2019.
- [DDG⁺22] Marina Danilova, Pavel Dvurechensky, Alexander Gasnikov, Eduard Gorbunov, Sergey Guminov, Dmitry Kamzolov, and Innokentiy Shibaev. Recent theoretical advances in non-convex optimization. In *High-Dimensional Optimization and Probability: With a View Towards* Data Science, pages 79–163. Springer, 2022.
 - [Dem97] James W Demmel. Applied numerical linear algebra. SIAM, 1997.
- [DGVPR23] Shuvomoy Das Gupta, Bart PG Van Parys, and Ernest K Ryu. Branch-and-bound performance estimation programming: a unified methodology for constructing optimal optimization methods. *Mathematical Programming*, pages 1–73, 2023.
 - [DH19] Simon S Du and Wei Hu. Linear convergence of the primal-dual gradient method for convex-concave saddle point problems without strong convexity. In *The 22nd International Conference on Artificial Intelligence and Statistics*, pages 196–205. PMLR, 2019.
- [DHLGL18] Aris Daniilidis, Mounir Haddou, Erwan Le Gruyer, and Olivier Ley. Explicit formulas for $C^{1,1}$ Glaeser-Whitney extensions of 1-Taylor

fields in Hilbert spaces. *Proceedings of the American Mathematical Society*, 146(10):4487–4495, 2018.

- [DK06] Etienne De Klerk. Aspects of semidefinite programming: interior point algorithms and selected applications, volume 65. Springer Science & Business Media, 2006.
- [dKGT17] Etienne de Klerk, François Glineur, and Adrien B Taylor. On the worst-case complexity of the gradient method with exact line search for smooth strongly convex functions. *Optimization Letters*, 11:1185–1199, 2017.
- [DKGT20] Etienne De Klerk, Francois Glineur, and Adrien B Taylor. Worstcase convergence analysis of inexact gradient and Newton methods through semidefinite programming performance estimation. *SIAM Journal on Optimization*, 30(3):2053–2082, 2020.
 - [Dro14] Yoel Drori. *Contributions to the complexity analysis of optimization algorithms*. PhD thesis, Universitat Tel-Aviv, Tel Aviv, Israel, 2014.
 - [Dro20] Yoel Drori. On the properties of convex functions over open sets. Journal of Convex Analysis, 27(4):1303–1314, 2020.
 - [DS20] Yoel Drori and Ohad Shamir. The complexity of finding stationary points with stochastic gradient descent. In *International Conference on Machine Learning*, pages 2658–2667. PMLR, 2020.
 - [DT14] Yoel Drori and Marc Teboulle. Performance of first-order methods for smooth convex minimization: a novel approach. *Mathematical Programming*, 145(1-2):451–482, 2014.
 - [DT20] Yoel Drori and Adrien B Taylor. Efficient first-order methods for convex minimization: a constructive approach. *Mathematical Programming*, 184(1-2):183–220, 2020.
- [DTdB21] Radu-Alexandru Dragomir, Adrien B Taylor, Alexandre d'Aspremont, and Jérôme Bolte. Optimal complexity and certification of Bregman first-order methods. *Mathematical Programming*, pages 1–43, 2021.
 - [DY16] Wei Deng and Wotao Yin. On the global and linear convergence of the generalized alternating direction method of multipliers. *Journal of Scientific Computing*, 66(3):889–916, 2016.

- [DY17] Damek Davis and Wotao Yin. Faster convergence rates of relaxed Peaceman-Rachford and ADMM under regularity assumptions. *Mathematics of Operations Research*, 42(3):783–805, 2017.
- [FM81] Masao Fukushima and Hisashi Mine. A generalized proximal point algorithm for certain non-convex minimization problems. *International Journal of Systems Science*, 12(8):989–1000, 1981.
- [FOP20] Alireza Fallah, Asuman Ozdaglar, and Sarath Pattathil. An optimal multistage stochastic gradient method for minimax problems. In 2020 59th IEEE Conference on Decision and Control (CDC), pages 3573–3579. IEEE, 2020.
 - [FP03] Francisco Facchinei and Jong-Shi Pang. *Finite-dimensional variational inequalities and complementarity problems*. Springer, 2003.
- [FRV18] Guilherme Franca, Daniel Robinson, and Rene Vidal. ADMM and accelerated ADMM as continuous dynamical systems. In *International Conference on Machine Learning*, pages 1559–1567. PMLR, 2018.
- [GB16] Pontus Giselsson and Stephen Boyd. Linear convergence and metric selection for Douglas-Rachford splitting and ADMM. *IEEE Transactions on Automatic Control*, 62(2):532–544, 2016.
- [GBC16] Ian Goodfellow, Yoshua Bengio, and Aaron Courville. *Deep Learning*. MIT Press, 2016.
- [Gha19] Saeed Ghadimi. Conditional gradient type methods for composite nonlinear and stochastic optimization. *Mathematical Programming*, 173:431–464, 2019.
- [GM75] Roland Glowinski and Americo Marroco. Sur l'approximation, par éléments finis d'ordre un, et la résolution, par pénalisation-dualité d'une classe de problèmes de dirichlet non linéaires. ESAIM: Mathematical Modelling and Numerical Analysis-Modélisation Mathématique et Analyse Numérique, 9(R2):41–76, 1975.
- [GM76] Daniel Gabay and Bertrand Mercier. A dual algorithm for the solution of nonlinear variational problems via finite element approximation. *Computers & mathematics with applications*, 2(1):17–40, 1976.

- [GMG⁺22] Baptiste Goujaud, Céline Moucer, François Glineur, Julien Hendrickx, Adrien Taylor, and Aymeric Dieuleveut. PEPit: computerassisted worst-case analyses of first-order optimization methods in Python. *arXiv preprint arXiv:2201.04040*, 2022.
 - [GMS13] Donald Goldfarb, Shiqian Ma, and Katya Scheinberg. Fast alternating linearization methods for minimizing the sum of two convex functions. *Mathematical Programming*, 141(1):349–382, 2013.
 - [GO82] Gene H Golub and Michael L Overton. Convergence of a two-stage Richardson iterative procedure for solving systems of linear equations. In *Numerical analysis*, pages 125–139. Springer, 1982.
 - [GO88] Gene H Golub and Michael L Overton. The convergence of inexact Chebyshev and Richardson iterative methods for solving linear systems. *Numerische Mathematik*, 53(5):571–593, 1988.
- [GOSB14] Tom Goldstein, Brendan O'Donoghue, Simon Setzer, and Richard Baraniuk. Fast alternating direction optimization methods. *SIAM Journal on Imaging Sciences*, 7(3):1588–1623, 2014.
 - [GOY17] Roland Glowinski, Stanley J Osher, and Wotao Yin. *Splitting methods in communication, imaging, science, and engineering*. Springer, 2017.
- [GPAM⁺20] Ian Goodfellow, Jean Pouget-Abadie, Mehdi Mirza, Bing Xu, David Warde-Farley, Sherjil Ozair, Aaron Courville, and Yoshua Bengio. Generative adversarial networks. *Communications of the ACM*, 63(11):139–144, 2020.
 - [GRC09] Gilles Gasso, Alain Rakotomamonjy, and Stéphane Canu. Recovering sparse signals with a certain family of nonconvex penalties and DC programming. *IEEE Transactions on Signal Processing*, 57(12):4686– 4698, 2009.
 - [Gri23] Benjamin Grimmer. Provably faster gradient descent via long steps. *arXiv preprint arXiv:2307.06324*, 2023.
 - [Gru09] Erwan Le Gruyer. Minimal Lipschitz extensions to differentiable functions defined on a Hilbert space. *Geometric and Functional Analysis*, 19(4):1101–1118, 2009.

- [GSW23] Benjamin Grimmer, Kevin Shu, and Alex L Wang. Accelerated gradient descent via long steps. *arXiv preprint arXiv:2309.09961*, 2023.
- [GTG22] Eduard Gorbunov, Adrien Taylor, and Gauthier Gidel. Last-iterate convergence of optimistic gradient method for monotone variational inequalities. *Advances in Neural Information Processing Systems*, 35:21858–21870, 2022.
- [GTHG23] Eduard Gorbunov, Adrien Taylor, Samuel Horváth, and Gauthier Gidel. Convergence of proximal point and extragradient-based methods beyond monotonicity: the case of negative comonotonicity. In *International Conference on Machine Learning*, pages 11614–11641. PMLR, 2023.
 - [GTT18] Jun-ya Gotoh, Akiko Takeda, and Katsuya Tono. DC formulations and algorithms for sparse optimization problems. *Mathematical Programming*, 169:141–176, 2018.
 - [GVL13] Gene H Golub and Charles F Van Loan. *Matrix computations*. JHU Press, 2013.
 - [GY20] Guoyong Gu and Junfeng Yang. On the dual step length of the alternating direction method of multipliers. *arXiv preprint arXiv:2006.08309*, 2020.
 - [GY22] Guoyong Gu and Junfeng Yang. Tight ergodic sublinear convergence rate of the relaxed proximal point algorithm for monotone variational inequalities. *Journal of Optimization Theory and Applications*, pages 1–15, 2022.
 - [HA21] Erfan Yazdandoost Hamedani and Necdet Serhat Aybat. A primaldual algorithm with line search for general convex-concave saddle point problems. *SIAM Journal on Optimization*, 31(2):1299–1329, 2021.
 - [Han22] De-Ren Han. A survey on some recent developments of alternating direction method of multipliers. *Journal of the Operations Research Society of China*, 10(1):1–52, 2022.
 - [Har59] Philip Hartman. On functions representable as a difference of convex functions. *Pacific Journal Mathematics*, 9(3):707–713, 1959.

- [HJ12] Roger A Horn and Charles R Johnson. *Matrix analysis*. Cambridge University Press, 2012.
- [HL17] Mingyi Hong and Zhi-Quan Luo. On the linear convergence of the alternating direction method of multipliers. *Mathematical Programming*, 162(1):165–199, 2017.
- [HSL21] Bin Hu, Peter Seiler, and Laurent Lessard. Analysis of biased stochastic gradient descent using sequential semidefinite programs. *Mathematical Programming*, 187:383–408, 2021.
- [HSS20] Oliver Hinder, Aaron Sidford, and Nimit Sohoni. Near-optimal methods for minimizing star-convex functions and beyond. In *Conference on Learning Theory*, pages 1894–1938. PMLR, 2020.
- [HSZ18] Deren Han, Defeng Sun, and Liwei Zhang. Linear rate convergence of the alternating direction method of multipliers for convex composite programming. *Mathematics of Operations Research*, 43(2):622–637, 2018.
- [HT99a] Kaj Holmberg and Hoang Tuy. A production-transportation problem with stochastic demand and concave production costs. *Mathematical Programming*, 85:157–179, 1999.
- [HT99b] Reiner Horst and Nguyen V Thoai. DC programming: overview. *Journal of Optimization Theory and Applications*, 103:1–43, 1999.
- [HTW15] Trevor Hastie, Robert Tibshirani, and Martin Wainwright. Statistical learning with sparsity. *Monographs on Statistics and Applied Probability*, 143:143, 2015.
 - [HU85] J-B Hiriart-Urruty. Generalized differentiability/duality and optimization for problems dealing with differences of convex functions. In Convexity and Duality in Optimization: Proceedings of the Symposium on Convexity and Duality in Optimization Held at the University of Groningen, The Netherlands June 22, 1984, pages 37–70. Springer, 1985.
- [HUL13] Jean-Baptiste Hiriart-Urruty and Claude Lemaréchal. *Convex analysis and minimization algorithms I: Fundamentals*, volume 305. Springer science & business media, 2013.

- [HY12] Bingsheng He and Xiaoming Yuan. On the O(1/n) convergence rate of the Douglas-Rachford alternating direction method. *SIAM Journal on Numerical Analysis*, 50(2):700–709, 2012.
- [Jac46] C.G.J. Jacobi. Uber ein leichtes verfahren die in der theorie der sacuuirstorungen vorkommenden gleichungen numerisch aufzulosen. *J. Reine Angew. Math*, 30:51–94, 1846.
- [JBK⁺18] Kaisa Joki, Adil M Bagirov, Napsu Karmitsa, Marko M Makela, and Sona Taheri. Double bundle method for finding Clarke stationary points in nonsmooth DC programming. SIAM Journal on Optimization, 28(2):1892–1919, 2018.
 - [JGR23] Uijeong Jang, Shuvomoy Das Gupta, and Ernest K Ryu. Computerassisted design of accelerated composite optimization methods: Optista. *arXiv preprint arXiv:2305.15704*, 2023.
 - [JK⁺17] Prateek Jain, Purushottam Kar, et al. Non-convex optimization for machine learning. Foundations and Trends[®] in Machine Learning, 10(3-4):142–363, 2017.
 - [JM22] Ruichen Jiang and Aryan Mokhtari. Generalized optimistic methods for convex-concave saddle point problems. *arXiv preprint arXiv:2202.09674*, 2022.
 - [KF16] Donghwan Kim and Jeffrey A Fessler. Optimized first-order methods for smooth convex minimization. *Mathematical Programming*, 159:81–107, 2016.
 - [KF18a] Donghwan Kim and Jeffrey A Fessler. Another look at the fast iterative shrinkage/thresholding algorithm (fista). SIAM Journal on Optimization, 28(1):223–250, 2018.
 - [KF18b] Donghwan Kim and Jeffrey A Fessler. Generalizing the optimized gradient method for smooth convex minimization. SIAM Journal on Optimization, 28(2):1920–1950, 2018.
 - [KF21] Donghwan Kim and Jeffrey A Fessler. Optimizing the efficiency of first-order methods for decreasing the gradient of smooth convex functions. *Journal of optimization theory and applications*, 188(1):192–219, 2021.

- [KHG23] Yassine Kamri, Julien M Hendrickx, and François Glineur. On the worst-case analysis of cyclic coordinate-wise algorithms on smooth convex functions. In 2023 European Control Conference (ECC), pages 1–8. IEEE, 2023.
- [Kim21] Donghwan Kim. Accelerated proximal point method for maximally monotone operators. *Mathematical Programming*, 190(1-2):57–87, 2021.
- [KNS16] Hamed Karimi, Julie Nutini, and Mark Schmidt. Linear convergence of gradient and proximal-gradient methods under the Polyak-Łojasiewicz condition. In Machine Learning and Knowledge Discovery in Databases: European Conference, ECML PKDD 2016, Riva del Garda, Italy, September 19-23, 2016, Proceedings, Part I 16, pages 795–811. Springer, 2016.
 - [LB16] Thomas Lipp and Stephen Boyd. Variations and extension of the convex–concave procedure. Optimization and Engineering, 17:263– 287, 2016.
- [LFN18] Haihao Lu, Robert M Freund, and Yurii Nesterov. Relatively smooth convex optimization by first-order methods, and applications. *SIAM Journal on Optimization*, 28(1):333–354, 2018.
- [LJJ20] Tianyi Lin, Chi Jin, and Michael I Jordan. Near-optimal algorithms for minimax optimization. In *Conference on Learning Theory*, pages 2738–2779. PMLR, 2020.
- [LL19] Huan Li and Zhouchen Lin. Accelerated alternating direction method of multipliers: An optimal O(1/k) nonergodic analysis. *Journal of Scientific Computing*, 79(2):671–699, 2019.
- [LLF22] Zhouchen Lin, Huan Li, and Cong Fang. Alternating Direction Method of Multipliers for Machine Learning. Springer, 2022.
- [Löf04] J. Löfberg. YALMIP : A toolbox for modeling and optimization in MATLAB. In *In Proceedings of the CACSD Conference*, Taipei, Taiwan, 2004.

- [LRP16] Laurent Lessard, Benjamin Recht, and Andrew Packard. Analysis and design of optimization algorithms via integral quadratic constraints. *SIAM Journal on Optimization*, 26(1):57–95, 2016.
 - [LS19] Tengyuan Liang and James Stokes. Interaction matters: A note on non-asymptotic local convergence of generative adversarial networks. In *The 22nd International Conference on Artificial Intelligence* and Statistics, pages 907–915. PMLR, 2019.
- [LSW⁺22] Hongli Liu, Yawei Shi, Zheng Wang, Liang Ran, Qingguo Lü, and Huaqing Li. A distributed algorithm based on relaxed ADMM for energy resources coordination. *International Journal of Electrical Power* & Energy Systems, 135:107482, 2022.
 - [LT93] Zhi-Quan Luo and Paul Tseng. Error bounds and convergence analysis of feasible descent methods: a general approach. *Annals of Operations Research*, 46(1):157–178, 1993.
- [LTHPD18] Hoai An Le Thi, Van Ngai Huynh, and Tao Pham Dinh. Convergence analysis of difference-of-convex algorithm with subanalytic data. *Journal of Optimization Theory and Applications*, 179(1):103– 126, 2018.
 - [LTN17] Hoai An Le Thi and Manh Cuong Nguyen. DCA based algorithms for feature selection in multi-class support vector machine. *Annals of Operations Research*, 249(1-2):273–300, 2017.
 - [LTPD18] Hoai An Le Thi and Tao Pham Dinh. DC programming and DCA: thirty years of developments. *Mathematical Programming*, 169(1):5– 68, 2018.
 - [LTPD21] Hoai An Le Thi, Duy Nhat Phan, and Tao Pham Dinh. DCA based approaches for bi-level variable selection and application for estimate multiple sparse covariance matrices. *Neurocomputing*, 466:162–177, 2021.
 - [LV12] Monique Laurent and Frank Vallentin. Semidefinite optimization. Lecture Notes, available at http://page. mi. fu-berlin. de/fmario/sdp/laurentv. pdf, 2012.

- [LV21] Luke Lozenski and Umberto Villa. Consensus ADMM for inverse problems governed by multiple PDE models. *arXiv preprint arXiv:2104.13899*, 2021.
- [LY⁺84] David G Luenberger, Yinyu Ye, et al. *Linear and nonlinear programming*, volume 2. Springer, 1984.
- [LYZZ18] Yongchao Liu, Xiaoming Yuan, Shangzhi Zeng, and Jin Zhang. Partial error bound conditions and the linear convergence rate of the alternating direction method of multipliers. SIAM Journal on Numerical Analysis, 56(4):2095–2123, 2018.
 - [LZ19] Zhaosong Lu and Zirui Zhou. Nonmonotone enhanced proximal DC algorithms for a class of structured nonsmooth DC programming. *SIAM Journal on Optimization*, 29(4):2725–2752, 2019.
- [LZOX15] Yifei Lou, Tieyong Zeng, Stanley Osher, and Jack Xin. A weighted difference of anisotropic and isotropic total variation model for image processing. SIAM Journal on Imaging Sciences, 8(3):1798–1823, 2015.
 - [LZS19] Zhaosong Lu, Zirui Zhou, and Zhe Sun. Enhanced proximal DC algorithms with extrapolation for a class of structured nonsmooth DC minimization. *Mathematical Programming*, 176:369–401, 2019.
 - [Mel86] D Melzer. On the expressibility of piecewise-linear continuous functions as the difference of two piecewise-linear convex functions. *Quasidifferential Calculus*, pages 118–134, 1986.
- [MHA20] Anirudha Majumdar, Georgina Hall, and Amir Ali Ahmadi. Recent scalability improvements for semidefinite programming with applications in machine learning, control, and robotics. *Annual Review of Control, Robotics, and Autonomous Systems*, 3:331–360, 2020.
- [MKL15] Ramtin Madani, Abdulrahman Kalbat, and Javad Lavaei. ADMM for sparse semidefinite programming with applications to optimal power flow problem. In 2015 54th IEEE Conference on Decision and Control (CDC), pages 5932–5939. IEEE, 2015.

- [MNG17] Lars Mescheder, Sebastian Nowozin, and Andreas Geiger. The numerics of GANs. Advances in Neural Information Processing Systems, 30, 2017.
- [MOP20a] Aryan Mokhtari, Asuman Ozdaglar, and Sarath Pattathil. A unified analysis of extra-gradient and optimistic gradient methods for saddle point problems: Proximal point approach. In *International Conference on Artificial Intelligence and Statistics*, pages 1497–1507. PMLR, 2020.
- [MOP20b] Aryan Mokhtari, Asuman E Ozdaglar, and Sarath Pattathil. Convergence rate of O(1/k) for optimistic gradient and extragradient methods in smooth convex-concave saddle point problems. SIAM Journal on Optimization, 30(4):3230–3251, 2020.
 - [MS13] Renato DC Monteiro and Benar F Svaiter. Iteration-complexity of block-decomposition algorithms and the alternating direction method of multipliers. *SIAM Journal on Optimization*, 23(1):475– 507, 2013.
 - [MTB23] Céline Moucer, Adrien Taylor, and Francis Bach. A systematic approach to Lyapunov analyses of continuous-time models in convex optimization. SIAM Journal on Optimization, 33(3):1558–1586, 2023.
 - [Nes03] Yurii Nesterov. *Introductory lectures on convex optimization: A basic course*, volume 87. Springer Science & Business Media, 2003.
 - [Nes12] Yu Nesterov. Efficiency of coordinate descent methods on huge-scale optimization problems. SIAM Journal on Optimization, 22(2):341– 362, 2012.
 - [Nes13] Yu Nesterov. Gradient methods for minimizing composite functions. *Mathematical Programming*, 140(1):125–161, 2013.
 - [Nes18] Yurii Nesterov. *Lectures on convex optimization*, volume 137. Springer, 2018.
- [NLR⁺15] Robert Nishihara, Laurent Lessard, Ben Recht, Andrew Packard, and Michael Jordan. A general analysis of the convergence of ADMM.

In International Conference on Machine Learning, pages 343–352. PMLR, 2015.

- [NNG19] Ion Necoara, Yu Nesterov, and Francois Glineur. Linear convergence of first order methods for non-strongly convex optimization. *Mathematical Programming*, 175:69–107, 2019.
- [NW06] Jorge Nocedal and Stephen J. Wright. *Numerical Optimization*. Springer, New York, NY, USA, 2e edition, 2006.
- [NY83] A. S. Nemirovsky and D. B. Yudin. Problem Complexity and Method Efficiency in Optimization. Wiley-Interscience, New York, NY, USA, 1st edition, 1983.
- [NYZ21] Jiawang Nie, Zi Yang, and Guangming Zhou. The saddle point problem of polynomials. *Foundations of Computational Mathematics*, pages 1–37, 2021.
 - [Pol63] Boris T Polyak. Gradient methods for the minimisation of functionals. USSR Computational Mathematics and Mathematical Physics, 3(4):864–878, 1963.
 - [PR23] Jisun Park and Ernest K Ryu. Accelerated infeasibility detection of constrained optimization and fixed-point iterations. arXiv preprint arXiv:2303.15876, 2023.
- [PRA17] Jong-Shi Pang, Meisam Razaviyayn, and Alberth Alvarado. Computing b-stationary points of nonsmooth DC programs. *Mathematics of Operations Research*, 42(1):95–118, 2017.
- [PVZ21] Javier Peña, Juan C Vera, and Luis F Zuluaga. Linear convergence of the Douglas-Rachford algorithm via a generic error bound condition. *arXiv preprint arXiv:2111.06071*, 2021.
- [RGP22] Teodor Rotaru, Franois Glineur, and Patrinos Panagiotis. Tight convergence rates of the gradient method on hypoconvex functions. *arXiv preprint arXiv:2203.00775*, 2022.
 - [Ric11] Lewis Fry Richardson. IX. the approximate arithmetical solution by finite differences of physical problems involving differential equations, with an application to the stresses in a masonry dam. *Philo*-

sophical Transactions of the Royal Society of London. Series A, Containing Papers of a Mathematical or Physical Character, 210(459-470):307–357, 1911.

- [Roc97] R Tyrrell Rockafellar. Convex analysis, volume 11. Princeton University Press, 1997.
- [ROF92] Leonid I Rudin, Stanley Osher, and Emad Fatemi. Nonlinear total variation based noise removal algorithms. *Physica D: Nonlinear Phenomena*, 60(1-4):259–268, 1992.
- [RTBG20] Ernest K Ryu, Adrien B Taylor, Carolina Bergeling, and Pontus Giselsson. Operator splitting performance estimation: Tight contraction factors and optimal parameter selection. SIAM Journal on Optimization, 30(3):2251–2271, 2020.
 - [RW09] R Tyrrell Rockafellar and Roger J-B Wets. *Variational analysis*, volume 317. Springer Science & Business Media, 2009.
 - [Saa03] Yousef Saad. Iterative methods for sparse linear systems. SIAM, 2003.
- [SBG⁺20] Bartolomeo Stellato, Goran Banjac, Paul Goulart, Alberto Bemporad, and Stephen Boyd. OSQP: An operator splitting solver for quadratic programs. *Mathematical Programming Computation*, 12(4):637– 672, 2020.
- [SCZZ19] Shiliang Sun, Zehui Cao, Han Zhu, and Jing Zhao. A survey of optimization methods from a machine learning perspective. *IEEE transactions on cybernetics*, 50(8):3668–3681, 2019.
- [SPPMD19] John W Simpson-Porco, Bala Kameshwar Poolla, Nima Monshizadeh, and Florian Dörfler. Input–output performance of linear–quadratic saddle-point algorithms with application to distributed resource allocation problems. *IEEE Transactions on Automatic Control*, 65(5):2032–2045, 2019.
 - [SS22] Kaizhao Sun and Xu Andy Sun. Algorithms for difference-of-convex programs based on difference-of-moreau-envelopes smoothing. *IN-FORMS Journal on Optimization*, 2022.

- [ST22] Shoham Sabach and Marc Teboulle. Faster Lagrangian-based methods in convex optimization. *SIAM Journal on Optimization*, 32(1):204–227, 2022.
- [TA97] Pham Dinh Tao and LT Hoai An. Convex analysis approach to DC programming: theory, algorithms and applications. *Acta Mathematica Vietnamica*, 22(1):289–355, 1997.
- [Tay17] Adrien B Taylor. Convex interpolation and performance estimation of first-order methods for convex optimization. PhD thesis, Catholic University of Louvain, Louvain-la-Neuve, Belgium, 2017.
- [TB19] Adrien Taylor and Francis Bach. Stochastic first-order methods: nonasymptotic and computer-aided analyses via potential functions. In *Conference on Learning Theory*, pages 2934–2992. PMLR, 2019.
- [TD23] Adrien Taylor and Yoel Drori. An optimal gradient method for smooth strongly convex minimization. *Mathematical Programming*, 199(1-2):557–594, 2023.
- [THG17a] Adrien B Taylor, Julien M Hendrickx, and François Glineur. Exact worst-case performance of first-order methods for composite convex optimization. SIAM Journal on Optimization, 27(3):1283–1313, 2017.
- [THG17b] Adrien B Taylor, Julien M Hendrickx, and François Glineur. Performance estimation toolbox (pesto): Automated worst-case analysis of first-order optimization methods. In 2017 IEEE 56th Annual Conference on Decision and Control (CDC), pages 1278–1283. IEEE, 2017.
- [THG17c] Adrien B Taylor, Julien M Hendrickx, and François Glineur. Smooth strongly convex interpolation and exact worst-case performance of first-order methods. *Mathematical Programming*, 161:307–345, 2017.
 - [THG18] Adrien B Taylor, Julien M Hendrickx, and François Glineur. Exact worst-case convergence rates of the proximal gradient method for composite convex minimization. *Journal of Optimization Theory and Applications*, 178:455–476, 2018.

- [THH⁺98] Hoang Tuy, Tuy Hoang, Tuy Hoang, Viêt-nam Mathématicien, Tuy Hoang, and Vietnam Mathematician. *Convex analysis and global optimization*. Springer, 1998.
 - [Tol79] John F Toland. A duality principle for non-convex optimisation and the calculus of variations. *Archive for Rational Mechanics and Analysis*, 71:41–61, 1979.
- [TVSL18] Adrien Taylor, Bryan Van Scoy, and Laurent Lessard. Lyapunov functions for first-order methods: Tight automated convergence guarantees. In *International Conference on Machine Learning*, pages 4897– 4906. PMLR, 2018.
- [UBTG23] Manu Upadhyaya, Sebastian Banert, Adrien B Taylor, and Pontus Giselsson. Automated tight Lyapunov analysis for first-order methods. *arXiv preprint arXiv:2302.06713*, 2023.
 - [VB96] Lieven Vandenberghe and Stephen Boyd. Semidefinite programming. *SIAM review*, 38(1):49–95, 1996.
 - [Wel73] John C Wells. Differentiable functions on Banach spaces with Lipschitz derivatives. Journal of Differential Geometry, 8(1):135–152, 1973.
 - [Whi34] Hassler Whitney. Analytic extensions of differentiable functions defined in closed sets. *Transactions of the American Mathematical Society*, 36(1):63–89, 1934.
 - [WKK20] Alex L Wang and Fatma Kilinc-Karzan. A geometric view of SDP exactness in QCQPs and its applications. *arXiv preprint arXiv:2011.07155*, 2020.
 - [WL20] Yuanhao Wang and Jian Li. Improved algorithms for convex-concave minimax optimization. *Advances in Neural Information Processing Systems*, 33:4800–4810, 2020.
- [WMZT20] Qi Wang, Yue Ma, Kun Zhao, and Yingjie Tian. A comprehensive survey of loss functions in machine learning. *Annals of Data Science*, pages 1–26, 2020.

- [Wri15] Stephen J Wright. Coordinate descent algorithms. *Mathematical Programming*, 151(1):3–34, 2015.
- [WSV12] Henry Wolkowicz, Romesh Saigal, and Lieven Vandenberghe. *Handbook of semidefinite programming: theory, algorithms, and applications*, volume 27. Springer Science & Business Media, 2012.
- [XZXL23] Zi Xu, Huiling Zhang, Yang Xu, and Guanghui Lan. A unified single-loop alternating gradient projection algorithm for nonconvex– concave and convex–nonconcave minimax problems. *Mathematical Programming*, 201:635–706, 2023.
- [YND⁺20] Mengjiao Yang, Ofir Nachum, Bo Dai, Lihong Li, and Dale Schuurmans. Off-policy evaluation via the regularized lagrangian. Advances in Neural Information Processing Systems, 33:6551–6561, 2020.
- [YPWL12] Ian EH Yen, Nanyun Peng, Po-Wei Wang, and Shou-De Lin. On convergence rate of concave-convex procedure. In Proceedings of the NIPS 2012 Optimization Workshop, pages 31–35, 2012.
 - [YZZ20] Xiaoming Yuan, Shangzhi Zeng, and Jin Zhang. Discerning the linear convergence of ADMM for structured convex optimization through the lens of variational analysis. J. Mach. Learn. Res., 21:83–1, 2020.
- [ZAdK23] Moslem Zamani, Hadi Abbaszadehpeivasti, and Etienne de Klerk. The exact worst-case convergence rate of the alternating direction method of multipliers. *Mathematical Programming*, 2023.
- [ZAdK24] Moslem Zamani, Hadi Abbaszadehpeivasti, and Etienne de Klerk. Convergence rate analysis of the gradient descent-ascent method for convex-concave saddle-point problems. Optimization Methods and Software, 2024.
- [ZBLG21] Guodong Zhang, Xuchan Bao, Laurent Lessard, and Roger Grosse. A unified analysis of first-order methods for smooth games via integral quadratic constraints. *The Journal of Machine Learning Research*, 22(1):4648–4686, 2021.
 - [ZG23] Moslem Zamani and François Glineur. Exact convergence rate of the last iterate in subgradient methods. arXiv preprint arXiv:2307.11134, 2023.

- [ZHZ22] Junyu Zhang, Mingyi Hong, and Shuzhong Zhang. On lower iteration complexity bounds for the convex concave saddle point problems. *Mathematical Programming*, 194(1-2):901–935, 2022.
- [ZWLG22] Guodong Zhang, Yuanhao Wang, Laurent Lessard, and Roger B Grosse. Near-optimal local convergence of alternating gradient descent-ascent for minimax optimization. In International Conference on Artificial Intelligence and Statistics, pages 7659–7679. PMLR, 2022.

CENTER DISSERTATION SERIES

No.	Author	Title	ISBN	Published
672	Joobin Ordoobody	The Interplay of Structural and Individual Characteristics	978 90 5668 674 1	February 2022
673	Lucas Avezum	Essays on Bank Regulation and Supervision	978 90 5668 675 8	March 2022
674	Oliver Wichert	Unit-Root Tests in High-Dimensional Panels	978 90 5668 676 5	April 2022
675	Martijn de Vries	Theoretical Asset Pricing under Behavioral Decision Making	978 90 5668 677 2	June 2022
676	Hanan Ahmed	Extreme Value Statistics using Related Variables	978 90 5668 678 9	June 2022
677	Jan Paulick	Financial Market Information Infrastructures: Essays on Liquidity, Participant Behavior, and Information Extraction	978 90 5668 679 6	June 2022
678	Freek van Gils	Essays on Social Media and Democracy	978 90 5668 680 2	June 2022
679	Suzanne Bies	Examining the Effectiveness of Activation Techniques on Consumer Behavior in Temporary Loyalty Programs	978 90 5668 681 9	July 2022
680	Qinnan Ruan	Management Control Systems and Ethical Decision Making	978 90 5668 682 6	June 2022
681	Lingbo Shen	Essays on Behavioral Finance and Corporate Finance	978 90 5668 683 3	August 2022
682	Joshua Eckblad	Mind the Gales: An Attention-Based View of Startup Investment Arms	978 90 5668 684 0	August 2022
683	Rafael Greminger	Essays on Consumer Search	978 90 5668 685 7	August 2022
684	Suraj Upadhyay	Essay on policies to curb rising healthcare expenditures	978 90 5668 686 4	September 2022

CentER for Economic Research, Tilburg University, the Netherlands

No.	Author	Title	ISBN	Published
685	Bert-Jan Butijn	From Legal Contracts to Smart Contracts and Back Again: An Automated Approach	978 90 5668 687 1	September 2022
686	Sytse Duiverman	Four essays on the quality of auditing: Causes and consequences	978 90 5668 688 8	October 2022
687	Lucas Slot	Asymptotic Analysis of Semidefinite Bounds for Polynomial Optimization and Independent Sets in Geometric Hypergraphs	978 90 5668 689 5	September 2022
688	Daniel Brosch	Symmetry reduction in convex optimization with applications in combinatorics	978 90 5668 690 1	October 2022
689	Emil Uduwalage	Essays on Corporate Governance in Sri Lanka	978 90 5668 691 8	October 2022
690	Mingjia Xie	Essays on Education and Health Economics	978 90 5668 692 5	October 2022
691	Peerawat Samranchit	Competition in Digital Markets	978 90 5668 693 2	October 2022
692	Jop Schouten	Cooperation, allocation and strategy in interactive decision-making	978 90 5668 694 9	December 2022
693	Pepijn Wissing	Spectral Characterizations of Complex Unit Gain Graphs	978 90 5668 695 6	November 2022
694	Joris Berns	CEO attention, emotion, and communication in corporate financial distress	978 90 5668 696 3	November 2022
695	Tom Aben	The (long) road towards smart management and maintenance: Organising the digital transformation of critical infrastructures	978 90 5668 697 0	December 2022
696	Gülbike Mirzaoğlu	Essays in Economics of Crime Prevention and Behavior Under Uncertainty	978 90 5668 698 7	February 2023
697	Suwei An	Essays on incentive contracts, M&As, and firm risk	978 90 5668 699 4	February 2023
698	Jorgo Goossens	Non-standard Preferences in Asset Pricing and Household Finance	978 90 5668 700 7	February 2023

No.	Author	Title	ISBN	Published
699	Santiago Bohorquez Correa	Risk and rewards of residential energy efficiency	978 90 5668 701 4	April 2023
700	Gleb Gertsman	Behavioral Preferences and Beliefs in Asset Pricing	978 90 5668 702 1	May 2023
701	Gabriella Massenz	On the Behavioral Effects of Tax Policy	978 90 5668 703 8	May 2023
702	Yeqiu Zheng	The Effect of Language and Temporal Focus on Cognition, Economic Behaviour, and Well-Being	978 90 5668 704 5	May 2023
703	Michela Bonani	Essays on Innovation, Cooperation, and Competition Under Standardization	978 90 5668 705 2	June 2023
704	Fabien Ize	The Role of Transparency in Fairness and Reciprocity Issues in Manager-Employee Relationships	978 90 5668 706 9	June 2023
705	Kristel de Nobrega	Cyber Defensive Capacity and Capability: A Perspective from the Financial Sector of a Small State	978 90 5668 707 6	July 2023
706	Christian Peters	The Microfoundations of Audit Quality	978 90 5668 708 3	June 2023
707	Felix Kirschner	Conic Optimization with Applications in Finance and Approximation Theory	978 90 5668 709 0	July 2023
708	Zili Su	Essays on Equity Incentive and Share Pledging in China	978 90 5668 710 6	September 2023
709	Rafael Escamilla	Managing the Nanostore Supply Chain: Base-of-the-Pyramid Retail in Emerging Markets	978 90 5668 711 3	September 2023
710	Tomas Jankauskas	Essays in Empirical Finance	978 90 5668 712 0	August 2023
711	Tung Nguyen Huy	Fostering Sustainable Land Management in Sub-Saharan Africa: Evidence from Ghana and Burkina Faso	978 90 5668 713 7	September 2023
712	Daniel Karpati	Essays in Finance & Health	978 90 5668 714 4	September 2023
713	Mylène Struijk	IT Governance in the Digital Era: Insights from Meta-Organizations	978 90 5668 715 1	September 2023

No.	Author	Title	ISBN	Published
714	Albert Rutten	Essays on Work and Retirement	978 90 5668 716 8	November 2023
715	Yan Liu	Essays on Credit Rating Agencies in China	978 90 5668 717 5	October 2023
716	Xiaoyue Zhang	Distortions and industrial upgrading in China	978 90 5668 718 2	September 2023
717	Andries van Beek	Solutions in multi-actor projects with collaboration and strategic incentives	978 90 5668 719 9	October 2023
718	Andries Steenkamp	Polynomial Optimization: Matrix Factorization Ranks, Portfolio Selection, and Queueing Theory	978 90 5668 720 5	October 2023
719	Luis Vargas	Sum-of-Squares Representations for Copositive Matrices and Independent Sets in Graphs	978 90 5668 721 2	November 2023
720	Zhenshu Wu	Essays in Corporate Finance and ESG	978 90 5668 722 9	November 2023
721	Frank de Meijer	Integrality and Cutting Planes in Semidefinite Programming Approaches for Combinatorial Optimization	978 90 5668 723 6	November 2023
722	Chayanin Wipusanawan	Standard-Essential Patents, Innovation, and Competition	978 90 5668 724 3	November 2023
723	Ke Wang	Essays in Corporate Risk	978 90 5668 725 0	November 2023
724	Mustafa Ahci	Essays on corporate disclosures, innovation, and investments	978 90 5668 726 7	November 2023
725	Frederique van Leeuwen	Motifs: From Theory to Practice	978 90 5668 727 4	December 2023
726	Lkhagvaa Erdenesuren	Expectation, Anticipation, and Identification: Essays on Subjective Expectations and Economic Decision- Making	978 90 5668 728 1	December 2023
727	Wouter van Eekelen	Distributionally Robust Views on Queues and Related Stochastic Models	978 90 5668 729 8	December 2023
728	Jia Bi	Essays on Extreme Returns and Investor Behavior	978 90 5668 730 4	January 202

No.	Author	Title	ISBN	Published
729	Jesse van der Geest	Economic Effects of Tax Avoidance and Compliance	978 90 5668 731 1	January 2024
730	Zeng Yu	Essays on Incentive Contract and Corporate Finance	978 90 5668 732 8	January 2024
731	Joost de Kruijff	Commitment-Based Smart Contracts using CommitRuleML	978 90 5668 733 5	February 2024
732	Simona Hannon	Essays On Consumer Finance	978 90 5668 734 2	January 2024
733	Bart Dees	Individualized Pension Contracts: Risks, Welfare Losses and Investment Choices	978 90 5668 735 9	April 2024
734	José Gabriel Carreño Bustos	Three Essays on Wage Compensation and Flexible Contracts	978 90 5668 736 6	March 2024
735	Lennart Dekker	Essays on Asset Liquidity and Investment Funds	978 90 5668 737 3	April 2024
736	Valentijn Stienen	Analytics for Humanitarian Networks in Uncertain and Data-Scarce Environments	978 90 5668 738 0	May 2024
737	Constanza Martinez Ventura	Essays on Liquidity Provision in Wholesale Funding Markets and on Financial Fragility	978 90 5668 739 7	May 2024
738	Kartik Chawla	Guarding the Digital Cookie Jar: An Interdisciplinary Study of Automated Privacy Preference Negotiation, Monitoring, and Enforcement	978 90 5668 740 3	May 2024
739	Koen Peters	Operationalizing analytics to improve food security	978 90 5668 741 0	June 2024
740	Јоусе Кох	Navigating Crises in Family Businesses: Unraveling the Role of the Family	978 90 5668 742 7	June 2024
741	Hazal Sezer	Historical and Contemporary Perspectives in Labor	978 90 5668 743 4	June 2024
742	Hadi Abbaszadehpeivasti	Performance Analysis of Optimization Methods for Machine Learning	978 90 5668 744 1	October 2024

HADI ABBASZADEHPEIVASTI (Tabriz, Iran, 1989) earned his bachelor's degree in Applied Mathematics from the University of Zanjan. Subsequently, he pursued and obtained his Master's degrees in Industrial Engineering from both Sharif University of Technology and Sabancı University. Following this, he continued as a Ph.D. candidate in the Department of Econometrics and Operations Research at Tilburg University, funded by the NWO ENW-GROOT project Optimization for and with Machine Learning (OPTIMAL).

Nowadays, complex real-world optimization problems, especially in Machine Learning, are gaining importance. Understanding which algorithm has the best performance for a specific problem is crucial. To answer this, it is important to understand how different algorithms behave. This dissertation focuses on studying the behaviors of certain algorithms widely used in optimization. We utilized performance estimation methodology that relies on semidefinite programming. In this thesis, we thoroughly studied five different algorithms.

ISBN: 978 90 5668 744 1 DOI: 10.26116/tisem.46749783

